Assessing the potential of biochar from crop residues to sequester CO₂: Scenarios to 2100.

Ву

Jayne Helen Windeatt

Submitted in accordance with the requirements for the degree of Doctor of Philosophy

February 2015

Energy Research Institute, The School of Chemical and Process Engineering, University of Leeds

and

The School of Earth and Environment, University of Leeds

The candidate confirms that the work submitted is her own, except where work which has formed part of jointly authored publications has been included. The contribution of the candidate and the other authors to this work has been explicitly indicated below. The candidate confirms that appropriate credit has been given within the thesis where reference has been made to the work of others.

Some of the work in Chapter 4, 5 and 7 of the thesis appears in publication as follows:

Windeatt, J.H., A.B. Ross, P.T. Williams, P.M.Forster, M.A. Nahil, S. Singh. 2014. Characteristics of biochars from crop residues: Potential for carbon sequestration and soil amendment. Journal of Environmental Management. **146** (15), pp 189-197.

The candidate, Jayne Windeatt, was responsible for all laboratory analysis, data interpretation and interpretation of the work within the publication. The joint authors had supervisory and advisory roles. Within this thesis Chapter 4 contains some of the published results from the laboratory analysis detailed in the above publication, whilst Chapters 5 and 7 contain some of the same methodology as the publication, although the applications to data are different. This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement.

© 2015 The University of Leeds and Jayne Helen Windeatt

The right of Jayne Helen Windeatt to be identified as Author of this work has been asserted by herself in accordance with the Copyright, Designs and Patents Act 1988.

Abstract

Amongst the suite of current or developing climate change mitigation tools, biochar is discussed within the literature as a method for long-term carbon sequestration (CS). The biochar field is rapidly developing, though there are uncertainties and limitations for which understanding could be improved.

The aim of this thesis was to assess the potential of biochars from crop residues to sequester carbon, under the land-use pathways of the Representative Concentration Pathways, to 2100. Eight crop residue feedstocks and their biochars were fully characterised to examine the effects of feedstock and process conditions on biochar characteristics. Biochar yield, carbon content and recalcitrance values from this experimental work were utilized in the global modelling of scenarios exploring future carbon sequestration potential.

Biochars produced were Class B or Class C, using the recalcitrance classification of Harvey et al. (2012), and classed as moderately or highly degradable. Recalcitrance increased with increasing pyrolysis temperature. The recalcitrance index of Harvey et al. (2012) may underestimate recalcitrance in high alkali metal content biochars. The carbon sequestration (CS) potential of the biochars was affected by the yield and content of stable carbon content of the biochar and predicted to be between 21.3 % and 32.5 %. The feedstock carbon remaining in the biochars decreased with increasing pyrolysis temperature although carbon stability increased with temperature. Biochar CS potential decreased with increasing pyrolysis temperature, despite increased stability and is due to the decreasing yields observed. A new equation was developed, using feedstock volatile content, as an alternative to the CS equation of Zhao et al. (2013).

The Representative Concentration Pathways (RCPs) were used alongside the experimental results for biochar yield, carbon content and recalcitrance, and secondary data such as future crop yield and crop residue trends to project the CS potential of crop residues from 2005 to 2100. Scenarios of biochar production and carbon storage were developed, built around the RCPs, investigating biochar potential under changing crop land area and exploring parameters such as biochar characteristics and biochar systems. Scenario 1 used the mean or most likely values from experimental data and literature, Scenarios 2 to 7 explored parameter assumptions and Scenarios 8 and 9 explored the impact of climate change on crop yields and subsequent biochar CS potential. Global biochar production in Scenario 1 for the four RCPs over 95 years (2005 to 2100) was: RCP 2.6, 138.4 Gt biochar; RCP 4.5, 132.3 Gt biochar; RCP 6, 173.2 Gt biochar and RCP 8.5, 217.9 Gt biochar. Although the carbon mitigation potential of biochar in the scenarios generally increased from RCP 2.6 to RCP 8.5, the quantity of emissions requiring

mitigation also increased. Scenario 1 saw 49.0, 45.8, 60.9 and 77.2 GtC sequestered over the 95 year period for the four RCPs respectively. These are reductions of 11 %, 5 %, 5 % and 4 % on the RCPs carbon emissions pathways. The maximum and minimum carbon emission mitigation potentials achievable under the assumptions of scenarios 1 to 7 were 22.5 %, 10.8 %, 10.0 %, 8.3 % and 4.7 %, 2.2 %, 1.9 %, 1.5 % for the four RCPs respectively. Climate change generally resulted in a decreasing carbon sequestration potential from RCP 2.6 up to RCP 8.5. This negative impact also increased over time. The maximum impact on mitigation potential in 2100 was - 0.14 GtC yr⁻¹ for RCP 2.6, this increased to - 0.72 GtC yr⁻¹ for RCP 8.5.

Biochar has the potential to sequester carbon in all of the scenarios explored, however the magnitude of this sequestration potential is dependent on a number of factors of which many are currently subject to large amounts of uncertainty. Reduction in these areas of uncertainty would be a valuable area of further work following this study.

Acknowledgements

Firstly I would like to thank my supervisory team, Professor Piers Forster, Dr Andrew Ross and Professor Paul Williams for their support, advice and guidance throughout. Following on from this I would like to extend these thanks to the Doctoral Training Centre staff, in particular James McKay, David Haynes, Rachael Brown and Emily Bryan-Kinns for their support and organisation which has been a great help. Thank you also to my DTC colleagues, who have provided support, thought provoking discussion and entertainment in my time at the centre. These have been some of the toughest yet most enjoyable years of my life so far. I would like to add a special mention for Zarashpe Kapadia and Tom Lynch who have been most excellent technical advisors and without whom I would undoubtedly have thrown my computer out of the window on a number of occasions! Thank you also to the rest of Cohort 2, Gemma Brady, Philippa Usher, Ramzi Cherad, Zarashpe Kapadia and David Wyatt. It's been tough guys but we made it..!

I could also not have achieved this without the unending support of my family. Special thanks to Mum. You are amazing and I owe you so much. Your confidence in me has helped in those times when I really needed it, and your logistical support with school runs, packed lunches, dinners, teas, washing, ironing etc.. etc.. has been above and beyond. Neil, I am endlessly grateful for your help with all of these things and also the early morning and late night cups of tea that have kept me going! Impeccable footman duties! I probably owe you a few thousand brews... Thank you also to Karl for making me laugh when I have needed it most, for reminding me when needed that the world does not revolve around biochar, and for making me many nice teas en route. Thank you also for your patience and understanding when I have been working hard. Hopefully I will now be able to repay you in dinners ⁽²⁾. And Jovi, you have been my inspiration and motivation throughout, always making my world a little shinier and far happier. I would not have begun this journey without you, and I can't thank you enough. I hope that if, in the future, you find yourself struggling to achieve a goal, you can use this journey of ours as inspiration and cups of tea. Love you always.

Contents

Abstractiii				
Acknowledgementsv				
Tab	ole of t	figure	25	x
Tab	ole of a	acron	yms	xiv
1	Intro	oduct	ion	1
-	l.1	Rese	earch objectives	2
2	Back	kgrou	nd literature	4
4	2.1	Anth	propogenic climate change: A driver for biochar technology?	4
4	2.2	Bioc	har production, characterisation, uses and environmental impact	6
	2.2.	1	What is biochar?	6
	2.2.	2	Biochar production: Types of feedstock and production methods	9
	2.2.3	3	Characteristics of biochar	21
	2.2.4	4	Biochar uses	26
	2.2.	5	Biochar addition to soil	26
	2.2.	6	Sustainable biochar systems	38
	2.2.	7	Regulation of biochar production and deployment	39
2	2.3	Scen	narios of future biochar production and use	43
	2.3.	1	Biomass potential	43
	2.3.	2	Estimation of carbon storage/emissions reductions through biochar use	46
2	2.4	Futu	re projections of climate change	51
	2.4.	1	The representative concentration pathways (RCPs)	51
	2.4.	2	Global mean temperature and the RCPs	54
	2.4.	3	Crop yield impacts of RCP projected climate change	56
3	Cha	racte	risation of biochars from crop residues: Methodology	58
	3.1	Intro	oduction	58
	3.2	Mat	erials and methods	59
	3.2.	1	Feedstock selection	59
	3.2.2	2	Slow pyrolysis	60
	3.2.3	3	Analytical methodology	61
	3.3	Sum	mary	66
4	Cha	racte	risation of biochars from crop residues: Results and discussion	67
4	1.1	Intro	oduction	67
	4.2	Fe	eedstock composition	67

	4.3	Biochar yields and characteristics	70
	4.4	Oil yields and characteristics	83
	4.5	Gas characteristics	84
	4.6	The effects of pyrolysis conditions on product yields and characteristics	86
	4.6.	1 Product yields	87
	4.6.2	2 Proximate analysis	88
	4.6.	3 Ultimate analysis	89
	4.6.4	4 рН	91
	4.6.	5 Surface area	92
	4.6.	6 Biochar recalcitrance	93
	4.6.	7 Macro-nutrient content	94
	4.6.	3 Gas characteristics	95
	4.7	Limitations and opportunities for further work	98
	4.8	Summary and conclusions	99
5	Bioc	har production potential within the RCPs: Methodology	101
	5.1	Introduction	101
	5.2	Model overview	102
	5.3	Model development	103
	5.3.	1 Dominant crop residues	104
	5.3.2	2 Baseline data	105
	5.3.	3 RCP land-use data	107
	5.3.4	4 Crop production	107
	5.3.	5 Crop residues	109
	5.3.	6 Competition for crop residues	112
	5.3.	7 Biochar production	114
	5.3.	Biochar carbon content	115
	5.4	Alternative scenario drivers and assumptions	116
	5.4.3	1 Scenario 2: Crop yields	117
	5.4.2	2 Scenario 3: Land use	118
	5.4.3	3 Scenario 4: Residue to product ratio (RPR)	119
	5.4.4	Scenario 5: Crop residue availability	121
	5.4.	5 Scenario 6: Biochar yield	122
	5.4.	5 Scenario 7: Biochar carbon content	122
	5.4.	7 Scenarios 8 and 9: Climate change impacts (with and without adaptation)	122

	5.5	Scenario limitations and uncertainties12		
	5.6	Summary		
6	Bioc	Biochar production potential within the RCPs: Results and discussion		
	6.1	Introduction		
	6.2	Cropland within the RCPs	130	
	6.3 Scenario 1			
	6.3.1	1 Commodity production	138	
	6.3.2	.2 Residue production and availability	140	
	6.3.3	.3 Yield of biochar	144	
	6.3.4	.4 Carbon in biochar	152	
	6.4	Scenario 2: Changes in crop yields	156	
	6.4.1	.1 Scenario 2a: No crop yield increase	156	
	6.4.2	.2 Scenario 2b: Optimistic crop yield increases	158	
	6.4.3	.3 Scenario 2c: Crop yield increase convergence at 0.25 % yr ⁻¹ in 2050 for RCP 4.5	159	
	6.5	Scenario 3: Land use change	160	
	6.5.2	.1 Scenario 3a: No land use change	160	
	6.5.2	.2 Scenario 3b: Land for biofuels	161	
	6.6	Scenario 4: Residue to product ratio (RPR)	162	
	6.7	Scenario 5: Residue availability	163	
	6.7.2	Scenarios 5a to 5c: 25 %, 50 % and 75 % residue available for biochar conversion	on164	
	6.7.2	2 Scenario 5d and 5e: Residue availability assumptions of Woolf et al. (2010)	164	
	6.8	Scenario 6: Alternative biochar yield	165	
	6.9	Scenario 7: Alternative biochar carbon content	166	
	6.10	Scenario 8 and 9: Climate change effects on crop yields and the impacts of adaptati	on167	
	6.10	0.1 Effect of crop yield change uncertainty	174	
	6.11	Summary	178	
7	Bioc	char carbon sequestration potential: Methodology	181	
	7.1	Introduction	181	
	7.2	Methodology	181	
	7.2.1	.1 Rates of biochar application	181	
	7.2.2	.2 Carbon sequestration potential	182	
	7.3	Summary	186	
8	Bioc	char carbon sequestration potential: Results and discussion	187	
	8.1	Introduction	187	

8.2	Rate	es of biochar addition to soil	. 187
8.3	Car	bon sequestration potential (CS) equation	. 188
8	3.3.1	Total carbon storage	. 188
8	3.3.2	The Impacts of climate change on carbon sequestration potential	. 192
8.4	Rec	alcitrance from volatile content (RVC) equation	. 196
8	3.4.1	Projection of total carbon storage potential	. 196
8	3.4.2	Projection of carbon storage potential of different biochars	. 197
8.5	Car	bon sequestration potential – 2 pool method	. 199
8	3.5.1	95 Year sequestration potential	. 199
8	3.5.2	Longer-term carbon storage potential	. 200
8	3.5.3	Comparison of the three methodologies	. 203
8.6	CO2	Reduction Potential	. 204
8.7	Imp	act of Biochar on RCP CO ₂ Emissions Pathways	. 204
8	3.7.1	Scenario 1	. 204
8	3.7.2	Mitigation potential of the alternative biochar scenarios	. 206
8	3.7.3	Climate change impacts on emissions reductions	. 208
8.8	Unc	ertainties and limitations	. 210
8.9	Sun	nmary	. 212
9 C	Conclusi	ons and recommendations for further work	. 215
I. A	Annexe 2	1	. 222
II. A	Annexe 2	2	. 224
a.	Regior	nal classifications	. 224
b.	Crop y	vields within the RCPs	. 227
с.	Residu	ue availability	. 230
d.	Crop y	/ields	. 232
e.	Biocha	ar carbon content	. 232
f.	Climat	e change impacts	. 233
III.	Annex	e 3: Biochar production: Results tables	. 234
IV.	Annex	e 4: Carbon storage: Results tables	. 265
V. A	Annexe S	5: Figure licences	. 271
VI.	Biblio	graphy	. 274

Table of figures

Figure 2-1: Schematic, adapted from Sohi et al. (2009), showing typical biomass feedstocks,	
biomass thermochemical conversion processes, typical products (including biochar), and product	
uses	7
Figure 2-2: Schematic of the negative emissions process which may enable biochar systems to	
remove CO ₂ from the atmosphere and store it in soils for long time periods	8
Figure 2-3: Distribution of lignocellulosic components (cellulose, hemicellulose and lignin) in the	
plant cell wall	.11
Figure 2-4: Schematic of an example static batch slow pyrolysis reactor.	.15
Figure 2-5: Field trial where the effects of biochar on crop yield is tested	.27
Figure 2-6: Schematic of typical biochar degradation rates	.30
Figure 2-7: Overview of biochar carbon sequestration showing the process inputs and outputs	.38
Figure 2-8: Life cycle assessment for carbon abatement of a number of biomass systems	.49
Figure 2-9: Carbon abatement efficiency of waste biomass feedstocks as calculated using LCA	
(Shackley et al., 2010).	.50
Figure 3-1: Schematic of the slow pyrolysis reactor	.60
Figure 4-1: Correlation analysis of the typical feedstock lignocellulosic composition (%)	
(cellulose (top left), hemicellulose (top right) and lignin (bottom)) and the feedstock carbon	
content (%)	. 69
Figure 4-2: The relationship between phosphorus (P) and Magnesium (Mg) content (mg kg ⁻¹) of	
the feedstocks	.70
Figure 4-3: Correlation analysis of feedstock C, H, N and O content and corresponding biochar	
content	.72
Figure 4-4: Correlation analysis of feedstock moisture, volatile, fixed carbon and ash content (%)	
with corresponding biochar content.	.73
Figure 4-5: Van Krevelen Diagram of O/C and H/C ratios of raw feedstocks and biochars	.74
Figure 4-6: Macro-nutrient content (g kg ⁻¹) (dry basis (db)) of (a) raw feedstocks and (b) biochars	.76
Figure 4-7: Correlation analysis between biochar pH and feedstock potassium content and biochar	
potassium content	.78
Figure 4-8: Temperature programmed oxidation (TPO) profiles for biochars	.80
Figure 4-9: Thermograms indicating the oxidation temperatures of washed and unwashed wheat	
straw biochar samples	.82
Figure 4-10: Gas species concentration (%) for the syngas produced from each feedstock	.85
Figure 4-11: Yields of biochar, bio-oil and syngas from the pyrolysis of sugarcane bagasse at (a)	
temperatures of 400 °C. 600 °C and 800 °C and (b) heating rates of 5 °C min ⁻¹ . 20 °C min ⁻¹ and 50	
$^{\circ}$ C min ⁻¹ at 600 $^{\circ}$ C.	. 87
Figure 4-12: Proximate analysis (moisture, volatiles, fixed carbon and ash content) of (a) biochars	
produced from the pyrolysis of sugarcane bagasse at temperatures of 400 °C, 600 °C and 800 °C	
(top) and (b) heating rates of 5 °C min ⁻¹ . 20 °C min ⁻¹ and 50 °C min ⁻¹ (bottom) at 600 °C.	. 88
Figure 4-13: Elemental composition (carbon (C) hydrogen (H) nitrogen (N) sulphur (S) and	
oxygen (O) content (%)) of biochars produced from the pyrolysis of sugarcane bagasse at peak	
final temperatures of 400 °C, 600 °C and 800 °C (top) and heating rates of 5 °C min ⁻¹ 20 °C min ⁻¹	
and 50 $^{\circ}$ C min ⁻¹ .	. 89
Figure 4-14: Biochar pH in biochars produced from sugarcane bagasse at the different peak	,
pyrolysis temperatures and heating rates.	.91
r J - J	·

Figure 4-15: Temperature programmed oxidation (TPO) profiles of biochars produced from (a)
sugarcane bagasse at pyrolysis temperatures of 400 °C, 600 °C and 800 °C and (b) heating rates of
5 °C min ⁻¹ , 20 °C min ⁻¹ and 50 °C min ⁻¹
Figure 4-16: Macro-nutrient content (g kg ⁻¹) (dry basis) of sugarcane bagasse biochars produced
under different pyrolysis temperature regimes and different pyrolysis heating rate regimes
(bottom)
Figure 4-17: Syngas composition (%) from the pyrolysis of sugarcane bagasse at different
pyrolysis peak temperatures of 400 °C, 600 °C and 800 °C
Figure 4-18: Syngas composition (%) from the pyrolysis of sugarcane bagasse at different
pyrolysis heating rates of 5 °C min ⁻¹ , 20 °C min ⁻¹ , and 50 °C min ⁻¹
Figure 5-1: Overview of the model stages and inputs used to calculate the biochar production
potential for each future scenario
Figure 5-2: Map of average income groups used to classify current developed and developing
nations
Figure 5-3: Schematic showing the calculation stages of residue availability within the framework
model
Figure 6-1: Hectares of crop land used globally for crop cultivation in 2005 for all RCPs
Figure 6-2: Percentage grid cell area under cropland in 2000.
Figure 6-3: Total global cropland (Mha vr^{-1}) for the four RCPs over the scenario period of 2005 to
2100
Figure 6-4: Fraction of land area used as cropland in 2100 for RCP 2.6. RCP 4.5. RCP 6. RCP
8.5
Figure 6-5: Distribution of the different crop types by crop category in 2005
Figure 6-6: Plot detailing progress towards peak cereal crop yields
Figure 6-7: Plot detailing progress towards achieving maximum rice vields
Figure 6-8: Global cropland (Mha yr ⁻¹), by crop group, for the four RCPs over the 95 year
assessment period.
Figure 6-9: Total global commodity production (Gt vr ⁻¹) for the four RCPs over the 95 year time
period
Figure 6-10: Total commodity production (tonnes) in 2100 for all crop categories for RCP 2.6
(top left), RCP 4.5 (top right), RCP 6 (bottom left) and RCP 8.5 (bottom right)
Figure 6-11: Global annual crop residue yield (Gt yr ⁻¹) (solid line) and unused residue (Gt yr ⁻¹)
(dashed line) for the four RCPs over the 95 year time period for Scenario 1 140
Figure 6-12: Global annual residue and unused residue quantities by crop group for each RCP 141
Figure 6-13: Spatial distribution of total crop residues (Mt vr^{-1}) in 2100 for each RCP and total
unused crop residues (Mt vr ⁻¹) in 2100 for each RCP 143
Figure 6-14: Total biochar vield (Mt yr^{-1}) from all crop residue types assessed for the baseline
vear (2005) of all RCP scenarios for Scenario 1
Figure 6-15: Global annual biochar vield (Gt vr^{-1}) for the four RCPs as assessed under Scenario 1 145.
Figure 6-16: Regional biochar vield (Mt vr ⁻¹) from all crop residues in 2100 under the
assumptions of Scenario 1 for the different RCPs 146
Figure 6-17: Global biochar production potential ($Ct xr^{-1}$) differentiated by crop category over
time for the four PCPs under the assumptions of Scenario 1
Figure 6-18: Regional biochar vield (Mt vr ⁻¹) from cereal crop residues (excluding rice) in 2100
for the four RCPs under the assumptions of Scenario 1
Figure 6-19: Biochar yield (Mt yr ⁻¹) from rice crop residues in 2100 for the four PCDs under the
assumptions of Scenario 1
ussumptions of Sechario 1

Figure 6-20: Biochar yield (Mt yr ⁻¹) from fibre crop residues in 2100 for the four RCPs under the	
assumptions of Scenario 1.	.150
Figure 6-21: Biochar yield (Mt yr ⁻¹) from oil crop residues in 2100 for the four RCPs under the	
assumptions of Scenario 1.	.151
Figure 6-22: Biochar yield (Mt yr ⁻¹) from sugarcane crop residues in 2100 for the four RCPs	
under the assumptions of Scenario 1	.152
Figure 6-23: Total biochar production (Gt yr ⁻¹) and total carbon content (Gt yr ⁻¹) of the biochars	
produced from all residues in Scenario 1	.153
Figure 6-24: Biochar quantity (Mt yr ⁻¹) and carbon content (Mt yr ⁻¹) of the biochar produced from	
residues from each crop type, over time, for the four RCPs, under the assumptions of Scenario 1	.154
Figure 6-25: Carbon content (Mt yr ⁻¹) of the total biochar produced in 2100 for the four RCPs	
under the assumptions of Scenario 1	.155
Figure 6-26: Global biochar yield (Gt yr ⁻¹) for Scenario 1 and Scenario 2a	.157
Figure 6-27: Global annual biochar yield (Gt yr ⁻¹) for Scenario 1 and Scenario 2b	.158
Figure 6-28: Biochar production (Gt) over the 95 year period for the range of crop yields	
projected	.159
Figure 6-29: Total biochar yield (Gt yr ⁻¹) for RCP 4.5 for Scenario 1 and Scenario 2c	.160
Figure 6-30: Global biochar yield (Gt yr ⁻¹) for Scenario 1 and Scenario 3	.160
Figure 6-31: Global biochar production for Scenario 1 and Scenario 3b.	.161
Figure 6-32: Biochar production (Gt) over the 95 year period for each RCP	.162
Figure 6-33: Biochar production (Gt) over the 95 year period for each RCP with altered residue	
availability	.164
Figure 6-34: Total biochar production (Gt) for Scenario 1, Scenario 5d and Scenario 5e	.165
Figure 6-35: Biochar production (Gt) under the assumptions of Scenario 1	.165
Figure 6-36: Total biochar carbon content (GtC) under the assumptions of Scenario 1 and	
Scenario 7b	.167
Figure 6-37: Biochar production potential (Gt yr ⁻¹) for each RCP with no temperature change,	
minimum projected temperature change and maximum projected temperature change for each	
RCP with no adaptation measures (Scenario 8).	.168
Figure 6-38: Biochar production potential (Gt yr ⁻¹) for each RCP with no temperature change,	
minimum projected temperature change and maximum projected temperature change with the	
simple crop based adaptation measures assumed in Scenario 9	. 169
Figure 6-39: Total biochar production (Gt yr ⁻¹) in 2100 for Scenario 1 and the mean climate	
change temperature projection with no adaptation (Scenario 8b)	.170
Figure 6-40: Total biochar production (Gt yr ⁻¹) in 2100 for Scenario 1 and the median climate	
change projection with adaptation	. 171
Figure 6-41: Total cumulative biochar production (Gt) for Scenario 1, Scenario 9b and Scenario	
8b over the 95 year period.	.173
Figure 6-42: Annual biochar production (Gt yr ⁻¹) for the four RCPs under the assumptions of	
Scenario 1, Scenario 8b (the mean yield and temperature change scenario using the IPCC	
projections of yield impacts) and Scenario 8f (using the crop yield impacts of climate change	
detailed in Kyle et al. (2014)	.174
Figure 6-43: Total biochar production (Gt) for the four RCP scenarios under the assumptions of	
Scenario 1, Scenario 8e2, and Scenario 8d2	.175
Figure 6-44: Total biochar production (Gt) for the four RCP scenarios under the assumptions of	
Scenario 1, Scenario 9e2, and Scenario 9d2.	.176
Figure 6-45: Summary of biochar produced annually for the four RCPs under the different	
scenario assumptions for Scenarios 1 to 7.	.178

Figure 7-1: Relationship between feedstock volatile content (% daf (drv. ash free)) and biochar	
R_{50} index values	184
Figure 8-1: Annual carbon storage potential (GtC vr ⁻¹) of the biochar produced within the RCPs	
under the assumptions of Scenario 1 and the cumulative carbon storage potential (GtC) of this	
biochar over the scenario period	188
Figure 8-2: Total carbon storage potential (GtC) of Scenarios 1 to 7 using the CS equation of	
Zhao et al. (2013)	190
Figure 8-3. Total carbon storage potential of the RCPs under the climate change impact	170
assumptions of Scenario 8d and 8e	192
Figure 8-4: Total carbon storage potential of the RCPs under the climate change impact	172
assumptions of Scenario 9d and 9e	194
Figure 8-5: Total carbon storage potential of the RCPs under the climate change impact	171
assumptions of Scenario 8f	195
Figure 8-6: Carbon storage potential of the four RCPs under the assumptions of biochar scenario	175
1 using the two variations of CS equation	196
Figure 8-7: Carbon storage potential of the different biochar types under the assumptions of	170
hiochar scenario 1 using the two variations of CS equation	197
Figure 8-8: Annual carbon storage potential (GtC yr^{-1}) and the cumulative carbon storage	177
notential (GtC) (right) of this biochar over the scenario period	199
Figure 8-9: The effect of variance in the assumed half-life of a) the recalcitrant fraction and b) the	177
labile fraction of biochar on the long term carbon sequestration potential of the biochars produced	
in RCP 2.6 under the assumptions of Scenario 1	200
Figure $8-10$: The effect of variance in the size of the labile fraction of biochar on the long term	200
carbon sequestration potential of the biochars produced in RCP 2.6 under the assumptions of	
Scenario 1	201
Figure 8-11: The effect on the long term carbon sequestration potential of the minimum and	201
maximum assumptions (labile and recalcitrant fraction half-life and fraction size) of biochar	
recalcitrance for biochars produced in BCP 2.6 under the assumptions of Scenario 1	202
Figure 8 12: Carbon equivalent of the CO, emissions projections of the PCPs under the	202
assumptions of Sconorio 1, and the notantial reduction in amissions which can be achieved for	
assumptions of Sechario 1, and the potential reduction in emissions when can be achieved for	205
Eigure 8, 12: Detential amission reductions for DCD2.6, relative to the earbon amission nothway of	205
register a CD2 6	206
projection RCP2.0	200
Figure 8-14. The impact of chinate change on the carbon emissions reduction potential of the	
CS restertial	200
CS potential.	209
Figure 8-15: Summary of the mitigation potential of the blochar scenarios, in relation to the	010
carbon emission pathways of the RCPs as calculated using the CS equation.	213
Figure 1v-1: Potential emission reductions for RCP4.5, relative to the carbon emission pathway of	260
projection KCP4.5.	268
Figure 1v-2: Potential emission reductions for RCP6, relative to the carbon emission pathway of	200
projection KCP6.	269
Figure 1V-3: Potential emission reductions for RCP8.5, relative to the carbon emission pathway of	070
ргојесиоп кСР8.5.	270

Table of acronyms

BOM	Biochar quality mandate
CDM	Clean Development Mechanism
CDR	Carbon dioxide removal
CEC	Cation exchange capacity
CFR	Certified emissions reduction
CFI	Carbon Farming Initiative
CH4	Methane
CO_2	Carbon dioxide
CS	Carbon sequestration potential
CV	Calorific value
FBC	European biochar certificate
FAO	Early pearly section section Early and Agriculture Organisation
FAOSTAT	Food and Agriculture Organisation Statistical Database
FID	Flame ionisation detector
GC	Gas chromatography
GHG	Greenhouse gas
GIS	Geographic Information System
ha	Hectares
hø	Hectograms
HHV	Higher heating value
н	Harvest index
НТС	Hydrothermal carbonisation
IBI	International Biochar Initiative
IPCC	Intergovernmental Panel on Climate Change
LCA	Life cycle assessment
LRTAP	Long-range Transboundary Air Pollution
N ₂ O	Nitrous oxide
NPP	Net primary production
PAH	Polyaromatic hydrocarbon
RCP	Representative Concentration Pathway
RPR	Residue to product ration
RVC	Recalcitrance from volatile content
SOM	Soil Organic Matter
SRES	Special Report on Emissions Scenarios
SRM	Solar radiation management
SSA	Specific surface area
TCD	Thermal conductivity detector
TCRE	Transient climate response to cumulative carbon emissions
TGA	Thermogravimetric analysis
TPO	Temperature programmed oxidation
UKBRC	UK Biochar Research Centre
UNCCD	United Nations Convention to Combat Desertification
UNFCCC	United Nations Framework Convention on Climate Change
VCS	Verified carbon standard
VER	Voluntary Emission Reduction
WFD	Waste Framework Directive

1 Introduction

Emissions of carbon dioxide (CO_2) into the atmosphere, from sources such as industrial and domestic processes and deforestation, are a key driver of anthropogenic climate change, with CO₂ accounting for 76 % of global anthropogenic greenhouse gas (GHG) emissions in 2010 (IPCC, 2013). Mitigation pathways discussed in the latest IPCC assessment report which aim to keep the increase in global mean surface temperature below 2°C are characterised by substantial reductions in GHG emissions by 2050 which are achieved mainly through changes to energy systems and land-use (IPCC, 2014b). Carbon dioxide removal (CDR) technologies are often employed within these scenarios (typically after 2050 for large scale deployment), relying on technologies including bioenergy with carbon capture and storage (BECCS) (up to -20 GtCO₂ eq yr^{-1} in 2100) and afforestation technologies (up to -16 GtCO₂ eq yr^{-1} in 2100) to achieve negative emissions. Scenarios with high levels of mitigation without carbon capture and storage (CCS) technologies are projected to be far more costly than those with CCS (IPCC, 2014b). Biochar is one of a number of CDR technologies discussed as a potential methodology for removing CO_2 from the atmosphere and storing it, as a stable form of carbon, for long time periods in soils (Royal Society, 2009). This removal and storage requires a combination of natural and anthropogenic processes. Naturally, CO_2 is removed from the atmosphere through biomass photosynthesis. The biomass retains a large proportion of this carbon, releasing it back to the atmosphere through respiration or decomposition. Biochar systems aim to limit the release through decomposition by the thermal conversion of the biomass to a charcoal like material termed biochar. Thermal conversion processes include pyrolysis, the heating of the biomass without oxygen, or more traditional carbonization methods such as charcoal kilns. Biochar carbon is often much more stable than in the raw biomass, therefore biochar addition to soils may provide a long-term carbon storage sink (Lehmann et al., 2009). Discussion of the potential of biochar to improve agricultural soils exists within historic scientific literature including Retan (1915) and Trimble (1851). Interest in the carbon storage potential of biochar developed from research into high-carbon, highly fertile 'dark-earth' soils such as the 'Terra Preta' soils of the Amazon (Marris, 2006, Sombroek et al., 2003). Evidence of ancient addition of charcoal to these soils, adding to increased soil fertility, led to the publication of a number of key articles including Glaser et al. (2001), Lehmann et al. (2003), Lehmann and Rondon (2006), Lehmann et al. (2006) and Krull et al. (2008) from which the dedicated, global field of biochar research for carbon sequestration and soil amendment began. Prior to this the term biochar was used mainly in connection with charcoal production, for example by (Demirbas, 2001, Demirbas and Arin, 2002). Biochar literature has expanded in the past decade, to include analysis of production methodologies, biochar characteristics, biochar effects in soils and on vegetation, carbon storage potential, and overarching fields such as economic and life-cycle analysis (Lehmann and Joseph, 2009). This is still a developing field, where many important research contributions can be made. The overarching aim of this thesis is to contribute to the biochar and CDR research fields by providing projections of the long-term carbon sequestration potential of biochar from crop residues under four different land-use scenarios spanning from 2005 to 2100.

1.1 Research objectives

The research has used an interdisciplinary approach to assess the global potential of biochar from crop residues to sequester carbon up to 2100. The methodology aimed to achieve a number of sub-objectives, each contributing to the overarching objective of quantifying the carbon storage potential of biochars from crop residues using the land-use scenarios of the four Representative Concentration Pathways (RCPs). The sub-objectives are:

- To produce and characterise biochars from eight crop residues under uniform pyrolysis conditions, examining the effect of feedstock characteristics on subsequent biochar characteristics.
- To characterise biochars from one crop residue, sugarcane bagasse, produced under varied pyrolysis conditions (peak temperature and heating rate), examining the effects of process conditions on subsequent biochar characteristics.
- 3. To assess the recalcitrance of biochars produced using the R_{50} Index described by Harvey et al. (2012).
- 4. To examine the potential influence of alkali metal content on biochar degradation, assessing possible conservative recalcitrance estimates in high alkali biochars.
- 5. The development of scenarios of biochar production using the land-use projections of the four RCPs, exploring the uncertainties and variation in biochar characteristics and biochar systems on these production potentials.
- Development and evaluation of a new method for estimating long-term carbon storage, based on the CS equation of Zhao et al. (2013) and incorporating the experimental data from biochar characterisation.
- Assessment of the long-term carbon storage potential of the biochars produced within these scenarios, using the CS methodology of Zhao et al. (2013) and the two-pool methodology of Woolf et al. (2010).

Experimental and modelling techniques were used alongside current biochar literature to develop a number of scenarios exploring aspects of uncertainty and variation in biochar systems.

Experimental work adds to the current knowledge on biochar characteristics, documenting the characteristics of eight crop residue biochars produced under uniform slow pyrolysis conditions. Biochar characteristics, such as biochar yields, recalcitrance and CS potential were used to inform the development of biochar availability scenarios to 2100. Biochar from one crop residue, sugarcane bagasse, was characterised after production under a range of alternative pyrolysis conditions, investigating the effects of temperature and heating rate. The experimental research further investigated the potential of the R₅₀ recalcitrance index of Harvey et al. (2012), using temperature programmed oxidation (TPO) to investigate the influence of alkali metals on the thermal degradation of biochar.

Chapters 5 and 7 detail the development of a number of scenarios of biochar production and CS potential from 2005 to 2100. A small number of assessments of the potential for biochar production and CS are detailed within the literature, with a fraction of these studies projecting these figures into the future (see for example Lehmann et al. (2006) and Woolf et al. (2010)). Studies of biochar for carbon sequestration within the literature also do not examine varied pathways of land-use change, projecting mainly biochar potential with current land-use distribution (Harvey et al., 2012). This thesis provides an assessment of biochar CS potential to 2100, looking at the effects of changing land use, crop yields and climate upon factors including feedstock availability, biochar production and CS potential. The RCPs, four emissions scenarios and related socio-economic drivers, were used to project crop residue availability and related biochar production methods for comparison. Two of these methodologies were taken from the literature (Woolf et al., 2010, Zhao et al., 2013) and one method was developed using the experimentally derived feedstock volatile content, providing an alternative method of estimating the CS potential of biochars.

The thesis is structured as follows: Chapter 2 details the surrounding literature regarding biochar and the RCPs, Chapters 3 and 4 respectively detail the methodology and results of the biochar production and characterisation, Chapters 5 and 6 respectively detail the methodology and results for the development of biochar production scenarios, Chapters 7 and 8 respectively detail the methodology and results for the assessment of the carbon sequestration potential of the biochar scenarios, and Chapter 9 provides an overall summary and discussion.

2 Background literature

2.1 Anthropogenic climate change: A driver for biochar technology?

Anthropogenic climate change is, in large, caused by the increased concentrations of GHGs in the atmosphere which are occurring as a result of actions such as fossil-fuel burning and deforestation (IPCC, 2013). Atmospheric carbon dioxide (CO₂) concentrations have increased by 40 % from pre-industrial levels, reaching 391 ppm in 2011, with CO₂ accounting for 76 % of total anthropogenic GHG emissions in 2010 (IPCC, 2013, IPCC, 2014b). Increased GHG concentrations in the atmosphere affect the Earths radiative balance, with the impacts of GHGs and other natural and anthropogenic climate drivers measured by radiative forcing changes. Radiative forcing, usually measured at the tropopause in W m⁻², is the difference in incoming short wave radiation (sunlight) and outgoing long wave radiation and is defined as:

'..the change in net (down minus up) irradiance (solar plus longwave in W m⁻²) at the tropopause after allowing for stratospheric temperatures to adjust to radiative equilibrium, but with surface and tropospheric temperatures and state held fixed at the unperturbed values'

(Ramaswamy et al., 2001)

Changes in this radiative balance cause temperature changes in the atmosphere and can cause subsequent changes in other climate systems (Hartmann et al., 2013). Climate change is likely to have impacts which will vary spatially and temporally, and will also be dependent on the manifestation and the magnitude of the changes. The potential impacts of increased GHGs on the climate system are projected, to different levels of certainty and understanding, in the latest report of the Intergovernmental Panel on Climate Change (IPCC) which assesses the current scientific knowledge and evidence on climate change. Warming of the climate has already been detected, with resulting reductions in cover of snow and ice and increased sea-level (IPCC, 2013). The future severity of climate change will depend on the concentrations of greenhouse gases reached, and the sensitivity of the climate to these concentrations. With regard to the sensitivity of the climate to GHG changes, the equilibrium climate sensitivity is a measure of the increase in temperature which would be seen with a doubling of atmospheric CO₂, once the climate had reached equilibrium. Current estimates for the equilibrium climate sensitivity are a global mean surface temperature increase of between 1.5 °C to 4.5 °C, with a probability of at least 0.66 (IPCC, 2013). This highlights both the uncertainty in some areas of climate science and a need to further understand and mitigate potential changes (as mentioned previously, atmospheric CO₂ concentrations are now 40 % higher than in 1790). The potential manifestations of climate change, in addition to an increase in global mean surface temperature, include more frequent and more intense extreme weather events (including droughts and storms), increased or decreased precipitation locally and increased variability in weather (IPCC, 2014c). These impacts are expected to vary spatially and temporally with, for example, some regions expected to see impacts such as increased precipitation and other regions facing greater risk of drought. Changes in global mean surface temperature are already impacting food production systems, with, for example, reductions identified in production of wheat and maize in some main production regions. Some positive impacts on crop productions have also been seen in regions of higher latitude such as the United Kingdom and Northeast China (Porter et al., 2014). These positive effects may be attributed to a number or combination of factors, including more favourable temperatures during the crop development cycle and the CO₂ fertilization effect (see below and Section 2.4.3 for further discussion). Any positive effects of climate change which may benefit crop production, for example more favourable growing temperatures, are likely to decline with increasing magnitude of climate change, and with increased frequency and magnitude of extreme events (Porter et al., 2014).

Climate change research now has a number of fields, with areas including diagnostics of climate processes and projections of potential future climate change, mitigation processes to limit emissions of GHG and influence of other climate change drivers, and climate change adaptation which seeks to develop methods of resilience in the face of impending change. There are many complexities within each of these fields, which often include social and economic drivers alongside the physical science aspects. The IPCC acts as an advisory body which aims to provide the most current knowledge on climate change. Alongside its physical science report on climate change, it publishes a volume on the current knowledge of climate change impacts, adaptation and vulnerability, a volume on climate change mitigation and a synthesis report (IPCC, 2014a). The United Nations Framework Convention on Climate Change (UNFCCC) aims to develop international mechanisms to tackle climate change, with 195 parties to the convention. Mechanisms such as the Kyoto Protocol have been developed to achieve international goals such as emissions reductions and the development of adaptation funds (UNFCCC, 2014). There is often a sluggish response to the forecasts of climate scientists with many national and international targets to reduce emissions falling short of the reductions deemed necessary by scientists. In light of the often slow response of governments, businesses and communities to reduce emissions, a number of methods of climate engineering have been proposed and are at varied levels of research and development. There are many important considerations related to the development of these 'geoengineering' technologies, beginning with whether such developments may reduce the focus of actors to reduce emissions in the near term. Other considerations include the effectiveness, cost and ease of deployment of these technologies. In terms of the engineering and physical impacts of the geoengineering technologies, they are often divided into two categories; solar radiation management and carbon dioxide removal technologies (Royal Society, 2009). Solar radiation management (SRM) technologies aim to reduce the amount of incoming short wave radiation through various methods including reflectors in space, surface albedo modification and stratospheric aerosol injection. SRM technologies do not reduce the build-up of CO₂ in the atmosphere so other effects such as ocean acidification would still occur, and the SRM technology would need to be continually deployed otherwise an abrupt change in climate may occur. Carbon dioxide removal (CDR) technologies aim to reduce the amount of CO₂ in the atmosphere, through negative emission technologies such as bioenergy with carbon capture and storage (BECCS), biochar, ocean fertilization and enhanced rock weathering. As mentioned previously, these SRM and CDR technologies are in different stages of research and development, with a number of potential regulatory barriers to deployment also existing.

Biochar is one method of CDR which is currently deployed on the small scale, mainly for soil amendment purposes rather than CO₂ removal. The impact on soil and CO₂ sequestration potential of biochar is not currently fully understood with research into areas such as production methods, soil and plant effects, and the potential for larger scale deployment for CS currently dominating the biochar literature. Section 2.2 and 2.3 includes a discussion of the most current biochar literature, from production processes to the projections of CS potential.

2.2 Biochar production, characterisation, uses and environmental impact

2.2.1 What is biochar?

Biochar, a recalcitrant form of carbon made by the thermo-chemical conversion of biomass (Lehmann and Joseph, 2009, Brown, 2009), is defined by the UK Biochar Research Centre (UKBRC) as:

"...a carbon rich solid product of the thermal stabilisation of organic matter, that is safe and potentially beneficial when stored in soil.."

(UK Biochar Research Centre, 2011)

Biochar is increasingly discussed as a potential tool in areas such as negative emissions technologies, climate change mitigation, soil quality and food security (International Biochar Initiative, 2012d, International Biochar Initiative, 2012a, Lehmann and Joseph, 2009, Sohi et al., 2009). The production and use of biochar as a soil additive may offer an opportunity to tackle a number of these issues simultaneously (Lee et al., 2010).

Biochar can be produced from a number of biomass types including wood, grasses, energy crops, residues and wastes. Currently a number of types of biomass are used for biochar production either commercially or for research purposes, including wood, grasses and energy crops (International Biochar Initiative, 2011b). Production of biochar from materials such as municipal waste and algae is currently less common. Biomass can be processed via a number of thermal conversion routes including pyrolysis, gasification and combustion, amongst other methods (see Figure 2-1) (Bridgewater et al., 2001). Of these processes the production of biochar is normally associated with pyrolysis and gasification technologies which are conducted in conditions of zero or limited oxygen (see Section 2.2.2.2).



Figure 2-1: Schematic, adapted from Sohi et al. (2009), showing typical biomass feedstocks, biomass thermochemical conversion processes, typical products (including biochar), and product uses. These lists are not exhaustive but give an indication of the range of processes and applications of biomass conversion.

As illustrated by Figure 2-1, each biomass conversion process can have a number of products, for example the production of biochar by pyrolysis also yields oil and gas products which can be used as a renewable energy source (Fagbemi et al., 2001). The yields, quality and composition of these products, including biochars, are influenced by a number of factors including feedstock type, process type and operating conditions. Biochar production processes, and effects on biochar yields and characteristics, are discussed further in Sections 2.2.2 and 2.2.3.

Interest in biochar has been growing, in-part due to its potential to remove carbon dioxide (CO_2) from the atmosphere (see Figure 2-2). Biochar contains an enhanced carbon content compared to the feedstock. A high percentage of this carbon is recalcitrant, meaning it is more resistant to degradation allowing it to remain stable for timescales of hundreds to thousands of years (Lehmann et al., 2009). Due to this high recalcitrance, the long-term storage of carbon from biochar in soil has the potential to create a carbon negative system where CO_2 is removed from the atmosphere and stored (see Section 2.2.5).



Figure 2-2: Schematic of the negative emissions process which may enable biochar systems to remove CO_2 from the atmosphere and store it in soils for long time periods.

As well as the potential for CO₂ sequestration, there are also other potential benefits of the addition of biochar to soils, such as improvement in the quality of degraded soils resulting in increased crop yields through mechanisms such as reduced nutrient loss, increased water holding capacity, soil pH modification, and soil microbial habitat improvement. There may also

be reductions in soil emissions of other greenhouse gas species such as N_2O (International Biochar Initiative, 2012d). The effects of biochar in soil are discussed further in Section 2.2.5.1.

2.2.2 Biochar production: Types of feedstock and production methods

2.2.2.1 Types of feedstock

Biochar can be produced from a number of different biomass feedstock types, with the availability of feedstock, char production process and desired char characteristics all influencing the choice of feedstock. Biochar can be produced from biomass such as agricultural wastes, forestry residues, energy crops, manure and municipal waste (Sohi et al., 2009). Current commercial pyrolysis projects typically use forestry residues, crop residues (including straw, nut and rice residues), switch grass, bagasse (from sugarcane), olive waste, chicken and dairy manure, sewage sludge and paper sludge from the pulp industry (Das et al., 2008, Shinogi et al., 2002, Sohi et al., 2009, Yaman, 2004). Current commercial scale biochar production projects are small in number and often utilize available local waste streams. A number of research-scale projects have experimented with using different types of biomass feedstock to assess variance in biochar characteristics, such as yield and composition of biochar, and to explore research areas such as lifecycle assessments of biochar systems and the impacts of optimizing the pyrolysis process for either energy or biochar production (Gaunt and Lehmann, 2008). When determining the suitability of a biomass feedstock for biochar production, it is important to consider a number of factors including its biochemical content, contaminants, availability of the feedstock, and the economic costs of processing. The lignocellulosic composition (i.e. the cellulose, hemicellulose and lignin content) of the biomass feedstock influences the yield of oils, gas and char from a feedstock, which has implications for the type of biochar system and amount of fossil fuel offset that could be achieved from each biomass type (Antal, 1985).

As discussed previously, biochar can potentially be produced from biomass waste. Questions have been raised about the safety of adding these biochars to soils as well as biochars produced from other potentially contaminated feedstocks such as painted wood. Traces of contaminants, such as heavy metals, may remain in the biochar and be added to soil (International Biochar Initiative, 2012c). Heavy metal contamination is discussed further in Section 2.2.5.3. The use of some feedstocks may also be regulated under waste management regulations, creating a barrier for their use for biochar production. The regulation of biochar, including feedstock choice and addition to soils, is discussed further in Section 2.2.7.

2.2.2.2 Biochar production

Biomass can be converted using a number of thermal processes, such as pyrolysis, gasification, hydrothermal carbonisation. The yields and characteristics of the products of these processes vary greatly, with gasification and combustion products being low in carbon and high in ash and pyrolysis and hydrothermal carbonisation products being high in carbon and low in ash (Capareda, 2011). Biochar technologies span a range of technical complexity and a variety of scales, from individual pyrolysis/gasification cooking stoves to large scale bio-refineries (International Biochar Initiative, 2011a). The main technology currently discussed and used for biochar production is pyrolysis, of which there are again a number of methods. The conditions of pyrolysis can be optimized to favour different products and yields. Systems of energy production, such as gasification can also produce biochar high in ash and low in carbon and typically in smaller yields than pyrolysis. Typical yields for each process are discussed in the following sections.

Due to the optimal yields of biochar being from pyrolysis, this technology is discussed in most detail here, although an overview of all biochar production processes is given.

2.2.2.2.1 Pyrolysis

Pyrolysis is the thermal decomposition of organic matter in the absence of oxygen (Peng et al., 2011). Pyrolysis of biomass degrades the hemicellulose, cellulose, lignin and other organic components, producing products including gas, bio-oil and biochar. A number of different operating conditions can be used, resulting in different yields of these main products (Bridgewater et al., 1999). The operating variables include temperature, residence time, feedstock particle size, heating rate and pressure (DiBlasi, 1996). Of these, the main operating conditions usually controlled in pyrolysis, notably the heating rate and final temperature, have led to the classification of process into either 'fast', 'intermediate' or 'slow' pyrolysis. Typical biochar yields are around 12 %, 25 % and 35 % for fast, intermediate and slow pyrolysis respectively (Table 2-1), although these yields can be highly variable depending on factors such as feedstock type and other process conditions. Pyrolysis is used for a range of commercial applications including production of fuel for transport or storage (densification), as a precursor process to gasification, or for indirect co-firing with conventional fuels (where biomass is converted to gas, bio-oil or char and then co-fired in a traditional combustion chamber). (See Section 2.2.2.2.1 for further discussion of the different types of pyrolysis).

	Product yield (%)		
Pyrolysis Process	Biochar	Bio-oil	Syngas
Fast	12	75 (25 % water)	13
Intermediate	25	50 (50 % water)	25
Slow	35	30 (70 % water)	35

Table 2-1: Approximate yield composition (%) of pyrolysis products by process type (fast, intermediate and slow pyrolysis). Adapted from Sohi et al. (2009).

2.2.2.2.1.1 Pyrolysis chemistry

The three main constituents of biomass, collectively known as lignocellulose, give supporting structure to the roots, leaves and stalks of the biomass material (Figure 2-3). Lignocellulose is composed of cellulose, hemicellulose and lignin, with typically 38-50%, 15-30% and 23-32% composition respectively dependent on biomass type (Society for Biological Engineering, 2011).



Figure 2-3: Distribution of lignocellulosic components (cellulose, hemicellulose and lignin) in the plant cell wall. Adapted from National Science Foundation (2014).

Cellulose and hemicellulose are both carbohydrate molecules. Cellulose consists of long glucose polymer chains. Hemicellulose consists of shorter polymer chains of 5 carbon sugars such as arabinose, glucose, mannose and xylose (Scheller and Ulvskov, 2010)). Lignin is a non-uniform three dimensional polymer, made from propyl-phenol monomer units which are arranged differently depending on plant type. Structurally, the cellulose and hemicellulose polymers make up fibre like strands called microfibrils, which are arranged into larger groups of macrofibrils. The macrofibrils are surrounded by the lignin polymers which act as a glue and structural support (Lange, 2007).

Pyrolysis of biomass results in a number of parallel and consecutive reactions (Balci et al., 1993). The temperature, residence time, and feedstock composition, influences the types of reactions that take place.

Table 2-2: Approximate temperatures of the different stages of pyrolysis and gasification, and the typical main products of each stage (Preto, 2008).

Temperature (°C)	Process	Main Products	
< 200	Drying	H ₂ O	
230 – 250	Depolymerisation	Acetic acid, methanol, CO_2 ,	
		СО	
250 – 280	Torrefaction	Extractives, CO ₂ , CO	
280 – 500	Devolatilization	Organics, Tars, CO ₂ , CO	
500 - 700	Dissociation/Carbonization	CO, H ₂	
> 700	Gasification	H ₂ , CO	

Low temperature pyrolysis (~300°C) and long residence time favours biochar formation through de-polymerisation of the carbohydrates. Higher temperature pyrolysis (~400 °C to 700 °C) favours the production of oil through the release of volatile species. Further increase in pyrolysis temperatures (>700°C) leads to breaking (cracking) of many of the C-C bonds present and the formation of a gas (syngas) (Bridgewater et al., 1999, Lange, 2007). The syngas produced is typically composed of CO, CO_2 , H_2 and CH_4 in different concentrations, alongside smaller amounts of other hydrocarbon species, dependent on feedstock and process conditions.

Two phases of biochar formation during pyrolysis are discussed within the literature. Primary biochar formation begins at relatively low pyrolysis temperatures, after moisture is driven off. Depending then, on residence time, further reactions can cause the slow decomposition of the primary biochar and tars to produce a secondary, less reactive, biochar. Neves et al. (2011) reported that the first phase of pyrolysis, termed primary pyrolysis, is completed at temperatures below 500 °C, with formation of a primary biochar, permanent gas species and tars. Above these temperatures, primary volatiles may undergo secondary reactions, forming secondary products. A number of secondary reactions may take place, including reforming, cracking, oxidation, dehydration, polymerization, gasification and water-gas shift. During the secondary pyrolysis phase the biochar may catalyse the conversion of tarry vapours into both light hydrocarbon gas species and secondary char (through cracking and polymerization)

reactions respectively). During the secondary pyrolysis phase, biochar may also be converted to gaseous species through gasification reactions. Neves et al. (2011) discuss that the secondary conversion of volatile species occurs at a much faster rate than the secondary conversion of char. For optimization of bio-oil yields, a flash pyrolysis process produces the best results, as the very short residence time does not allow the secondary cracking of semi-volatiles into gaseous species (Demirbas, 2004b).

The proportions of cellulose, hemicellulose and lignin in the biomass feedstock also affect the yields of oil, gas and biochar as this influences the proportion of volatile and semi-volatile compounds which are released from the biomass feedstock to form oil and gas products and the quantities of stable carbon which remains in the solid biochar product (Sohi et al., 2009). Neves et al. (2011) also found that yields of pyrolysis products, including biochar, liquids, water, total gas and individual gas species, are highly dependent on the peak temperature of pyrolysis. They found that the yields and properties of pyrolysis products follow general trends in relation to temperature, despite the variety of biomass types, processes and reactors available.

2.2.2.2.1.2 Fast pyrolysis

Fast pyrolysis is achieved using high peak temperatures and short residence times to optimize production of oil or gas as the end product (Bridgewater et al., 1999). To optimize the process for liquid production, the residence time of product must be short (typically 1 to 5 seconds dependent on reaction temperature) so the further reactions which form gaseous products cannot occur. To maximise the yield of oil or gas products, the aim of fast pyrolysis is to limit exposure of the biomass particles to the lower temperatures that would form biochar (Bridgewater et al., 2001). The process does still yield small quantities of biochar at typically around 12 % (Table 2-1). Fast pyrolysis may, therefore, be more economically viable than slow pyrolysis for biochar production in systems where the production of oil and/or gas is in high demand for energy use (DECC, 2011, International Energy Agency, 2010).

A number of different pyrolysis reactor designs are used for fast pyrolysis, including fluid beds, circulating fluid beds, entrained reactors, rotating cone, ablative reactors and vacuum reactors (Bridgewater et al., 2001, IEA Bioenergy, 2011). The type of reactor used affects the yield and type of products due to differences in heat transfer, residence time and other factors (Bridgewater et al., 1999). In a fluidized bed reactor, the biomass is first dried and ground and then introduced into the reactor where it is passed through a granular material such as sand. The material is introduced at high velocity, which causes it to behave as a fluid. Heat is also added to the system. Fluidized beds crack the polymer chains of the lignocellulosic components,

resulting in degradation and the formation of oil, gas and char in varying proportions. Circulating fluidized beds operate on a similar principle but have shorter residence times for gas and vapours. Processing times in a circulating fluidised bed are often shorter than for the basic fluid bed reactor and they can be more effective on materials that are difficult to fluidize (Bridgewater et al., 1999). Ablative pyrolysis uses a hot reactor wall to pyrolyse the biomass. High pressure is used and biomass is fed along the reactor to the reactor wall. A rotating cone reactor involves biomass particles being fed into the bottom of a heated cone. The particles are then swept upwards along the side of the heated cone surface. The close proximity of the particles to the heated cone surface allows for heat transfer to the biomass (BTG, 2011). Vacuum pyrolysis, performed at very low pressure, limits the secondary decomposition reactions of the gaseous products. In a fast vacuum pyrolysis reactor vapour and gases are removed from the reactor by the vacuum pump into the condensers, limiting secondary reactions, and any biochar formed remains in the reactor chamber until the end of the process (Roy et al., 1990). Product yields from vacuum pyrolysis are not currently as high as other methods, and costs of equipment and operation may be currently higher than the other methods as vacuum pyrolysis is in the earlier stages of development than some of the other fast pyrolysis technologies (Bridgewater et al., 1999, Roy et al., 2011).

Biochar produced by fast pyrolysis reactors can typically either be collected for utilization elsewhere (for example as a soil amendment or fuel) or can be recycled and combusted to provide energy for the pyrolysis process (Bridgewater et al., 2001).

2.2.2.2.1.3 Slow pyrolysis

A detailed description of the slow pyrolysis process is given here as this is the process which is used for biochar production in the experimental section of the thesis. Slow pyrolysis occurs at lower temperatures than fast pyrolysis, and with longer residence time of the biomass in the reactor. The lower temperatures, longer heating rates and longer residence times of slow pyrolysis produces higher biochar yields and lower oil and gas yields than fast pyrolysis (Williams and Besler, 1996, Xu et al., 2011). There is still a large amount of variation in the yields and characteristics of biochars produced by slow pyrolysis, dependent on the slow pyrolysis reactor type, process conditions and feedstock type. Examples from the literature of biochar yields from slow pyrolysis are: 25 - 62 % yield using different types of charcoal kiln in research by Antal (2003) and 26 - 63% biochar yield dependent on process peak temperature and residence time by Peng et al. (2011). The lowest yields from Peng et al. (2011) were found at their higher experimental temperatures (450 °C) and long residence times (8 hours), highest yields were

found at lower temperatures (250 °C) and shorter residence times (2 hours). Demirbas (2004b) reported a reduction in yield of biochar (produced from olive husk and corncob respectively) of 56.4 % and 81.4 % when pyrolysis temperature was increased from 177 °C to 977 °C. Williams and Besler (1996) produced biochar yields of between 16.2 % and 60.8 % from the slow pyrolysis of wood at process temperatures ranging from 300 °C to 720 °C and heating rates of between 5 ^oC min⁻¹ and 80 ^oC min⁻¹. The study found that biochar yields were higher at lower temperatures and with lower heating rates, although the highest biochar yields (produced at 300 °C) may indicate incomplete charring of the biomass. Although it is well documented that increasing temperature results in decreased yield of char, it is also discussed that the quality of char (fixed carbon content and other characteristics) may increase with increasing process temperature. Lange (2007) discuss that depolymerisation of the hemicellulose and cellulose chains begins at low temperatures (< 200 °C) and a residence period of a number of hours is required for this reaction to be completed at these relatively low temperatures. At higher temperatures (~ 300 °C +) dehydration begins and a series of other reactions begin to form unsaturated polymers and biochar. If the process hold time is not sufficient for these reactions to occur fully then the biomass material may not be fully charred.

There are a number of reactor types suitable for the slow pyrolysis process, including fixed bed reactors, multiple hearth kilns, screw kilns, drum kilns and rotary kilns (International Biochar Initiative, 2011a, Williams and Besler, 1996).

In a static batch reactor the feedstock is pyrolysed in batches in a reactor chamber (see Figure 2-4). The process is not continuous.



Figure 2-4: Schematic of an example static batch slow pyrolysis reactor from Williams and Besler (1996).

The batch process typically results in long residence times of the biomass and products during the pyrolysis process. In the batch reactor shown in Figure 2-4, biochar is collected from the reaction chamber at the end of the pyrolysis process whilst the volatile and semi-volatile compounds are removed throughout the process with the nitrogen (or other carrier gas) flow. Batch reactors may contain more than one 'hearth'. The feedstock is introduced at the top of the reactor and moved from one hearth down to the next during the pyrolysis process in a continuous process. The rotary kiln slow pyrolysis system consists of a tilted rotating drum which moves feedstock along the kiln during the pyrolysis process through gravity. Research using rotary kilns has found that altering feedstock input rates and/or operating temperature can control production yields of oils and gases whilst keeping the yield of biochar relatively constant at around 20 – 24 % (Klose and Wiest, 1999). The drum kiln reactor also uses a feeding mechanism to allow for continuous operation. The basic drum kiln system is similar in design to the rotary kiln, except a series of paddles inside the kiln move the biomass along the kiln. This allows biomass to be pyrolysed faster than in the batch reactor process, but is still classed as a slow pyrolysis process. The screw kiln pyrolysis mechanism, also similar in design to the rotary and drum kilns, uses a screw mechanism to move the biomass along the kiln. The heat source can be provided externally or by adding a heated substrate such as sand to the kiln to heat the biomass. The screw feeder removes the char at the end of the rotary kiln, again oil and gas products can be collected for use as an energy source. These slow pyrolysis kilns, except for the batch reactor, typically operate with a biomass hopper which can provide a constant flow of feedstock for continuous operation.

2.2.2.2.1.4 Microwave pyrolysis

Microwave pyrolysis uses microwaves to directly heat the feedstock, or to heat carbon that has been added to aid the process, which then transfers heat to the biomass through conduction. Carbon may be added as it has high microwave absorbency, and so can aid the process if the feedstock does not have the desired absorbency. As the biomass begins to carbonize, this increases the microwave absorbency of the materials and microwaves begin to produce reactions within the biomass (Dominguez et al., 2007). Current research focusing on the microwave pyrolysis process, includes improving the efficiency of electricity to microwave conversion and also the optimization of certain microwave pyrolysis products, in particular the yield of H₂ gas within the syngas produced (Dominguez et al., 2007, Zhao et al., 2011).

2.2.2.2.1.5 Energy intensity of pyrolysis systems

There are a number of necessary energy inputs for a biomass pyrolysis system. Initially energy may be required to dry biomass. Moisture in biomass is dependent on the biomass type and any pre-treatment, but most types of biomass will need some drying (in some cases natural drying may be sufficient). Example moisture content of some biomass types are 8 – 20 % moisture (wheat straw), 50 - 80 % (rice straw) and 30 - 60 % (wood bark) (Basu, 2010).

The pyrolysis process also requires energy input to provide a heat source for the reactions to occur. This energy may be provided from an external source, or after an initial energy input, may be provided through processing and burning of the secondary products. An energy penalty may apply for upgrading the pyrolysis products, for example, bio-oil is often rich in oxygen, acidic and corrosive and further treatment is often needed which would incur an energy penalty (Beurskens et al., 2000).

2.2.2.1.6 Emissions from pyrolysis

Pollution that may result from pyrolysis can be placed into three categories, ash, liquid tars and gases (Jauhiainen et al., 2005). In most large scale pyrolysis projects, the gases produced would be captured for energy use, although the further processing of this gas may mean that some fractions are un-used and disposal would therefore be needed. Emissions from the pyrolysis of biomass would depend on both the type of biomass used and the process type and conditions. As a variety of feedstocks can be used, projects would need to be assessed individually to determine the gaseous species which may be released. Examples of emissions from some feedstocks are discussed further here as an illustration. An assessment by Jauhiainen et al. (2005) details emissions of volatile and semi-volatile species from the pyrolysis of olive pomace at temperatures of 750 °C to 1050 °C. Emissions of some volatile gas species changed with temperature, whilst others remained relatively constant. They found that with increasing temperature, methane production decreased initially and then increased to return to initial production levels. Ethene emissions decreased with increasing temperatures. Formation of semi-volatile compounds from olive pomace pyrolysis tended to follow a trend of increase with temperature, but the magnitude of production and increase rate for each compound varies with temperature. Increasing temperature of olive pomace pyrolysis increased conversion of biomass to oil and gas. These oil and gas products may be classed as pollutants if not utilized for energy production within the system, or collected and utilized or disposed of properly. Pyrolysis syngas has also been used to replace the N_2 atmosphere once the process is underway, with the N_2 flow reduced proportionally to the inflow of syngas. This would supply direct heat to the process,

although the presence of other gases in the reaction chamber can act as co-reactants in the process, altering the reactions occurring (Mante et al., 2012). Pyrolysis on a smaller scale may not have the technical capacity to capture the oil and gases formed and therefore emissions of some pollutants, such as CO and CO_2 , would occur (International Biochar Initiative, 2011a).

2.2.2.2.1.7 Scale and efficiency of pyrolysis technologies

Globally, there is a large variation in types of biomass pyrolysis system. In developed countries, the use of large scale pyrolysis systems (bio-refineries) is, and is projected to be, the predominant method of biomass pyrolysis. In developing countries more small scale, localised systems such as farm scale pyrolysis units are currently more likely to be used (Demirbas, 2009, Ramachandra et al., 2000). The efficiency and energy demand of each systems type differs and this has implications for the viability and productivity of biochar pyrolysis systems.

The efficiency of larger, industrial scale pyrolysis systems depends on a number of factors. Energy inputs to the system may differ depending on the type and location of the system, biomass may need to be processed by drying and/or grinding into smaller particle sizes (Zafar, 2011) and the harvesting and transport of biomass can add a substantial energy penalty to the life cycle of the whole system (Gaunt, 2012). The use of pyrolysis systems which operate on a smaller scale, for example farm scale, or can be mobilized to travel to a number of farms in a locality have been discussed as a possibility for deployment in some areas, particularly in developing nations or remote locations. Depending on the sophistication of these mid-sized systems, it may be possible to capture the oil and gas products for use within the system or as a fuel for other purposes. The close proximity of the reactor to the biomass source and potentially also to the biochar distribution site would improve the carbon footprint of the overall biochar system by reducing transport energy requirements (Gaunt, 2012).

A large portion of the world's population (over 2 billion people) currently use basic stoves or open fires for cooking and heating requirements which leads to emissions of pollutants which may affect health and climate. A transition to pyrolysis/gasification stoves would improve efficiency and decrease emissions of these pollutants (International Biochar Initiative, 2011a). The efficiency of traditional pyrolysis kilns for charcoal production may still be low compared to their theoretical efficiency, due to issues such as infiltration of O₂ into the reaction zone, where gasification would then convert a large percentage of the biochar to CO and CO₂ (Brown, 2009). This, alongside the pollution and inability to capture oil and gaseous products makes traditional kilns undesirable compared to other more advanced technologies. Yield of biochar is not the only measure of the effectiveness of a biochar pyrolysis system which must be considered. As detailed in Section 2.2.2.2.1 differences exist in not only the yield of biochar with change in reactor type and process conditions, but in the carbon content and other characteristics of the biochars. Where biochar is being produced for carbon sequestration purposes, the yield, carbon content and recalcitrance of the biochar are all important factors when determining the optimum pyrolysis system and process conditions. These factors highlight that there are many important considerations when determining the most suitable pyrolysis reactor(s) and wider system, including the availability and suitability of feedstocks, energy demands and outputs, potential pollution issues, desired biochar characteristics (if any), and the economic costs and benefits of the system.

2.2.2.2.2 Gasification

The gasification process primarily creates a syngas, mainly consisting of carbon monoxide (CO), hydrogen (H_2) and carbon dioxide (CO₂), with some methane (CH₄). The composition of syngas can be typically around 35 % CO₂, 30 % CO and 20 % H_2 with smaller amounts of other gases present (Haryanto et al., 2009). A number of reaction stages make up the gasification process (Hosoya et al., 2008).

The initial stages of gasification are the same as the pyrolysis reactions which occur at increasing temperatures and include drying, depolymerisation, devolatilization and carbonization (see Table 2-2). Following this, gasification reactions take place in the presence of oxidising agents such as oxygen (O_2) , air or steam. The volatiles and a portion of the biochar react with the controlled levels of oxidizing agent added to the reaction chamber to produce CO₂ and some CO. The CO is combusted to provide heat for the gasification reactions, whereas the char is reacted with steam to give carbon monoxide and hydrogen. The reversible water-gas shift reaction can reduce the CO content of the syngas, increasing H₂ content (Biomass Energy Centre, 2014a). Gasification can be classed as low temperature (~700 °C to 1000 °C) or high temperature (~1200 °C to 1600 °C) gasification. Syngas from low temperature gasification will have more hydrocarbon species present and can be burned directly or further processed to remove tars. Higher temperature syngas will have less hydrocarbon species and more hydrogen and carbon monoxide present. Further upgrading of the high temperature syngas, using the Fischer-Tropsch process can produce a synthetic diesel if the correct proportions of hydrogen to carbon monoxide are present (Biomass Energy Centre, 2014a).

The gasification of biomass, although optimized for syngas production, often yields small amounts of biochar, with yields reported in the region of 10 wt % from sugarcane bagasse, 10.9

wt % from mulberry stems, 12.8 wt % using cassava stems and 13.7 wt % using coconut shells (Rodriguez et al., 2009). 5 to 10 wt % yield of biochar from the gasification of switchgrass and corn stover was reported by Brewer et al. (2009). As with pyrolysis, the technology required for gasification can operate on a small scale or a large scale, ranging from cooking stoves to industrial scale bio-refineries (biochar.org, 2011).

2.2.2.2.3 Hydrothermal carbonization

Hydrothermal carbonization (HTC) is a process in which wet biomass is heated to between 170 and 250 °C, and subjected to increased pressure, in a closed chamber. The residence time of biomass within the chamber can range from hours to days. A number of conversion reactions take place during HTC, resulting a solid hydrochar, and process water containing nutrients and polar organics (Schneider et al., 2011). One of the main advantages of this process is that the biomass feedstock does not need to be dried before the subsequent carbonisation process, and is therefore particularly suitable for feedstocks with very high moisture content. Using this process, Schneider et al. (2011) produced a yield of 45 % biochar from bamboo biomass. Hydrochars produced by hydrothermal carbonization often have a high carbon content, but the recalcitrance of this carbon is often far lower than pyrolysis biochars (Schimmelpfennig and Glaser, 2012). Some studies have suggested hydrochars have an increased eco-toxicity when added soils due to adsorbed tars, for example Wagner and Kaupenjohann (2014) found that plant biomass growth was reduced in soils which had been amended with hydrochars.

2.2.2.2.4 Traditional carbonisation methods

Carbonisation is a general term used for a number of pyrolysis processes which resemble traditional charcoal manufacturing methods. Auto-thermal carbonisation is a group of simple processes used for small scale charcoal production and is widely used in rural communities. Kilns for charcoal production may be made from materials such as tin drums, concrete pipes or bricks, which provide an affordable, simple method of charcoal production (Okimori et al., 2003). Kilns designed in this way usually admit a small amount of oxygen to the process, in order to burn some of the biomass to supply the required process heat, therefore may more closely resemble a gasification process than pyrolysis (Brown, 2009). Yields of biochar produced using traditional kiln methods with forestry waste feedstock were found to typically produce 24 % biochar with a 76 % carbon content at 400 - 500 °C and 28 % char with 89 % carbon content at 600°C (Okimori et al., 2003).

2.2.3 Characteristics of biochar

Biochar characteristics can vary greatly and are affected by both the feedstock used and the process conditions. Within the literature there are still a number of uncertainties in the properties of biochar, the causal factors of some biochar properties, and also in the effects of biochar application to soil. Further research to examine these biochar characteristics and the effects of biochar application to soil is pivotal to assessing how much CO₂ could be removed from the atmosphere and stored using biochar systems. Sections 2.2.3 to 2.2.5 discuss the characteristics of biochars, the effects of biochar addition to soil and the potential feedback effects within the carbon cycle that may be seen from this addition. As pyrolysis is the main method currently used for biochar production the characteristics of biochars from pyrolysis are the main focus here, some notable effects on biochar characteristics and effects caused by other production processes are discussed where relevant.

2.2.3.1 Biochar properties

Different feedstocks and pyrolysis processes lead to different characteristics in biochars, such as yield, carbon content, elemental composition, structure, surface area, porosity and pore volume, pH and calorific value. Both the physical and chemical characteristics of biochars may be important depending on the end use of the biochar. Physical properties are discussed in Section 2.2.3.1.1 and chemical properties are discussed in Section 2.2.3.1.2.

A number of biochar properties can potentially be used to classify biochars including pH, volatile content, ash content, water holding capacity, density, pore volume and specific surface area (Kuwagaki and Tamura, 1990). Efforts to design a biochar classification system have been undertaken by a number of groups in order to enable biochar systems to be considered for large scale agronomic and carbon storage purposes. These classification systems are discussed further in Section 2.2.7. Chapters 3 and 4 further explore the production of, and variability in, biochars from different feedstocks and slow pyrolysis conditions through experimental research.

2.2.3.1.1 Physical characteristics of biochar

Biochar yield is influenced by the feedstock characteristics, with high lignin feedstocks producing higher biochar yields due to the increased thermal stability of lignin in comparison to hemicellulose and cellulose (Brown, 2009, Sohi et al., 2010). The physical characteristics of biochars are important as when biochar is added to soils, they may affect the soil properties such as soil structure, water holding capacity and also the microbial communities present in the soil. Characteristics such as the biochar structure, pore volume, surface area and water holding capacity can be affected by both the feedstock properties and pyrolysis conditions. The chemical transformations that occur during biochar formation occur along a temperature gradient, where biomass is converted to partly charred matter, biochar and then soot (Table 2-2).

Biochar structure tends to be more stable than the structure of its feedstock. During the pyrolysis process some restructuring of the elemental composition occurs, generally reducing the ratios of hydrogen (H) to carbon (C) (H/C) and oxygen (O) to carbon (O/C) highlighting increased aromaticity. This aromatic structure indicates an increased stability in structure (Downie et al., 2009). The research of Peng et al. (2011) found that both H/C and O/C ratios decrease with increasing pyrolysis temperature, indicating an increase in aromatic structure and a related increase in structural stability with increasing temperature (University of East Anglia, 2011). The composition and structure of biochar and other products may have been found to change with pyrolysis conditions such as heating rate and residence time (Asadullah et al., 2010, Peng et al., 2011). Research by Asadullah et al. (2010) discusses the effect of heating rate on yield composition, finding that higher pyrolysis heating rates typically yield more volatiles than lower heating rates. Williams and Besler (1996) examined the calorific value (CV) of biochar, finding that CV was not affected by the heating rate of the pyrolysis process. The average CV for biochar in the study was 32 MJ kg⁻¹. A number of studies within the literature discuss an increase in both fixed carbon and corresponding calorific value with increased peak pyrolysis temperature (Demirbas, 2004a, Peters, 2011). Surface area of biochars is influenced by both both feedstock and process conditions. Surface area is generally increased, from that of the feedstock, during pyrolysis with tars being removed and increasing porosity. Graber et al. (2012) reported a range of specific surface areas for biochars, from 3.6 m² g⁻¹ up to 242 m² g⁻¹. The initial feedstock structure (i.e. cellular and capillary structure of biomass) is often retained in the resulting biochar, with a high surface area feedstock typically producing a high surface area biochar. Process conditions also influence biochar surface area, with surface area generally increasing with temperature (Downie et al., 2009). Maximum biochar surface areas were identified by Uchimiya et al. (2011) at production temperatures between 500 °C and 900 °C. Further increases in surface area can be achieved by including a further activation process, for example steam activation, resulting in an activated carbon. At higher temperatures (~> 900 °C) pore structure may begin to break down resulting in reduced surface area.

Biochar may, when added to soils, alter the physical structure of the soil due to the biochars porosity and surface area characteristics. This can offer improved habitat to soil microorganisms. Warnock et al. (2007) discuss that many species of bacteria and fungi may be

22
protected from predators within the pore space of biochars. Microorganisms in soil perform a variety of functions relating to both soil properties and plant function. These functions include the decomposition of organic matter, nutrient cycling, removal of contaminants and the increase and decrease of greenhouse gas emissions from soil. The surface area and porosity characteristics of biochars can also make them useful adsorbents, where they can be used to immobilize chemicals and toxins whilst also potentially preventing the leaching of nutrients from soil thus increasing nutrient availability to plants (Warnock et al., 2007, Zheng et al., 2010). This could also, in some circumstances, be a detrimental feature as toxins and undesired species could accumulate in soils due to their immobilization by biochar. Herbicides can also, in some cases, be rendered ineffective by their adsorption to biochars (Graber et al., 2012). Biochar colour is another physical property which could be of potential import when added to soils. Biochar is typically a very dark coloured material which has the potential to modify the land surface colour if added to soils in large quantities and where vegetation cover is not constant. A number of studies have begun to examine the potential albedo effect of biochar. Meyer et al. (2012) determined a reduction of 13 - 22 % in the overall climate mitigation potential of a biochar system where albedo is incorporated into the calculation. Genesio et al. (2012) determined a reduction in surface albedo of 20 - 26 % when biochar from durum wheat was added to soils. They determined that a large decrease in this effect was seen in year 2 after application.

2.2.3.1.2 Chemical and nutrient characteristics of biochar

A number of biochar characteristics are chemical in nature, including the biochar composition, pH, calorific value and effect on soil nutrients. The chemical composition of biochars is dependent on the feedstock material and also process conditions (Krull et al., 2009). Sohi et al. (2009) discuss that the elemental composition of the feedstock has a large influence on the composition of the resulting biochar. Biochar elemental composition is also affected by pyrolysis conditions. Krull et al. (2009) discuss that one common feature of biochars produced at different temperatures is a high aromatic carbon content, but that composition and the uniformity of the biochar structure can vary greatly with process conditions.

Ultimate analysis of biochars gives information about the C, H, N, S and O content of biochars. From this, the ratio of hydrogen to carbon (H/C) and oxygen to carbon (O/C) is often calculated to give an indication of the aromaticity present. H/C and O/C ratios tend to decrease with increasing process temperature and increased periods of heating, resulting in a more aromatic structure. A literature review by Downie et al. (2009) of H/C and O/C ratios between various biomass feedstock and biochars detailed increasing aromaticity during biochar production.

The yield of carbon in biochar is affected by both feedstock composition and pyrolysis conditions. With relation to feedstock effects, carbon yield is related to the concentration of carbon in the feedstock and also the ash content. Feedstock with lower ash content tends to have higher biochar carbon content. Increasing pyrolysis temperature also increases carbon content, for example Tanaka (1963) saw a reduction in biochar yield of 41 wt % whilst carbon content was increased by 37 wt % when peak pyrolysis temperature was increased from 300 °C to 800 °C. Sohi et al. (2009) note that, generally, the carbon content of a biochar is inversely related to the biochar yield.

Biochar is also known within the literature for providing nutrient benefits to plants (Sohi et al., 2009, Yin Chan and Zhihong, 2009). These effects may be either through the direct supply of nutrients or via indirect attraction of nutrients and retention on the biochar surface resulting in reduced fertilizer loss. Chan and Xu (2009) discuss that nutrient content of biochars can be highly variable, reporting concentrations of total nitrogen (N) of 1.8 g kg⁻¹ to 56.4 g kg⁻¹, total phosphorus (P) of 2.7 g kg⁻¹ to 480 g kg⁻¹ and total potassium (K) of 1.0 g kg⁻¹ to 58 g kg⁻¹. Biochar nutrient concentrations are, again, dependent on both feedstock composition and process conditions. Different feedstocks naturally have large variation in nutrient concentrations, for example animal manures are high in total P and sewage sludge biochars tend to be high in total N. Although the total concentration of a nutrient may be high within a biochar, much of this content may be inaccessible to the plant. Potassium, an important macro-nutrient, has often been found to be in high plant available concentrations in biochars, resulting in increased uptake by plants after biochar addition to soils (Yin Chan and Zhihong, 2009). Chan and Xu (2009) analysed biochar content of both total and available P, finding a range of 0.2 to 73 g kg⁻¹ total P and 15 to 11,600 mg kg⁻¹ available P. The biochar with 0.2 g kg⁻¹ total P had 15 mg kg⁻¹ available P, and the biochar with 25.2 g kg⁻¹ total P had 11,600 mg kg $^{-1}$, indicating that available P increases with total P, although the limited reporting of either total P and/or available P makes trend determination difficult.

Biochar has also been seen to improve plant nutrient uptake by increasing the cation exchange capacity (CEC) of soils, improving the nutrient retention capacity of the soil. The effects of process conditions on biochar nutrient properties have been seen, for example, to alter biochar surface charges through increased process temperature. This would affect the CEC of the biochar, having implications for its nutrient retention effectiveness (Yin Chan and Zhihong,

2009). Biochars may also contain heavy metal species, retaining heavy metal species if present in the feedstock, or may act as an adsorbent due to this CEC mechanism creating a build-up of heavy metals in soils. This may cause toxicity to soils and plants if the biochar is added to soils, depending on the metal species and concentration. The risk of heavy metal presence in biochars is increased by using feedstocks, such as sewage sludge or painted wood, which may be contaminated. Conversely, evidence shows that biochar added to contaminated soil may immobilize heavy metals and other toxins therefore being a potential substrate for soil remediation (Afionis, 2011). Biochars may also act as a carrier to provide nutrients to plants, after being added onto the biochar through a separate process.

Other indirect nutrient effects, relating to changes in soil structure and function, involve the removal of constraints to plant growth by mechanisms such as improving water holding capacity and increasing soil pH. Biochar pH has been linked to increased plant growth due to the increase or maintenance of soil pH levels (Hoshi, 2001, Van Zweiten et al., 2007, Yin Chan and Zhihong, 2009). Alkaline biochars have the potential to buffer excess soil acidity. Application of biochar to soil has been found to have an overall liming effect, increasing soil pH (Biederman and Harpole, 2013), although knowledge of the initial pH of both biochar and soil is necessary to amend soil pH using biochar. The liming effect of adding alkaline biochars to soils can help to raise pH and overcome potentially toxic effects of acidic soils on plants. The addition of alkaline biochars to neutral or already alkaline soil may have a negative effect by suppressing nutrient availability to plants (Yin Chan and Zhihong, 2009). Increasing the pH of acidic soils has been seen to increase microbial activity, increasing soil organic matter mineralization and increasing nutrient availability to plants. This may, in some circumstances, cause a priming effect resulting in the increased emission of CO₂ from soils and may also have only a short term effect on microbial activity (Sohi et al., 2009). Previous literature reports biochar pH values (without further processing of the chars) of between pH 4 and pH 12, with typical values being above pH 7. Zhao et al. (2013) determined a biochar pH range of between 8.8 and 10.8, dependent on feedstock type. They also reported that the pH of biochars increases with increasing pyrolysis temperature due to the enrichment of ash as the temperature increases.

With regards to carbon storage, the variability of biochar properties has implications for estimating the amount of carbon that could be stored using biochar systems, and also for the accounting and monitoring systems that would be necessary to validate incentives for carbon storage using biochar, such as carbon credits traded within either the compliance market, for example Certified Emissions Reductions (CERs) or within the voluntary market, for example to

Verified Carbon Standard (VCS) credit) (see Section 2.2.7.2 for further discussion of biochar in carbon credit schemes). Sohi et al. (2009) discuss that the biogeochemical characterisation of biochar, alongside techniques to measure and track biochar in soil, will be necessary to enable large scale application and use, and to develop economic incentives such as incorporation into carbon trading schemes.

2.2.4 Biochar uses

The focus of biochar utilization within this study is for carbon sequestration. Financial or regulatory incentives for carbon removal and storage would help to drive scenarios where biochar is produced for carbon sequestration. Such incentives are currently limited by factors such as the need to develop accurate accounting methods for long term retention of biochar carbon in soils (see Section 2.2.7.2 for further discussion). Alongside CS and utilization as a soil improver there are a number of other commercial or theoretical uses and storage options for biochar. The suitability of biochars to be used, as produced or after upgrading, as an activated carbon or an adsorbent have been noted in the literature (Mohan et al., 2011). Utilization as an additive to construction materials has also been proposed (Okimori et al., 2003) and in some regions is current practice (Phonphuak and Thiansem, 2012). Biochar can be burnt to fuel the pyrolysis/gasification process or transported and used to fuel other processes (Williams, 2013). In future scenarios which are reliant on biofuels the use of biochar to fuel bio-oil and/or syngas production may be desirable. Biochar produced in excess of that which can be safely/technically applied to soils could be buried or stored (Shackley et al., 2010). Storing biochar, in disused mines for example, would allow the biochar to be utilized at a later date. This would add to the cost of the biochar system and is unlikely as no immediate economic benefit would be gained.

2.2.5 Biochar addition to soil

2.2.5.1 The effects of biochar in soil

Biochar, when added to soil, forms part of the soil organic carbon (SOC) pool. Soil organic carbon is the largest component of soil organic matter (SOM) and is mainly formed through the decomposition of plant and animal material, ranging from freshly deposited to highly decomposed material (Schumacher, 2002).

The addition of biochar to soils enables carbon from biomass to be stored in the soil carbon pool with more recalcitrance than carbon from un-charred material (see Sections 2.2.3.1.1 and 2.2.5.2 for further discussion of biochar stability). The biochar may also have a number of benefits to both soil quality and plant growth (Sohi et al., 2009). Adding biochar to soil can have co-benefits by improving soil quality by mechanisms such as improving microorganism habitat,

improved nutrient retention and cycling, and increased water retention. A number of studies have suggested that these improvements in soil quality and function, as well as the potential improvement in efficiency of fertilizer use, may lead to some increase in crop yields (see Figure 2-5) (Atkinson et al., 2010, Major et al., 2010, Van Zweiten et al., 2010). The variability of biochar properties, soil properties, environmental and climatic conditions, and plant requirements means that a uniform effect does not occur when biochar is added to soil.



Figure 2-5: Field trial where the effects of biochar on crop yield is tested. Crop yield is shown with no biochar (top left) and with high biochar content soil (top right). Soil profiles are also shown for each test plot with an untreated oxisol (bottom left), and Terra Preta de Indio soil (bottom right). The two samples were taken in close proximity in the Amazon Basin (Glaser et al., 2001, International Biochar Initiative, 2014).

As discussed further in Section 2.2.5.2, historical analogues for the addition of biochar to soil are seen in natural and anthropogenic additions of black carbon and biochar to soils. The Terra Preta anthrosols of the Amazon show increased fertility when compared with the fertility of surrounding untreated soils (Glaser et al., 2001). Characteristics of the Terra Preta soil are often

higher SOM, phosphorus content, pH and CEC than surrounding soils. The Terra Preta soils can support a more diverse array of crop and plant variety and have also been seen to increase crop yield in some, but not all, studies (Cornell University, 2006). The Terra Preta analogue can only indicate the effects of charcoal addition to soils. The production methods of Terra Preta soils are somewhat unknown, with a number of theories about their formation, but conclusive evidence is lacking. Research suggests that other additives, such as mineral residues and decomposed organic material, were often used alongside biochar to supplement soils (Woods, 2003). The original soil type and climatic conditions of the Terra Preta soils may have unique effects on the formation and evolution of Terra Preta soils, their effects on plant growth, and carbon storage potential. These effects may not be directly transferable to other soil types and climatic conditions.

Any differences in the effects of adding char to soils of different types and climatic conditions are currently not well researched and documented due to a limited number of studies undertaken, therefore transposing the available data to situations with different combinations of soil type, climate and biochar type is problematic and further studies to broaden this knowledge would be beneficial (Verheijen et al., 2010). A meta-analysis of current data by Verheijen et al. (2010) determined that there is an overall trend of increasing plant productivity with biochar addition, with the magnitude of this increase varying between soil and cultivar type, and with no impact or negative change seen in some cases. Acidic soils saw the greatest crop yield increase, potentially due to increase in soil pH from biochar addition, whilst a number of other soil types saw little or no increase. The meta-analysis showed a trend of increasing productivity as pH increased with biochar addition. This is described by Verheijen et al. (2010) as a possible liming effect. A liming value of approximately 30 % CaCO₃ was seen in a study where biochar produced from paper mill waste was added to soils by Van Zweiten (2012). This liming effect was attributed to Calcium mineral formation. The magnitude and longevity of this effect is currently unknown, and the effect of liming on alkaline soils may be detrimental to plant growth by increasing pH past the threshold for healthy plant growth. Further investigation into these effects could enable biochar addition to be targeted in areas where the greatest benefits would be seen, such as acidic soils or used with particular crop types and cultivars.

The removal of biomass for biochar production would mean that less SOC is formed through the natural decomposition of this biomass. The formation of SOC in this way is an important part of the carbon cycle and this reduction may have adverse effects that should be further identified before large scale residue removal should occur. The removal of plant material from soils may

also remove nutrients that would otherwise have been returned to the soil through decomposition. As discussed by Lindstrom (1986), a conservative estimate of 70 % residue is required to remain in situ for soil health and nutrient recycling and has been applied here for these in-field residues. This is a highly generalised figure. Acceptable levels of biomass removal must be identified to ensure that enough biomass remains to add essential nutrients and SOC to the soil in a sustainable manner. Evidence from preliminary research has also shown that the addition of biochar to soils may lead to an increased rate of decomposition in existing soil organic matter (SOM), potentially affecting emissions of greenhouse gases such as CO₂ and N₂O from soils (Verheijen et al., 2010).

2.2.5.2 Lifetime of biochar carbon in soil

Biochar carbon is often more recalcitrant than that of un-charred biomass due to the more stable structure of the biochar in relation to that of the raw biomass (see Section 2.2.3.1.1). The charring process tends to increase the aromaticity of carbon in biochar, making it more resistant to degradation, with the extent of this being dependent on both feedstock composition and structure, and pyrolysis conditions (Downie et al., 2009). The stability of biochar is also dependent on external factors such as aggregation within soil, soil type and climate (Foereid et al., 2011). A number of mechanisms can remove or degrade biochars in soil. Abiotic and biological degradation play a large role in the degradation of biochar in soils, but this degradation is thought to occur at a far slower rate than degradation of non-charred material (Verheijen et al., 2010). Examples of abiotic degradation include chemical oxidation, photo-oxidation and solubilisation whilst biological degradation examples include microbial incorporation or the respiration of carbon by organisms (Zimmerman, 2010). Biochar can also be eroded and/or washed out of soils, where it would still have carbon sequestration potential but may not have soil improvement qualities. Figure 2-6 is an example of the remaining carbon from un-charred and charred biomass over a five year period.



Figure 2-6: Example of typical biomass and biochar degradation. Figure shows the amount of carbon remaining, over time, in biochar and un-charred organic matter after addition to soil. The schematic is a generalisation of the recalcitrance of these materials, for illustrative purposes (Lehmann et al., 2006).

Although the un-charred biomass initially contains 100 % of the original biomass carbon remaining, compared with ~ 50 % for the biochar (with ~ 50 % lost as volatile and semi-volatile matter during the production process), the rate of decomposition of the biochar carbon is much slower than that of the un-charred biomass. Soil organic matter has a mean residence time of 50 years, whereas biochar may have a mean residence time of 1000+ years (Hammond et al., 2011). Much variation still exists in the estimates of biochar lifetime in soils within the literature. A number of studies have attempted to determine the longevity of biochars in soil, with particular focus on the lifetime of the biochar carbon. Determining these long timescales has obvious difficulties due to the long timeframes of any assessment period. A number of different methods have been applied to determine or estimate the longevity of biochar in soil using analogues, proxies, short scale laboratory tests, field experiments and modelling techniques (International Biochar Initiative, 2010). Due to the timescales required for long term carbon sequestration purposes it is not possible to undertake laboratory or field studies which span the full timescales considered. A standardised method of observing and accounting biochar ageing in soils does not yet exist and the difficulties in observing and simulating very long term changes has contributed to the uncertainties regarding the potential of biochar to sequester carbon in the long-term.

Historical analogues used to infer the lifetime of biochar carbon in soil include studies of soils where biochar has been added either anthropogenically or naturally in the past. This includes the study of the Terra Preta de Indio ('dark earth') soils of the Amazon and the Terra Preta Australis anthrosols of Australia (Glaser et al., 2001). These soils, having a very high carbon content compared to that of neighbouring soils, were anthropogenically altered by the addition of substances including biochar up to several thousand years ago. Carbon dating techniques have indicated that biochar in these soils may be stable for thousands of years (Kuzyakov et al., 2009). Amazonian Terra Preta soils have been dated to 6,850 years old (Acutuba, Brazil), 9,000 years old (Jaguariuna, Brazil), 1,775 ± 325 years old (Santarém, Brazil) and 740 to 2460 years old (Colombia and Venezuela respectively) (Appenzeller, 1998, Glaser et al., 2001, Lee et al., 2010, Saldarriaga and West, 1986). Australian Terra Preta soils contained biochar dated to around 650 and 1609 years old (Downie et al., 2011). The extrapolation of biochar lifetime in soils from research on the Terra Preta soils of South America and Australia, although useful, requires an air of caution. As the soils were often altered by the addition of other substances alongside the biochar addition, and soil and climate properties may influence the lifetime, these studies may not be representative of biochar addition to soils under different circumstances (Glaser et al., 2001). Natural historical analogues of biochar residues or 'black carbon' from forest fires have been found to date back 1000's to 10,000 years (Saldarriaga and West, 1986, Schmidt et al., 2002). Zimmerman (2010) calculated a lifetime of 266 to 1600 years, with 80 Gt of black carbon present in soils today and a rate of accumulation, from natural biomass burning events, of 0.05 to 0.3 GtC year⁻¹. This gives a half-life for black carbon from natural burning events of approximately 100 to 1000 years. (Zimmerman, 2010) also inferred from the amount of black carbon found in soils today, compared with that which should have been produced from natural burning events over time, that some losses of black carbon must be occurring.

As well as observing analogues for biochar degradation rates, researchers have also looked to simulate long term degradation over shorter time periods in laboratory incubation experiments and field experiments. Biochar degradation rates are often classified into two pools, looking at degradation of the labile (easily degraded) fraction and the recalcitrant (more stable) fraction separately (Foereid et al., 2011). The labile fraction of biochar will often degradation of the quickly, giving an initial impression of fast decay. After this initial period the degradation of the recalcitrant fraction is thought to occur over much longer timescales (Cheng et al., 2008). Within the literature, short term degradation experiments have been used to determine the rates of decay for the labile and recalcitrant biochar fractions, which have then been used to project the lifetime of the biochar (Lehmann et al., 2009). A number of studies have carried out short term

incubations at fixed temperature, in different mediums, to determine degradation rates. A range of results have been seen within the literature for these incubation tests. For example, one study by Hamer et al. (2004) saw 0.3 % and 0.8 % carbon loss for two biochars, which were produced at 800 °C and 350 °C respectively, over 60 days when incubated at 20 °C. Cheng et al. (2008) saw mineralization of 0.4 % of biochar carbon in a 50 day incubation experiment. Incubation for 48 days, of biochars pyrolysed at 350 °C and 400 °C, saw mineralization of 0.5 % to 4 % of biochar (Hilscher et al., 2009). Some longer term experiments have also been conducted, looking at the mineralization of biochar over periods of months to a few years. Although biochar properties due to feedstock characteristics cause variation in the effects of biochar in soils (see Section 2.2.3.1.1), it is discussed by Zimmerman (2010) that the conditions of pyrolysis are most important for the short term stability of biochar in soil. The study discussed that the rate of mineralization of biochar carbon slowed with increasing pyrolysis temperature. The carbon loss over 100 year period, extrapolated from the short term study, was 3 % to 26 %. The half-lives of the biochars assessed were determined to range from 10^2 to 10^7 years. Forestry experiments, where biochar was buried in porous bags, has also revealed large variation in carbon mineralization, from no mass loss to between 16 % and 51 % mass loss over ten years and two years respectively. Lehmann et al. (2009) discuss that these variations may be due, in large, to different analytical techniques as well as the inherent variation in biochars from different feedstocks and process conditions. This supports the theory that the labile fraction of biochar may decompose quickly, with degradation then slowing as the recalcitrant fraction remains (Lehmann et al., 2009). Harvey et al. (2012) used thermogravimetric analysis (TGA) to classify biochars by their stability in relation to that of graphite. They developed the R_{50} index which classifies biochars by the temperature of degradation, using the temperature at which 50 % of the biochar remains and the temperature at which 50 % of a graphite sample remains (see Section 3.2.3.4, Equation 3-1). Comparison of R₅₀ values with carbon mineralization rates led Harvey et al. (2012) to classify biochars, by their degradation potential, into three classes: $R_{50} \ge 0.7$, most recalcitrant; $0.5 \le R_{50} < 0.7$, minimal degradation; $R_{50} < 0.5$, more degradable. The R_{50} index does not give a quantity of carbon that will remain stable, or a timeframe for this stability. Further to the work by Harvey et al. (2012), the R_{50} index was used by Zhao et al. (2013) to develop a method of calculating the long-term carbon sequestration (CS) potential of a biochar (See Section 3.2.3.5, Equation 3-2). The CS potential equation determines the amount of carbon, from that of the original feedstock, which will remain stable for long time periods. The length of this storage period is not detailed in the literature regarding the CS potential, which is an area of potential further research.

Other methods of projecting the stability of biochar use modelling techniques, which may draw on data from the analogues and short-term studies discussed above. Efforts to model the lifetime of biochars in soils often use two mineralization rates, one for each of the labile and recalcitrant fractions. Woolf et al. (2010) used this two pool method to calculate the carbon storage potential of biochars, assuming a range of labile and recalcitrant fraction sizes and halflives (See Section 7.2.2.3, Equation 7-4). This method enables a time frame to be determined for the carbon storage potential of the biochars. Further research to reduce the uncertainty of the size of the two pools, and rate of decay of each pool would improve the accuracy of prediction using this method. Cheng et al. (2008) discuss that stability of biochar in soils reduces as mean annual temperature increases. They calculated a half-life of 80 years for biochar incubated at the temperature and moisture contents optimum for promoting degradation. They propose, therefore, that degradation in the natural environment would be slower (estimated at half-life of 925 years with a mean annual temperature of 10°C) than in their laboratory experiment, due to less than optimum conditions for degradation. Foereid et al. (2011), using a two-pool model to assess the lifetime of biochar in soils, determined that 9.8 %, 11.7% and 20.7 % of biochar would decompose within 2, 100 and 2000 years respectively. They also determined that 0.02 %, 0.13 % and 0.52 % would be lost due to movement down the soil profile and 49.1 %, 76.1 % and 74.8 % would be lost due to runoff over the same time periods (2, 100 and 2000 years). In total 58.9 %, 87.9 % and 96.0 % of biochar was degraded or removed from its position over 2, 100 and 2000 years respectively. These results highlight the importance of the erosion mechanisms in the lifetime of biochar. The method of application of biochar may also have a large impact on the rate of erosion from the point of deposition. The variation in biochar characteristics, combined with variation in soil types and climate makes the assessment of biochar carbon stability complex. Individual biochar systems could be assessed on a case to case basis, but for the purposes of large scale estimations of biochar stability and carbon storage some generalisations and assumptions about the characteristics of the biochars and the environment of biochar addition must be made.

2.2.5.3 Environmental impacts

2.2.5.3.1 Possible toxicity of biochars

There is some evidence that the type of feedstock or process type used for producing biochar can lead to toxins being present in the biochar and thus added to soils. Contaminants such as heavy metals that may be present in some feedstock types, such as some processed and waste feedstocks, may remain in the biochar upon conversion. There is also some evidence that some types and conditions of thermochemical process used for biochar production can lead to formation of toxic substances such as dioxins, furans and polyaromatic hydrocarbons (PAHs). PAHs are a range of organic compounds, of which a number are highly carcinogenic. Dioxins and furans can be present in soils, air and water. They can become concentrated along the food chain and may cause adverse health effects such as cancers and changes in hormone production in the body (DECC, 2011). Evidence of biochar toxicity in soils has been seen in some studies (Kookana et al., 2011). Freddo et al. (2012) studied the potential for toxic elements in biochars to exceed regulatory thresholds in soils and found that, for the biochars tested, levels were not likely to be exceeded with a biochar application rate of < 100 t ha⁻¹. Some studies also suggest that earthworm activity may be affected by the addition of biochar to soils, dependent on biochar type and application rate (Liesch et al., 2010). In a number of cases, earthworm activity has been seen to increase in biochar soils, but this is not true of all studies (Verheijen et al., 2010).

A number of standards for biochar classification have been developed by different biochar groups, including the Biochar Quality Mandate (BQM) of the British Biochar Foundation, the IBI Biochar Standards of the International Biochar Initiative and the European Biochar Certificate (EBC) of the European Biochar Foundation (British Biochar Foundation, 2013, European Biochar Foundation, 2013, International Biochar Initiative, 2012e). The guidelines developed for standardising biochar are often designed to ensure safe use of biochar, for example setting thresholds for permitted heavy metal contents. Guidance is given on suitable feedstocks, production methods and recording, and the laboratory testing of biochars. Biochar properties including the total and fixed carbon content, molar H/C and O/C ratios, volatile organic compound (VOC) content, major nutrient content (N, P, K, Mg and Ca), heavy metal content, pH, bulk density, moisture and ash content, and specific surface area must be reported and must adhere to the thresholds set for the biochar to gain certification. The requirements for assessment and reporting of positive biochar characteristics, such as nutrient content, are often optional, and where they are a requirement are generally specified as a declaration not within thresholds. Guidance is also set for when testing should occur.

2.2.5.3.2 Effects on emissions of greenhouse gas emissions

The addition of biochar to soils, as well as storing carbon away from the atmosphere, may affect emissions of CO_2 and other greenhouse gases (GHGs) from soils. Research is currently being undertaken to develop further understanding of the effects of biochar on direct emissions of GHGs such as methane (CH₄) and nitrous oxide (N₂O). Current research, although limited, has shown a reduction in emissions of N₂O from soils and an increase in CH₄ uptake by soils (Rondon et al., 2006, Yanai et al., 2007). Maximum emissions of N₂O were seen to decrease by 90 % compared to a control soil, in a seven day incubation study, and by 85 % when large amounts of biochar were added to soil by Yanai et al. (2007) Emissions of N₂O were reduced by between 21 % and 51 % following the addition of 40 t ha⁻¹ wheat straw biochar to a rice paddy. Rondon et al. (2006) determined that biochar properties play a large role in the effects on emissions of N₂O, with high temperature biochars significantly reducing emissions of N₂O but low temperature biochars increasing N₂O emission by >100 % from that of the control soil. Proposed mechanisms for the reduction of N₂O production through nitrification, changes in the de-nitrifier species composition and/or a reduction in N₂O from de-nitrification due to increases in soil pH (Van Zweiten et al., 2009).

Emissions of CH₄ from soils were found to be completely suppressed with applications of biochar produced from Calliandra, of the pea family, at 15 g kg⁻¹ and 30 g kg⁻¹ by Rondon et al. (2006). Conversely, Zhang et al. (2010) found that application of 40 t ha⁻¹ wheat straw biochar to a rice paddy increased CH₄ emissions by 34 % to 41 %. Proposed mechanisms for the reduction of CH₄ emissions and potential increased uptake to soils are the reduction of anaerobic zones suitable for the production of CH₄ by methanogenic bacteria and the stimulation of CH₄ uptake into soils through physical soil changes.

The literature also highlights a potential for emissions reductions through the improved efficiency of fertilizer use when added with biochar. Gaunt and Lehmann (2008) discuss that the reduction in N₂O emissions often seen with the addition of biochar to soils indicates an increased efficiency of fertilizer use by plants. As discussed in Section 2.2.3.1.2, the improved efficiency of plant fertilizer use is related to the improvement of soil cation exchange capacity through biochar addition. Scenarios within their research explore 50 %, 10 % and 0 % reduction in fertilizer requirement to maintain current yields. Schulz and Glaser (2012) found a significant increase in plant growth when biochar and fertilizer where added together, when compared to the yield increases of fertilizer addition alone. This suggests a potential for indirect reductions in GHG emissions due to the potential increase in the efficiency of fertilizers with biochar addition, meaning less fertilizer addition may be required for plant growth. Less energy would therefore be needed for the production and transport of fertilizers. Emissions associated with these processes, and also with the application of fertilizer to soils, would be reduced if a reduced fertilizer application rate is required with biochar (Lehmann et al., 2009). This effect on fertilizer efficiency may also be a key economic benefit to the use of biochar as a soil additive.

2.2.5.3.3 Climate effects of biochar use

The previous discussion has detailed a number of possible effects on soils and vegetation that may result from the production of biochar and its addition to soils. These biochar systems, if large enough in scale, may have wider influence on the climate.

The introduction of large scale biochar systems could lead to either an increase or decrease in GHG concentrations, depending on the design and sustainability of the system, and this may affect the climate through changes in radiative forcing. The initial production of biochar would see a removal of CO₂ from the atmosphere through plant photosynthesis and the subsequent conversion of this plant biomass to biochar (see Figure 2-2). Without conversion to biochar much of the CO₂ would be released back into the atmosphere during plant respiration and decomposition, and so carbon storage in biochar results in a net removal from the atmosphere. Depending on the type of biochar system (i.e. slow pyrolysis, fast pyrolysis, gasification) some of the original biomass carbon will be stored temporarily in the bio-oil and syngas products and released back to the atmosphere upon combustion. Transportation of the biomass and/or biochar within the system will lead to some emission of CO₂ and other GHGs depending on the type and distance travelled from the source of biomass, and production and end use of biochar.

A number of possible indirect effects regarding the addition of biochar to soils have been discussed in the literature. The addition of biochar to soil is reported to result in increased biomass production, through increased crop yields or through the re-use of degraded or abandoned land (Sohi et al., 2009) If this were to occur, this would result in an increase in the net removal of CO₂ from the atmosphere. As discussed in Section 2.2.5.3.2, the literature also discusses the possible effects of biochar on direct emissions of GHGs from soils, such as CO₂, CH₄ and N₂O. These effects are not currently well understood. More research is needed on the effects of biochar on different soil and plant types growing under different conditions, to reduce uncertainty and to improve the generalizations required to make meaningful projections. As discussed in Section 2.2.5.3.2, the possible increased efficiency of nutrient uptake and reduced fertilizer loss with addition of biochar may result in reductions in emissions, for instance, of N₂O (Gaunt and Lehmann, 2008). Studies quantifying these effects are limited in number, with generalizations often required to incorporate these systems and processes into large scale studies.

One of the main benefits of biochar systems is the potential for them to be carbon negative, resulting in a net removal of CO_2 from the atmosphere. There are circumstances where the biochar system may not be carbon negative. This, for example, would be in cases where the

carbon input (for example energy requirements for growing biomass, the pyrolysis process or to transport the biomass or biochar) to the system is larger than the amount of carbon sequestered by the system. Considerations made when designing a biochar system must include factors such as the energy intensity of feedstock production, of the pyrolysis system (including the amount and quality of oil and gas produced as fossil fuel offset), and the distance and mode of transport of both biomass and biochar. The energy demand of feedstock may not be a factor if the feedstock is a waste or residue product of a biomass grown for other purposes.

An example of biochar system assessments, where some systems have been carbon negative and others carbon positive, is the life cycle assessment study conducted by Roberts et al. (2010). The study showed occasions where the carbon balance of a switch grass biochar system was either carbon positive or negative depending on how land use was accounted in the study. As long as a biochar system has been properly designed by assessing the full life cycle of the process then it is feasible that the system can result in a net reduction in atmospheric CO_2 (Glaser et al., 2009). Figure 2-7 shows an overview of a biochar system, showing the different stages, inputs, processes and impacts on emissions involved. A number of different inputs and process scales can be applied, resulting in potentially very different impacts on the emissions balance of the system. This indicates that the net effect of biochar on atmospheric concentrations of greenhouses gases will, therefore, be dependent on a number of factors including the type and scale of system, the effects of biochar on crop yields and on soil and plant processes such as N₂O and CH₄ emission and fertilizer uptake efficiency. A further consideration of the impact on radiative forcing of biochar systems may be a decrease in surface albedo through the darkening of surfaces due the addition of biochar to soils (Meyer et al., 2012).



Figure 2-7: Overview of biochar carbon sequestration showing the process inputs and outputs. The schematic also shows possible factors within the biochar system which may affect climate, such as changes in emissions of GHGs including transport emissions, biofuel emissions, avoided fossil fuel emissions, and increased net primary production (NPP) (Woolf et al., 2010).

2.2.6 Sustainable biochar systems

The amount of carbon that can be sequestered by a biochar system is not the only consideration necessary for its design. The long-term sustainability of the system must also be considered in order to ensure no unforeseen damage to the environment, societal systems, or economic systems occur. Issues relating to the sustainability of the system that must be considered include the sustainability of the biochar feedstock supply, competition for land use, competition with food supply and the economic viability of the system (International Biochar Initiative, 2012a). Different scales of biochar system may encounter different sustainability issues, for example a large pyrolysis plant may be more energy efficient than a small or medium unit but may have a larger impact on the biomass resource, or larger emissions from transport demands than the smaller plants as the latter would tend to be located closer to the biomass supply and/or biochar distribution location. A number of the biochar guidance reports discuss issues of sustainability within the biochar system and hold this as a key assessment criterion (see Section 2.2.7.1).

A number of studies of biochar systems do not include traditional carbonization methods used currently in many developing nations to produce charcoal due to the emissions of black carbon and other pollutants to the atmosphere which are often associated with their operation. The development of sustainability criteria to introduce safer, cleaner small scale pyrolysis units in these areas could increase the potential of biochar to sequester carbon in a sustainable way on a more localized scale. This could be an important step towards sustainable development goals such as those in the Rio Declaration as it would also provide opportunities to increase food and energy security for those in developing nations whilst also potentially improving health through the improvement of localised air quality (United Nations Environment Programme, 1992).

2.2.7 Regulation of biochar production and deployment

2.2.7.1 Environmental and sustainability regulation

There are a number of regulatory considerations required when designing and operating a biochar system. Depending on the biochar feedstock and production method, and how current waste regulation is interpreted, biochar may be classified as a waste and regulated by legislation such as the European Union's (EU's) Waste Framework Directive (WFD) (Directive 2008/98/EC) (European Commission, 2011b). Also, the classification and regulation of biochar systems is important to ensure that they are operated in a sustainable manner. Regulation and monitoring is important to ensure that the feedstocks and processes used do not produce harmful toxins that may enter the soil or atmosphere. In order for the sequestration of carbon to be monitored, and perhaps incentivised, then further regulation and monitoring techniques need to be developed to allow this. Section 2.2.7, here, discusses the current regulation which may manage biochar systems, and then moves on to discuss gaps in regulation and how these could be approached as necessary.

Under current legislation, biochar from some systems may be classed as a waste product, and would therefore be subject to strict regulation on handling and disposal. This could inhibit the use of biochar from some systems as a soil amendment. If biochar is classed as a waste product then in would need to be managed in accordance with waste regulations such as the EU's WFD or the US's Resource Conservation or Recovery Act 1976 (RCRA) (Zhang, 2011). The wording of these legislations can be open to interpretation over whether biochar would, or would not, be classified as a waste. The EU has an inclusive list of wastes, listing those which are regulated, and the US has an exclusive list of wastes, listing substances which are not regulated. An example of how biochar may be classed as a waste under EU law is the inclusion of *'...wastes from incineration or from pyrolysis of wastes'*, which leaves the question of whether biochar is a

waste product of pyrolysis. If pyrolysis is conducted for the production of biochar then it is not a waste, but pyrolysis for the production of energy would yield biochar as a waste product, and current pyrolysis systems operate mainly for the latter purpose. If biochar is classified as a waste product under these guidelines then further consideration is needed to determine whether biochar could, subsequently, be re-classified from a waste to a desired end product of the process. Van den bergh, (2009) suggests that four further questions must then be considered for this reclassification: Will biochar be commonly used for a specific purpose? Does a market exist for biochar? Does biochar conform to existing technical and legislative requirements? Would the addition of biochar to soil cause adverse environmental or health impacts? In many cases it seems that a case could be easily made for biochar as a viable product rather than a waste, but the variation seen between biochar systems, inputs and outputs would mean that detailed assessment of cases may be required. A biochar system using virgin biomass material, and where biochar is a main product should not encounter these difficulties. As biochar systems evolve, and if biochar is recognised as a product of pyrolysis and not a waste, these problems of waste regulation may recede.

Another consideration regarding biochar as a waste is whether certain biochars would be classed as hazardous wastes. Certain feedstocks or production conditions may lead to the presence of toxins within the biochar, which could pose issues for soil, water or air contamination upon application to soils (see Section 2.2.5.3.1). Any biochar containing toxic substances such as PAHs would be regulated under legislation such as the Hazardous Waste Directive (Directive 91/689/EEC) (European Commission, 2011b). As this is dependent on feedstock and process type the IBI propose standards for testing of biochars from pyrolysis plants on a basis of feedstock type, feedstock throughput volume and/or annual testing to ensure that safe levels are maintained (International Biochar Initiative, 2012e). Due to the scientific and regulatory barriers, biochar testing regimes and updated regulation would be needed if feedstocks such as municipal wastes or sewage sludge were to be used to create biochar for addition to soils due to the increased risk of contaminants entering the system. The regulation of sustainable feedstock supplies for each biochar system will depend on the type of feedstock used. Some mechanisms may currently allow for the regulation of sustainable use, for example, feedstocks from forestry waste may be regulated by guidelines such as the Forest Stewardship Council's standards for forest management (Forest Stewardship Council, 2014).

There are, to date, no legally binding environmental or sustainability regulations specifically for biochar systems. Examples of aspects of biochar systems which may be regulated by existing

legislation are air and water quality requirements. The Sustainability Protocols designed by the US Biochar Initiative (2011) highlight a number of areas where existing regulation may be able to provide some guidance, discussing that biochar systems should adhere to the following principles:

'...biochar production does not contaminate water and utilizes water resources efficiently.'

and

'...biochar production and use improves air quality and does not lead to increased air pollution as compared to fossil fuels.'

(US Biochar Initiative, 2011)

A number of environmental and sustainability principles may, therefore, be regulated by existing wider guidance and legislation at either national or international level. For example, water pollution is regulated nationally and multi-nationally in many regions, and air pollution is regulated both nationally and internationally through a number of mechanisms such as the UKs Clean Air Act (Crown, 1993), the Montreal Protocol (UNEP, 2012), the Kyoto Protocol (UNFCCC, 2012) and the Geneva Convention on Long-range Transboundary Air Pollution (LRTAP Convention) (United Nations Economic Commission for Europe, 1979). These regulatory systems may apply to different points of the biochar systems, with, for example, air quality regulation likely to apply to the biochar production process and water quality regulation more likely to apply to factors surrounding the addition of biochar to soils. Any potential environmental impacts of a proposed biochar system must be considered and necessary regulatory guidance taken into consideration.

As discussed above, a range of environmental and social aspects of biochar systems may be regulated by existing regulation. There may be regional issues where some nations do not have adequate regulation or enforcement procedures in place to ensure that biochar systems are conducted in a sustainable manner. Where national, rather than international, regulation will guide the biochar system there may also be some differences between these national guidelines leading to regional differences in standards for biochar systems.

2.2.7.2 Incorporation into carbon credit/reduction schemes

In order to optimise the viability of biochar systems, there is a need for mechanisms to support and incentivise these systems. There are three main options to develop the economic viability of these systems: compliance market carbon credits, voluntary market carbon credits or adding economic value through soil benefits. One example of this is inclusion into emissions reduction or carbon removal and storage mechanisms such as the United Nations Framework Convention on Climate Change's (UNFCCC's) Clean Development Mechanism (CDM). A number of nations have requested to the UNFCCC that biochar is included into such mechanisms to incentivise CO₂ removal and carbon storage. In order for biochar to be successfully incorporated into schemes such as carbon credit schemes there needs to be further development of the monitoring schemes and quantification techniques of biochar over time to determine how much carbon is being stored in each system. This would involve analysing the carbon content of biochars, monitoring rates and location of addition to soils, and further research into quantifying the lifetime of biochar carbon in soils. The variation seen in biochar characteristics from different feedstocks and processes also adds another layer of complexity to this. Such monitoring schemes could potentially incorporate quantification of emission reductions from reduced fertilizer requirements and soil greenhouse gas emission changes. Any fossil fuel offset achieved by the system could also be credited, along with the negative emissions from the photosynthetic removal of CO_2 from the atmosphere. These additions to carbon credit calculations would further incentivise the use of biochar systems. There remain a number of challenges to the development of the monitoring techniques necessary for a robust, accurate methodology to be installed. A 'sister' framework convention to the UNFCCC, the United Nations Convention to Combat Desertification (UNCCD), has set goals for the development of biochar technologies and stated its support for inclusion of biochar into future climate mitigation agreements. To reach these goals, the UNCCD recognises that classification of biochar within the UNFCCC's definitions such as additionality, permanence and leakage are necessary (International Biochar Initiative, 2012b, UNCCD, 2009). Determining the permanence of biochar carbon, when added to soils, involves a number of uncertainties and assumptions. The various methods currently used to determine the lifetime of biochar are discussed in Section 2.2.5.2. Calculations of the permanence of biochar under schemes such as the UNCCD would also require the determination of biochar lifetime under a changed climate, which is currently an area of scant literature. Determining the additionality of biochar systems, that is the extra impact the system has on carbon storage which would not have otherwise occurred, may also be very difficult to document. Steiner (2010) details that the addition of biochar to soils leads to the addition of carbon to the inactive carbon pool, rather than the active pool which is currently the main focus of similar projects (including afforestation and reforestation) for the UNCCD. He discussed,

therefore, that these issues of addition, leakage and permanence may not be of such high priority.

The voluntary market currently includes schemes such as the Verified Carbon Standard (VCS) which has been developed by a number of partners including the World Economic Forum and The Climate Group (VCS, 2014). The VCS scheme allows businesses and schemes which are able to sequester carbon (which would not otherwise have been sequestered) to gain carbon credits for each tonne of carbon stored. These credits can then be sold to other businesses or interested parties who wish to reduce the carbon emissions footprint of their activities. These systems operate on a voluntary participation basis. A number of other voluntary trading schemes exist including the Gold Standard Voluntary Emission Reduction (VER) credits (Gold Standard, 2014). Currently methodologies for the assessment of biochar systems within voluntary carbon schemes are rare. The VCS and VER schemes detailed above do not yet have approved methodologies for biochar systems to be accountable. One example of a carbon reduction scheme which has incorporated a biochar methodology into its potential project portfolio is the Carbon Farming Initiative (CFI) in Australia (Australian Government, 2014).

2.3 Scenarios of future biochar production and use

2.3.1 Biomass potential

The biochar literature details a limited number of assessments which have been conducted into different aspects of biochar potential. These include assessing the sustainable availability of biomass to produce biochar, the carbon storage potential of biochar under particular economic and physical constraints and the comparison of biochar systems with other types of energy production scenario. The assessments of biomass availability range from a localised assessment of a particular project, to a large scale assessment of one or more biomass types regionally or globally. The assessments of the potential biomass resource can be allocated into two categories, demand driven and resource focused assessments. Demand driven assessments are used to estimate the amount of biomass that would be needed to fulfil a particular requirement, such as a biofuels obligation, or to analyse the competitiveness of different biomass based fuels or processes. Resource based studies focus on the available resource (Berndes et al., 2003, Ericsson and Nilsson, 2006). A review of studies of biomass potential, both demand driven and resource based, for the potential bio-energy production was undertaken by Berndes et al. (2003) and shows estimates ranging from 47 EJ yr⁻¹ to 450 EJ yr⁻¹. Although these studies are focused on biomass for bio-energy, the methodology used is also applicable to biomass resource for biochar production. The literature study relates the large variation in estimates of biomass

potential to differences in the methodologies and the assumptions made, including assumptions of land availability and crop yields.

Resource based methods will be the focus of further discussion here, as an assessment of the available crop residue resource for biochar production is a key aim of this study. An example of a small scale resource based study assessed the available biomass resource for a number of power plants in the Tennessee area of the US (Noon and Daly, 1996). The study used a Geographic Information System (GIS) platform to develop a map of the biomass resource available to the Tennessee plants. This GIS platform then provided economic data for biomass supply, related to factors such as the biomass type, location of source and end use of biomass and related infrastructure (e.g. transport). This kind of methodology would be useful to assess the feasibility of individual biochar projects or pyrolysis plants, with regard to factors such as sustainability and economic viability of supply. An example of using a global scale methodology to assess biomass resource would be to assess the available land that is, or could be, utilized for biomass production, and then applying biomass productivity for different biomass types relating to this available land area (Hoogwijk et al., 2009, Hoogwijk et al., 2005, Lehmann, 2007). This methodology has been used to estimate the current global biomass resource, and could also be used to project scenarios for the available biomass resource in the future. As discussed in Section 2.2.6, when assessing biomass potential, the criteria used must apply sustainability principles throughout biomass production and utilization (Rogulska and Kunikowski, 2006). This means finding a good balance between environmental, economic and societal issues, which in practice may be very difficult and is often ill-defined. In order to address these issues of sustainability within biochar scenarios, a number of factors should be considered when making assessments of available biomass. A key issue is that such an assessment must account for competition for the resource, such as for manufacturing, food production, bio-energy production, forestry and other land uses (De Meester et al., 2011). The level of inclusion and detail of these factors will affect the accuracy of biomass resource estimates (Berndes et al., 2003). Factors such as land area for production and distribution, crop yields and resource competition, particularly for future scenarios, are also often difficult to project and quantify. Assumptions and generalisations made regarding these parameters can add significant uncertainty to the projection and are often a main cause of differences between studies (Slade et al., 2011). Other variations between studies may arise from differences in the foci of the studies. Studies to assess the potential of biochar have often focused on one feedstock or one type of feedstocks (e.g. forestry waste), or a particular location, to make an assessment. This often makes the assessment more manageable, but comparability and transparency between

studies is often difficult due to the different considerations within the studies. A number of previous studies began by assessing the full biomass resource, and then subsequently applying the limiting factors such as competition for the resource. Hoogwijk et al. (2005) made an assessment of the global biomass resource after identifying gaps and issues with previous assessments, which they attempted to address. They discussed that previous assessments had a mainly regional aggregation, which may have led to oversights in the spatial distribution of biomass. This could have had implications for the economic viability assessment of the biochar systems as a large proportion of the cost is often transportation costs (McCarl et al., 2009). Hoogwijk et al. (2005) also discussed that the inclusion of land use competition is an important factor in the assessment. The work used an assessment of land use scenarios (adapted from the IPCC SRES scenarios (IPCC, 2000)) to determine the geographical potential of the biomass resource globally. They use a method of 'five categories of potentials' to assess the availability of biomass (for bio-energy production) by including assumptions at different levels, including economic and social factors. The five categories of potentials used for the assessment were:

I. The theoretical potential

This is the theoretical upper limit for total Net Primary Productivity (NPP) of biomass at the earth's surface produced by photosynthesis.

II. The geographical potential

This is the potential land that is available for the production of biomass. A 'land claim exclusion factor' was applied at grid cell level to estimate the land that would be available for biomass production.

III. The technical potential

The technical potential accounts for efficiency losses during the conversion of primary biomass to product (energy/stored carbon).

IV. The economic potential

The economic potential assesses the technical potential that can be utilised whilst maintaining profitability.

V. The implementation potential

The implementation potential uses the economic potential and applies factors to account for the timescale of implementation, for example incorporating governance barriers and incentives.

Making an assessment using all five levels of potential is a very detailed, lengthy task, but the number of steps included in an assessment could greatly influence the resulting amount of

biomass predicted within the scenario. The Hoogwijk et al., (2005) study went to level 3, the technical potential.

Projecting biomass scenarios into the future is possible using the same methodology as the Hoogwijk et al. (2005) 'categories of potential' assessment, but complex and often uncertain factors such as the effects of future land use change and future policy implementation must be estimate and incorporated. An example of this is the Rogulska and Kunikowski (2006) study of EU energy crop potential to 2030. The study made assumptions about future environmental regulation and obligations, future market development and future regulation such as agricultural and renewable energy legislation. One of their main findings was that much of the land available for future bio-fuel production could come from a reform of the EUs Common Agricultural Policy (CAP) (European Commission, 2011a). The necessity for assumptions made during the scenario development make the research more a projection of a possible future that a prediction of the likelihood of a certain outcome. The inherent assumptions related to making future projections make producing projections of global resources difficult as socio-economics, regulation and environmental constraint will change both spatially and temporally. The further into the future the projection timescale, the more difficulty and uncertainty arises in projecting what changes will take place.

2.3.2 Estimation of carbon storage/emissions reductions through biochar use

Estimations have been made within the literature, using various methodologies, for the potential of biochar to reduce emissions of CO_2 and other GHGs. The research which, using resource based approaches, provides projections of emissions reductions using biochar systems and shows a great deal of variation in results dependent on the methodology, assumptions and uncertainties of each assessment.

One assessment, which focussed on the sustainability of biochar systems and accounted issues such as food security and soil conservation, estimated a maximum total net emissions reduction of 1.8 Pg CO_2 -C_e yr⁻¹ and a 130 Pg CO_2 -C_e over 100 years. The study included emissions of CO_2 , CH₄ and N₂O, was global in scale and included a number of feedback affects in the analysis (Woolf et al., 2010). The research involved the development of an assessment framework incorporating the effects of a number of processes, but did not include factors such as biochar under future climate change, land use change, technological development or population changes. A study of biochar potential by Lehmann et al. (2006) included the potential of biochar to sequester carbon from charcoal production wastes, bio-fuel production, agricultural wastes, forestry croppings and alterations to current shifting cultivation practices. They calculated that

0.56 Pg C yr⁻¹ could be produced from these feedstocks currently, with 0.16 Pg C yr⁻¹ current storage potential from agricultural waste biochars. The agricultural feedstocks considered are forest and mill residues, rice husks, groundnut shells, and urban waste (i.e. garden wastes). A number of crop residues are not considered within the study due to them being deemed as unsuitable for biochar production due to lower lignin content. Exceptions to this are rice husks, sugarcane bagasse and nut shells. Experimental research undertaken for this study and detailed in Chapter 6 discusses how many of the excluded crop wastes are in fact suitable for biochar production, storing a similar amount of carbon, and having a similar recalcitrance in soils, to rice husk biochar. The projections by Lehmann et al. (2006) also look at conversion of current biofuel production to pyrolysis systems, with biochar as a by-product which they projected could sequester 0.18 Pg C yr⁻¹. This could increase to between 5.5 and 9.5 Pg C yr⁻¹ if biofuel production, up to the year 2100, is met using pyrolysis (using biofuel projections from literature). With regards to the rate of addition of biochar to soils the research estimates that sequestering 140 Mg C ha⁻¹ in the 1,600 Mha of global cropland would achieve 224 Pg C stored, and 175 Pg C could be stored in the 1,250 M ha of temperate grassland available. The rate of application is discussed by Lehmann et al (2006) as a very high application rate, but one which has not seen detrimental effects on soil and plant health in their experimentation. They highlight that some studies within the literature have seen detrimental effects at lower rates of application. The Lehmann et al. (2006) study has also come under some scrutiny, for example by Paul et al. (2009), for the large amount of land use change which is assumed within the scenarios, as incentives which promote land-use change from food production to biofuels production may impact on the long-term sustainability of food production and the integrity of vital ecosystems within the scenario. Another biochar study by Matovic (2011) projected that the conversion of 10 % of global net primary productivity (NPP) to biochar, with 50 % biochar yield and 30 % oil and gas yield for energy use, would sequester 4.8 Pg C yr⁻¹. The 50 % assumption of biochar yield may be an overestimation of potential biochar yields on such a scale. Biochar yields from within the literature from experimental data are discussed further in Sections 2.2.3 and 4.3.1 respectively. The study by Matovic (2011) also does not include estimations of feedback effects such as potential increases in NPP or reduced fertilizer requirements. Lenton and Vaughan (2009) used the radiative forcing change relating to the reduction in atmospheric CO_2 to calculate the climate cooling potential of a number of different methods of climate engineering. They discuss biochar production potential from a number of different studies, including a projection of 15 Pg C storage potential by 2035, increasing to 52 Pg C by 2060 from an unpublished study by Reid (2009), discussed in the published work of Reid (2008). The Reid

(2008) study does not focus on the use of crop residues, as is the main focus of the research detailed in Chapters 3 to 8 here, instead assessing the potential of producing biochar from dedicated timber/bio-energy plantations, mostly on non-arable land. Lenton and Vaughan (2009) use the values from the Reid and Parshotam (2008) study to calculate a C removal from the atmosphere of 11 Pg C in 2030 and 31 Pg C in 2060. The quantity of atmospheric carbon removed is lower, in comparison to the carbon stored in the biochar, due to the reaction of the other land and ocean sinks which may respond by releasing larger amounts of CO₂ into the atmosphere. They discuss that, for small removals of atmospheric CO₂, 92 % would be still be removed after 1 year, 64 % after 10 years, and 34 % would still be removed from the atmosphere after 100 years. Lenton and Vaughan (2009) summarised that the long term global carbon storage capacity of cropland, using biochar, is 224 Pg C, and of temperate grasslands is 175 Pg C. This carbon storage would result in a reduction in atmospheric CO₂ of 34 ppm resulting in a radiative forcing of -0.52 W m⁻². This projection assumes that the biochar carbon stocks stored in soils is replenished as it decays.

On a UK scale, Shackley et al. (2010), using life cycle assessment (LCA) methodology, concluded that available virgin biomass in the UK could abate between 3.59 - 11.15 Mt CO₂eq yr⁻¹ (between 0.00359-0.01115Pg CO₂eq yr⁻¹). The scope of an LCA is defined as ensuring;

'...that all environmental burdens connected with a product or service have to be assessed, back to the raw materials and down to waste removal.'

(Kloppfer, 1997)

LCAs are very specific to the product or service assessed and highly sensitive to the parameters chosen. Changes in the depth and/or breadth of the assessment parameters can alter the results dramatically. The UK Biochar Research Centre (UKBRC) conducted LCAs of a number of scenarios of biochar production from various feedstocks, incorporating factors such as the energy requirements of feedstock production and transport and looking at both 'large' and 'small' scale production (Shackley et al., 2010).



Figure 2-8: Life cycle assessment for carbon abatement of a number of biomass systems. Top panel shows the abatement potential (kg CO_2 eq) per tonne of feedstock. Bottom panel shows Life cycle assessment of carbon abatement (t CO_2 eq) per hectare. 'Large' and 'small' relates to large and small scale production facilities respectively (Shackley et al., 2010).

The study assessed a number of feedstocks and estimated that approximately 7 – 30 t CO_2 eq ha⁻¹ yr⁻¹ could be abated using biochar systems. Brownsort (2009) concluded that within the LCA analysis, the largest portion of the carbon abatement was achieved through the storage of carbon in soils through biochar addition. Other relatively large abatement factors were the indirect effects of biochar addition to soil, for instance reduced GHG emissions from soil, and the offset of fossil fuel emissions (Hammond et al., 2011). A similar LCA conducted for biochar produced from corn stover, yard waste and switch grass determined a carbon abatement of 864 kg CO_2 eq t⁻¹ and 885 kg CO_2 eq t⁻¹ for corn stover and yard waste respectively (Roberts et al.,

2010). Adverse carbon abatement potential, where CO_2 emissions were increased due to the system, was found in some cases for the switch grass biochar system depending on the methods of accounting used for land use change. The assessment of carbon abatement by hectare by Shackley et al. (2010) (see Figure 2-8 (bottom)) shows that even under the LCA assumptions of one study there can be large variation (~ 30 t CO_2 eq ha⁻¹) between the carbon abatement potential of biochar systems using different feedstocks and processes. Shackley et al. (2010) also conducted an assessment of potential carbon abatement from biochar production of nonvirgin biomass feedstocks (see Figure 2-9). The assessment resulted in estimates of abatement potential of between 300 kg CO_2 eq per tonne of feedstock and over 1700 kg CO_2 eq per tonne of feedstock, dependent on feedstock choice. The abatement potentials calculated for garden and green waste, food waste and wood waste are higher than the estimates for virgin biomass.





The scenario uncertainties, which are dependent on feedstock and process type assumptions amongst other factors, are increased for the non-virgin biomass. The LCAs carried out in the Shackley et al. (2010) study were designed with a focus on carbon abatement potential of systems within the UK and therefore have assumptions such as production type, transport methods, available biomass types and quantities, and types of energy to be offset which may not be applicable to other systems internationally. The carbon abatement values discussed by Roberts et al. (2010) are lower than those of Shackley et al. (2010). This is likely to be due to differences in the accounting methods and assumptions made within the development of each LCA. This highlights that LCA analysis is difficult, if not often impossible, to apply to other cases due to the very complex network and interactions of the system assumptions. The highly specific and sensitive nature of LCAs, to the assumptions made and parameters chosen, means that an LCA conducted for another area or system is likely to produce different results. However, the values in the two LCA studies indicate well the potential of individual scenarios to abate carbon emissions and the contributions of individual factors to the total abatement. They also help to give a general impression of how biochar may be used to abate CO₂ emissions.

2.4 Future projections of climate change

The climate modelling community have developed a number of projections of future climate change. These projections have, generally, improved over time as understanding of the climate system and drivers of climate change also improves. The IPCC have used a range of future scenarios in their previous assessments. In 2000 the IPCC published their Special Report in Emissions Scenarios (SRES) which detailed a number of future emissions pathways developed by the IPCC to guide future research and to assist in the harmonization of other research (IPCC, 2000). The SRES emissions pathways were used in the 2007 IPCC report to direct discussion of potential climate change (IPCC, 2007). The most recent IPCC report uses future emissions scenarios developed by modelling communities, including physical science and socio-economic groups, to assess potential future climate change. These scenarios are the Representative Concentration Pathways (RCPs).

2.4.1 The representative concentration pathways (RCPs)

The RCPs are four scenarios of future emissions, each resulting in a different radiative forcing change in 2100 (van Vuuren et al., 2011a). The four radiative forcing end-points explored are 2.6 W m⁻², 4.5 W m⁻², 6 W m⁻² and 8.5 W m⁻², and the RCPs are named after their respective radiative forcing targets: RCP2.6; RCP4.5; RCP6; RCP8.5. Each RCP, designed by a different climate modelling group, aims to give one example of how the radiative forcing could be achieved in 2100. The radiative forcing levels covered by the four RCPs encompass the range seen within the climate change literature. Each pathway is designed around an underlying socio-economic scenario and comprises of datasets for the evolution of emissions of well-mixed greenhouse gases, other species important to radiative forcing, land-use changes, radiative forcing and atmospheric concentrations. The research detailed in Chapters 3 to 8 uses the land-use projections for each RCP to investigate how much biochar could be produced from crop residues within each scenario. Section 2.4.1 outlines the main assumptions of each RCP and Chapter 3 details how the assumptions of each RCP were used to develop the biochar pathways.

2.4.1.1 RCP 2.6

RCP 2.6 was designed by the IMAGE modelling team and is a 'peak and decline' scenario where radiative forcing peaks at around 3 W m⁻² mid-century and then declines to the target of 2.6 W m⁻² in 2100. It was designed to represent those scenarios within the literature which have very ambitious radiative forcing targets, as a representation of scenarios which aim to limit global mean temperature increase to 2 °C. Van Vuuren et al. (2011a) detailed that the scenario involves the reduction of cumulative emissions by 70 % from 2010 to 2100, and emissions reduction of 95 % by 2100 when compared to the 2100 emissions of the baseline scenario. The RCP 2.6 scenario relies heavily on bio-energy and reforestation to achieve these objectives. Emissions of CO_2 from the energy sector are negative by 2100, with the use of nuclear energy and renewable energy technologies (solar PV and wind) and carbon capture and storage on both fossil-fuel and biomass energy production. Emissions of CO_2 from land are increased, relative to the baseline, due to the large increases in biofuel production. Emissions of other species (e.g. N₂O, CH₄) mostly decrease in RCP 2.6, relative to the baseline, although within this some sectors where emissions reductions are difficult, see increases (e.g. livestock production).

The RCP scenario was developed using an integrated assessment model, from a baseline scenario of medium development (using population, income, energy and land use as indicators). Agricultural production within RCP2.6 is based on assumptions of demand and trade, then calculated using the input of production data taken from the Adapting Mosaic scenario of the Millennium Ecosystem Assessment (2005). The initial distribution of agricultural land is taken from the History Database of the Global Environment (HYDE) (Klein Goldewijk et al., 2011). The changing distribution of agricultural land is then calculated using data on agricultural productivity, proximity of existing agriculture, proximity to water sources and urban areas, and a random factor. Van Vuuren et al. (2011a) discuss that crop land for food production increases moderately to 2050, levelling off after this period. The increase in agricultural land projected for RCP 2.6, relative to the baseline scenario, is attributed to increased use of bio-energy and a reduced CO₂ fertilization effect. The increase in agricultural productivity projected is achieved mainly through increases in crop yields. Van Vuuren et al. (2011a) also note a shift in agricultural production from regions of high income to regions of low income, with biofuel production occurring near areas of current agricultural production and in particular in the regions abandoned in high income areas.

2.4.1.2 RCP 4.5

The RCP 4.5 scenario was developed by the Global Change Assessment Model (GCAM) group at the Joint Global Change Research Institute (2013) and details a pathway which reaches a

radiative forcing of 4.5 W m⁻² in 2080 which then remains constant to 2100. Thomson et al. (2011) discuss that this relates to atmospheric concentrations of approximately 650 ppm CO₂-equivalent, with 526 ppm atmospheric CO₂ in 2100, compared with 792 ppm CO₂ by 2100 in the reference scenario (Clarke et al., 2007). Descriptions of scenario development are detailed in Wise et al. (2009) and Thomson et al. (2011). Global greenhouse gas emission pricing is used within the scenario to prompt reductions in emissions, with agriculture and land-use emissions included in the system. The scenario assumes that efforts are made in unison, globally, to achieve emissions reductions through the pricing scheme. In terms of energy systems, the scenario sees deployment of carbon capture and storage (for both fossil and biofuels), as well as a general shift towards electricity and low emissions technologies for energy production as well as energy efficiency measures. Thomson et al. (2010) detail the changing use of land within RCP 4.5 across the period, with decreasing crop and pasture land and increasing biofuel and forest cover. A reduction in pasture land is seen within the scenario due to a shift away from beef consumption. Afforestation occurs due to the inclusion of carbon storage in land sinks, and of reduced emissions from land-use change, into the carbon pricing scheme.

2.4.1.3 RCP 6

RCP 6 reaches a radiative forcing of 6 W m^{-2} in 2100. The scenario, developed by the Asia-Pacific Integrated Model (AIM) team, includes a global market for emissions credits resulting in limits on emissions, achieving 13 GtC per year by the end of the century, with emissions peaking around 2060 and declining to 2100. The scenario is projected to reach atmospheric CO₂equivalent concentrations of 855ppm. Cumulative CO₂ reductions are 463 GtC lower in RCP 6 than the reference scenario (Masui et al., 2011). No climate intervention policies are assumed in the reference scenario, which sees emissions increase to 27.7 GtC yr⁻¹ and reaches a radiative forcing of 7.0 W m⁻² in 2100 (Masui et al., 2011). The baseline scenario for RCP 6 is based upon the SRES B2 scenario which assumes intermediate levels of economic development and technological change (Nakicenovic et al., 2000). RCP 6 sees an increase in renewable energy production and in electricity use compared with other final energy types. A shift from coal to gas and nuclear power is also seen for electricity production. The use of non-fossil sources for electricity production and the increased use of carbon capture and storage see a decrease in CO_2 emissions from electricity generation from 2060 onwards. Crop land is seen to increase by 26 % over time due to increased demand for food and energy crops within the scenario (Masui et al., 2011).

2.4.1.4 RCP 8.5

RCP 8.5, reaching a radiative forcing of 8.5 W m⁻² in 2100, was developed by the MESSAGE modelling team using the IIASA Integrated Modelling Framework (Riahi et al., 2011). This RCP is also used by modelling communities as the baseline climate scenario as it does not include any specific climate mitigation actions although some air pollutants are regulated for air quality rather than climate purposes. As such there is no baseline scenario to the RCP against which emissions reductions are made, as has been seen with the other RCPs. A number of the main drivers and assumptions for RCP 8.5 were derived from a revised version of the SRES A2 scenario, named the A2r scenario (Nakicenovic et al., 2000, Riahi et al., 2007). The main revision of the A2 scenario, to A2r, was the replacement of future demographic projections with more current data, including a decrease in projected population from 15 billion down to 12 billion in 2100. Riahi et al. (2011) discuss that, of the literature regarding business as usual scenarios, RCP 8.5 is a conservative projection of future development, with high population, low incomes and high energy demand. RCP 8.5 assumes some reduction in emissions intensity after 2030 due to the assumed link between welfare and environmental conditions, as shown with the Environmental Kuznets Curve theory (Riahi et al., 2011, Stern, 2004). Fossil fuels, predominantly coal, are the most economically viable and dominant energy source within the scenario, with unconventional fossil fuel sources also being utilised. The share of nuclear and hydro energy also increases towards the end of the scenario. Riahi et al. (2011) discuss that much of the potential for increased agricultural land exists in South America and Africa, with some other regions, notably Asia, seeing constraints in the amount of expansion possible. They project that agricultural land in developed nations will decline over the period, whilst net increases will be seen in the developing nations. Increased yields and intensification of agricultural production are projected to provide most of the increase in agricultural production, with crop land expansion making a smaller contribution.

2.4.2 Global mean temperature and the RCPs

Within the literature projections of changes in future temperature with climate change and the impacts of climate change on crops, within each of the RCPs and wider climate change scenarios, are determined using climate models and their associated inputs. Working Group I of the IPCC has collated and summarised current knowledge of the physical science of climate change, including projections of future temperature change under the RCP emissions scenarios (Collins et al., 2013, IPCC, 2013, Kirtman et al., 2013). There are many uncertainties and challenges involved in the projection of future climate, including natural variability, non-linear response to drivers, future emissions pathways (of the many species which can influence radiative forcing,

for example of emissions of short lived pollutants such as sulphate, nitrate and black carbon aerosols and carbon monoxide (CO) and methane (CH₄)), and the climate response to these forcings. Despite these uncertainties, climate model projections are becoming more robust and are among the best indicators, currently, of the changes in climate which may manifest in the future due to anthropological emissions.

Near term projections of the 5 to 95 % range for changes in global mean surface air temperature for 2016-2035 (from the 1986-2005 reference period) are 0.47 °C to 1.0 °C using CMIP5 (climate model intercomparison project 5) projections (Kirtman et al., 2013). Kirtman et al. (2013) discuss that, between 2016 – 2035, global mean surface temperature is likely to be more than 1°C above the 1850-1900 mean, but not likely to be more than 1.5°C above this mean. Using the Allen, Stott and Kettleborough (ASK) method of weighting models in relation to their quality by considering the accuracy of previous model predictions to observations, the 5 % to 95 % range for the same period is 0.39 °C to 0.87 °C. The lower projections of the ASK method are mainly attributed to the weighting towards models which agree more closely with the temperature hiatus seen in recent observations (Kirtman et al., 2013). Collins et al. (2013) detail the longer term projections for changes in annual mean surface temperature which are summarised in **Table 2-3**.

Table 2-3: Projected changes in annual mean surface temperature (°C) from the 1986-2005 reference period for the four RCPs in different spatial and temporal regions (global 2046-2065, global 2081-2100, land and tropics). Adapted from Table 12.2 in Collins et al. (2013). Values shown are the multi-model mean, ± 1 standard deviation and, in brackets, the 5 and 95 % ranges of the distribution.

		RCP 2.6	RCP 4.5	RCP 6	RCP 8.5
		ΔT in ^o C			
Global	2046-	1.0±0.3(0.4,1.6)	1.4±0.3(0.9,2.0)	1.3±0.3(0.8,1.8)	2.0±0.4(1.4,2.6)
	2065				
Global	2081-	1.0±0.4(0.3,1.7)	1.8±0.5(1.1,2.6)	2.2±0.5(1.4,3.1)	3.7±0.7(2.6,4.8)
	2100				
Land	2081-	1.2±0.6(0.3,2.2)	2.4±0.6(1.3,3.4)	3.0±0.7(1.8,4.1)	4.8±0.9(3.4,6.2)
	2100				
Tropics	2081-	0.9±0.3(0.3,1.4)	1.6±0.4(0.9,2.3)	2.0±0.4(1.3,2.7)	3.3±0.6(2.2,4.4)
	2100				

2.4.3 Crop yield impacts of RCP projected climate change

As each RCP pathway would be expected to induce different levels of climate change, due to the different radiative forcings of each scenario, the changes in climate induced by each RCP would be expected to affect crop yields differently. A number, if not all, of the impacts of climate change may affect the potential for biochar production and subsequent carbon sequestration within the scenarios discussed in the previous sections. The impacts of climate change on crop yields and residue production is one of the main effects which may impact biochar potential within the scenarios, and may occur due to a number of climate impacts. Impacts may be due to changes in temperature, precipitation, extreme events such as drought or flooding, and other factors. For example, many plants and crops have thresholds for water and/or temperature stress for all or part of their development and maturation cycles, which, if not achieved or exceeded, may limit development of the crop (Porter et al., 2014). From this, it would be expected that crops growing within the RCP scenarios with higher projected global mean surface temperatures (e.g. RCP 8.5) may suffer more from temperature stress than those crops in scenarios of lower projected global mean temperature change (e.g. RCP 2.6). Countering this effect, to some extent, may be the influence of increased atmospheric CO_2 concentrations on crop yields, where some research has found increased crop yields with increased atmospheric CO₂ (Lobell and Field, 2008). RCP 2.6 has a lower atmospheric CO₂ concentration pathway than RCP 8.5, meaning that any increases in crop yield through the CO_2 fertilization effect are likely to be lower in RCP 2.6 than RCP 8.5. The results of studies into this CO₂ fertilization effect are highly variable. The CO_2 fertilization effect has often been found to be stronger in the C_3 crops, such as wheat, rice and cotton, than C_4 crops due to the increased responsiveness of the C_3 photosynthetic pathway to increased CO₂ concentrations (Leakey, 2009, Porter et al., 2014). Laboratory and enclosed CO₂ fertilization experiments often see greater increases in crop yields than open air experiments (termed Free Air Concentration Enrichment (FACE) studies), potentially due to a number of factors including temporal fluctuations in CO₂ or difficulty in controlling other experimental variables (Porter et al., 2014). Although high uncertainty exists around the potential effect of CO_2 fertilization on crop production, Lobell and Field (2008) determined an average effect on the C_3 crops rice, wheat and maize of 0.1 % yield increase for each 1 ppm CO_2 increase. Porter et al. (2014) summarise that, with CO_2 fertilization effects and without adaptation, negative impacts will be seen for all crop yields with a temperature increase above 3°C. Projections of many climate change impacts in the near-term (to 2035) are very similar for the four RCPs. For this reason, the projections discussed within the literature mainly focus on the impacts of RCP 4.5 as this is an intermediate scenario (Kirtman et al., 2013). In the longer term the projected impacts become increasingly divergent, and so longer term projections discuss the RCPs separately (Collins et al., 2013).

A summary of the current literature regarding projections of the impacts of increasing temperatures on crop yields is detailed in the most recent IPCC Working Group II report (Porter et al., 2014). The report details projections, from a review of the available literature, of the impacts of temperature increases between 1 °C and 5 °C on three major crops: maize, wheat and rice. The projections include yield changes with and without adaptation for both tropical and temperate regions. The specific methods of adaptation are not discussed by Porter et al. (2014) in relation to the figure, but are described as methods of 'simple agronomic adaptation'.

Kyle et al. (2014) detailed the change in crop yield which may occur due to the temperature change projected for each RCP, relative to the projected change without temperature change, for corn, wheat, rice, fibre, sugar and bioenergy crops. The data in Kyle et al. (2014) is given for every tenth year from 2015 to 2095 for most species, both with no adaptation, and with some relocation of species to more suitable land as climate change progresses. The study uses the GCAM model, used for the development of RCP 4.5, and a crop growth model to assess the impacts of each RCP pathway on crops. They discuss that including the potential effects of climate change into assessment of the RCP could significantly change the evolution of the RCP pathways. The Kyle et al. (2014) study determined decadal average changes in crop yields for each RCP including climate change effects, relative to baseline crop yields. The baseline crop yield data, representing a 'present climate', was sourced from the FAO and so is compatible with the baseline data used in Scenario 1. The study summarised that climate impacts were, on average, negative for all crops except sugar and dedicated bioenergy crops. The lowest yields of cereals, including rice and wheat, were seen towards the end of the assessment period (2100) and under the highest emission scenarios. The study discussed that re-distribution of crop-types across available land would be likely to mitigate, to some extent, the reduction in crop yield from that expected with no change in crop distribution.

3 Characterisation of biochars from crop residues: Methodology

3.1 Introduction

Biochar characteristics including carbon content, surface area, ash content, nutrient content, pH and cation exchange capacity (CEC) vary due to both feedstock properties (Manya, 2012) and process conditions including temperature, residence time and pyrolysis atmosphere (Demirbas, 2004b, Zhao et al., 2013). The extent to which these factors influence biochar characteristics is not fully understood, with the drivers of variation in biochar characteristics requiring further investigation (Lehmann and Joseph, 2009).

The thermal decomposition of biomass into biochar can be achieved using a number of processes including (slow, intermediate or fast) pyrolysis, gasification, hydrothermal carbonisation (HTC), torrefaction and traditional carbonisation methods (Bridgwater, 2003) (see Section 2.2.2.2). Slow pyrolysis generally produces higher biochar yields relative to the other processes, and therefore is considered further here for biochar production. During pyrolysis the biomass feedstock is heated in the absence of oxygen so that full combustion does not occur. Volatiles and semi-volatiles are released from the biomass, as oil and gas products, leaving the biochar product (Antal, 2003).

Various feedstocks exist for biochar production including agricultural and forestry residues, municipal wastes, animal manures and purpose grown biomass. A number of factors such as desired biochar characteristics, sustainability requirements, possible toxicity effects and desired biochar end-use must be considered when determining the suitability of particular feedstocks (Joseph et al., 2009). One prominent focus for biochar use is the addition of biochar to soils, which may have both agronomic and climate change mitigation benefits, enabling the removal and storage of atmospheric carbon whilst potentially acting as a soil improver with benefits such as increased plant growth (Biederman and Harpole, 2013, Lehmann and Joseph, 2009, UK Biochar Research Centre, 2011).

Many of the investigations reported within the literature, for biochar production from the pyrolysis of biomass, have used uniform pyrolysis conditions and focus only on one or a small number of biomass types (Chan et al., 2008, Cheah et al., 2014, Das et al., 2008, Hossain et al., 2011, Kim et al., 2012, Peng et al., 2011). Production conditions and feedstock details are also often not reported fully. These factors often make comparison between biochar studies difficult. Improving knowledge of biochar characteristics and their relationship to feedstock characteristics and process conditions will enable further understanding of soils and vegetation
effects, and of biochar for carbon sequestration. This study adds insight to the current biochar characterisation literature through documenting biochars produced from eight agricultural crop residues using a fixed-bed slow pyrolysis reactor, exploring the effects of feedstock characteristics on biochar yield and characteristics. The biochars were documented by yield, carbon content and recalcitrance, pH, nutrient content, surface area and porosity, elemental composition, moisture, ash, fixed carbon and volatile matter content, and calorific value. Sugarcane bagasse biochars produced under different pyrolysis temperatures and heating rates also gave insight into how the conditions of pyrolysis affect biochar characteristics. Yields and calorific values of oils and gases were also documented, alongside identification of individual gas species present.

3.2 Materials and methods

3.2.1 Feedstock selection

Agricultural crop residues are an often underutilized resource which, if large scale conversion to biochar and co-products was achieved, have the potential to offer both agronomic and carbon sequestration functions. Eight globally predominant agricultural residue types were selected for analysis using crop production quantities from the Food and Agriculture Organisation's (FAO) statistical database FAOSTAT (FAO, 2013a). Residue to product ratio (RPR) values, indicating the amount of residue produced alongside the commodity production detailed by the FAO data, were calculated from averages from literature (ARNAB, 1989, Lal, 2005, Penn State College of Agricultural Sciences, 2012, US National Research Council Board on Agriculture, 1983). Crops with high regional production quantities and high RPRs were chosen for analysis, the eight residues being: coconut fibre, coconut shell, cotton stalk, olive pomace, palm shell, rice husk, sugarcane bagasse, and wheat straw. Wheat straw and rice husk samples were sourced from fields in the Faisalabad District, Punjab province, Pakistan (31 21 N, 72 59 E), Sugarcane bagasse was sourced from Samundri, Pakistan (30 48 N, 71 52 E). Samples were transported in plastic bags and, on receipt, ground and sieved to 1.4 to 2.8 mm particle size. Coconut husk and shell, and palm kernel shells were sourced from the waste streams of coconut and palm kernel oil processing in the western region of Ghana. The coconut and palm kernel shells, as received, ranged from 3.35 to 10 mm particles. Cotton stalks were sourced from Northern Syria. Olive pomace was received in powdered form, with particle size < 2mm. All samples were stored in air tight containers after grinding, prior to pyrolysis.

3.2.2 Slow pyrolysis

A laboratory scale fixed-bed slow pyrolysis reactor was used to pyrolyse the feedstocks (Figure 3-1). The reactor, 250 mm in length by 30 mm internal diameter, was externally heated by a 1.2 kW tube furnace. The furnace was controlled to produce the desired heating rate, final temperature and residence time at peak temperature. A stainless steel crucible was used to hold 6 g of each biomass sample.



Figure 3-1: Schematic of the slow pyrolysis reactor (Windeatt et al., 2014).

To isolate and investigate relationships between feedstock and biochar characteristics, the conditions of pyrolysis were kept uniform. From this point these are termed the 'standard conditions'.

Under standard conditions the eight biomass types were pyrolysed with a heating rate of 5 °C min⁻¹ and final temperature of 600 °C which was held constant for one hour. The sugarcane bagasse feedstock was also pyrolysed under alternative conditions, exploring the effects of pyrolysis temperature and heating rate on biochar characteristics (Table 3-1 summarises the temperature and heating rate conditions used).

Table 3-1: Experimental conditions: final temperature and gas collection period. The table shows the standard conditions used for all feedstocks (top) and those of altered final temperature (middle) and heating rate (bottom) used in experiments on the sugarcane bagasse feedstock.

Final Temperature (°C)	Heating Rate	Gas collection start	Gas collection period after				
	(°C min⁻¹)	(x min after	final temperature hold time				
		experiment start)	(x min)				
Standard Conditions (all fe	edstocks)						
600	5	30	15				
Conditions with altered fir	nal temperature (sugarcane bagasse on	ly)				
400	5	30	15				
800	5	30	15				
Conditions with altered heating rate (sugarcane bagasse only)							
600	20	10	15				
600	50	10	15				

Pyrolysis was undertaken in a nitrogen (N_2) atmosphere with a flow rate of 200 ml min⁻¹ at 20°C and 1 bar pressure. The N₂ carrier gas was introduced 10 minutes before heating commenced to purge oxygen from the system. Oil was collected using a three condenser system, with glass wool used to remove uncondensed semi-volatiles from the gas stream.

The crucible and condenser system were weighed before and after each experiment to calculate mass balance of the feedstocks and products. Biochar and oil yields were determined by weight. Product gases were collected in a tedlar bag and analysed off-line by gas chromatography (see 3.2.3.12).

3.2.3 Analytical methodology

3.2.3.1 Moisture, carbon, volatile and ash content

Proximate analysis, used to determine the moisture, fixed carbon, volatile matter and ash content of the raw feedstocks and biochars was conducted using a muffle furnace method. Samples, in a ceramic crucible with lid, were dried in an oven at 105 °C for 2 hours, then heated to 550 °C, and held for 4 hours. The lids were then removed and the samples held at 550°C for 1 hour to combust the fixed carbon, leaving the residual ash. Moisture, volatiles and ash were calculated by direct weight loss and fixed carbon content calculated by difference.

3.2.3.2 Elemental composition

Carbon, hydrogen, nitrogen, sulphur and oxygen (C, H, N, S, O) content of the feedstocks and biochars was determined by ultimate analysis methodology using a Flash 2000 organic element analyser with thermal conductivity detector. 3mg of each sample was analysed, in duplicate,

with vanadium pentoxide (V_2O_5) to aid combustion. During ultimate analysis the samples, combusted in oxygen, produce N_2 , H_2O , CO_2 and SO_2 , which is quantified using chromatography to give the elemental composition. Oxygen was calculated by difference. Carbon yield, defined here as the carbon retained from the original biomass carbon, was calculated from the ultimate analysis.

3.2.3.3 Metal and nutrient content

Metal and nutrient content of the feedstocks and biochars was analysed using inductively coupled plasma mass spectrometry (ICP-MS). 0.2 g of either feedstock or biochar was added to 8 ml of nitric acid and heated, in either a microwave reactor or using a sand bath, to aid sample digestion. Concentrations of phosphorus (P), potassium (K), calcium (Ca) and magnesium (Mg) were determined as these are key macro-nutrients for healthy plant growth and soil quality (Maathuis and Diatloff, 2013).

3.2.3.4 Biochar recalcitrance

There are a number of methods of estimating the recalcitrance of biochar carbon including using analogues and laboratory enhanced weathering methods. Section 2.2.5.2 discusses the methods currently used to estimate the stability of biochar carbon in more detail. One method of estimating biochar recalcitrance is the R_{50} index developed by Harvey et al. (2012) which assesses the recalcitrance of a biochar in relation to that of graphite and is described in equation 3-1.

$$R_{50,x} = T_{50,x} / T_{50,graphite} \tag{3-1}$$

Where $T_{50 x}$ and $T_{50 graphite}$ are the temperatures at which 50 % of the material was oxidized for biochar and graphite respectively. $T_{50 x}$ was determined by Temperature Programmed Oxidation (TPO), conducted by thermogravimetric analysis (TGA) using a Mettler Toledo TGA/DSC1 analyser with alumina crucible and Al lid. 5 mg of sample was heated in air to 900 °C at 10 °C min⁻¹. R₅₀ values were calculated from the TPO data using equation 3-1 (Harvey et al., 2012). A value of 886 °C for $T_{50 graphite}$, as reported by Harvey et al. (2012), was used here. The R₅₀ index only gives a measure of the recalcitrance of the biochar in relation to that of other biochars and graphite, it does not give a timescale or rate of degradation. Methodology for using the R₅₀ index to indicate a timescale for degradation is discussed in Section 3.2.3.5.

We hypothesised that high concentrations of alkali metals, such as potassium (K), in biochars may influence the oxidation temperature during TPO, due to this effect being reported for other

materials such as raw biomass (Jiang et al., 2013). Alkali metals are likely to be rapidly leached from biochars upon addition to soil (Lindstrom, 1986) which, if high alkali content catalyses the oxidation of biochars, as hypothesised, the R_{50} value for that biochar would be conservative. The biochar may have a higher stability in soil than predicted due to this rapid leaching. To test this hypothesis a wheat straw biochar (high in alkali content) was washed in deionised water at 80 °C for 2 hours, removing the majority of the alkali metals present. TPO was performed as described above, in duplicate, on the washed and unwashed biochars for comparison.

3.2.3.5 Carbon sequestration potential

The carbon sequestration (CS) potential of a biochar is defined by Zhao et al. (2013) as a measure of the amount of feedstock carbon that would be retained in the soil long-term upon addition as biochar. The CS potential methodology, as detailed in equation 3-2, was developed by Zhao et al. (2013) and is calculated as:

$$CS(\%) = (M x Ch x C_{Ch} x R_{50}) / M x C_{F}$$
(3-2)

Where, M is the mass of feedstock (g), Ch is the yield of biochar (%), C_{Ch} is the carbon content of the biochar (%), R_{50} is the recalcitrance and C_F is the carbon content of the feedstock (%). The CS potential was calculated for each of our biochars. The methodology described by Zhao et al. (2013) does not detail the timeframe of 'long-term' sequestration, but does discuss that biochar may be recalcitrant for <100 to >1000's of years. The CS potential is therefore assumed here to be estimate long-term storage above the 95 year period of the biochar scenarios within this thesis (see Chapter 5).

3.2.3.6 Surface area and pore volume

Biochar surface area and pore volume was determined using a Quantasorb continuous flow gas adsorption unit with N_2 adsorbate. Vacuum outgassing was conducted at 200 °C for 3 hours. Surface area was determined using the BET method of isotherm analysis (Brunauer et al., 1938, Osborne, 2004). Combined micropore and mesopore volume was determined using the BJH cumulative adsorption method (Ceram, 2013).

3.2.3.7 Calorific value

Calorific value, determined as the heating value (HV), of the biochars was determined using a PARR 6200 bomb calorimeter, determining the energy contained within the sample by detecting a temperature rise in surrounding water (University of Waterloo, 2012). Some samples required the addition of kerosene to aid burning, with the calorific value of the accelerant factored into

the calculations. Calorific value was also calculated for the raw feedstocks and biochars using the Dulong equation:

$$HV (kJ/kg) = 33,800 * C + 144,000 * (H_2 - O_2/8) + 9,270 * S$$
(3-3)

(Capareda, 2011)

Where C, H, O and S are elemental mass fractions from the ultimate analysis (Section 3.2.3.2).

3.2.3.8 pH

pH was determined using a pH meter, following the method described by Yao et al. (2010). A solution of 1:20 biochar to deionised water was shaken for 30 minutes then left to stand for 10 minutes before testing.

3.2.3.9 Lignocellulosic content

Typical cellulose, hemicellulose and lignin content were determined for each feedstock as an average of the values available from literature (Table 4-1).

3.2.3.10 Statistical analysis

Correlation and regression analysis was used to determine significant relationships between characteristics of the raw feedstock and biochars, and also between the different biochar characteristics. The data was analysed using Microsoft Excel correlation and regression tools and a confidence level of 95% was used.

3.2.3.11 Oil analysis

Oil yields were calculated by weight. Calorific value of the oil products was analysed using the same method as for biochars which is discussed in Section 3.2.3.7.

3.2.3.12 Gas analysis

Gas collection are summarised in Table 3-1. The gases were collected into a gas tedlar bag throughout the experiment and then analysed offline by gas chromatography. The initial gas flow was not collected as volatile formation does not begin until around 300 $^{\circ}$ C, therefore this initial flow was assumed to be the N₂ carrier gas. A 'flushing period' at the end of the pyrolysis hold time was maintained to ensure collection of any remaining gases.

Gas chromatography (GC) was used for the analysis of gas species type and concentration. Three gas chromatographs were used to separately determine CO₂, permanent gases (H₂ and CO) and hydrocarbon gases (CH₄, C₂H₄, C₂H₆, C₃H₆, C₃H₈, C₄H₈, C₄H₆, C₄H₁₀). A 1 ml glass syringe was used for all gas injections. The CO₂ and permanent gas chromatographs used thermal conductivity detectors (GC/TCD), with argon as the carrier gas. The column oven temperature was 30 °C, injector and detector temperature was 120 °C and the filament temperature was 160 °C. The hydrocarbon chromatograph used a flame ionisation detector (GC/FID) with nitrogen gas carrier. The initial temperature was 60 °C for 3 minutes, increased by 10 °C min⁻¹ to 100 °C and held for 3 minutes, then increased by 20 °C min⁻¹ to 120 °C and held for 3 minutes. The injector temperature was 150 °C and detector temperature was 200 °C.

The calorific value of each gas species was calculated and multiplied by the mole fraction in the gas using:

$$CV = \sum (x_1 * CV_1 + x_2 * CV_2 + \dots x_n * CV_n)$$
(3-4)

The compression factor (Z) was applied to modify the ideal gas law for the behaviour of real gas, giving the volume of the gas at given pressure and temperature divided by the volume under the same conditions under the ideal gas law.

$$Z = \frac{p}{\rho R_{specific}T} \tag{3-5}$$

Where ρ is the gas density, $R_{specific} = R/M$ (specific gas constant where R is the molar gas constant and M is the molar mass) and T is temperature. The final calorific value, corrected to 15 °C and 1 atmosphere of pressure, is then given by:

$$CV = \frac{CV}{Z} \tag{3-6}$$

This methodology is detailed further in Ulbig and Hoburg (2002).

3.3 Summary

Methodologies for the production and characterisation of biochar, bio-oil and syngas have been detailed in Sections 3.2.2 to 3.2.3.12. Characterisation of biochars from eight crop residues under uniform pyrolysis conditions enabled relationships between feedstock and biochar characteristics to be analysed. Sugarcane bagasse biochars produced under alternative pyrolysis conditions enabled relationships between pyrolysis conditions and biochar characteristics to be determined. Feedstock characterisation was by elemental composition, moisture, volatile, carbon and ash content, H/C and O/C content and calorific value. This provides valuable documentation of biochar feedstock properties, as in-depth reporting of feedstock characteristics is often under-reported within the literature and the analysis here will both add to the limited literature (Downie et al., 2009). Biochar characterisation included the same methodologies as for the raw feedstocks, with the addition of pH, pore volume, surface area, stored carbon, recalcitrance (R₅₀ index) and the carbon sequestration (CS) potential. These characterisations gave a well-rounded documentation of the biochars, enabling analysis of relationships between feedstocks, biochar characteristics and process conditions. Analysis to determine whether the degradation of some biochars is catalysed by high alkali metal content, potentially making the R₅₀ classifications of these biochars conservative, was also undertaken through TPO of washed and unwashed biochars. Biochar yield, carbon content, R₅₀ and CS potential data was fed into the later work assessing the global potential of biochar for carbon sequestration (see Chapter 5 onwards).

4 Characterisation of biochars from crop residues: Results and discussion

4.1 Introduction

Pyrolysis of eight crop residue types (see Table 4-1), under the standard pyrolysis conditions of 5 °C min⁻¹ heating rate, to 600 °C and 1 hour hold time, was used to produce biochars, bio-oils and syngas for analysis. Sugarcane bagasse was pyrolysed under alternative conditions, with peak temperatures of 400 °C and 800 °C and heating rates of 20 °C min⁻¹ and 50 °C min⁻¹, to examine the effects of alternate pyrolysis conditions of the characteristics of pyrolysis products.

The feedstocks and biochars were assessed for elemental composition and moisture, volatile, fixed carbon and ash content by ultimate and proximate analysis respectively. The H/C and O/C ratios of the feedstocks and biochars were determined from the ultimate analysis. The Dulong equation was used to determine the calorific value of the feedstocks and biochars, alongside bomb calorimetry of the biochars and bio-oils. Biochars were also characterised by pore volume and surface area using a gas adsorption method using N₂ adsorbate, and pH determined using a pH meter. ICP-MS was used to determine the macro-nutrient content of the feedstocks and biochars, in particular the P, K, Ca and Mg content. Syngas species composition, concentration and calorific value was determined using gas chromatography (GC). The carbon retained in the biochar from that of the feedstock, R₅₀ values and the (long-term) carbon sequestration (CS) potentials were calculated for the biochars. Temperature programmed oxidation of washed and unwashed wheat straw biochars was conducted to determine whether high alkali metal content influences the degradation temperature of a biochar. These analysis techniques, excluding TPO on washed biochar, were conducted on the alternate conditions biochars to examine any effects these conditions have on biochar characteristics.

4.2 Feedstock composition

The typical lignocellulosic composition of the feedstocks is listed in Table 4-1, illustrating that the composition of biomass feedstocks are often very different in their basic composition.

Feedstock characteristics are detailed in Table 4.2. Feedstock carbon content was highest in the palm shell feedstock (53.1 wt %) and lowest in the rice husk feedstock (42.5 wt %). The rice husk feedstock had correspondingly high ash content at 19.6 wt %. The lowest ash content was seen in the coconut shell raw feedstock (0.6 wt %) which had the second highest carbon content (52.6 wt %), palm shell had the second lowest ash content (2.0 %).

	Cellulose (%)	Hemicellulose (%)	Lignin (%)	References
Palm shell	30	18	53	(Daud and Ali, 2004, Okoroigwe and Saffron, 2012)
Sugarcane bagasse	39	26	24	(Aguilar et al., 2002, Pandey et al., 2000)
Rice husk	38	18	22	(Kadam et al., 2000, Nguyen et al., 2010)
Coconut shell	20	49	30	(Daud and Ali, 2004)
Wheat straw	35	25	19	(Kaparju et al., 2009, Kristensen et al. 2007, McKendry 2002)
Cotton stalk	35	39	21	(Akpinar et al., 2007, Goksu et al., 2007, Kang et al., 2012, Ververis et al., 2004)
Olive pomace	34	15	20	(Ayrilmis and Buyuksari, 2010)
Coconut fibre	46	15	33	(Justiz-Smith et al., 2008, Khedari et al., 2004, Tomczak et al., 2007)

Table 4-1: Typical cellulose, hemicellulose and lignin content of raw feedstocks (%).

Regression analysis showed some significant relationship between carbon content and ash content ($r^2 = 0.53$, p = 0.04). A significant correlation was seen between the typical feedstock cellulose content and feedstock carbon content ($r^2 = 0.64$, p = 0.02), with the cellulose content declining with increasing carbon content (Figure 3-1). No significant correlation was seen between typical hemicellulose or lignin content and carbon content. The lignocellulose values used were typical values from the literature. Analysis of the lignocellulosic composition of the actual feedstocks assessed here would provide more accurate representation of correlation between these factors. This was not possible here due to time constraints, but would be a useful area for further study.



Figure 4-1: Correlation analysis of the typical feedstock lignocellulosic composition (%) (cellulose (top left), hemicellulose (top right) and lignin (bottom)) and the feedstock carbon content (%).

The main macro-nutrient content (P, K, Ca and Mg) of the feedstocks are shown in Figure 4-6 alongside the macro-nutrient content of the biochars, showing the effects of pyrolysis on macronutrient content. Table I-1 in Annexe I shows the data values for macro-nutrient concentration. The coconut shell feedstock had very low concentrations of the nutrients assessed, relative to most of the other feedstocks. Wheat straw also had relatively low concentrations of P, Ca and Mg, though concentrations of K were relatively high when compared with other feedstocks except olive pomace. A significant correlation was seen between the P and Mg content of feedstocks ($r^2 = 0.77$, p = 0.01), where, if concentrations of P are high then concentrations of Mg tend to be correspondingly high (Figure 4-2).



Figure 4-2: The relationship between phosphorus (P) and magnesium (Mg) content (mg kg⁻¹) of the feedstocks.

No other significant correlations were seen between the different macro-nutrient species in the feedstocks.

4.3 Biochar yields and characteristics

The proximate analysis, ultimate analysis and calorific value of the feedstocks and biochars (produced under standard conditions) are listed in Table 4.2 together with the pH, calorific value, surface area, pore volume, stored carbon, R_{50} values and carbon sequestration (CS) potential of the standard biochars. The characteristics of biochars produced under altered conditions are detailed in Section 4.6.

Table	4-2:	Characteristics	of	raw	feedstocks	(top)	and	biochars	(bottom)	produced	under
stando	ard co	nditions.									

occ off training of the second s	jä
Raw materials	, ,
Ultimate Analysis	
C (ar) % 53.1 45.9 42.5 52.6 48.1 46.0 49.2 4	14.7
H (ar) % 7.1 6.7 6.5 6.2 6.8 7.6 6.8	7.5
N (ar) % 0.7 0.9 1.3 2.0 1.8 5.6 2.0 (0.8
S (ar) % 0.0 0.2 0.0 0.0 0.0 0.0 0.0 0.0	0.0
O (by diff) % 46.8 59.2 46.0 53.1 49.0 54.5 45.8 6	51.8
H/C 1.6 1.7 1.8 1.4 1.7 2.0 1.6	2.0
O/C 0.7 1.0 0.8 0.8 0.9 0.7	1.0
Proximate Analysis	
Moisture % 3.0 5.8 5.7 5.6 6.1 5.7	7.5
Volatiles (daf) % 74.1 85.3 80.9 77.2 85.9 93.1 80.5 81	35.3
Fixed Carbon (daf) % 25.9 14.7 19.1 22.8 14.1 6.9 19.5 14.3	L4.7
Ash (db) % 2.0 4.4 19.6 0.6 7.9 4.2 4.5	5.3
HHV MJ kg ⁻¹ 19.9 14.6 15.5 17.2 17.3 16.8 18.2 14	L4.9
inits alm shell ugar cane agasse agasse ice husk ice husk ice husk ice husk ice nusk ice nusk ice onut traw ofton talk traw	ibre
<u>Chars</u>	
Product Yields	
Char % 31.8 27.7 39.0 28.2 30.3 28.0 30.5 30	80.8
Oil % 50.3 50.3 33.5 43.7 50.0 53.6 44.8 4	17.8
Gas % 17.9 23.6 21.8 28.1 17.6 18.4 29.2 2	25.1
Ultimate Analysis	
C (ar) % 90.6 88.6 54.5 93.9 75.3 83.2 71.8 83	32.6
H (ar) % 2.8 2.8 2.1 3.0 2.6 3.2 2.8	2.7
N (ar) % 0.9 1.3 1.1 0.4 1.0 4.8 1.9	2.4
S (ar) % 0.0 0.1 0.0 0.0 0.2 0.1 0.0	0.0
O (by diff) % 7.9 13.7 5.4 2.6 4.5 14.2 11.6 12	L2.8
H/C 0.4 0.4 0.5 0.4 0.4 0.5 0.5 0	0.4
O/C 0.1 0.1 0.1 0.02 0.04 0.1 0.1	0.1
Proximate Analysis	
Moisture % 2.2 3.7 5.7 7.1 8.1 8.5 10.0 10	L0.4
Volatiles (db) % 11.5 30.1 13.9 8.1 21.2 28.8 20.9 2	25.1
Fixed Carbon (daf) % 88.5 69.9 86.1 91.9 78.8 71.2 79.1 74	74.9
Ash (db) % 6.7 13.0 47.0 4.1 23.4 9.5 18.1 1	L3.5
Other Characteristics	
HHV MJkg ⁻¹ 33.6 30.1 19.3 33.7 26.5 31.4 24.1 20	26.6
Total pore volume $cc g^{-1}$ 0.16 0.18 0.10 0.15 0.01 0.05 0.00 0.	0.04
Surface Area (BET) M ² g ⁻¹ 220.0 149.1 114.9 222.5 6.3 121.2 1.2 2	23.2
pH 6.1 8.6 9.9 8.5 11.6 10.3 10.5	9.6
stored carbon (db) % 54.3 53.5 50.0 50.3 47.4 50.6 44.5 50	56.9
$R_{ro} = 0.61 0.53 0.54 0.59 0.46 0.50 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 0.58 $).49
CS (sequestration	
potential) % 32.5 27.3 26 28.7 21.3 23.8 24.5 24	26.8
(ar) = as received, (db) = dry basis, (daf) = dry ash free, HHV = higher heating value	

4.3.1 Biochar yields

Biochar yields, when produced under the standard conditions, ranged from 27.7 wt % (sugarcane bagasse) to 39 wt % (rice husk) showing that feedstock characteristics can have a large influence on biochar yield. Correlation between biochar yield and biochar carbon content ($r^2 = 0.67$, p = 0.01).

4.3.2 Ultimate analysis

Biochar carbon content varied from 55 wt % (rice husk) to 94 wt % (coconut shell). Between 45 wt % and 57 % of the original feedstock carbon was contained in the biochar after pyrolysis, with an average value of 51 %, comparing well with values of 49 % from Woolf et al. (2010) and 50% from Xu et al. (2012).



Figure 4-3: Correlation analysis of feedstock C (top left), H (top right), N (bottom left) and O (bottom right) content and corresponding biochar content.

Correlation analysis of the ultimate analysis results for the feedstocks and biochars (Figure 4-3) highlighted strong correlation between the nitrogen content of feedstocks and biochars ($r^2 = 0.64$, p = 0.02). Correlation between feedstocks and biochar content for carbon, hydrogen and oxygen was low, showing no significance.

4.3.3 Proximate analysis

Biochar volatile content was from 8 wt % to 30 wt %. Good correlation was found between the proximate analysis results of the feedstocks (moisture, volatiles, fixed carbon and ash) and the corresponding characteristic in the biochars (Figure 4-4). Ash content of the biochars had a large range, from 4.1 wt % (coconut shell) to 47 wt % (rice husk).



Figure 4-4: Correlation analysis of feedstock moisture (top left), volatile (top right), fixed carbon (bottom left) and ash content (bottom right) (%) with corresponding biochar content.

Correlation between feedstock and biochar moisture content was $r^2 = 0.58$, p = 0.03. Feedstock and biochar volatile content ($r^2 = 0.71$, p = 0.01) and feedstock and biochar fixed carbon content ($r^2 = 0.71$, p = 0.01) were strongly correlated. Feedstocks with high ash content tended to result in lower biochar yields ($r^2 = 0.76$, p = 0.005), and also in biochars with higher ash content ($r^2 =$ 0.97, p = 0.0001). Coconut shell biochar had both the highest carbon content and lowest ash content. Enders et al (2012) suggested that feedstocks with high ash content produce biochars with lower fixed carbon, which they attribute to the high ash content inhibiting the formation of aromatic carbon structures. Clearly there is a shift from high volatile, low fixed carbon content in the raw biomass to low volatile, high fixed carbon for the product biochars. Enders et al. (2012) reported ash, volatile and fixed carbon contents for a range of biochars from the literature, confirming these trends for a number of other biochars. There was also variation in the proportions of volatiles, fixed carbon and ash in the resulting biochars, some of which could be attributed to feedstock characteristics and some of which may be related to process conditions (see Section 4.6).

4.3.4 Hydrogen to carbon (H/C) and oxygen to carbon (O/C) ratios



Figure 4-5: Van Krevelen Diagram of O/C and H/C ratios of raw feedstocks and biochars. Eight feedstocks and biochars are presented, although some values are overlying.

The H/C atomic ratios plotted against the O/C atomic ratios for the feedstocks and biochars are shown by van Krevelen diagram in Figure 4-5. H/C and O/C ratios decreased in all biochars when compared with their feedstocks. This decrease is related to the removal of H and O during pyrolysis and indicates increased aromatic structure. Figure 4-5 clearly indicates the loss of

volatile hydrocarbon and oxygenated hydrocarbon species and increasing carbon content during pyrolysis.

Krull et al. (2009) reported that a decreasing biochar H/C ratio indicates increasing aromatic structure in the biochar potentially increasing the stability of the biochar and the recalcitrance of carbon. Increased carbon content also relates to an increased calorific value of the material. Kim et al. (2012) reported H/C and O/C ratios for biochars produced from the fast pyrolysis of pitch pine at 500 °C which were similar to those reported found here using slow pyrolysis at 600 °C. Lee et al. (2013) also reported a decrease in H/C and O/C ratios upon slow pyrolysis of Miscanthus to 500 °C.

4.3.5 Calorific value

Biochar may also be utilised as a solid fuel. Biochar calorific value (CV) followed the expected trend with the higher carbon, lower ash biochars having the highest CVs. The lowest CV was for rice husk biochar at 19 MJ kg⁻¹, and the highest for coconut shell biochar at 34 MJ kg⁻¹. Feedstocks with higher hemicellulose and lignin contents also tended to have higher CV. The biochar CVs ranging between 26-34 MJ kg⁻¹ are in a similar range to bituminous-anthracite grade coal and may therefore be a useful solid fuel commodity. Biochars with a lower CV are unlikely to be a commercially competitive solid fuel source but could potentially be used for household fuel use (i.e. cooking or heating).

4.3.6 Surface area and porosity

Biochar surface area ranged from $1.2 \text{ m}^2 \text{ g}^{-1}$ for olive pomace biochar to 223 m² g⁻¹ for coconut shell biochar. Total pore volume of the biochars ranged from negligible for olive pomace biochar to 0.18 cc g⁻¹ for sugarcane bagasse biochar. Significant correlation was seen between biochar surface area and total pore volume (r² = 0.80, p = 0.003). Lee et al. (2013) reported a surface area of 293 m² g⁻¹ for biochar produced from Miscanthus at a pyrolysis temperature of 600 °C. This is higher than, but within a similar range to, the highest surface areas seen in this study. The surface area and porosity properties of biochars can influence biochar effects in soil. The physical properties of soil, such as structure, water holding capacity and soil biology, such as microbial communities and earthworm presence, may be altered by biochar application (Biederman and Harpole, 2013, UK Biochar Research Centre, 2010). The high surface areas and porosities found in some biochar here, for example coconut shell biochar, indicate that the biochar may have positive soil impacts, such as increasing water holding capacity and providing habitat for microbial communities. Biochars may also be subjected to a further processing stage, for example chemical or steam activation, to further increase the surface area and produce activated carbons (Wu et al., 2012). Azargohar and Dalai (2006) increased the internal surface area of a biochar by 50 times through potassium hydroxide chemical activation.

4.3.7 Macro-nutrient concentrations

Biochars have been noted to contain varying concentrations of nutrient species (Chan and Xu, 2009). The macro-nutrient contents of the feedstocks and biochars assessed in this study are shown in Figure 4-6. P, K, Ca and Mg contents were assessed due to their importance for healthy plant growth (Maathuis and Diatloff, 2013). Nitrogen (N) content, which is another important macronutrient, was also examined by ultimate analysis (see Table 4.2), and is discussed further within this section.



Figure 4-6: Macro-nutrient content (g kg⁻¹) (dry basis (db)) of (a) raw feedstocks and (b) biochars. Species, from left to right, are phosphorus (P), potassium (K), calcium (Ca) and Magnesium (Mg). N.B. Scales vary between different nutrient species.

Large variations in some nutrient concentrations were found between the different raw feedstocks and also between the different biochars. A number of the nutrient species examined were concentrated within the biochar during the pyrolysis process. Those feedstocks with lower macro-nutrient concentrations had relatively lower concentrations when converted to biochar. Total biochar N content ranged from 4.3 g kg⁻¹ for coconut shell to 47.8 g kg⁻¹ for cotton stalk. Total P content of the biochars was between 0 g kg⁻¹ for palm shell and coconut shell to 4.2 g kg⁻¹

for olive pomace. Total biochar K ranged from 0.6 g kg⁻¹ for palm shell to 60 g kg⁻¹ for wheat straw. Total N and P concentrations of the biochars were low compared to some values found in literature (Chan and Xu, 2009). This may be due to the types of biomass used for biochar production, for example manure biochars used by Chan and Xu (2009) are known to have relatively high nutrient content when compared to the woody biomass biochars used here. The literature on biochar nutrients details that biochar total nitrogen (N), phosphorus (P) and potassium (K) concentrations are often under-reported, though a large range exists within those values available. For example, Chan and Xu (2009), in a meta-analysis of the current literature, reported concentrations of total N of 1.8 to 56.4g kg⁻¹, total P of 2.7 to 480g kg⁻¹ and total K of 1.0 to 58g kg⁻¹. Total N and total K contents are fairly well matched with those determined within this study. Biochar P content was over 100 times greater than the highest values of P content determined here. These high concentrations of P are attributed by Chan and Xu (2009) to feedstocks of animal origin such as sewage sludge, broiler litter and manures, rather than from plant based feedstocks as used in this study. The values reported both here and by Chan and Xu (2009) are for total N, P and K concentrations. N, P and K can be present in different forms, only some of which may be available for uptake and use by plants. N is taken up as NO₃, NH_4^+ , and N_2 in gas from the atmosphere or ions from soil solution; P is taken up as phosphates from soil solution; K, Ca and Mg are taken up as ions from soil solution (Kirkby, 2012). Although total N, P and K concentrations may be high in some biochars, this does not always correlate with high plant available N, P and K concentrations. Often, concentrations of mineral N are low even where total N is high, as in biochars produced from poultry manures, conversely where total K is high, available K is often also high (Chan et al., 2007). Chan et al. (2007) detailed both total and available P for a number of biochars, finding a range of 0.2 to 73 g kg⁻¹ total P and 15 to 11,600 mg kg⁻¹ available P. The biochar containing 0.2 g kg⁻¹ total P had 15 mg kg⁻¹ available P, and the biochar containing 25.2 g kg⁻¹ total P had 11,600 mg kg⁻¹. This indicates that plant available P may increase with total P, though available P concentrations remain much lower than total P concentrations. The limited reporting of total and/or available nutrient concentrations for biochars makes it difficult to determine trends between the two. Despite these limitations it appears that the highest concentrations of both total and available N, P and K are often seen in biochars from nutrient rich feedstocks such as poultry manures and sewage sludge (Chan et al., 2007).

The nutrients required for optimal plant growth are dependent on factors including soil type, soil nutrient concentrations and plant specific requirements. In some soils an excess of nutrient may already exist. Knowledge of the nutrient content of biochars from different feedstocks would

assist the tailoring of biochar production, complimenting efforts to achieve soil optimum nutrient concentrations (Roberts et al., 2010). Methods of biochar application to soils include adding biochar to organic fertilizers such as composts, during and after the composting process, and anaerobic digestate (Lehmann et al., 2009, Schulz et al., 2013). Such processing of biochar, before addition to soil, aims to increase the nutrient content of biochars to encourage optimum plant growth. The use of biochar to manage nutrient concentrations within soils could potentially reduce mineral and organic fertilizer requirements due to the addition of nutrients, increased fertilizer efficiency and reduce fertilizer run-off with biochar addition (Lehmann, 2007). As the production of mineral fertilizer is both energy and resource intensive, and its overuse can be damaging to the environment, any reduction in mineral fertilizer use could be environmentally and economically beneficial (Brentrup et al., 2004).

As well as the potential beneficial effects on soil nutrients, biochar addition to soil can also have detrimental effects such as the introduction of toxic metals and organic contaminants. Contaminants may be introduced to biochar from the feedstock or through the production process (McHenry, 2009). As with macronutrient species some heavy metal species, if present in the feedstock, may be concentrated within the biochar during the pyrolysis process. Some biochars may carry these toxic metal species into soils and potentially into vegetation through plant uptake mechanisms (Kirkby, 2012). None of the biochars tested here were above the thresholds for heavy metal species specified in the IBI Biochar Standards (International Biochar Initiative, 2012e).



4.3.8 Biochar pH

Figure 4-7: Correlation analysis between biochar pH and feedstock potassium content (left) and biochar potassium content (right).

Biochar pH ranged from pH 6.1 for palm shell biochar to pH 11.6 for wheat straw biochar, with the majority of biochars being slightly alkaline. Higher concentrations of alkali metals (P, K, Ca, Mg) generally corresponded to higher pH of the biochar. Biochar pH correlated well with feedstock ($r^2 = 0.74$, p = 0.01) and biochar ($r^2 = 0.70$, p = 0.02) potassium content.

No other correlations were found between feedstock or biochar alkali content and biochar pH. Palm shell produced slightly acidic biochar and was found to be low in alkali metals and high in lignin content. The thermal degradation of lignin produces phenolic compounds which may contribute to the acidity of biochars (Nonier et al., 2006). Previous literature reports biochar pH values (without further processing of the biochars) of between pH 4 and pH 12, with values normally found to be above pH 7 (Lehmann and Joseph, 2009). All of the biochars produced here were within this range. Zhao et al. (2013) detailed biochar pH of 8.8 to 10.8 depending on biomass feedstock type. They reported increasing biochar pH with increasing pyrolysis temperature. The maximum temperature of 600 °C used in the standard conditions here may, therefore, promote biochar alkalinity. The application of biochar to soils has been reported in some cases to produce a liming effect, increasing soil pH (Biederman and Harpole, 2013). The UK Biochar Research Centre (2010) reported that alkaline biochars have the potential to buffer excess soil acidity. Increasing the pH of acidic soils has been seen to increase microbial activity, increasing soil organic matter mineralization and increasing nutrient availability to plants. This mineralization may result in the emission of more CO_2 from soils, termed a priming effect. Some of the biochars characterised here may have potential for reducing soil acidity, including the rice husk, wheat straw, cotton stalk and olive pomace biochars. Knowledge of the initial pH of both biochar and soil would be necessary to amend soil pH effectively using biochar, due to the potential ranges of both biochar and soil pH which may be encountered, and the relationship may involve other factors and could by highly complex. Due to the uncertainties of using biochar to alter soil pH, further large scale and scenario specific (i.e. particular biochar/soil combinations) analysis should be undertaken before large scale application to soils for this purpose.

4.3.9 Biochar recalcitrance

Temperature programmed oxidation (TPO) to determine the R_{50} recalcitrance index of each biochar showed a range of degradation profiles (Figure 4-8). 50 % weight loss was achieved between 400 °C and 560 °C. Biochars produced from physically hard feedstocks, such as palm shell and coconut shell, tended to have higher oxidation temperatures than biochars from feedstocks which were not as physically hard, for example wheat straw and coconut fibre.



Figure 4-8: Temperature programmed oxidation (TPO) profiles for biochars showing the weight loss (%) of each biochar with increasing temperature (°C). (Thermograms have been corrected for moisture and ash content following the method of Harvey et al. (2012)).

The TPO profiles of Figure 4-8 were used to calculate the recalcitrance index (R₅₀ values) of each biochar following the method of Harvey et al. (2012). Comparison of R_{50} values with carbon mineralization rates led Harvey et al. (2012) to classify biochars, by their degradation potential, into three classes where: $R_{50} \ge 0.7$ = Class A: most recalcitrant biochar; $0.5 \le R_{50} < 0.7$ = Class B: minimal degradation; R_{50} < 0.5 = Class C: more degradable biochar. Using the same classification system two of our biochars were Class C biochars ('more degradable') and six biochars were Class B ('minimal degradation'). None of the biochars were class A biochars (most recalcitrant). Palm shell biochar (R₅₀ = 0.60) would be most resistant to degradation and wheat straw biochar (R_{50} = 0.45) would be least resistant. The R_{50} recalcitrance index was developed by comparison of R₅₀ values with rates of microbial degradation for 12 biochars (Harvey et al., 2012). Their comparison of these two characteristics showed that, over an incubation period of 1 year, Class A biochars experienced negligible amounts of carbon mineralisation, Class B biochars experienced between 0.2 % and 1.3 % carbon mineralisation and Class C biochars experienced between 0.8 % and 3 % carbon mineralisation. The biochars classified by Harvey et al. (2012) as Class C were all produced at temperatures below 400 °C, which was not the case in our study. Our R₅₀ analysis indicated that biochars produced at 600°C can also exhibit Class C degradation rates, with both wheat straw and coconut fibre residues producing Class C biochars within this study. Zhao et al. (2013) calculated a range of R₅₀ values of 0.54 to 0.83 for wheat straw and shrimp hull respectively whilst examining a number of biochars produced at 500 °C from various biomass types including agricultural residues, manures and algae. Wheat straw, which is the only comparable feedstock between the two studies, had a R₅₀ value of 0.46 within this study, which is lower than the value of 0.54 found by Zhao et al. (2013). The wheat straw biochar was classified here as a Class 3 biochar, but achieved Class 2 in the Zhao et al. (2013) study. This highlights that variability may be seen between biochars from the same feedstock type as factors such as geographical location and growing conditions may affect biomass characteristics. Such variation may also be caused by pyrolysis conditions. Zhao et al. (2013) used a faster heating rate, and the Zhao et al. (2013) and Harvey et al. (2012) studies both used longer residence times at peak temperature than the standard pyrolysis conditions used here. The higher R₅₀ values seen by Zhao et al. (2013) and Harvey et al. (2012) may be related to the longer residence time of the biochars within those studies, allowing for longer exposure of the feedstock, perhaps allowing the charring process, including formation of aromatic structure, to develop further. This is investigated further in Section 4.6.

4.3.10 Catalysis of biochar degradation by alkali metal content

Some inorganic metal species including potassium, calcium and magnesium have been reported to lower the degradation temperature of biomass during thermogravimetric analysis (TGA). Nowakowski et al. (2007) discuss that biomass samples prepared with high concentrations of potassium had lower degradation temperatures than samples which had been demineralised by an acid washing pre-treatment. In-Yong et al. (2011) also reported this behaviour in biomass samples pyrolysed using TGA. A number of the biomass samples in the latter study were washed with water rather than acid, suggesting that the metals and salts can be easily removed by water and that more complex pre-treatment is not needed to see the effect of metal removal on oxidation temperature. Major et al. (2009) observed the leaching of nutrients from soils as water percolation caused nutrient displacement to areas outside the rooting zone, removing them from areas of potential plant uptake. They discuss that cations such as K, Ca and Mg are easily leached. Our analysis of thermograms from washed and unwashed wheat straw biochars showed that high alkali metal concentrations may reduce the oxidation temperature of biochars, and washing may remove sufficient alkali metal content to remove or reduce this effect. This has implications for the recalcitrance index (R_{50}) values calculated for the biochars in this study (Table 4.2), where the biochars may have high alkali content.



Figure 4-9: Thermograms indicating the oxidation temperatures of washed and unwashed wheat straw biochar samples, showing weight loss (%) of each biochar with increasing temperature. Thermograms have been corrected for moisture and ash content following the method of Harvey et al. (2012).

The washed biochar had 92 % lower potassium (K) content than the unwashed biochar, at 0.2 g kg⁻¹ and 2.2 g kg⁻¹ respectively. The oxidation temperature of the washed biochar was increased in relation to that of the unwashed biochar (Figure 4-9). The R₅₀ value of wheat straw biochar was increased from 0.46 to 0.55 by the washing pre-treatment to reduce alkali content. This reclassified the wheat straw biochar as a Class B biochar (minimum degradation) where previously it was Class C (most susceptible to degradation). As leaching is likely to occur quickly when biochars are added to soils there is likely to be a significant reduction in the concentration of alkali metal species in the biochars in a short time-frame, potentially reducing the degradation potential of some biochars in soils. R₅₀ values determined by TPO for unwashed biochars with high alkali content may therefore provide a conservative estimate of recalcitrance, with stability increased soon after addition to soil.

4.3.11 Carbon sequestration (CS) potential

The carbon sequestration (CS) potentials of the biochars indicate that between 21.3 % and 32.5 % of the feedstock carbon would be retained in soil long term upon conversion to biochar and addition to soil. Zhao et al. (2013), investigating the production of biochars from 12 different

biomass feedstocks, reported CS potentials from 21.1 % to 47.1 %. The CS potential of the biochars studied here are within the same range as those biochars assessed by Zhao et al. (2013), but tend to be towards the lower end of their range. There are also caveats to be considered when applying this method to biochar analysis. Firstly, the method and definition for carbon sequestration potential described by Zhao et al. (2013) does not define a time frame for the 'long-term' carbon storage. As the Harvey et al (2012) study, which developed the R_{50} index, discussed, the lifetime of biochar in soils can range from under a century to several millennia, therefore it is assumed throughout the rest of this study that the carbon retained would be stored for a period of longer than 100 years. This is an acceptable assumption as 100 years is at the very bottom end of the lifetime range discussed in Harvey et al. (2012). Secondly, as discussed in Section 2.2.5.2, there is no single favoured method of estimating the recalcitrance or carbon storage potential of biochars. Harvey et al. (2012) and Zhao et al. (2013) used thermal degradation to indicate a biochars stability, other methods include chemical enhancement of degradation or estimates of degradation using proxies such as charcoal from natural forest fires. The thermal degradation method of Harvey et al. (2012), incorporated into the assessment of carbon sequestration potential by Zhao et al. (2013) is used here due to a number of factors including the relatively short timescale of carbon storage required for the scenarios of Chapters 5 to 8 (95 years), the ease of comparison and ranking of biochar stability, and the relative simplicity of the assessment which made analysis possible within the study timeframe and constraints.

4.4 Oil yields and characteristics

The highest product yield for all feedstocks was oil, achieving between 33.5 % and 53.6 % yield under standard pyrolysis conditions. Oil CV was between 9 and 15 MJ kg⁻¹ which is low compared to that of, for example, conventional oil at 42.5 MJ kg⁻¹ (Biomass Energy Centre, 2014b). Higher heating values (HHVs) of 20 MJ kg⁻¹ are reported by Qi et al. (2007) for pyrolysis oils from wood and agricultural residues. They also detail that a number of characteristics including high oxygen content, high water content, high acidity, low HHV and variable viscosity often make pyrolysis oils a poor substitute for conventional hydrocarbon fuels. Upgrading to reduce oxygen content and address other problems such as water content can make them a more suitable fuel source. Oils produced from the process could, therefore, be used as a fuel for the process or for other energy needs after upgrading, although this would add an energy penalty and further technological complexity to the system. After processing, energy from the pyrolysis oils could be utilized with the pyrolysis system or as a renewable replacement within other fossil fuel systems.

4.5 Gas characteristics

Syngas yield ranged from 17.6 % to 29.2 %, from wheat straw and olive pomace residues respectively. The CV and composition of each syngas are detailed in Table I-2, Annexe I. The syngas produced had an average CV of 12.4 MJ m⁻³ which is lower than the average 40 MJ m⁻³ of natural gas (Demirbas and Arin, 2002, National Grid, 2013). The syngas of lowest CV was from coconut shell, at 10.2 MJ m⁻³, whilst the highest CV, at 15.0 MJ m⁻³, was seen in syngas from rice husk. The dominant gas species produced from all feedstocks under standard pyrolysis conditions was CO₂ which constituted between 42.4 % and 54.8 % of the total syngas (these values are from palm shell and cotton stalk pyrolysis respectively). The other dominant gas species produced from monoxide (CO), methane (CH₄) and (H₂). Small concentrations of other hydrocarbon (HC) species were present in varying quantities. All of these gases, except CO₂, can be used as a fossil fuel offset source either at the pyrolysis plant or elsewhere, although some upgrading may be required to increase the CV per unit of gas which would incur an energy penalty.



Figure 4-10: Gas species concentration (%) for the syngas produced from each feedstock. Gas species detrmined are carbon dioxide (CO₂); permanent gases: hydrogen (H₂) and carbon monoxide (CO); and hydrocarbon gases: methane (CH₄), ethene (C₂H₄), ethane (C₂H₆), propene (C₃H₆), propane (C₃H₈), butene (C₄H₈), butane (C₄H₁₀)). Feedstocks are: coconut shell (c sh), cotton stalk (c st), palm shell (ps), coconut fibre (cf), olive pomace (op), rice husk (rh), wheat straw (ws) and sugarcane bagasse (sc). N.B. Scales vary: some species are displayed on a 0 – 50 % scales, whilst others are displayed on a 0 – 4 %.

4.6 The effects of pyrolysis conditions on product yields and characteristics

The standard pyrolysis conditions used to explore the effect of feedstock on biochar yields and characteristics were varied to examine the effects of alternative pyrolysis conditions, particularly peak temperature and heating rate. One feedstock type, sugarcane bagasse, was pyrolysed under final temperatures of 400 °C, 600 °C and 800 °C, and heating rates of 5 °C min⁻¹, 20 °C min⁻¹ and 50 °C min⁻¹. These alternative pyrolysis conditions used are discussed in more detail in Section 3.2.2 and Table 3-1.

Table 4-3: Characteristics of biochars produced from the slow pyrolysis of sugarcane bagasse under different pyrolysis temperature and heating rate regimes. The results attained at 600 °C and 5 °C min⁻¹ are those determined under standard conditions to allow for comparison of the effects of altered conditions against the standard conditions.

		Final temperature (°C)			Heating rate (°C min ⁻¹)			
Characteristics	Units	400	600	800	5	20	50	
Product Yields								
biochar	%	34.0	27.7	23.0	27.7	23.3	22.8	
oil	%	49.0	50.3	44.0	50.3	59.5	56.2	
gas	%	21.8	20.9	24.0	20.9	25.5	19.2	
Ultimate Analysis								
С	%	85.0	87.0	81.9	87.0	82.2	81.7	
Н	%	0.9	2.1	1.0	2.1	2.2	2.2	
Ν	%	0.5	1.3	0.5	1.3	0.6	0.8	
S	%	0.1	0.1	0.0	0.1	0.1	0.1	
O (by diff)	%	17.8	13.2	20.9	13.2	18.5	18.8	
H/C		0.1	0.3	0.1	0.3	0.3	0.3	
O/C		0.2	0.1	0.2	0.1	0.2	0.2	
Proximate Analysis								
moisture	%	4.1	3.7	4.1	3.7	3.6	3.5	
volatiles (daf)	%	28.1	30.1	17.2	30.1	29.3	18.0	
carbon (<i>daf</i>)	%	71.9	69.9	82.8	69.9	70.7	82.0	
ash(db)	%	14.5	15.5	16.5	17.5	18.5	19.5	
Other Characteristics								
HHV	MJ kg ⁻¹	30.0	30.1	31.1	30.1	32.5	30.8	
Total pore volume	cc g ⁻¹	0.0	0.2	0.0	0.2	0.0	0.1	
Surface Area (BET)	$M_2 g^{-1}$	2.8	149.1	15.7	149.1	77.6	110.4	
pH		6.9	8.6	10.6	8.6	9.2	9.2	
stored carbon	%	63.0	53.5	41.0	53.5	41.8	40.6	
R ₅₀		0.47	0.51	0.56	0.51	0.52	0.51	
CS (sequestration potential)	%	29.6	27.3	23.0	27.3	21.7	20.7	

4.6.1 Product yields



Figure 4-11: Yields of biochar (left), bio-oil (centre) and syngas (right) from the pyrolysis of sugarcane bagasse at (a) temperatures of 400 °C, 600 °C and 800 °C and (b) heating rates of 5 °C min⁻¹, 20 °C min⁻¹ and 50 °C min⁻¹ at 600 °C.

Biochar yields decreased with increasing temperature, with a clear trend seen in Figure 4-11. This can be linked to the evolution of more semi-volatile and volatile material from the biochar at increasing temperatures. Biochars produced at 400 °C may not be fully charred (Williams and Besler, 1996), potentially leading to higher degradation rates, upon addition to soil, than fully charred material. Biochar yield also decreased slightly with increasing heating rate, although this effect was not as pronounced as that of peak temperature. Bio-oil production increased slightly as temperature increased from 400 °C to 600 °C indicating the evolution of highermolecular weight hydrocarbon species from the biomass, occurring at ~ 500 °C (Neves et al., 2011). 600 °C is not sufficient to convert many of these hydrocarbons to gaseous species. 50.3 % bio-oil yield was achieved from pyrolysis of sugarcane bagasse at peak temperature of 600 °C, with 49 % and 44 % yields achieved at 400 °C and 800 °C respectively. Oil yields decreased at 800 °C indicating the presence of secondary reactions at higher pyrolysis temperatures, for example the increased breakdown of the high-molecular weight species into lower molecular weight gaseous species. A heating rate of 20 °C min⁻¹ increased oil yields by almost 10 % to 59.5 %. Further increase to 50 °C min⁻¹ caused a reduction in yields. Gas yields followed the opposite trend, seeing a slight yield decrease when peak temperature was increased from 400 °C to 600 °C, and a yield increase when peak temperature was increased to 800 °C. Again, this is due to the higher pyrolysis temperatures breaking down more higher-molecular weight species into gaseous products. Gas production was highest at the heating rate of 20 °C min⁻¹ and lowest at 50 °C min⁻¹. A pyrolysis heating rate of 20 °C min⁻¹ was optimal for both oil and gas yields. In addition to impacts on yields, biochar, oil and gas characteristics may also vary with pyrolysis conditions. Sections 4.6.2 to 4.6.7 examine variance in biochar conditions with pyrolysis conditions. Effects on the composition syngas are discussed in Section 4.6.8.

4.6.2 Proximate analysis

Proximate analysis determined the moisture, volatile, fixed carbon and ash content of the biochars from sugarcane bagasse.



Figure 4-12: Proximate analysis (moisture, volatiles, fixed carbon and ash content (left to right)) of (a) biochars produced from the pyrolysis of sugarcane bagasse at temperatures of 400 °C, 600 °C and 800 °C (top) and (b) heating rates of 5 °C min⁻¹, 20 °C min⁻¹ and 50 °C min⁻¹ (bottom) at 600 °C. N.B. Scales vary between characteristics.

Biochar moisture content was initially reduced as peak temperature was increased from 400 °C to 600 °C, and then increased at 800 °C. It was also reduced by increasing heating rate. There was a slight increase in biochar volatile concentration at 600 °C, relative to that of 400 °C, and then a marked decrease in volatile content at 800 °C. This may be explained by the secondary pyrolysis processes discussed by Neves et al. (2011) and discussed in Section 4.6.1. As temperatures are raised above ~ 500 °C secondary processes including polymerisation may cause volatiles to be reformed on the biochar. As temperature is increased further the

likelihood of cracking and the evolution of volatile species from the biochar increases, resulting in the lower volatile content seen at 800 °C. Fixed carbon content was initially reduced very marginally (-2 %) by the increase peak temperature to 600 °C then increased by a further 12.7 % at 800 °C. This indicates the removal of other species from the biochar and the increased aromaticity of the biochar structure at higher temperatures. Although biochar yields were reduced at 800°C, the fixed carbon proportion of this biochar increased. Ash content increased with both increased peak temperature and heating rate, with a range of 2 % between biochars produced at the highest and lowest of each pyrolysis condition tested. This indicates that ash is concentrated within the biochar as other species, such as volatiles, are evolved from the biochar with both increasing temperature and heating rate.

4.6.3 Ultimate analysis

Ultimate analysis determined the carbon, hydrogen, nitrogen, sulphur and oxygen content of the sugarcane bagasse biochars.



Figure 4-13: Elemental composition (carbon (C), hydrogen (H), nitrogen (N), sulphur (S) and oxygen (O) content (%) (left to right)) of biochars produced from the pyrolysis of sugarcane bagasse at peak final temperatures of 400 °C, 600 °C and 800 °C (top) and heating rates of 5 °C min⁻¹, 20 °C min⁻¹ and 50 °C min⁻¹ (bottom). N.B. scales vary between elemental species.

A slight increase of 2 % was seen in carbon content with increasing peak temperature from 400 °C to 600 °C. Further increase to 800 °C decreased elemental carbon by 5.1 %. This is different from the trend seen by proximate analysis for fixed carbon content (Figure 4-12). It was expected that elemental carbon content would continue to increase with increasing

temperature, therefore further investigation of sugarcane bagasse biochar produced at different temperatures would be beneficial, but was not possible within the timescale of this study. Elemental hydrogen and nitrogen concentrations increased markedly as peak temperature was increased from 400 °C to 600 °C, reducing to near 400 °C concentrations as peak temperature was increased further to 800 °C. Again this may correlate with the increased volatile content of biochars produced at 600 °C (Figure 4-12). Oxygen content of the biochars was reduced and then increased as peak pyrolysis temperature was increased from 400 °C to 600 °C and 800 °C, at 17.8 %, 13.2 % and 20.9 % respectively. High oxygen content in the biochar produced at 800 °C may be attributable to inorganic oxygen content in ash (Brewer et al., 2009). Sulphur concentration was 0.1 % for biochars produced at both 400 °C and 600 °C, reducing to 0 % at 800 °C. The sulphur contents were low, in comparison to that of coal, potentially making biochar a suitable low sulphur fuel. Removal of the sulphur at peak temperature between 600 °C and 800 °C and 800 °C can be related to the evolution of sulphur as hydrogen sulphide (H₂S) in syngas with increasing temperature (Cheah et al., 2014).

Increasing the pyrolysis heating rate resulted in a small reduction in elemental carbon content, with a reduction of 3.1 % seen between carbon in biochars produced at the lowest and highest heating rates. The carbon reduction with increased heating rate follows the same trend as the reduction in biochar yield. This may mean that slightly more elemental carbon is removed from the feedstock as volatile matter during pyrolysis with increasing heating rate. Hydrogen and sulphur contents were largely unaffected by heating rate. Nitrogen content reduced from 1.3 % to 0.6 % when heating rate was increased from 5 °C min⁻¹ to 20 °C min⁻¹, then increased slightly to 0.8 %, at 50 °C min⁻¹. Oxygen content was increased with increasing heating rate, from 5 °C min⁻¹ to 20 °C min⁻¹ and then remained unaffected with a further increase in heating rate to 50 °C min⁻¹.

Where peak pyrolysis temperature was increased from 400 $^{\circ}$ C to 600 $^{\circ}$ C the ratio of H/C was increased from 0.1 to 0.3 and then reduced again to 0.1 at 800 $^{\circ}$ C.

4.6.4 pH



Figure 4-14: Biochar pH in biochars produced from sugarcane bagasse at the different peak pyrolysis temperatures (left) and heating rates (right).

Biochar pH increased from 6.9 to 10.6 in the biochars produced at peak temperatures of 400 °C and 800 $^{\circ}$ C respectively. The r² value of 0.997 (p = 0.03) indicates an almost perfect correlation between pyrolysis temperature and biochar pH. The significance of this correlation analysis would benefit from a larger sample size. The increase in pH of 3.7 seen across the range of peak pyrolysis temperatures (400 °C) indicates that each 10 °C increase in pyrolysis temperature would result in an increase in biochar pH of 0.09. Biochar pH showed a smaller increase with increasing heating rate, being pH 8.6 and 9.2 in the biochars produced at 5 °C min⁻¹ and 50 °C min⁻¹ respectively. The increase in pH seen with temperature may be related to the increasing ash and alkali species content in the higher temperature biochars (Figure 4-12, Figure 4-16 and Section 4.3.8). The increasing ash content and pH of biochars with increasing peak temperature and heating rate was consistent with the discussion of Enders et al. (2012) who reported a relationship of increasing biochar ash content and pH. They did not find correlation with any individual alkali metal or alkali earth metal species. The rate of increase in pH seen here with increasing temperature, in relation to that of increased hearing rate, could be explained by the larger increases seen in potassium, magnesium and calcium seen with increasing temperature than with increases in heating rate (Section 4.6.7). When examining the effects of feedstock characteristics on biochar pH in Section 4.3.8, biochar pH was positively correlated with concentrations of potassium in the eight different feedstock types and resulting biochars. These correlations between the potassium content and pH of biochars from different feedstocks indicate that the increased potassium content seen with increasing pyrolysis temperature, and to a lesser extent with increasing heating rate, may be related to the increased pH values seen here.

4.6.5 Surface area

Surface area increased markedly, from 2.8 m² g⁻¹ to 149 m² g⁻¹ when final pyrolysis temperature was increased from 400 °C to 600 °C. Pagnanelli et al. (2008) discuss the temperature dependent evolution of volatile matter from pore spaces, causing an increase of biochar surface area through unblocking of the pore spaces. Increasing peak temperature further to 800 °C reduced surface area to 15.7 m² g⁻¹. The reduction seen at 800 °C may be related to a breakdown of structure at high temperatures. Uchimiya et al. (2011) reported a breakdown in biochar structure at higher temperatures, resulting in a drastically lowered surface area. The breakdown of biochar structure was related to the lignin content of biochars, with the structure of lower lignin biochars degrading at lower temperatures. They reported this effect at temperatures above ~ 900 °C. The results here indicate that this effect may occur at lower temperatures (~800 °C) for some biochars. Sugarcane bagasse has typically low lignin content, when compared to feedstocks such as palm shell (Table 4-1). Future research into the magnitude of this effect in other feedstocks, and at a wider range of peak pyrolysis temperatures would be beneficial. Surface area was reduced from 149 m² g⁻¹ to 77.6 m² g⁻¹ by an increase in heating rate to 20 °C min⁻¹. Surface area was increased again, to 110.4 m² g⁻¹ with a heating rate of 50 °C min⁻¹. These results imply that heating rate may have some effect on biochar surface area, though these effects were not as marked as those seen with increasing temperature.

4.6.6 Biochar recalcitrance



Figure 4-15: Temperature programmed oxidation (TPO) profiles of biochars produced from (a) sugarcane bagasse at pyrolysis temperatures of 400 °C, 600 °C and 800 °C (left) and (b) heating rates of 5 °C min⁻¹, 20 °C min⁻¹ and 50 °C min⁻¹ (right). Thermograms were corrected for moisture and ash following the method of Harvey et al. (2012).

Biochars produced from sugarcane bagasse at higher peak pyrolysis temperatures showed increased oxidation temperatures (Figure 4-15). This was reflected in the recalcitrance index R_{50} values of the biochars, which increased from 0.47 to 0.56 for biochars produced at 400 °C and 800 °C respectively. This reclassified the sugarcane bagasse biochar from Class C of the R_{50} classification index where the biochar is 'most susceptible to degradation', to Class B with 'some susceptibility to degradation' (Harvey et al., 2012). This increase in oxidation temperature and subsequent R_{50} may be indicative of increased aromaticity with increasing peak pyrolysis temperature. Heating rate did not have an effect on the oxidation temperature of the sugarcane bagasse biochars, indicating that heating rate does not significantly affect the aromaticity or stability of the biochars. The R_{50} values were very similar for all three biochars produced at different heating rates.

The carbon sequestration (CS) potential of the biochars did not exhibit the same relationship with peak pyrolysis temperature as seen between biochar R₅₀ and pyrolysis temperature (Table 4-3). CS potential is also related to biochar yield and carbon retained from that of the original feedstock. Although the biochars produced at higher temperatures may have higher R₅₀ values, containing carbon which is more stable, the production of larger quantities of biochar at lower

pyrolysis temperatures and slower heating rates resulted in biochars with higher CS potentials. The biochar produced at 5 °C min⁻¹ heating rate had the highest CS potential of the three biochars assessed due to higher biochar yield combined with higher carbon content.

Harvey et al. (2012), through examining the R_{50} value of a number of biochars and the amount of carbon mineralization which has occurred after 1 year, determined that between 0.8 % and 3 % of a Class C biochar would be mineralized after 1 year, compared with 0.2 % to 1.3 % of a Class B biochar. This indicates that although increasing the temperature of pyrolysis would reduce the amount of carbon added to soil in biochar, this would also reduce the carbon mineralised over time after soil addition. Discussion regarding mineralization rates of biochar by Cheng et al. (2008) and Foereid et al. (2011) indicates that the degradation of biochar may occur at two rates, which is often modelled as a 'two pool' method (Section 2.2.5.2). Degradation initially occurs at a fast initial rate until any labile fraction is degraded, and the remaining recalcitrant fraction then degrades over much longer timescales. Foereid et al. (2011), for example, determined that carbon mineralization over timescales of 2, 100 and 2000 years would see 9.8 %, 11.7 % and 20.7 % of biochar degraded respectively. This highlights rapid initial degradation, followed by a marked reduction in degradation rate over time. This implies that, where biochars are produced solely for carbon storage purposes, the required timescale for carbon storage must be considered when determining which pyrolysis conditions should be used. Where a storage period of decades to hundreds of years is required then a Class C biochar produced at 400 °C may be sufficient, whereas if carbon storage of thousands of years is required then higher pyrolysis temperatures, resulting in lower initial carbon quantities, may be necessary.

4.6.7 Macro-nutrient content

Pyrolysis peak temperature and heating rate affected the concentrations of some macronutrient species in biochars produced from sugarcane bagasse, whilst having little or no effect on other species.


Figure 4-16: Macro-nutrient content (g kg⁻¹) (dry basis) of sugarcane bagasse biochars produced under (a) different pyrolysis temperature regimes and (b) different pyrolysis heating rate regimes. Macro-nutrient species shown (left to right) are phosphorus (P), potassium (K), calcium (Ca) and magnesium (Mg). N.B. Scales vary between nutrient species.

Most notably, biochar phosphorus concentration was increased with increasing pyrolysis temperature. Hossain et al. (2011) related increasing phosphorus content and pyrolysis temperature to the inorganic nature of the phosphorus contained within the feedstock, meaning phosphorus is retained and concentrated with pyrolysis. A similar but smaller trend was seen with calcium concentration. Potassium content increased as peak temperature was increased from 400 °C to 600 °C, but no further increase was seen at 800 °C, perhaps due to the onset of volatilisation. A small decrease in calcium concentration was seen with increasing heating rate which may indicate that calcium is increasingly evolved from the biochar at faster pyrolysis heating rates.

4.6.8 Gas characteristics

As well as affecting yields, changing the peak temperature of the pyrolysis process also affected the composition of the syngas produced. Calorific value of the syngas from sugarcane bagasse increased with increasing peak pyrolysis temperature from 6.6 MJ m⁻³ at 400 °C, to 14.5 m⁻³ at 800 °C. This was attributed to reductions in CO and CO₂ production and increases in production of H₂, CH₄ and other hydrocarbon species.



Figure 4-17: Syngas composition (%) from the pyrolysis of sugarcane bagasse at different pyrolysis peak temperatures of 400 °C, 600 °C and 800 °C. Gas species determined are carbon dioxide (CO_2); permanent gases: hydrogen (H_2) and carbon monoxide (CO); and hydrocarbon gases: methane (CH_4), ethene (C_2H_4), ethane (C_2H_6), propene (C_3H_6), propane (C_3H_8), butene (C_4H_8), butane (C_4H_{10})). N.B. Scales vary between gas species.



Figure 4-18: Syngas composition (%) from the pyrolysis of sugarcane bagasse at different pyrolysis heating rates of 5 °C min⁻¹, 20 °C min⁻¹, and 50 °C min⁻¹. Gas species determined are carbon dioxide (CO_2); permanent gases: hydrogen (H_2) and carbon monoxide (CO); and hydrocarbon gases: methane (CH_4), ethene (C_2H_4), ethane (C_2H_6), propene (C_3H_6), propane (C_3H_8), butene (C_4H_8), butane (C_4H_{10})). N.B. Scales vary between gas species.

Increasing the peak temperature of pyrolysis increased the production of hydrogen (H₂), methane (CH₄), ethane (C₂H₆) and propane (C₃H₈) within the syngas. Production of carbon dioxide (CO₂) was reduced with increasing peak temperature, with a large decrease of 25 % seen as peak temperature increased from 400 °C to 600 °C and a further decrease of 3.8 % at 800 °C. Carbon monoxide (CO) production increased slightly as peak temperature was increased from

400 °C to 600 °C, but was then reduced by 15.5 % as peak temperature increased to 800 °C. Increasing heating rate led to increased CO₂ production and reduced production of CO and H₂. CH₄ production was increased at 50 °C min⁻¹ after an initial reduction when conditions were altered from 5 °C min⁻¹ to 20 °C min⁻¹. A number of hydrocarbon species were not affected by changing heating rate, including C₂H₆, C₃H₆ and C₃H₈. Due to the minimal increase seen in CH₄ production combined with the increase in CO₂ and slight decreases seen in CO and H₂, the optimum heating rate for production of a gas useful for energy utilization, of those assessed here, was 5 °C min⁻¹.

4.7 Limitations and opportunities for further work

The nature of experimental work carries a risk of human or technical error introducing error into the results. Mass balances were calculated for each pyrolysis experiment in order to reduce the potential for experimental error, with a ± 5 % allowance made. Any experiments having a mass balance outside this were discarded and repeated. A wood chip feedstock was pyrolysed three times under standard conditions to check that variation between results was acceptable. Other experimental methodology was conducted using tested methodology which was, where possible, peer reviewed. All experimental equipment, such as gas chromatographs and pH meters were calibrated before use. All results were considered against trends and the wider literature, with anomalous results scrutinised to determine whether experimental or data reporting error was the causal factor.

The study assessed the characteristics of agricultural residues and their resulting biochars. This was due to the later focus of this research on the global potential of crop residue biochars to sequester carbon. The biochar characterisations made here are not, therefore, representative of biochar characteristics produced from other biomass feedstocks such as animal or municipal wastes. The results detailed here have been discussed in the context of the wider literature where possible, including that of biochars produced from other feedstocks. Analysis of other feedstock types using the methodology detailed in Chapter 3 would further add to the literature.

This research also offers insight into the effects of pyrolysis conditions on pyrolysis yields and conditions, but could be expanded in a number of areas to offer further insight. The alternative pyrolysis conditions were applied here to one feedstock type, sugarcane bagasse. This, whilst offering insight into the effects of pyrolysis conditions, may not be representative of the behaviour of other feedstocks under the alternative pyrolysis conditions. This could be explored in future research, to supplement the current literature, by assessing a number of other feedstocks under the different pyrolysis conditions used here. Three variations were explored

within this research for each of the pyrolysis conditions explored. It would be useful to widen the ranges of pyrolysis conditions explored outside those examined here. Determining any threshold temperatures which may have large impacts on a particular characteristic, for example determining whether a threshold peak temperature exists for the breakdown of biochar structure would also be a useful addition to this research. The limited number of pyrolysis conditions explored (i.e. three peak pyrolysis temperatures and three heating rates on one feedstock type) makes inference of correlations between experimental variables and product characteristics difficult, although the results here are indicative of relationships which could be further investigated. Whilst correlation analysis has been used to suggest potential relationships, an increase in sample size would help to further define these relationships.

4.8 Summary and conclusions

Biochars from eight crop residues exhibited variation in a number of characteristics. Under the standard pyrolysis conditions biochar yields were 28 wt % to 39 wt %, with relationships seen between biochar yield and ash content of both feedstock and biochar. Overall, the nutrient contents of the feedstocks and biochars were low, with the majority of nutrient species being concentrated during pyrolysis. Biochar pH range (pH 6.1 to pH 11.6) showed the majority of biochars were slightly alkaline, with relationships seen between K content and pH. A strong correlation indicated a relationship between increased peak pyrolysis temperature and increased biochar pH. The average carbon quantity retained in biochar from the original feedstock carbon was 51 %. This stored carbon decreased with increasing peak pyrolysis temperature and to a lesser extent with increased heating rate, although recalcitrance of the remaining carbon increased with increasing peak temperature. High lignin feedstocks produced high carbon biochars with high recalcitrance. Feedstock and biochar H/C and O/C ratios indicated increasing aromaticity upon charring. The biochars were either moderately or more highly degradable, with palm shell and wheat straw biochars having the highest and lowest recalcitrance values respectively. None of the biochars were of the most recalcitrant classification. Recalcitrance values for biochars with high alkali content were found to be conservative due to lowering of the oxidation temperature by these alkali species. CS potentials of the biochars were between 21.3 % and 32.5 %, and were affected by the yields of biochar and carbon content, as well as the stability of this carbon. CS potential of the biochars were decreased by increasing both peak pyrolysis temperature and heating rate.

Oil yields were between 33.5 % and 53.6 %. The oil produced would be likely to require upgrading, due to high water and oxygen content, before it would be a useful fuel source.

Syngas yields ranged from 17.6 % to 29.2 %. All of the syngases produced contained some hydrocarbon species alongside CO and H₂ which could all be utilised for energy purposes. The most dominant species in each syngas was CO₂. CO₂ content was decreased by increasing peak pyrolysis temperature, with quantities of H₂ and some hydrocarbon species increasing concurrently. This corresponded to an increase in syngas CV. CVs of the oil and gas products were considerably lower than their fossil-fuel alternatives (fuel oil and natural gas). Their energy could still be utilised as a fuel source, after some upgrading, to fuel the pyrolysis process or fulfil other fuel requirements. Any bio-oil and syngas produced within a biochar system would also represent carbon neutral emissions (disregarding any transport of feedstock or fuel to point of use) which could help to reduce emissions if replacing a fossil-fuel energy source.

In conclusion, much variability is seen in biochars which can be attributed to both feedstock variability and the influence of process conditions. Producing a biochar of particular characteristics requires consideration of both of these factors, though some characteristics may be influenced more by feedstock characteristics or process conditions than others. A number of characterisations were made here which can be used to select feedstocks or process conditions to produce biochars with particular traits. The R₅₀ index is a useful tool to estimate the stability of biochar carbon. Knowledge of the alkali content of a biochar would enable assessment of whether its R₅₀ value is a conservative estimate for that biochar. The long term CS potential of the biochars is also a useful tool for determining which biochars would be useful in biochar systems designed for carbon storage. The CS potential, dependent on the biochar yields, carbon content and stability can be used in the assessment of biochar scenarios for carbon sequestration, although a number of other considerations such as the availability of feedstock also need to be accounted for.

5 Biochar production potential within the RCPs: Methodology

5.1 Introduction

Chapter 4 discusses the potential yields and characteristics of biochars produced from a number of agricultural residues. This research and literature such as Shackley et al. (2010) and Lehmann et al. (2009) indicates that biochar from crop residues has substantial carbon content and that this carbon may have a high stability over long periods, alongside having other characteristics which may be beneficial to soils and plant growth. Discussion within Chapter 4 highlights that there can be a large amount of variability between different biochars, and often a large element of uncertainty regarding biochar characteristics and the effects of biochars in soils. Chapters 5 and 7 detail the development of a framework for the global assessment of the long-term potential for biochar production to sequester carbon in soils. The framework uses data reported in Chapter 4 alongside other current knowledge of biochar characteristics and behaviour in soils. A number of scenarios were created to explore the effects of current uncertainty and variability in biochar systems on biochar production and carbon storage potential.

Within the literature few projections have been made for the potential of carbon sequestration using biochar. Section 2.3 discusses a number of projections detailed in the literature and discusses the strengths and limitations of these previous studies. One major limitation of the current literature on carbon sequestration using biochar, as a whole, is that the current literature does not investigate in detail the effects of future changes in land use and related socio-economic metrics on biochar potential. Most studies instead use static current scenarios of land-use, or very closely related land distributions. One example of this is the study by Woolf et al. (2010) which estimates a theoretical upper potential of biochar for climate change mitigation under current conditions. Although their study aims to assess a potential for future mitigation, the only land-use conversion considered within the study is of the production of dedicated 'biomass for biochar' crops on currently abandoned or degraded agricultural land. Further to this, their study does not consider the potential impact of changing agricultural land area over time, only looking at the maximum potential under existing land-use regimes. It is highly unlikely that land use will remain unchanged into the future, therefore alternative scenarios must be explored (Alcamo et al., 2008). Efforts to model potential future changes in land-use are reliant on a number of assumptions, both physical and socio-economic in nature, which results in a number of uncertainties within the scenarios. In addition to considering only current land-use, the study by Woolf et al. (2010) does not consider the effect of changing climate on biochar production potential or stability in soils. This is another area which should be

investigated as potential manifestations of climate change, including changes in temperature and precipitation, may impact on biochar scenarios.

The Representative Concentration Pathways (RCPs) are one set of projections, developed by the climate modelling community, which provide pathways of how land use may change under different socio-economic drivers from 2005 to 2100 (van Vuuren et al., 2011a). The pathways also have projections of emissions of gas species such as long-lived greenhouse gases and aerosols which, when combined with the emissions and sinks associated with the land-use pathways, results in projections of changes in radiative forcing. Further discussion of the literature surrounding the background and assumptions of each RCP can be found in Section 2.4.1. and Annexe II.b. Sections within this chapter discuss how the RCP land-use and CO₂ emissions projections were used, alongside the biochar literature and experimental data, for the development of the biochar model and scenarios. Chapter 6 details the resulting projections of the rabon sequestration potential from a number of different scenarios, allowing for analysis of the carbon sequestration potential of the scenarios, which is detailed in Chapters 7 and 8.

5.2 Model overview

An overview of the stages of model development is shown in Figure 5-1 and the details and assumptions of each stage are discussed in Sections 5.2 to 5.5.

Key inputs to our biochar assessment model were baseline production values for 2005, projected yield changes to 2100, and the RCP land use scenarios to 2100, all used to determine potential global agricultural production for cereals crops, fibre crops, sugarcane and oil crops. Residue to product ratios (RPRs) and unused residue coefficients were then applied to agricultural residue feedstock availability for biochar production. Biochar literature and primary experimental data, including biochar yield and carbon content, were applied to determine potential yields of biochar, the carbon content of this biochar and related carbon dioxide (CO_2) values for each year from 2005 - 2100.



Figure 5-1: Overview of the model stages and inputs used to calculate the biochar production potential for each future scenario. These stages were modified slightly depending on the focus of the sub-scenario under assessment. These modifications are discussed in the relevant summary of individual scenarios.

5.3 Model development

Sections 5.2 to 5.5 detail the model development for the assessment of biochar production potential and carbon content. The subsequent methodology for the assessment of long-term carbon stability within these scenarios is detailed in Chapter 7. The parameters discussed in Sections 5.3.1 to 5.3.8 are those used in Scenario 1 unless otherwise stated. Development of alternative scenario parameters is detailed in Section 5.4.

5.3.1 Dominant crop residues

The first step in model and scenario development involved determining which crops are globally dominant and which, of these dominant crops, produce relatively large quantities of residues which may be used for biochar production. The Food and Agriculture Organisation's (FAO) statistical database (FAOSTAT) was used to analyse commodity production values for different regions of the world (FAO, 2013a).

Table 5-1: The 5 world regions and 22 sub-regions used to determine dominant crop residues, total cropland (ha) and the fraction of this total cropland (where total cropland = 1) cultivated for each crop type. The individual countries within each region are detailed in Annexe II.a.

Region	Sub-region
Africa	Eastern Africa
	Middle Africa
	Northern Africa
	Southern Africa
	Western Africa
Americas	Northern America
	Central America
	Caribbean
	South America
Asia	Central Asia
	Eastern Asia
	Southern Asia
	South-Eastern Asia
	Western Asia
Europe	Eastern Europe
	Northern Europe
	Southern Europe
	Western Europe
Oceania	Australia & New Zealand
	Melanesia
	Micronesia
	Polynesia

A database of crops with high production and high regional importance was constructed to highlight globally dominant crops. Crop yields (in hectograms per hectare (hg ha⁻¹)) and the regional land area under crop production, in hectares (ha), were sourced from FAOSTAT for each of the main crop types (FAO, 2013a). The crop yield and production data were both averaged using data for 2004 to 2006 to give an average figure for 2005 excluding any very short term anomalies, for example changes caused by extremes in weather or agronomic conditions, or errors within the statistical collection and reporting system. RPR values were determined from the literature for these crops, determining which crop types yield the largest quantities of

residue. Discussion within the literature of global crop residues includes Woolf et al. (2010) who found that 75 % of crop residues in 2001 were from cereal crops and 8 % from sugarcane. Lal (2005) found that 74 % of total residue was cereal residues, 10 % sugar crop residues, and 3 % oil crop residues. Taken together, 87 % of total residue production was from these three crop types. Following this analysis of global residues it was decided to focus on the four crop groups which have the highest potential for global residue production: cereal crops, sugarcane, fibre crops and oil crops. Table 5-2 shows the main crop types within each of the groups examined.

Crop category	Main crop types
Cereals (excluding rice)	Wheat
	Maize
	Sorghum
	Barley
	Rye
	Oats
	Millet
Rice	Rice (paddy)
Sugarcane	Sugarcane
Fibre crops	Cotton
	Jute
Oil crops	Soya beans
	Groundnuts in shells
	Cottonseed
	Linseed
	Mustard seed
	Rape
	Sesame seed
	Sunflower seed
	Olives
	Coconuts (in shell)
	Palm nuts and kernels

Table 5-2: The five categories of crop groups used within the study and the main crop types within each category, as per FAO (2013b) categorisation.

Rice, although a cereal crop, was categorised as a separate group due to differences in the requirements for residues left in the field, compared to other cereal crops. This is discussed in more detail in Section 5.3.6.1.

5.3.2 Baseline data

Each biochar scenario was developed using 2005 baseline data of crop production including crop land area, crop type and crop yield. These baseline datasets for 2005 were developed using historical data from FAOSTAT and the baseline 2005 land-use dataset for the RCPs. Arable land is defined as:

'...land under temporary agricultural crops (with multiple-cropped areas counted only once), temporary meadows for mowing or pasture, land under market and kitchen gardens and land temporarily fallow'

(FAO, 2013b)

Permanent crop land is defined as:

'...land cultivated with long term crops which do not have to be regularly replanted for several years (such as cocoa and coffee).'

(FAO, 2013b)

Definitions of each crop type are detailed in the glossary section of the FAOSTAT website (FAO, 2013b).

Data was sourced from FAOSTAT detailing, for each of the 22 world regions (Table 5-1) for 2004 to 2006, the total cropland (total arable and permanent crop land) (in hectares (ha)) and the area harvested (ha) for each of the five crop groups. Comparison of FAOSTAT production data with national inventories by Kim and Dale (2004) determined that, although some discrepancies exist, the datasets are mostly consistent. FAOSTAT data was accepted for use here as it is globally representative, consistent with most national inventories and the only official, global data source available. The regional data for total crop land and area harvested per crop type was used to calculate the percentage of the total crop land that was cultivated under each of the five crop types in 2005. Regional variation was retained at the scale of the 22 sub-regional groups. Annexe II.a details the regional crop fractions applied for each crop group. Throughout the biochar scenarios, it was assumed that although total crop land may increase or decrease within a region, the percentage of this total regional cropland dedicated to each crop type remained constant. In reality, the percentage of a region's cropland used to cultivate each crop type would be very likely to change over time due to changes in economic, technological, climatic and social factors. There is a large amount of uncertainty involved in projecting how the contribution of each crop type to total crop production will change over time due to the complex interactions of these drivers of change, and the RCP literature does not discuss this in detail. It is beyond the scope of this research to project how drivers such as demand, climate and economic considerations may change for each crop type, unfolding into the future. The method used here gives a plausible projection of future production, but in reality the fraction of each crop type, as a fraction of total production, may alter.

5.3.3 RCP land-use data

As discussed in detail in Section 2.4, the RCP scenarios are four scenarios of future emissions and land-use pathways which are projected, using current knowledge of radiative forcing potentials and climate sensitivity, to each lead to a different radiative forcing in 2100. Each RCP consists of a number of datasets, for example emissions pathways for different climatically important species and land use datasets. This research used the land-use projections of each RCP to project available residue quantities within the biochar scenarios. Within the land-use data for each RCP there are a number of separate datasets detailing a) the types of land within each grid cell and b) changes between land-use types for each grid cell. The datasets extracted from the wider RCP data for use here were annual gridded data of the fraction of each grid cell used for cropland, for 2005 to 2100, for the four RCPs. The grid format of the data represents 0.5 x 0.5 degree global coverage. The cropped area of each 0.5 x 0.5 degree grid cell was found by computing the total area of each grid cell using cosine weighted latitudes and multiplying by the cropland fraction. The area of total cropland cultivated for each of the five crop groups (see Table 5-2) was determined for each year (2005-2100), for the four RCPs.

5.3.4 Crop production

Crop and scenario specific crop yields, in hectograms per hectare (hg ha⁻¹), were applied to the data of cultivated land area for each crop type, determining annual production quantities for each crop type. These values were converted to tonnes per grid cell, per year. The baseline (2005) regional crop yields (hg ha⁻¹ yr⁻¹), taken from FAOSTAT, were the regional average yield for that crop group and whilst crop group specific, were uniform across the four RCP scenarios. Regional crop yields were averaged for the years 2004 to 2006 to eliminate any short term anomalies, resulting in an average regional crop yield per crop group for 2005. This includes all production from cropping systems which operate more than one cropping cycle per year (i.e. dual or multiple harvests per year) as may be seen, for example, with some rice cultivars (Walcott et al., 1977) or where more than one crop is grown on one land area.

The development of the agricultural scenarios within each RCP used scenario drivers such as population, dietary demands and crop yields to project land use requirements and distributions. Crop yields within each RCP are assumed to change over time, simulating influences of factors within the scenario such as technological development or socio-economic factors, with each RCP having different assumptions. The literature surrounding each RCP was examined separately to determine its underlying assumptions affecting crop yields. The assumptions made to project future crop yields in Scenario 1 are based on the assumptions of each individual RCP pathway, and where necessary on the scenario assumptions and background literature underlying each RCP. Other scenarios of crop-yield change were also assessed to examine the effects of alternative crop yield pathways (Sections 5.4.1 and 5.4.7). Detailed analysis of crop yields within the RCPs can be found in Appendix II.b and a summary of the assumptions found in Table 5-3. Discussion of changes in crop yields within the RCP literature, and yield data derived from the surrounding literature, is often differentiated by the developed or developing status of nations, without explicitly stating these classifications, therefore the current development status of nations was determined. A number of organisations around the world classify the development status of nations. Figure 5-2 shows that many regions of the world are classified by the World Bank as developing, with the exception of North America, Canada, Western Europe, Southern Europe, Japan, Australia, New Zealand, Saudi Arabia and Oman. Those nations currently classed as developing by the World Bank (2011) are detailed in Annexe II.a.



Figure 5-2: Map of average income groups used to classify current developed and developing nations (World Bank, 2011).

The FAO's regional classification data details the developed regions of the world as: Northern America, Canada, Europe, Japan and Australia and New Zealand (United Nations Statistics Division, 2013). This correlates well with the World Bank classifications, with the exception of Saudi Arabia and Oman. Within this research, the development classifications from the more recent FAO classifications were used. Table 5-3 shows the annual crop yield change (% yr⁻¹) for

each 10th year from 2010. The values may change in the years between those shown, but the general trends can be seen from the decadal summary.

Table 5-3: Summary of the annual crop yield change (% yr⁻¹), for the four RCPs, for each 10th year.For each RCP, crop yields are differentiated by the developed or developing status of the region.

		Annual crop yield change (% yr ⁻¹)									
RCP	Development status	2010	2020	2030	2040	2050	2060	2070	2080	2090	2100
2.6	developed	0.55	0.55	0.55	0.55	0.55	0.36	0.36	0.36	0.36	0.36
2.6	developing	0.55	0.55	0.55	0.55	0.55	0.36	0.36	0.36	0.36	0.36
4.5	developed	0.9	0.9	0.9	0.81	0.71	0.62	0.53	0.44	0.34	0.25
4.5	developing	1.1	1.1	1.1	0.98	0.86	0.74	0.61	0.49	0.37	0.25
6	developed	1	1	1	1	1	1	1	1	1	1
6	developing	1	1	1	1	1	1	1	1	1	1
8.5	developed	1	1	1	1	1	1	1	1	1	1
8.5	developing	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5

The RCP scenarios do not include the potential impacts of climate change on crop yields therefore the crop yields used in biochar Scenario 1 (Table 5-3) also do not include climate change impacts. In reality, as the climate changes there may be a number of related impacts on crops. A set of alternative crop yield scenarios exploring the effects of different crop yield assumptions (without climate change impacts) and the related effects on biochar production are detailed in Section 5.4.1. Scenarios exploring the effects of climate change related crop yield impacts are detailed in Section 5.4.7.

5.3.5 Crop residues

The crop production quantities calculated represent the marketable commodity of the crop excluding the residues and losses during harvest (FAO, 2013b). The total residue from each crop group is therefore a related factor and was calculated from the production quantities using residue factors from the literature. Jolli and Giljum (2005) discuss two methods for estimating residue quantities from agricultural production, the first method calculating total (gross) residues and the second method calculating total unused residues. The first method involves applying a coefficient for the unused proportion of crop to that of the calculated commodity. This coefficient may be a harvest index (HI) which is defined within the literature as the ratio of grain yield to the total above ground biomass (Equation 5-1) or may be the related residue to product ratio (RPR), also known as the straw to grain ratio (Equation 5-3).

$$HI = \frac{X}{Y} = \frac{X}{X+S} \tag{5-1}$$

(Huehn, 1993)

Where X is grain yield, Y is the above ground biomass at harvest/maturity and S is the straw (residue) weight at crop maturity.

The grain to straw ratio (GSR), also called the crop to residue ratio is the ratio of harvested product to residue (Perlack et al., 2005).

$$GSR = \frac{X}{S} = \frac{X}{Y - X}$$
(5-2)

(Huehn, 1993)

Related to the GSR is the residue to product ratio (RPR), also called the residue to crop, straw to grain, or residue to grain ratio.

$$RPR = \frac{S}{X} = \frac{Y - X}{X} \tag{5-3}$$

(Huehn, 1993)

The terms HI and RPR are related, although differences between the methodologies used to take measurements and calculate the two values, such as the amount of plant harvested and height of stubble, can mean that the values are often not easily interchangeable (Huehn, 1993). Explicit detail of the calculation methodology is required for the RPR to be properly calculated from the HI and vice versa. The second method discussed by Jolli and Giljum (2005) directly applies a coefficient of unused crop residue to the harvest area. The 'unused crop' coefficient is determined using the HI of crops and the amount of crop which is not used for other purposes (technical availability). This unused crop coefficient is then applied directly to the land area, giving an amount of unused residue per area. Within our study the availability of residues for biochar production, within each scenario, is dependent on a number of factors including the amount of residue produced relative to commodity, the requirements for residues to remain in situ for soil nutrient purposes, and competing uses for the resource such as for fodder and biofuel production. The first method detailed by Jolli and Giljum (2005), applying a coefficient to calculate total (gross) residue, was used in our study to enable the total residue quantity and unused residue factors to be considered as separate parameters within the biochar model.

Within the literature on crop residues, a number of RPR values for different crop types are discussed. These values are often average figures and actual values may vary with crop strain and growing conditions (Smil, 1999). Projection of future changes in RPR within the literature range from decreasing to increasing RPRs (Hoogwijk et al., 2003). A decrease in RPR may be seen, for example, in scenarios where crops types with increased yields but current or reduced residue quantities in relation to the grain yield are utilised. An increase in RPR may be seen, for example, where crop varieties which obtain similar yields to present but produce more residue are utilized. This could be within a heavily biofuels or biochar focussed scenario, where crop residue production is desirable. Potential future changes in RPR are explored in Section 5.4.3.

To determine residue quantities for Scenario 1 RPR values were used to determine how much crop residue would be produced alongside the commodity values calculated. Ranges of RPR values from the literature were collated for each of the crop types.

Table 5-4: Average residue to product ratio (RPR) values used Scenario 1 and other scenarios unless specified. Rice residues were separated into rice straw and rice husk, sugarcane residues were separated into sugarcane leaf and sugarcane bagasse due to the very different RPR values for these residues. Averaging over these residues for the crop group would result in an unacceptable skewing of the RPR value that could be easily avoided due to the separate crop group categories for rice crops and for sugarcane crops throughout the model.

Crop group	Average residue to product				
	ratio (RPR)				
Cereals	1.22				
Rice (straw)	1.41				
Rice (husk)	0.27				
Fibre crops	0.2				
Oil crops	1.64				
Sugarcane (leaf)	0.1				
Sugarcane (bagasse)	0.4				

Although RPRs are specific to each crop, cultivar and growing conditions (for example: region, climate, environmental stresses), it was beyond the scope of this study to apply each specific crop RPR and therefore some generalisations were assumed. The values used, for Scenario 1 and other scenarios unless specified, were averages across the crop group (Table 5-4). Some

exceptions were made, for example separate RPR values were applied to rice straw and rice husk due to the very different RPR values, and primary experimental knowledge on biochar properties from these two residue types (see Chapter 4).

Smil (1999) discussed that some RPR values within literature do not account for stubble left in the field. This extra stubble was not accounted for here due to often poor reporting within the literature, often not detailing whether field stubble was included in the RPR value. It has been assumed here, therefore, that all RPR values are inclusive of field stubble. This may lead to a larger quantity of residue produced than that determined here. This assumption ensures that sufficient residues remain in the field for soil quality purposes but may, in some cases, lead to more residues remaining in the field than necessary. This was deemed to be an acceptable method of dealing with this uncertainty, avoiding the risk of developing a scenario which removes too much residue from the field.

5.3.6 Competition for crop residues

A number of sources of competition exist for agricultural residues, including for the maintenance of soil quality by leaving some residues in situ, use as fuel, fodder and within manufacturing processes. The following discussion details these sources of competition for crop residues and how they are addressed within this study.

5.3.6.1 Residues left in-situ

Some crop residue is often required to remain on soils after harvest to avoid wind and water erosion and to maintain soil organic matter, nutrients and soil structure. Residues may also be later returned to the field for nutrient recycling by mulching or after incorporation into substrates such as animal wastes or compost. The amount of in-situ residue necessary to maintain healthy soils varies from system to system, for example rice paddy agriculture does not require any residue to remain, whereas other systems may need up to 50 - 70 % of the residue to remain (Andrews, 2006, Perlack et al., 2005). Safe levels of residue removal are highly dependent on soil type, yield and management practices, with complex relationships existing between the quality of residues, climate, soil type, topography and soil management practices such as tillage and the addition of nitrogen fertilizers (Andrews, 2006). Lindstrom (1986) found that adverse effects of residue removal from corn (maize) systems decreased with decreasing residue removal, but this effect plateaued at 30 % residue removal and below. This indicates that 30 % of residues could be removed without a detrimental effect on soils. Lal (2005) also reported evidence that 20 - 40 % of residues can be safely removed from fields. Woolf et al. (2010) discuss a range of safe removal rates within the literature, from < 25 % for soils prone to

erosion, to 70 % in a no till system. Their biochar assessment used three levels of residue removal, at 25 %, 35 % and 45 % extraction, whilst noting that the higher removal rates may require changes in soil management practices to retain soil quality. The exact conditions of each system within this assessment, particularly into the future, are not currently known and therefore assumptions are made in order to generalise system types. Average values from the literature regarding the amount of residue required to remain in situ for a generalized system are used.

5.3.6.2 Residues burnt in the field

Within some agricultural systems crop residues are burnt in the field. This is common in ricegrowing systems, where residues are not required to provide cover from erosion, but the practice does occur in other growing systems (Kroeze et al., 1996). Burning may take place for a number of reasons, including to reduce the risk of pests and diseases remaining in soils between harvests, and to reduce the difficulties of preparing and planting a new crop with previous residues remaining in situ (Smil, 1999). The United Nations Environment Program (UNEP) estimated that approximately 25 % and 10 % of total crop residue is burnt in field for developing and developed nations respectively. These figures are conservative estimates as they do not include any residues burnt as fuel. The minimum estimates including residues burnt as fuel are 33 % and 15 %, and maximum estimates are 45 % and 25 % of residues burnt in developing and developed nations respectively (UNEP, 1995). In Scenario 1 crop residues usually burnt in the field were assumed instead to be collected for biochar production using the average values from the UNEP (1995) analysis.

5.3.6.3 Competing uses for crop residues

Apart from residues which must remain in situ and those burnt in the field, a number of other uses currently exist for agricultural crop residues. These uses vary regionally, with some residue types being utilised in some areas and not in others. Crop residues are commonly converted to biofuels, burnt as fuel, used as animal feed, building materials, pulp materials, as a mushroom cultivation substrate and for chemical extraction (Smil, 1999, Sud et al., 2008). For each RCP scenario a residue availability factor for each crop category was applied to the gross residue production quantities determining how much residue may be available, within each scenario, for biochar production.



Figure 5-3: Schematic showing the calculation stages of residue availability within the framework model.

Each residue availability factor is crop category specific, accounting for specialised uses. The utilization of crop residues will, in reality, vary spatially within the crop categories used here. It is beyond the scope of this study to project this spatial variation in crop residue utilisation with average values assumed here aggregated at the scale of developed and developing countries. Residue availability assumptions for the main biochar scenario (Scenario 1) are summarised in Table 5-5. Annexe II.c details the residue availability assumptions of Scenario 1 regarding residue availability, for the different crop categories.

Table 5-5: Residue availability assumptions for Scenario 1 (% of total (gross) residue). Values are differentiated by crop group and regional development status.

	Residue availability (%)							
Scenario	Cereals	Rice	Rice	Sugarcane	Sugarcane	OII	Fibre	
		(straw)	(husk)	(bagasse)	(trash)	crops	crops	
Developing	55	70	100	100	50	30	30	
Developed	40	70	100	100	50	30	30	

5.3.7 Biochar production

5.3.7.1 Yields of biochar

The biochar scenarios assumed a uniform production method for biochar throughout the scenario period (2005 – 2100), with crop specific biochar yields remaining constant both spatially and temporally. The pyrolysis process was assumed to be a modern technological process enabling higher biochar yields alongside oil and gas capture, also enabling good levels of yield and emissions control.

Chapter 2 detailed typical biochar yields from the literature, and also discussed the variation which may be seen in yields of biochar from pyrolysis, due to factors such as feedstock and process conditions. Average biochar yields determined experimentally from the slow pyrolysis process used in Chapter 4 were used to apply a coefficient of biochar yield to the scenarios of residue availability for Scenario 1. Chapter 4 saw a range of biochar yields from 28 % to 39 % for crop residues pyrolysed at 600 °C with a heating rate of 5 °C min⁻¹. The experimental results for biochar yield and other characteristics under pyrolysis conditions of 600 °C and 5 °C min⁻¹ were used to calculate the values used within biochar Scenario 1 and all other scenarios unless stated otherwise. Although higher values were determined for biochar fixed carbon content at peak pyrolysis temperatures of 800 °C, the consideration of other factors such as increased biochar yield and surface area of biochars produced at 600 °C was seen to justify a potential small forfeit in fixed carbon and recalcitrance (R₅₀). This would most likely optimise the soil amendment properties of the biochars, making the biochar product more desirable and economically competitive. Biochar yields of 30 % for cereal residues (excluding rice) and oil crop residues, 28 % for fibre crops and sugarcane residues, and 39 % for rice residues were assumed for Scenario 1.

5.3.8 Biochar carbon content

For Scenario 1 the experimental values for total elemental carbon content of 75.3 % for cereal biochars (excluding rice), 84.7 % for oil crop residue biochars, 54.5 % for rice residue biochars, 83.2 % for fibre crop residue biochars and 88.6 % for sugarcane residue biochars were used (See Section 4.3.2). Using the total elemental carbon content, from ultimate analysis, gives results for the total carbon that would enter the soil with the biochar, including any volatile material, whereas use of the total fixed carbon, from proximate analysis results, would disregard this volatile carbon content and therefor lead to an incorrect value for carbon entering the soil.

Following the calculation of the carbon content of the biochars, the related quantity of CO_2 was calculated. A conversion factor of 3.67 was applied to the carbon values, as one molecule of CO_2 is roughly 3.67 times the mass of one molecule of carbon (C). It must be noted that the CO_2 values are related to the carbon content of the biochars before addition to soils. Some of the biochar carbon may be released over time, as CO_2 , upon degradation of the biochar after addition to soil. The net carbon storage potential of the biochars, after addition to soil, and related CO_2 quantities are discussed in Chapter 7.

5.4 Alternative scenario drivers and assumptions

In addition to Scenario 1, a number of other scenarios were designed to explore the sensitivity of results to the assumptions of the main scenario. The parameters explored, and the related sub-scenario group, are detailed in Table 5-6. Further detail of each alternative scenario can be found in Sections 5.4.1 to 5.4.7. Within the alternative scenarios all assumptions are the same as for Scenario 1 unless specified.

Scenario	Scenario driver	Assumptions			
Main Scenario					
1	Main/mean scenario assumptions	See Section 5.3			
Crop yields					
2a	No crop yield change	0 % yr ⁻¹			
2b	Optimistic crop yield	Scenario 1 + 50 % yr ⁻¹			
2c	Alternative convergence point of	0.25 % yr ⁻¹ in 2050 for RCP 4.5			
	crop yield to 0.25 % yr ⁻¹ for RCP				
	4.5				
Land-use change					
За	No land-use change	2005 land use distribution kept			
		constant			
3b	RCP crop land without dedicated	Subtraction of biofuel land from total			
	biofuels	cropland for RCPs 2.6, 4.5 and 6			
Residue to product ratio					
4a1	Small RPR decrease	RPR of 0.75 (2005) to 0.71 (2100)			
4a2	Large RPR decrease	RPR of 0.75 (2005) to 0.14 (2100)			
4b	Increasing RPR	RPR of 0.75 (2005) to 1.28 (2100)			
Crop residue availabi	lity				
5a	Low availability	25 % availability			
5b	Medium availability	50 % availability			
5c	High availability	75 % availability			
5d	Conservative Woolf et al (2010)	See Table 5-7			
5e	Optimistic Woolf et al (2010)	See Table 5-7			

Table 5-6: Summary of biochar scenario groups indicating the parameters explored within each sub-scenario set.

Biochar yield		
6a	Low yield	25 %
6b	High yield	63 %
Biochar carbon conte	ent	
7a	Low carbon content	60.2 %
7b	Medium carbon content	72.5 %
7c	High carbon content	89.0 %
Climate change impa	acts	
8a	Min temp change, mean yields	See Section 5.4.7
8b	Mean temp change, mean yields	
8c	Max temp change, mean yields	
8d1	Min temp, min yields	
8d2	Mean temp, min yields	
9d3	Max temp, min yields	
8e1	Min temp, max yields	
8e2	Mean temp, max yields	
8e3	Max temp, max yields	
8f	Assumptions of Kyle et al (2014)	
Climate change with	adaptation	
9a	Min temp change, mean yields	See Section 5.4.7
9b	Mean temp change, mean yields	
9c	Max temp change, mean yields	
9d1	Min temp, min yields	
9d2	Mean temp, min yields	
9d3	Max temp, min yields	
9e1	Min temp, max yields	
9e2	Mean temp, max yields	
9e3	Max temp, max yields	

5.4.1 Scenario 2: Crop yields

Each RCP uses a prescribed crop yield and land-use combination to provide the agricultural resources needed to meet demand within the scenario. These projections of crop yields within the RCPs are not certain pathways of how crop yields will change into the future. Scenario 2 looks at the effect of alternative rates of crop yield change over time. This may mean that, used

alongside the prescribed land-use of the RCP, the demand for crops and other agricultural produce may not be met within the RCP. The analysis does give an indication of the impact of different rates of crop yields on crop production and subsequent biochar potential. It also indicates how sensitive production within the RCP is to the yield assumptions of the pathway. Three scenarios of alternate crop yields were developed: Scenario 2a, no crop yield increase; Scenario 2b, optimistic crop yield increase; Scenario 2c, alternative convergence period for RCP 4.5. The assumptions of each alternative crop yield scenario are summarised in Sections 5.4.1.1 to 5.4.1.3. These scenarios of alternative crop yields are not intended to simulate the impacts of climate change, which are explored in Scenarios 8 and 9, Section 5.4.5.

5.4.1.1 Scenario 2a: No crop yield increase

Scenario 2a used the baseline yields for 2005 throughout the scenario, keeping them constant to 2100. This determined how much of the biochar production potential of Scenario 1 could be attributed to the assumed crop yield increase over time.

5.4.1.2 Scenario 2b: Optimistic crop yield increase

Scenario 2b used rates of crop yield increase which are higher than those projected within the RPCs, perhaps due to factors such as new crop cultivars, genetic modification or higher rates of technological development than prescribed within the RCP. Thomson et al. (2011) discussed an optimistic yield scenario where the annual yield increase assumptions are increased by 50 %. For Scenario 2b this assumption of Scenario 1 crop yield increase + 50 % was applied to the rates of crop yield change (see Annexe II.d).

5.4.1.3 Scenario 2c: Convergence rates for RCP 4.5

In Scenario 1, the annual rate of crop yield increase for RCP 4.5 was assumed to converge to 0.25 % yr⁻¹ in the year 2100. Due to uncertainty in the RCP scenario literature about when this convergence occurs, Scenario 2c examined an alternative convergence point of 2050. This gave biochar production potential in RCP 4.5 using the earliest and latest potential yield rate convergence points. Convergence to 0.25% yr⁻¹ yield increase was implemented using an incremental change every 10 years for consistency with Scenario 1 yield assumptions.

5.4.2 Scenario 3: Land use

5.4.2.1 Scenario 3a: No change in land use

Scenario 3 applied the crop land distribution of 2005 to the entire biochar scenario period, from 2005 to 2100. The assumption that land-use will not change over time is unlikely to manifest in reality. However, the use of a static land-use scenario was used to determine how much of the

biochar production potential of Scenario 1 could be attributed to the changing agricultural land area projected within each of the RCPs.

5.4.2.2 Scenario 3b: Biofuels land

Scenario 3b examined the impact which subscribed biofuels land has on the biochar production potential of Scenario 1. Biofuels produced from crops (either purpose grown energy crops or crop residues) are considered as an energy source within all of the RCP scenarios. They are categorised in different ways within the development and datasets of each RCP scenario. Within RCP 2.6, RCP 4.5 and RCP 6 biofuels are categorised as crops and, as such, are included in the datasets of total crop fraction provided by the RCP modelling team and used here for scenario development. RCP 2.6 and RCP 6 treat biofuels as a number of different crop types, RCP 4.5 treats biofuel crops as herbaceous crop species. RCP 8.5 considers biofuels to be part of the wood harvest, which is not included in the dataset for total crop land fraction. Separate biofuels datasets of the gridded fraction of land used for biofuel cultivation are provided by the RCP modelling teams for RCPs 2.6, RCP 4.5 and RCP 6. The RCP 8.5 biofuels dataset is comprised of the wood harvest for fuel within the scenario. This was adapted by Hurtt et al. (2009) into a dataset of the carbon content of the wood harvested for energy production within RCP 8.5.

Whilst the main biochar scenario set (Scenario 1) uses the fraction of total cropland datasets as provided by the RCP development teams, Scenario 3b subtracts the biofuels land area from the total cropland area for RCPs 2.6, 4.5 and 6. The cropland dataset for RCP 8.5 is not altered in Scenario 3b as biofuels are not included in the original cropland datasets by the RCP 8.5 team.

5.4.3 Scenario 4: Residue to product ratio (RPR)

Scenario 4 examined the effect of alternative RPRs on the biochar production potential. Historically, a high crop RPR was desirable due to the high economic value of straw. More recently the economic value of grain has increased in comparison to that of straw, leading to a desire to produce crops with higher grain yields relative to straw yield (Reddy et al., 2003, Sinclair, 1998). This change in RPR has occurred through crop and cultivar selection alongside other management techniques (Reddy et al., 2003). Imhoff et al. (2004) discuss that RPRs of 0.14, 0.71 and 1.28 are low, intermediate and high estimations of generalised RPR. These RPR values are generalised across a number of crop types and are average values of current RPRs, which may not be representative of the potential RPRs of the future. Future projections of changes in RPR are not discussed widely within the available literature. It is possible that future crop RPRs may continue to decrease or may begin to increase, dependent on factors such as technological improvements, cultivar choice, genetic alteration, and crop management (Hoogwijk et al., 2003). The future economic potential of residues may be a main driver in the changes of RPRs, meaning that RPRs may increase in a future biochar or biofuels focussed scenario, where residues can be easily utilized for economic gain. In other future scenarios crop residues may become less desirable, for example where the majority of residues are regarded as waste and the main focus is on increasing yield of grain from the crop. Regional differences in RPR are beyond the scope of this work, with potential changes in the global average RPR of the different crop groups being examined within Scenario 4. An initial RPR of 0.75 (in 2005), which is the average value of the RPRs used for all of the crop types in Scenario 1, is assumed for the Scenario 4 sub-sets. For Scenario 1 separate RPR values were assumed for the different crop groups. For Scenario 4 one average RPR value for all crop groups is assumed, due to scant literature regarding projected changes in RPRs. Total rice residue and total sugarcane residue is therefore calculated, where Scenario 1 separates these groups into sub-residue types. Using the relative contribution of each residue type (i.e. rice straw and rice husk) to the total residue quantity for each crop type (i.e. total rice residues) from Scenario 1 enabled the application of availability factors for each residue sub-set to be applied in Scenario 4. The values used for the contribution of each residue type to total residues for their crop group in Scenario 4 were: rice straw, 84 %; rice husk, 16 %; sugarcane leaf, 20 %; sugarcane bagasse 80 %.

5.4.3.1 Scenario 4a: Decreasing RPR

A decrease in RPR over time was examined in Scenario 4a. This represents a continuation of the recent historical trend of decreasing RPR which has been attributed to faster increases in grain yield relative to residue increases (de Leeuw, 1997). The rate of this decrease is not discussed in detail within the literature so we make two assumptions below. The changes in RPR were made incrementally each decade so that the final RPR is reached in 2090 and maintained until 2100. This 10 year increment period is in-keeping with the changes applied to other scenario variables, such as crop yield increase. The initial RPR of 0.75 was reduced over the scenario period to the intermediate RPR of 0.71 discussed by Imhoff et al. (2004). The RPR of 0.75 was reduced over the scenario period to the low RPR of 0.14 discussed by Imhoff et al. (2004).

5.4.3.2 Scenario 4b: Increasing RPR

This scenario is representative of a future pathway where crop residues become more desirable, holding more economic value. The increased desirability of crop residue is not likely to occur at the expense of grain yields, with all parts of the plant expected to gain economic value in such a scenario (Lorenz et al., 2010). Because of this, only marginal increases in RPR would be expected in a future scenario of increasing RPR. The initial RPR of 0.75 was increased over the scenario period to the high RPR value of 1.28 discussed by Imhoff et al. (2004).

5.4.4 Scenario 5: Crop residue availability

Scenario 5 examined the effect of potential changes in the availability of crop residues on biochar production potential. Scenarios 5a to 5c used average residue availability values across all residue types, of 25 %, 50 % and 75 % availability respectively. Although these average values are generalisations of the availability of crop residues, which in reality will vary with crop type and spatial and temporal coverage, these generalisations were used to provide general estimations of crop residue availability under different assumptions. Scenarios 5d and 5e used the conservative and optimistic residue availability assumptions of Woolf et al. (2010) respectively to assess the impact of the assumptions of similar studies on the biochar production potential here. With regards to residue availability Woolf et al. (2010) discussed cereal, rice and sugarcane residues which are relevant to this study. They also discussed the availability of other 'biomass crops' which has been used here to represent oil and fibre crop residues. The values are aggregated to a global scale, rather than by developed and developing region status in Scenario 1 here, due to the global scale of aggregation in Woolf et al. (2010). The largest change in residue availability in Scenario 5d, relative to Scenario 1, was the reduction in cereal residue availability to 8 %. Rice straw, rice husk and sugarcane bagasse residues remained at the level of availability seen in Scenario 1, as this is the same as availability discussed in the conservative scenario of Woolf et al. (2010). Scenario 5e used 'the maximum potential' values from Woolf et al. (2010). The 20 % availability of cereal residues was lower than the values used for cereal residue availability in Scenario 1. Woolf et al. (2010) assume 45 % removal rates from the field of cereal crop residues, 25 % use as animal fodder, leaving 20 % for use in biochar production.

Table 5-7: Summary of crop residue availability (%) for Scenarios 5d and 5e, the conservative and maximum scenarios of residue availability using the assumptions of Woolf et al. (2010).

		Residue availability (%)							
Scenario	Cereals	Rice	Rice	Sugarcane	arcane Sugarcane Oil crops		Fibre		
		(straw)	(husk)	(bagasse)	(trash)		crops		
5d	8	70	100	100	25	30	30		
5e	20	90	100	100	75	75	75		

The cereal residue availability assumptions made in Scenario 1 included the use of residues which are currently burned in the field or used for energy production, which are not considered by Woolf et al. (2010), contributing to the difference in availability factors. Scenarios 5d and 5e

highlight the impact which alternative residue availability assumptions have on the biochar production potential of Scenario 1.

5.4.5 Scenario 6: Biochar yield

Scenario 6 explored the assumptions of biochar yield, using the average low biochar yield (25 %) and average high biochar yield (63 %) from the literature to give conservative and optimistic accounts of biochar production potential. The scenarios, termed Scenario 6a and 6b represent low and high biochar yield respectively. In reality variation in biochar yield will occur in biochars produced from different feedstocks, and both spatially and temporally depending on factors such as process type and conditions. The values here were used to give an indication of the variability which may occur around the biochar yield assumptions of Scenario 1.

5.4.6 Scenario 7: Biochar carbon content

Scenario 1 used the experimentally derived values for total elemental carbon content of biochars which are discussed in Section 5.3.8. These values assume that all biochars produced from each residue type will have the same carbon content as those derived experimentally. In reality this is dependent on a number of factors including the provenance of the residue, for example plant strain and growing conditions, and the biochar process type and conditions. To explore the effects that variation in biochar carbon content may have on total biochar carbon content, values representing low and high carbon content values from the literature were used to determine a possible range. Annexe II.e details carbon contents from the literature where biochars were produced between 500 °C and 600 °C under slow and fast pyrolysis. These values for total carbon content were used to determine minimum, mean and maximum values of 60.2 %, 72.5 % and 89.0 % which were applied to the biochar quantities for Scenarios 7a, 7 b and 7c respectively.

5.4.7 Scenarios 8 and 9: Climate change impacts (with and without adaptation).

The climate change scenarios (Scenarios 8 and 9) examined the potential effects of climate change impacts on the biochar production potential of Scenario 1. The baseline scenario assumptions, before the effects of climate change, were taken from scenario 1 except where specified. There may be some yield increases with small temperature increases, and some crops may experience yield increases at relatively higher temperatures, though the dominant impact is expected to be decreased yield (Porter et al., 2014). The impact on crop yields is expected to increase with increasing temperature, meaning that impacts are likely to increase in magnitude, from lower to higher, across RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively.

The work both of Kyle et al. (2014) and Porter et al. (2014) were used to determine changes in crop yield, related to the level of climate change projected for each RCP. These crop yield assumptions were applied within biochar Scenario 8 and 9 scenario sets, where Scenario 8 assessed the impact of climate change induced temperature change, relative to Scenario 1, with no adaptation measures employed and Scenario 9 assessed the impact of climate change induced temperature adaptation measures.

The crop yield impact projections of Porter et al. (2014) contribute to Scenarios 8a-e, and 9a-e. The literature on projected temperature change from the IPCC (2013) report was used to determine projected temperature changes up to 2100 for each of the RCPs. The IPCC literature was then used to apply crop yield impacts relating to these projected temperature changes. Scenario 8f was developed using the work of Kyle et al. (2014) which discusses the effects which climate change may have on the agricultural projections within each RCP. Section 2.4.3 details the discussion of Porter et al. (2014) and Kyle et al. (2014).

5.4.7.1 Scenarios 8a-e and 9a-e

To project the impact on crop yields of temperature change within each RCP the data from Porter et al. (2014) was used. The projected impact on crop yield was determined from Figure 7-4 of the Porter et al. (2014) report, which shows projected change in crop yield (%) from the baseline crop yields, at 1 °C intervals for a 1 °C to 5 °C temperature increase. This data was linearly extrapolated between each 1 °C temperature interval to give a projected yield change for each 0.1 °C increase in temperature. Using the 95 % confidence interval values were determined for the lowest, mean and maximum yield projections for crop yield impacts both with and without the 'simple agronomic' adaptation measures (this data is displayed in Annexe II.f). These changes in crop yield, relative to the crop yields of Scenario 1, were applied alongside the projected changes in global mean surface temperature projected for each RCP to project the changes in crop yield which may occur with the manifestation of climate change induced temperature change for each RCP pathway. Scenarios using both the non-adaptation and adaptation crop yield impact data were developed (Scenarios 8 and 9 respectively). Scenarios 8a to c and 9a to c assume the mean projections of crop yield impact with the minimum, mean and maximum temperature projection for each RCP (Scenarios a, b and c respectively). Scenarios 8 d and 9 d apply the lowest yield projections with the minimum, mean and max temperature projections (d1, d2 and d3) respectively. Scenarios 8 e and 9 e apply the highest yield projections with the min, mean and max temperature projections (d1, d2 and d3) respectively.

5.4.7.2 Scenario 8f

A similar methodology to that discussed in Section 5.4.7.1 was used to determine the impacts that the crop yield changes projected by Kyle et al. (2014) would have on the potential for biochar production. The data for crop yield impacts with no species relocation was used to determine a percentage change in crop yield for each crop type of the biochar scenarios, relative to the baseline changes of Scenario 1, for each RCP. The crop yield impacts of each RCP scenario were applied to the initial crop yield assumptions of Scenario 1. The decadal data was linearly extrapolated to determine an annual variation from the baseline yield. Projections of changes in corn and wheat yields were averaged for each year to produce a value for cereal crops. Bioenergy crop projections from Kyle et al. (2014) were applied to the yield projections for fibre crops in the biochar scenarios, as no fibre crop category was detailed in Kyle et al. (2014). This data projection set was chosen as a number of herbaceous fibre crops can also be used for bioenergy production, therefore it was deemed to be the closest fit of the crop groups available from Kyle et al. (2014). This was done only for the dataset without adaptation measures from Kyle et al. (2014), resulting in Scenario 8f (no adaptation). The dataset of projected crop yields with crop relocation adaptation measures from Kyle et al. (2014) was not used to project an adaptation scenario here as the land-use projections would then deviate from those of the RCPs.

5.4.7.3 Scenario 9: Climate change with adaptation

Scenarios 9 a-e used the same methodology and data sources as Scenarios 8a-e, as described in Section 5.4.7, except the values for crop yields impacts with adaptation (Annexe II.f) were used alongside the temperature projections for each RCP. This gives insight into how simple crop based adaptation measures may alter the potential of biochar production, relative to both Scenario 1 and Scenarios 8 a-d.

5.5 Scenario limitations and uncertainties

There are a number of limitations which must be considered when assessing and using the data provided by these biochar scenarios. Some of these limitations apply to all of the scenarios and some just to one or two scenarios. Many of the limitations are a result of the assumptions which have been made during the development of each scenario, often incorporating underlying uncertainty. These uncertainties exist in a number of factors such as biochar properties and effects, future radiative forcing pathways and the manifestation of future climate change. Many of these uncertainties are discussed during the sections on scenario development and a number of the main uncertainties are also detailed here in summary. These scenarios should be used as a set of plausible future scenarios, although a number of other potential future scenarios are

also possible. Many of the assumptions made during the development of these scenarios have been made to enable large-scale scenario analysis. For example, it has often been necessary to develop average values of spatial data.

It is not possible to determine how future land-use will vary spatially, therefore the RCP pathways have been used as possible pathways, but with no probability of occurrence attached. In reality a number of land-use scenarios could result in the same radiative forcing pathways as the RCPs, and many other land-use scenarios may contribute to a range of other radiative forcing pathways. As the development of the RCP land-use scenarios did not consider climate change, it is feasible that some of the land prescribed as cropland within each RCP scenario may become unsuitable for crop production under a changed climate (Kyle et al., 2014). Concurrently other areas may become more suitable for crop production. Detailed information on how climate change may impact the suitability of agricultural land for particular crop types is not available for the RCP scenarios. The land prescribed as crop land within each RCP is, therefore, not changed in Scenarios 8 and 9, thus giving a representation of the impacts of climate change on crop yields and biochar production without the relocation of crop species beyond the changes originally prescribed in each RCP scenario. The biochar assessment model can also be used to assess further land-use scenarios as uncertainties are reduced. The projections of land use change within each RCP scenario are interlinked with projections of crop yield. Each scenario, dependent on underlying drivers such as population and diet, was developed by the RCP teams around a certain demand for crop production. The RCP land-use development involved the prescription of crop yields and rates of crop yield change, combined with prescribed agricultural land area to satisfy this demand. The fractional contribution of each crop group to the total regional production was kept constant within the scenarios, with the exception of Scenario 3b which explores the prescription of biofuels land within the RCPs. Drivers such as changing socio-economic pathways and/or climate may, in the future, alter the fractional contribution of each crop type to the total crop production within each region of agricultural land area. It is beyond the scope of this study to project how these crop fractions may change over time. The main focus here regarding the effects of climate change on the biochar production potential of Scenario 1 is the effects of temperature changes on crop yields within each RCP. Although changing regional crop distribution due to climate change and changing demand for particular crop types are not explored here, these are both future potential research areas which could offer further insight and development of the biochar scenarios.

The datasets used for Scenario 1 are inclusive of biofuels land, though the fraction of total cropland cultivated by each crop type applied to each region remains the same from 2005. In reality, the biofuel crop fraction will change with time. The crop fraction value has been held constant within the study due to the aforementioned difficulties in projecting changing demand in crop types. The RCP literature does not specify in detail what crop types will be used for biofuels. As the crop fractions applied include cereal crops, oil crops, sugarcane and fibre crops this was assumed here to be inclusive of most biofuel producing species. Further research regarding the assumptions of future changing demand for different crop types, and the impact of this on the contribution of crop types to total crop production, would be beneficial.

The IPCC literature summarises the latest scientific research, projecting how the climate may change with each RCP emissions pathway. This includes, but is not limited to, projections of temperature, precipitation, cloud cover and extreme weather events. Ensembles of climate models, using detailed parameters for a number of Earth processes, are used to make these projections. There are still a number of uncertainties associated with prescribing these model parameters and how the parameters are interlinked to accurately represent the Earth's complex systems. Current climate models are able to reproduce historical large-scale changes in mean surface temperature to a very high confidence (correlation between the historical observations and model reproductions is \sim 0.99). The climate models often have more difficulty making accurate regional or smaller scale projections of observed temperature changes (Flato et al., 2013). Although representation of Earth processes such as the carbon cycle and cryosphere are improving within climate models, some large areas of uncertainty remain. In order to increase the likelihood of an accurate representation of potential future climate change assessment often uses an ensemble of models, creating a range of likely values. Flato et al. (2013) discuss that the simulation of global mean temperature change is mostly good, and is often represented more accurately that other parameters such as regional temperature change and precipitation. The IPCC ranges of potential mean surface temperature change for each RCP are accepted here as representing the most current and accurate projections available. They are used here due to the validation of the projected ranges by the use of an ensemble of models, and also due to the ability of the models in the ensemble to effectively simulate large scale mean historical surface temperature changes. In reality, the temperature changes which may manifest through climate change will vary regionally, dependent on a number of factors. Localised temperature changes are not applied within the biochar scenarios. The ability of climate models to predict localised changes can be limited (Flato et al., 2013), and the application of regional temperature changes to the biochar scenarios is beyond the scope of this study. The average global mean surface

temperature change is used when applying temperature changes to the effects of crop yields within the biochar scenarios.

The impact of the projected temperature changes on crop yields also has a number of related assumptions and uncertainties. The impacts on crop yields taken from the literature are related to changes in global mean surface temperature. On a more localised scale changes may be seen in localised surface temperature which may have larger range and variance than the changes in global mean surface temperature. Due to the large scale nature of the scenarios developed here, and the scarcity of localised temperature change projections from the RCPs, changes in the global mean surface temperature was deemed an acceptable aggregate scale for temperature change. The uncertainties in more localised temperature change projections to 2100 would add another layer of uncertainty to the assessment. The biochar assessment model could be used to assess a smaller regional area, with the application of local temperature change projections if required and as uncertainty in the projected data is reduced.

The values for crop yield impacts taken from Porter et al. (2014) are mean values from a review of the literature. A range of values exist around these mean values. In order to address this, the development of scenarios 8a-e and 9a-e used these mean values and also the 5 % and 95 % confidence intervals (as indicators of lowest and highest crop yield impact values), meaning that 95 % of the crop yield impact projections assessed within the IPCC review of literature are encompassed within the scenarios assessed here. All of these yield scenarios are subject to inherent uncertainties related to the prediction of future scenarios and also the generalisation necessary to develop manageable global scenarios. A number of factors other than temperature may also affect crop yields, such as precipitation changes and change in the frequency and/or magnitude of extreme events. The assessment of all of these drivers of change is outside the scope of this study, but would add valuable insight to the scenarios during future assessment. As further literature becomes available regarding the potential impacts of the many drivers of crop yield change, and potential likelihood of the manifestation of these future impacts, new scenarios may be examined.

As discussed in Section 5.4.3, a combination of impacts on crop grain yields and residue production may occur, potentially leading to changes in crop RPR values. Scenario 4 explores future scenario of both increasing and decreasing RPR. A range of maximum, mean and minimum values for the RPR have been used within these scenarios to determine a range of possible outcomes for the total residues within each scenario. In reality RPR values may exhibit more variation than this, with variation possible regionally, temporally and both between and

within crop groups and cultivars. It is beyond the scope of this study to examine the RPRs in more detail than has been conducted here. However, the addition of more detailed RPR values, perhaps on a crop type and regional scale, would enable the projection of more accurate results. Sufficient literature on future RPRs and the impacts of climate change on RPRs is not currently available to project how RPR may be change in Scenarios 8 and 9. These scenarios therefore assume the same RPR values as Scenario 1, assuming that they remain constant over time. This simulates a uniform effect of potential changes in global mean temperature on both commodity and residue production, which may be re-assessed as further literature becomes available regarding future projections of RPR.

Assessment of the biochar yield potential of each scenario has relied upon a number of parameter generalisations, such as the RPR values of crops, biochar yields and biochar carbon content. In reality each of the parameter values applied has some variability. It is beyond the scope of this study to apply this variability in its full complexity due to the global nature of the study. As with other studies of this type it has been necessary to make some generalisations about these parameters. Saying this, the potential variability in these parameters has been explored within individual scenarios, exploring the effect which maximum and minimum values derived from experimental work and/or the current literature has on the biochar production potential. The assessment of biochar yield potential also does not account for the impacts of any extreme events such as weather, pest or disease events which may impact land-use, crop or residue production or the ability to collect and convert residues. These extreme events were not included in the scenario analysis here due to the magnitude of current uncertainties. It was not possible to account for these events during the analysis here, though it is possible to say the projections should be seen as maximum potential biochar production scenarios, with the impacts of extreme events likely to affect biochar production in reality.

Many of the caveats addressed in this study are related to either uncertainty within the current knowledge relating to the parameter, or a high level of spatial or temporal complexity which has been generalised during this study due to the global nature of the assessment. Future research developments will allow for the assumptions relating to current uncertainty to be refined, updating the scenarios as new information is disseminated. The biochar assessment model may also be adapted in the future to apply higher levels of complexity to the spatial and temporal variations in parameter values.

5.6 Summary

The assessment of biochar production potential using the RCP land-use pathways incorporated a number of parameters from both experimental research and the surrounding literature. The development of a main scenario (Scenario 1) enabled the assessment of the biochar production potential using the main or most likely assumptions. Due to the high potential variation or uncertainty in a number of the parameters, such as crop yields, residue availability and biochar yields, a number of sub-scenarios were developed, each focussing on the variability of one parameter. The parameter variation or uncertainty assessed within the sub-scenarios looked at the potential variation in: crop yields, land-use change, RPR, residue availability, biochar yield, biochar carbon content and the impacts of climate change related temperature change on crop yields.

6 Biochar production potential within the RCPs: Results and discussion

6.1 Introduction

A number of scenarios which assess the potential for biochar production within the Representative Concentration Pathways (RCPs) (van Vurren et al., 2011) are presented. Scenario 1 explored the main parameter assumptions, using the mean or most likely parameter values throughout the assessment. Sub-scenarios 2 to 9 explored the range of variation and uncertainty surrounding a number of the parameters. Each sub-scenario focused on the variation and uncertainty of one parameter, these parameters being: crop yields, land-use change, RPR, residue availability, biochar yield, biochar carbon content and the impacts of climate change related temperature change on crop yields with and without adaptation measures. The initial results and discussion presented discuss the effectiveness of the biochar model, looking at projections of cropland and spatial accuracy. Following this the biochar production potential of Scenario 1 is discussed, looking in detail at the projections at each model stage. Variance from the projections of Scenario 1 induced by the assumption of alternative parameters are then detailed to outline the uncertainty in the projections of Scenario 1, and the variance in the biochar production quantities and carbon content which may occur.



6.2 Cropland within the RCPs

Figure 6-1: Hectares of crop land used globally for crop cultivation in 2005 for all RCPs.
Figure 6-1 shows the distribution of crop land in 2005 for all four RCPs. This is the baseline year for all scenarios, from which the RCPs diverge, along separate pathways of land-use to 2100. The baseline year of 2005 sees 1.56 billion hectares of land under cultivation in each RCP. This corresponds well with data from the FAO database FAOSTAT which details that 1.54 billion hectares of land were under the cultivation of permanent of arable crops in 2005 (FAO, 2014).



Figure 6-2: Percentage grid cell area under cropland in 2000. Adapted from SEDAC (2014) under the creative commons 3.0 Attribution Licence (http://creativecommons.org/licenses/by/3.0/legalcode).

When compared with data from NASA's Socioeconomic Data and Applications Centre (SEDAC) (Figure 6-2) (SEDAC, 2014) and data from a study of global cropland coverage in 2000 by Ramankutty et al. (2008) the spatial coverage of cropland is reproduced well in the RCP datasets for 2005. Small discrepancies can be seen between the datasets, for example the RCP dataset has more prescribed land coverage in North Eastern Australia than is detailed in the SEDAC (2014) and Ramankutty et al. (2008) datasets, but generally agricultural land use coverage is a good match between the datasets. Figure 6-3 below shows how the total global cropland area (Mha) changes over time within each RCP.



Figure 6-3: Total global cropland (Mha yr^{-1}) for the four RCPs over the scenario period of 2005 to 2100.

From the initial 1.56 billion hectares in 2005, the cropland area in 2100 is 2.10, 1.12, 1.93 and 1.84 billion hectares (B ha) for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively. RCP 4.5 consistently has the smallest area of land under crop cultivation of the four RCPs. It is the only scenario which sees a major and sustained decline in cropland area, resulting in less land used for crop cultivation in 2100 than in the baseline year of 2005. This is attributed to the scenario drivers of RCP 4.5, including afforestation measures and dietary changes (Thomson et al., 2011). RCP 6 and RCP 8.5 have similar crop land area pathways throughout the period, with RCP 6 overtaking RCP 8.5 in around 2075. RCP 6 has around 91.4 Mha of cropland more than RCP 8.5 in 2100. RCP 2.6 consistently has the largest land area under crop cultivation throughout the time period. RCP 2.6 sees 975 Mha yr⁻¹ more cropland under cultivation than RCP 4.5 in 2100. Figure 6-4 shows the distribution of cropland in 2100 for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively.



Figure 6-4: Fraction of land area used as cropland in 2100 for RCP 2.6 (top left), RCP 4.5 (top right), RCP 6 (bottom left), RCP 8.5 (bottom right).

The large-scale regional distribution of land used for crop production is similar in the four RCPs in 2100 (Figure 6-4). Within each region there is often large variation in the more localised areas of cultivation, with some RCPs utilising much more cropland than others. In all regions, RCP 4.5 appears to utilise the smallest area of cropland in 2100, with cropland area in South America and Africa particularly reduced in comparison to the other RCPs. In 2100, in a number of regions, RCP 6 has the second lowest area of cropland, with, in particular, less crop land under cultivation in Africa and South America than in RCP 2.6 and RCP 8.5. In other regions RCP 6 has more land under crop production in 2100 than the other three RCPs. Some areas of India have more land under crop cultivation in 2100 in RCP 6 than in RCP 2.6 or RCP 8.5.

The total land under cultivation of the crop groups assessed within this study (see Table 5-2) in 2005 is 1.16 Gha, this is 74 % of the total permanent and arable cropland prescribed for the RCPs in 2005. The total cropland area in each RCP for the years 2005, 2025, 2050, 2075 and 2100 is detailed in Annexe III.





Figure 6-5: Distribution of the different crop types by crop category in 2005. Plots show distribution of cereal crops (top left), rice crops (top right), fibre crops (mid-left), oil crops (mid-right) and sugarcane (bottom). This represents the baseline distribution of crops for the four RCPs.

The distribution of crop types generated by the biochar model for 2005 is illustrated in Figure 6-5. The spatial variation in crop groups simulated by the model, using RCP cropland data and calculated crop fractions is representative of the distribution of these crop types in reality.



Figure 6-6: Plot detailing progress towards peak cereal crop yields, taken from Mueller et al. (2012). The plot illustrates the current global distribution of cereal crop cultivation land, for comparison with cereal cropland generated by the biochar model (Figure 6-5, top left panel). See Annexe V for licencing details for use of this figure.

Comparison of current cereal cultivation land (Figure 6-6) with the model distribution of cereal land (Figure 6-5, top left panel) shows that the model recreates cereal land distribution well. There are a few small areas which may not be prescribed as crop land within the model, for example the North Eastern region of Brazil and Angola (South Western Africa), but overall the fit of cereal crop land in the model is seen to be good.



Figure 6-7: Plot detailing progress towards achieving maximum rice yields, taken from Ray et al. (2012). The plot illustrates the current global distribution of rice cultivation land. See Annexe V for licencing details for use of this figure.

Comparison of current rice cultivation land (Figure 6-7) with the model distribution of cereal land (Figure 6-5, top right panel) shows that the model also recreates rice cultivation land distribution well. The main rice cultivation areas are defined in the model, though the model misses some small areas of rice cultivation, for example Madagascar and the North American states of Arkansas, Louisiana and Texas are not prescribed as rice producing regions within the model. This is due to the aggregation of production data from the FAO into regional data, meaning that the small levels of production in these areas are overlooked due to the larger scale of the data (for example production was determined from FAOSTAT for the North American region, of which the vast majority is not rice producing). The oversight of these rice producing areas was deemed to be acceptable for the assessment here as the main rice producing regions of the world were prescribed well by the model (Asia currently produces over 90 % of the world's rice (FAO, 2000)). In 2007, the countries Brazil, India and China contributed more than 60 % of global sugarcane production, with other countries such as Mexico, Thailand and the Philippines also making important contributions (Fischer et al., 2008). The distribution of sugarcane crops projected by the model was, therefore, deemed to be representative of current distribution as the main areas of sugarcane production within the model are representative of these regions (Figure 6-5, bottom panel). The various types of oil crop, for example oil palm and rapeseed (see Table 5-2), are produced in different regions of the world, with one or more oil crop types being grown in almost every agricultural region (Oregon State University, 2004). For this reason, the representation of oil crops produced by the model was accepted here due to the good fit with global agricultural land. The representation of crop group distribution in 2005 simulated by the model was a good fit to the actual distribution.

Crop group	Area cultivated in 2005	Area cultivated in 2005	
	(model projection) (Mha)	(FAO data) (Mha)	
cereals	694	691	
rice	154	155	
oil	252	253	
fibre	38	38	
sugarcane	19	20	

Table 6-1: Comparison of the model prescribed cropland area (Mha) (left) and FAO data for the total area of cultivated cropland in 2005 (Mha) (right) for each crop group.

Table 6-1 details a comparison of the total cropland area (Mha) prescribed to each crop group within the biochar assessment model, and data for the actual area of cultivated cropland (Mha) for each crop group in 2005, taken from FAOSTAT. Analysis of the cultivated area prescribed within the model, in comparison to the data from FAOSTAT shows very good correlation between the two sets of values. Total cropland cultivated for these crop groups is the same for both datasets. The area of cropland prescribed for cereal land within the model is 3 Mha more than the dataset from the FAO. This is the largest discrepancy between the two, at 0.4 % larger than the actual cultivated area, with rice, oil and sugarcane cropland all being prescribed 1 Mha less in the model than in FAOSTAT. Fibre crops are prescribed the same amount of land for cultivation within both datasets. The model prescribed values were therefore determined to be a close fit to the actual data, allowing further use of the model to project the biochar scenarios from a realistic baseline.

The amount of land cultivated for each crop group differs within each RCP, spatially and temporally. Figure 6-8 shows the trends of cultivation area (Mha) for each crop group within each RCP. The differences seen in trends over time between the crop groups indicate the spatial differences in total cultivation land area between the RCPs. An example of this can be seen clearly in the cultivation areas of the different crop groups in RCP 4.5. To differing extents, the land area under cultivation of cereals (excluding rice), fibre, oil crops and sugarcane sees a decline to 2020, before an increase in cultivation area begins. This initial decline is not seen in the area of land under rice cultivation, indicating that the changes in total cropland area in RCP 4.5 affect areas where rice is cultivated in a different way to land under cultivation of the other crop groups. Figure 6-8 shows a number of such differences, indicating changing regional cropland areas over time, thus affecting the production of each crop category over time. Figure 6-8 also demonstrates the large differences seen in cultivation area, within each crop category, between the RCPs. RCP 2.6 consistently has the largest area of land under cultivation for all of the crop groups, although RCPs 6 and 8.5 are beginning to converge on the RCP 2.5 values in 2100 for some crop types (notably rice and sugarcane). RCP 4.5 has consistently declining cropland for all crop groups across the scenario period, with some variation in rates of decline seen temporally and between crop groups.



Figure 6-8: Global cropland (Mha yr⁻¹), by crop group, for the four RCPs over the 95 year assessment period. N.B. Scales vary between crop groups.

6.3 Scenario 1

Section 6.2 has detailed the land prescription for the production of each crop type for the biochar model. Sections 6.3.1 to 6.3.4 detail the model outputs at each stage of assessment of Scenario 1 which is the main scenario, using the most likely or average values for the parameters prescribed within the model. Variations in these parameters are discussed in Sections 6.4 to 6.11.

6.3.1 Commodity production

Using the crop land area prescribed for each RCP and the crop yield projections for each RCP resulted in commodity production quantities for each RCP. Figure 6-9 shows total annual

commodity production (Gt yr⁻¹) for the four RCPs and Figure 6-10 details the distribution of total commodity production in 2100 for each RCP.



Figure 6-9: Total global commodity production (Gt yr⁻¹) for the four RCPs over the 95 year time period.



Figure 6-10: Total commodity production (tonnes) in 2100 for all crop categories for RCP 2.6 (top left), RCP 4.5 (top right), RCP 6 (bottom left) and RCP 8.5 (bottom right).

Commodity production quantities within the RCPs do not follow the same trends as cropland area due to the different crop yield assumptions of each RCP (Figure 6-9). Although RCP 2.6 has a higher area of land under cultivation for all of the crop groups, RCP 6 and RCP 8.5 both produce more commodity due to the larger crop yield increases assumed within those RCPs.

A large quantity of the total commodity production in each RCP is produced in Asia. RCP 2.6 and RCP 4.5 have production hotspots in China, with large quantities of commodity produced there. RCP 6 and RCP 8.5 see these large production quantities in the wider Asian region, including India. The Corn Belt region of North America is also an area of consistent production throughout the four RCPs, with RCP 6 seeing the largest production quantities, of the four RCPs, for this region in 2100. Figure 6-10 shows that production in other regions of the world is highly variable, for example total commodity production in Africa is relatively low in RCP 2.6 and RCP 4.5 compared to that of RCP 6 and RCP 8.5. Total commodity production over the 95 year period was 430 Gt, 411 Gt, 541 Gt and 646 Gt for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively. Annexe III details the total commodity production in each RCP for the years 2005, 2025, 2050, 2075 and 2100.

6.3.2 Residue production and availability

Combined with the crop specific RPR values the total commodity quantity produced within each RCP would produce the residue quantities shown in Figure 6-11.



Figure 6-11: Global annual crop residue yield (Gt yr^{-1}) (solid line) and unused residue (Gt yr^{-1}) (dashed line) for the four RCPs over the 95 year time period for Scenario 1.



Figure 6-12: Global annual residue and unused residue quantities by crop group for each RCP. Plots show: Cereal residues, top left; rice residues, top right; oil residues, mid left; fibre crop residues, mid right; sugarcane residues, bottom left. Each plot shows total residues (Gt) (solid line) and total unused residues (Gt) (dashed line), by crop category under the assumptions of Scenario 1. N.B. Scales vary between crop groups.

The trends for total commodity and residue production are very similar across the period, as are the trends in total and unused residue quantities (Figure 6-11). Figure 6-11 also shows the total unused residue quantities for under the assumptions of biochar scenario 1. Total residues and unused residues are 4.65 Gt yr⁻¹ and 3.0 Gt yr⁻¹ respectively in 2005. Figure 6-12 shows the total residue production from each crop group for each RCP, alongside the unused portion of each residue type.

The trends in the production of residues are seen to differ between the crop groups, particularly in the production of crops in RCP 2.6 and RCP 4.5. The majority of crop groups loosely follow the trend of total residue production see in Figure 6-11. Rice residues in RCP 4.5 are one exception to this, with residue quantities from rice production increasing above those of RCP 2.6 after 2030, then declining to similar levels as RCP 2.6 at the end of the century. The variation seen in production trends between the different crop groups for RCP 2.6 and RCP 4.5 indicates that changing land-use has a dominant effect on these RCPs, which is not the case for RCP 6 and RCP 8.5. The crop yield increase assumptions of RCP 6 and RCP 8.5 appears to be the dominant factor in commodity production and residue trends for these two RCPs. This is explored further in Sections 5.4.1, 5.4.2 and 5.4.7.

Figure 6-13 details the distribution of total crop residues (left panels) and unused crop residues (right panels) in 2100 for the four RCPs. Some differences in the distribution of unused crop residues are seen, when compared to the total production of residues. For example, much of the residue produced in Europe becomes unavailable for conversion to biochar within the model, although much of the total residue produced in Asia is still available as unused residue. These differences in regional availability between total and unused residues are attributable to the different spatial distributions and availability of the different crop residue types. Within all RCPs the largest quantities of both total residues and available residues are in Asia. South America is consistently the second largest region of available residues. The potential of these residues for biochar production depends on a number of other factors such as biochar yield, biochar carbon content and other properties.



Figure 6-13: Spatial distribution of total crop residues (Mt yr⁻¹) in 2100 for each RCP (left panels) and total unused crop residues (Mt yr⁻¹) in 2100 for each RCP (right panels).

6.3.3 Yield of biochar

6.3.3.1 Total biochar yield (all crop categories)



Biochar Yield: 2005

Figure 6-14: Total biochar yield (Mt yr⁻¹) from all crop residue types assessed for the baseline year (2005) of all RCP scenarios for Scenario 1.

Biochar yield in 2005, under the assumptions of Scenario 1, is the same for each of the RCPs due to the baseline conditions being the same for each scenario. The total amount of biochar produced globally in 2005, within Scenario 1, is 932.28 million tonnes (Mt). As can be seen in Figure 6-14 the distribution of biochar varies regionally. As with residue availability, much of the biochar production is concentrated in Asia, particularly in India, China, and Indonesia. The maximum regional average production potential, per hectare, in 2005 is in Asia, at 1.7 t ha⁻¹ yr⁻¹. Central and South America (particularly Brazil and Argentina) also have some more intense regions of biochar production potential, producing an average of 0.8 t ha⁻¹ yr⁻¹ in 2005. Other regions, such as the Corn Belt of the USA and Europe also have some biochar production potential for biochar production within each RCP changes over time under the assumptions of Scenario 1.



Figure 6-15: Global annual biochar yield (Gt yr^{-1}) for the four RCPs as assessed under Scenario 1.

RCP 8.5 consistently has the largest global potential for biochar production under the assessment of biochar Scenario 1. RCP 4.5 has the lowest potential for biochar production for the majority of the assessment period, with one period (2050 to 2060) where biochar production potential in RCP 2.6 drops below that of RCP 4.5. RCPs 2.6 and 6 follow a very similar trajectory for global biochar production potential to around 2035, where divergence begins and RCP 6 then demonstrates a larger production potential throughout the rest of the assessment. As with commodity production, the crop yield increase assumptions of RCPs 6 and 8.5 appear to be the dominant factors in the biochar production trends for these two RCPs. The increased fluctuation in the trends of RCP 2.6 and RCP 4.5 indicate that other influences, such as land-use change, are more dominant in the biochar production potential of these two pathways.

Figure 6-16 shows the spatial potential for biochar production in 2100 for the four RCPs. RCP 4.5 generally manifests the lowest potential for total biochar production in many regions. There are a few small regions where biochar production may be greater in RCP 4.5 than RCP 2.6 (see, for example, the difference in the magnitude of biochar production in Eastern China and Southern India, where more biochar is produced in RCP 4.5 than in RCP 2.6). Although there are some regions of higher biochar production potential, per hectare, in RCP 4.5, there area of biochar production potential is more widespread in RCP 2.6 and the other RCPs.



Figure 6-16: Regional biochar yield (Mt yr⁻¹) from all crop residues in 2100, under the assumptions of Scenario 1, for the different RCPs. Biochar production is shown for RCP 2.6 (top left), RCP 4.5 (top right), RCP 6 (bottom left) and RCP 8.5 (bottom right).

The spatial distribution of total biochar yield within the RCPs in 2100 is closely matched with the distribution of total available residues (Figure 6-11), indicating that the dominant available crop residues are likely to produce the majority of biochar. Figure 6-17 to Figure 6-22 examines the contribution of each crop group to the biochar production potential of each RCP, highlighting in particular the different magnitudes of production potential between some crop groups.



6.3.3.2 Contribution of different crop categories to biochar production

Figure 6-17: Global biochar production potential (Gt yr⁻¹), differentiated by crop category, over time for the four RCPs under the assumptions of Scenario 1. Plot shows biochar from cereal residues (top left), rice residues (top right), fibre residues (mid-left), oil residues (mid-right) and sugarcane residues (bottom). N.B. Scales vary between biochar quantities from different crop residue types.

The different crop categories offer varied contributions to the total quantity of biochar which could potentially be produced within each RCP (see variation in scales in Figure 6-17). For all RCPs the largest contribution comes from cereal crops, producing 0.43 Gt yr⁻¹ of biochar in 2005. The other crop groups with relatively high biochar production potential in 2005 are rice and sugarcane crops at 0.35 Gt and 0.16 Gt of biochar respectively. Oil crops and fibre crops are projected to produce 0.02 Gt and 0.5 Mt each respectively in 2005. The large quantities and widespread distribution of biochar produced from cereal crops indicates that this biochar may be of high global significance for carbon sequestration. Although large quantities of biochar may be produced from some crop types, it does not always follow that this is the feedstock of highest importance and potential regionally. For example, biochar from rice residues would be mostly produced in Asia (see Figure 6-19) therefore the location of the addition to soil must also be considered. See Section 2.2.5 and Chapter 7 for further discussion on biochar addition to soil and carbon sequestration potential. The regional significance of the crop groups for biochar production may also change temporally. This regional analysis of each crop group is beyond the scope of this study but would offer important insight into biochar production potential as the product of further research.



Figure 6-18: Regional biochar yield (Mt yr^{-1}) from cereal crop residues (excluding rice) in 2100 for the four RCPs under the assumptions of Scenario 1. The panels show RCP 2.6 (top left), RCP 4.5 (top right), RCP 6 (bottom left) and RCP 8.5 (bottom right).

The biochar produced from cereal crop residues in Africa and South and Central America in RCP 8.5 covers a far wider geographical area than in these regions for the other RCPs. Many regions also see more biochar produced per hectare, from cereal crops, in RCP 8.5 than the other three RCPs. Regardless of this larger total geographical area of biochar production in RCP 8.5, and generally higher intensity of production, there are still some regions in the other RCPs where more biochar is produced from cereal residues than in the same region in RCP 8.5 due to higher utilization of cropland within the particular region. Examples of this are the North American and Western European regions in RCP 6, which produce more biochar in certain area due to a higher fraction of land used for crop production in these areas (see Figure 6-4). RCP 4.5 has the smallest geographical area of biochar production from cereal crop residues in 2100, at 522 Mha, but does have some areas of increased biochar production per hectare, when compared to RCP 2.6. This is particularly noticeable in China and India.



Figure 6-19: Biochar yield (Mt yr⁻¹) from rice crop residues in 2100 for the four RCPs under the assumptions of Scenario 1. The panels show RCP 2.6 (top left), RCP 4.5 (top right), RCP 6 (bottom left) and RCP 8.5 (bottom right).

Biochar production from rice residues is mainly concentrated in Asia, with some potential also in South America and Western Africa. Total production potential is larger in RCP 4.5 than RCP 2.6, at 52.6 Gt and 48 Gt respectively, as production per hectare in Asia is higher in RCP 4.5. Production potential from rice residues in the alternative production regions of Africa and South America is lowest in RCP 4.5, with increased geographical coverage in RCPs 2.6, 6 and 8.5 respectively.



0.0E+00 4.3E-05 8.0E-05 1.3E-04 1.7E-04 2.1E-04

Figure 6-20: Biochar yield (Mt yr⁻¹) from fibre crop residues in 2100 for the four RCPs under the assumptions of Scenario 1. The panels show RCP 2.6 (top left), RCP 4.5 (top right), RCP 6 (bottom left) and RCP 8.5 (bottom right).

Biochar can be produced from fibre crop residues in a number of the world's agricultural regions. As detailed in Figure 6-14, the potential for biochar production from fibre crop residues is limited in comparison to that of the other crop residues, due to the smaller amount of commodity and related residue which may be produced. There are no regions of the world, within these RCP scenarios, that produce fibre crop residues alone, therefore fibre crop residues are unlikely to be the main biochar producing residue of any region. However, they may add to a biochar production system if it is technologically possible and economically practicable to collect and convert the fibre crop residues.



Figure 6-21: Biochar yield (Mt yr⁻¹) from oil crop residues in 2100 for the four RCPs under the assumptions of Scenario 1. The panels show RCP 2.6 (top left), RCP 4.5 (top right), RCP 6 (bottom left) and RCP 8.5 (bottom right).

Biochars from oil crop residues are also produced in many of the world's agricultural regions, with the highest intensity of production often seen in South-Eastern Asia. This intensity of production is also seen in areas of Brazil and Argentina within RCP 8.5. Geographical distribution of biochar from oil crop residues is increasingly widespread from RCP 4.5 to RCPs 2.6, 6 and 8.5 respectively.

Biochar from sugarcane residues is focussed in Asia, South America, Central America and Eastern and Southern Africa (Figure 6.22). The distribution of cropland for sugarcane production follows the same trend as discussed for oil crops, with the smallest coverage in RCP 4.5, increasing for RCP 2.6, RCP 6 and RCP 8.5 respectively. Although RCP 4.5 has the smallest geographical coverage, there are some areas of more intense production than RCP 2.6, particularly in India. South America generally is the region of highest intensity production for all of the RCPs, with India also being a region of high production in RCP 8.5.



Figure 6-22: Biochar yield (Mt yr⁻¹) from sugarcane crop residues in 2100 for the four RCPs under the assumptions of Scenario 1. The panels show RCP 2.6 (top left), RCP 4.5 (top right), RCP 6 (bottom left) and RCP 8.5 (bottom right).

6.3.4 Carbon in biochar

Biochar carbon content is another necessary consideration when determining the suitability of different feedstocks for biochar production for carbon sequestration purposes. The biochar produced in 2005, under the assumptions of Scenario 1, would contain 0.49 GtC. Over time this would increase to 0.93, 0.78, 1.53 and 2.18 GtC for RCPs 2.6, 4.5, 6 and 8.5 respectively. Total carbon content of the biochar, over the 95 year period is 69.9, 65.7, 87.1 and 109.7 GtC for the four RCPs respectively. This follows the trend seen in residue production, residue availability and biochar production for the RCPs, with RCP 4.5 having the lowest potential for total biochar carbon content and RCP 8.5 having the highest potential.



Figure 6-23: Total biochar production (Gt yr⁻¹) (left) and total carbon content (Gt yr⁻¹) (right) of the biochars produced from all residues in Scenario 1.

Total biochar carbon content, by weight, is approximately 70 % of the quantity of biochar produced. Although biochar often has high carbon content, some biochars have high ash and/or moisture content, such as the 47 % ash content of the rice husk biochar produced experimentally and detailed in Chapter 4, Section 4.3.

Figure 6-24 shows the relative contribution of each biochar type to the total biochar production (left hand plots) and also to the total carbon content of the biochars (right hand plots). This plot shows the variation which can exist in biochar carbon content, for example, in the baseline year of 2005, sugarcane biochar is 89 % carbon, whereas rice biochar is 55 % carbon. This highlights the importance of considering the carbon content and other characteristics of the biochar, rather than the yield of biochar alone. The recalcitrance of the carbon in each biochar is also of high importance, and is discussed further in Chapter 7.



Figure 6-24: Biochar quantity (Mt yr⁻¹) (left) and carbon content (Mt yr⁻¹) (right) of the biochar produced from residues from each crop type, over time, for the four RCPs, under the assumptions of Scenario 1. N.B. Scales vary between plots.



Figure 6-25: Carbon content (Mt yr⁻¹) of the total biochar produced in 2100 for the four RCPs under the assumptions of Scenario 1. Panels show RCP 2.6 (top left), RCP 4.5 (top right), RCP 6 (bottom left) and RCP 8.5 (bottom right).

Carbon content of the total biochar produced is highest in Asia for all RCPs under the assumptions of Scenario 1. South and Central America also have relatively high potential, compared to the other biochar producing regions of the world. The regions of Africa and Europe have some potential, but this is limited compared to that of Asia and South/Central America in all RCPs. The Mt C per year are equivalent to between 335.8 Pg and 563.6 Pg CO₂ over the 95 year period, these values being for RCP 4.5 and RCP 8.5 respectively. Global anthropogenic CO₂ emissions in 2005 were around 29.7 Gt CO₂ (Le Quere et al., 2009). Within Scenario 1 the carbon within the biochar in 2005 is therefore equivalent to 8.4 % of global anthropogenic CO₂ emissions, and equivalent to between 13.4 % and 37.7 % of global anthropogenic CO₂ emissions levels).

Table 6-2: The annual biochar production, biochar carbon content and CO_2 equivalent of biochar carbon content for the years 2005 and 2100. Also shown are the cumulative values for the 95 year period, for each of the four RCPs under the assumptions of Scenario 1.

Year	Units	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production							
2005	Mt yr ⁻¹	969.79	969.79	969.79	969.79		
2100	Mt yr⁻¹	1835.49	1570.53	3022.82	4322.58		
95 year period	Gt	138.38	132.25	173.17	217.89		
Total carbon in biochar							
2005	Pg C yr ⁻¹	0.68	0.68	0.68	0.68		
2100	Pg C yr⁻¹	1.30	1.09	2.13	3.05		
95 year period	Pg C	97.87	91.48	121.69	153.56		
CO ₂ equivalent of carbon in biochar							
2005	Pg CO ₂ yr⁻¹	2.49	2.49	2.49	2.49		
2100	$Pg CO_2 yr^{-1}$	4.78	3.98	7.83	11.20		
95 year period	Pg CO ₂	359.20	335.75	446.61	563.55		

Addition of the biochar to soils would see some degradation leading to the release of some of the carbon detailed in Table 6-2 back into the atmosphere as CO_2 . This is discussed further in Chapter 7.

6.4 Scenario 2: Changes in crop yields

Scenario 2 examined the effect that the assumptions of crop yield increase made in Scenario 1 had on the production potential of biochar. Two alternative scenarios of crop yield assumptions, Scenario 2a and 2b, assumed no crop yield increase and optimistic yield increase respectively. Scenario 2c explored an alternative convergence year for crop yields in RCP 4.5.

6.4.1 Scenario 2a: No crop yield increase

Scenario 2a, simulating no crop yield increase over time, shows that biochar production is greatly enhanced in each RCP within Scenario 1 due to the assumptions of crop yield increase.



Figure 6-26: Global biochar yield (Gt yr⁻¹) for Scenario 1 (top left) and Scenario 2a (top right) and Scenario 2a with a smaller scale (bottom) for the four RCPs.

RCP 2.6 has the greatest biochar production potential when assuming no crop yield increase over time due to having the largest land area for crop production. The pathway of annual global biochar production is similar to that of total cropland area (Figure 6-3). The point at which RCP 6 overtakes RCP 8.5 occurs earlier for biochar production potential than in the total global cropland pathways. The difference between biochar production in 2100 for RCP 2.6 and RCP 6 is smaller than the difference in global cropland area. Analysis of the trends of global cropland by crop type (Figure 6-8) shows that the trend in total biochar production for Scenario 2a is influenced by the land area available for each crop type. Compared to Scenario 1, biochar production potential over the 95 year period is reduced by 30.5 Gt, 46.3 Gt, 70.9 Gt and 117.4 Gt for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively when no yield increase occurs over the assessment period. Using the experimental values from Chapter 4 for carbon content of these biochars, this is equivalent to 15, 22.9, 35.7 and 59.0 Gt C respectively.

6.4.2 Scenario 2b: Optimistic crop yield increases

A 50 % increase in the annual crop yield increase, relative to Scenario 1 crop yield increases, results in an increase in biochar production potential of 19.1 Gt, 33.9 Gt, 57.9 Gt and 120.5 Gt for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively over the 95 year period. This is 9.7, 16.8, 29.1 and 60.5 Gt more carbon in biochar for Scenario 2b, relative to Scenario 1. Increasing the crop yield assumptions by 50 % results in more biochar being produced both per annum and in total in RCP 4.5 than in RCP 2.6, which does not occur in Scenario 1.



Figure 6-27: Global annual biochar yield (Gt yr⁻¹) for Scenario 1 (left) and Scenario 2b (right) for the four RCPs.

The crop yield increases examined in Scenario 2b, although possible, are more unlikely to be achieved than those of Scenario 1. Historically, the rate of crop yield increase has been declining over time, therefore large changes in areas such as crop management, technology and genetic modification would be required to see the continual large increases in crop yields seen in Scenario 2b. Crop yields in many regions are also likely to be impacted by climate change within the assessment period, with RCPs 6 and 8.5 likely to see larger impacts than RCPs 2.6 and 4.5 due to the increased likelihood and severity of climate change in the former two RCPs (Porter et al., 2014). This makes it more likely that incorporating the effects of climate change into crop yield projections will see reduced crop yields, or at least slower rates of increase, than those tested in this scenario (Porter et al., 2014). The impacts of climate change on crop yields are examined in Scenarios 8 and 9 (Section 5.4.7).



Figure 6-28: Biochar production (Gt) over the 95 year period for the range of crop yields projected. Filled bars indicate the main projections of Scenario 1 and error bars indicate the range of production values estimated using the maximum and minimum crop yield scenarios (2a and 2b).

The variation in crop yields explored in Scenario 2a and 2b creates an increasing range of potential biochar production quantities across the RCPs (from RCP 2.6 to RCP 8.5 respectively). The increased range shown in RCP 6 and RCP 8.5 illustrate the larger assumptions of crop yield increase in these RCPs for Scenario 1, which leads to larger variation in scenarios of no yield increase and large yield increase.

6.4.3 Scenario 2c: Crop yield increase convergence at 0.25 % yr⁻¹ in 2050 for RCP 4.5

The rate of convergence of crop yield to 0.25 % yr⁻¹ in RCP 4.5 was explored in this scenario, due to uncertainty within the RCP literature regarding the point of convergence. Scenario 1 assumed that the convergence date is 2100. Scenario 2c sees this convergence date altered to 2050, exploring the effects of the earliest potential convergence date on biochar production.

The change in convergence date decreases the cumulative yield of biochar over the 95 year period by 6.5 Gt. Total carbon in biochar is reduced by 3.2 GtC. The point of convergence is of particular importance in the continued increase of biochar production rates for RCP 4.5 due to the reduction in cropland area seen over time in the RCP. Where convergence to 0.25 % yr⁻¹ is assumed in 2100, the rate of biochar production is seen to increase to around 2090. Where the point of convergence is in 2050 the production of biochar plateaus and then sees a decrease in annual biochar production to 2100.



Figure 6-29: Total biochar yield (Gt yr^{-1}) for RCP 4.5 for Scenario 1 and Scenario 2c. Plot illustrates the effect of different rates of crop yield increase convergence to 0.25 % yr^{-1} . The two convergence years assessed are 2050 (red (Scenario 2c)) and 2100 (blue (Scenario 1)).

6.5 Scenario 3: Land use change

6.5.1 Scenario 3a: No land use change



Figure 6-30: Global biochar yield (Gt yr⁻¹) for Scenario 1 (left) and Scenario 3 (right) for the four RCPs.

Scenario 3a explored the effect of land use change assumptions on biochar production potential. The area of land prescribed for agricultural production was kept constant at 2005 levels, with all other assumptions of Scenario 1 remaining the same, allowing for analysis of how the changing land use of the RCP pathways affects the total biochar potential of Scenario 1.

The Scenario 1 crop yield assumptions of RCP 6 and RCP 8.5 dominate the biochar production potential in Scenario 3. In Scenario 3a, RCP 4.5 produces similar amounts of biochar as RCP 6 up to 2055. RCP 2.6 has consistently the lowest biochar production potential. RCP 8.5 has consistently the largest production of biochar, relating to having the highest crop yield increases throughout the Scenario. Scenario 3a sees a reduction in total biochar production potential of 19.8 Gt, 18.1 Gt and 19.3 Gt over the 95 year period, relative to Scenario 1, for RCP 2.6, RCP 6 and RCP 8.5 respectively. RCP 4.5 would see an increase in biochar production potential of 12.1 Gt over the 95 year period, relative to Scenario 1, if land use remained the same throughout the RCP period.



6.5.2 Scenario 3b: Land for biofuels

Figure 6-31: Global biochar production for Scenario 1 and Scenario 3b. Scenario 3b shows the biochar production if the land prescribed for biofuels is disregarded for RCPs 2.6, 4.5 and 6.

The land prescribed as biofuels land follows different trends for each of the three RCPs which have biofuels land area datasets. When biofuel land area is subtracted from the total cropland area for the three RCPs, excluding RCP 8.5 due to differences in accounting biofuel land, RCP 2.6 sees the smallest reduction in biochar production consistently over time. RCP 4.5 sees the largest reduction in cropland area for most of the scenario period, indicating that RCP 4.5 is a heavily biofuels reliant scenario. RCP 6 sees a steady increase in biofuels cropland across the scenario period, up to around 2080 where biofuels cropland begins to increase markedly, resulting in a reduction in cropland area of 312 Mha in 2100. This is in comparison to reductions of 195 Mha and 301 Mha for RCPs 2.6 and 4.5 respectively. Uncertainty surrounding the types of biofuel crops used within each RCP prevents further analysis of the available residues from the prescribed biofuel land. The incorporation of potential biofuel crops, including oil crops and sugarcane, into the Scenario 1 analysis is as representative of biofuels production as is possible without further detail on crop types. This would be a good area for further study in the future.



6.6 Scenario 4: Residue to product ratio (RPR)

Figure 6-32: Biochar production (Gt) over the 95 year period for each RCP. Dark grey bars indicate total biochar production under the RPR assumptions of Scenario 1. Light grey bars indicate total biochar production under the RPR assumptions of Scenario 4a1. Error bars indicate the range of potential biochar production using the Scenario 4a2 and 4b as markers of low RPR and high RPR values.

A range of scenarios were explored looking at potential variation in RPR. Scenario 4a examined a future of declining RPR. Within this scenario, two scenarios (4a1 and 4a2) looked at a reduction in RPR to medium and low levels by 2100. Scenario 4b examined a future of increasing RPR.

Biochar production was reduced, from that of Scenario 1, by 25.3 Gt, 32.9 Gt, 35.7 Gt and 39.1 Gt of biochar for the RCPs under the medium RPR assumptions. The variation in biochar production, for each RCP, between the scenarios of low and high RPR is 87.2 Gt, 76.7 Gt, 115.8 Gt and 155.7 Gt respectively. When using the average values (across all crop types) for RPRs, assuming the high RPR values results in projections which are closest to the biochar production of Scenario 1. Using the medium RPR assumptions reduces the biochar production to be considerably below that of Scenario 1. The low RPR assumptions resulted in further considerable reductions in biochar production potential.

6.7 Scenario 5: Residue availability

Scenario 5 examined alternative scenarios of residue availability. Scenarios 5a, 5b and 5c examine assumptions (across all crop groups) of 25 % and 50 % and 75 % residue availability respectively. Scenarios 5d and 5e used the assumptions of Woolf et al. (2010) as alternative residue availability scenarios for comparison.

Scenario 5b, which assumes that 50 % of all residues are available, sees a reduction in biochar production of 9.9, 7.9, 11.5 and 18.1 Gt biochar relative to Scenario 1 for the four RCPs respectively. Biochar production potential varies by 80.1, 71.7, 98.9 and 120.3 Gt for the four RCPs between the assumptions of Scenarios 5a and 5c (which assume 25 % and 75 % residue availability for all crop groups respectively).

6.7.1 Scenarios 5a to 5c: 25 %, 50 % and 75 % residue available for biochar conversion



Figure 6-33: Biochar production (Gt) over the 95 year period for each RCP with altered residue availability. Dark grey bars indicate total biochar production under the residue availability assumptions of Scenario 1. Light grey bars indicate total biochar production under the residue availability assumptions of Scenario 5b (50 % availability). Error bars indicate the range of potential biochar production using the low and high residue availability assumptions of Scenario 5 Scenari

6.7.2 Scenario 5d and 5e: Residue availability assumptions of Woolf et al. (2010)

Scenarios 5d and 5e assessed the conservative and maximum potential scenarios of residue availability discussed by Woolf et al. (2010) respectively (see **Table 5-7**). Both scenarios result in smaller total biochar quantities than Scenario 1. The residue availability assumptions of Scenario 1 are, overall, more optimistic than even the most optimistic assumptions made in Woolf et al. (2010). Applying the most conservative residue availability assumptions of Woolf et al. (2010) (Scenario 5d) results in a reduction in biochar production potential, from that of Scenario 1, of 52.9, 49.6, 66.1 and 79.7 Gt for the four RCPs respectively. The rationale behind the assumptions made for residue availability of residues. The assumptions detailed in Woolf et al. (2010) are also plausible, with research currently providing little probabilistic analysis of likelihood for each set of assumptions.



Figure 6-34: Total biochar production (Gt) for Scenario 1 (black), Scenario 5d (light grey) and Scenario 5e (dark grey).



6.8 Scenario 6: Alternative biochar yield

Figure 6-35: Biochar production (Gt) under the assumptions of Scenario 1 (main bars). Error bars indicate biochar production using the assumptions of minimum and maximum (Scenarios 6a and 6b respectively) typical biochar yields from literature.

The biochar yield has the potential to greatly influence the quantities of biochar produced within the scenarios. If the maximum biochar yields of 63 % from the literature (Peng et al, 2011) were achieved for all feedstocks throughout the scenarios then biochar production could be increased by 132.7, 122.9, 164.6 and 207.4 Gt relative to Scenario 1 for the four RCPs respectively. It is unlikely that these very high yields of biochar would be achievable for all feedstocks, both spatially and temporally, throughout the scenarios due to factors such as system requirements for the production of high yields and the likely use of dual energy production systems which lead to the balancing of biochar yields against yields of oil and gas from the process. Using the lower value for average biochar yield from the literature (25 %) resulted in less variation from the biochar production quantities of Scenario 1, seeing a decrease of 30.8, 31.0, 39.1 and 49.1 Gt of biochar for the four RCPs respectively. In reality biochar yields can be highly variable and are dependent on both feedstock and process. The assumptions of Scenario 1 are, therefore, probably more representative of a possible future scenario as some variation due to feedstock is applied. Scenario 1 does not include variation which may occur through altering process type or conditions, or the range which may be seen within biochar yields from one feedstock type. Scenario 6, therefore, offers an indication of best and worst case scenarios for biochar yields from crop residues, with the likely production quantity lying somewhere between the two.

6.9 Scenario 7: Alternative biochar carbon content

Using the mean or high average value for the carbon content of biochars sees an increase in the total carbon content projections relative to Scenario 1. Using the low value for average carbon content sees a decrease in the carbon content of the total biochar produced. It is probable that the use of carbon content values for each biochar type result in a more accurate representation of the total carbon in biochar due to the large variation in carbon content which can exist between biochar types, alongside the different quantities of different biochars which contribute to the total biochar production. The use of an average value for biochar carbon content increased the total biochar carbon content by 2.5, 4.4, 3.9 and 4.4 GtC over the 95 year period for the four RCPs respectively.


Figure 6-36: Total biochar carbon content (GtC) under the assumptions of Scenario 1 (light grey bars), and Scenario 7b (dark grey bars) which assumes the average carbon content (determined across all crop groups from experimental data and literature). Error bars indicate the range of total biochar carbon content under the assumptions of the minimum and maximum typical biochar yields from literature (Scenarios 7a and 7c respectively).

6.10 Scenario 8 and 9: Climate change effects on crop yields and the

impacts of adaptation

The projected changes in temperature for the RCPs, relating to the response of the climate to the radiative forcing pathway of each RCP, are discussed in Chapter 2.4 and Section 5.4.7. These temperature changes are likely to impact crop yields, affecting the quantity of agricultural residues produced within each scenario. Scenario 8 explores the magnitude of impacts that these changes in temperature may have on biochar production potential within each RCP. Scenario 9 assessed how much impact measures of adaptation may have on the impacts of temperature change on crop yield. Those RCPs with larger emissions are projected to experience larger increases in mean global surface temperature. Resulting projections of larger related impacts on crop yields are also made for these RCPs within the literature. Section 5.4.7 details the development of temperature projections and related crop yield impacts for each biochar production scenario using the available literature.



Figure 6-37: Biochar production potential (Gt yr⁻¹) for each RCP with no temperature change (black), minimum projected temperature change (red) and maximum projected temperature change (blue) for each RCP with no adaptation measures (Scenario 8).

Figure 6-37, showing the impact of the range of projected temperatures for each RCP on biochar production potential, illustrates that the projected impacts on crop yields and resulting biochar production increases in severity with increasing radiative forcing scenario. RCP 2.6 shows only minimal impact on crop yields, and this impact is only seen when applying the maximum potential temperature change to the assumptions of Scenario 1. RCP 4.5, RCP 6 and RCP 8.5 all see larger decreases in crop yield, and resulting biochar production potential, even where the minimum projected temperature change is applied to the scenario. The temporal onset of these impacts becomes more rapid from RCP 2.6, in RCP 4.5, RCP 6 and then to RCP 8.5 when the maximum temperature change projections are applied. In RCP 4.5 the impacts of temperature increase can only be seen after 2090 when the minimum temperature change is applied to the scenario. The maximum projected temperature change begins to impact residue production also begins to influence the biochar production potential from 2040, with the minimum temperature

change impacting production from 2080, which is around 10 years earlier than in RCP 4.5. In RCP 8.5 the minimum temperature change begins to affect the production potential around 2050, whilst the maximum temperature change production has impacts from 2040.



Figure 6-38: Biochar production potential (Gt yr⁻¹) for each RCP with no temperature change (black), minimum projected temperature change (red) and maximum projected temperature change (blue) with the simple crop based adaptation measures assumed in Scenario 9.

Using crop based adaptation measures has the potential to reduce the impact of temperature changes on crop yields. Figure 6-38 shows the projected impacts of temperature changes on crop yields when these adaptation measures are applied. With these adaptation measures there is no impact on crop yield production in RCP 4.5 when the minimum temperature change projection is applied. The impacts of temperature change on biochar production for RCPs 6 and 8.5 are also greatly reduced, with impacts occurring at the same time as in the scenario with no adaptation, though these impacts are reduced in magnitude.

The biochar production potential in 2100 is reviewed here as biochar potential generally increases annually therefore 2100 indicates the largest production potential within the scenario period. With regard to the climate change scenarios, Scenarios 8 and 9, 2100 is also the year with the largest manifestation of temperature change related to climate change. Analysis of 2100 results therefore demonstrates the maximum impact on biochar production that this temperature change would have within the scenario period.



Figure 6-39: Total biochar production (Gt yr⁻¹) in 2100 for Scenario 1 (dark grey) and the mean climate change temperature projection with no adaptation (Scenario 8b) (light grey). Where no dark grey is shown (i.e. RCP 2.6) there is no reduction in biochar production projected under the mean climate change temperature scenario (relative to Scenario 1). Error bars show the range of projected biochar production related to the range in crop yield impacts relating to the minimum and maximum potential temperature change projections (Scenario 8a and 8c, 5% and 95% projection values respectively) related to each RCP scenario.

The mean temperature projection pathway has no impact on biochar production potential, in 2100, for RCP 2.6. A small impact is seen where the maximum projected temperature change is applied to RCP 2.6. The minimum temperature change projection for the RCP 4.5 pathway also sees relatively little reduction in biochar production potential relative to Scenario 1 production. In RCP 4.5 the maximum projected temperature change results in a reduction in biochar production potential of 106.8 Mt yr⁻¹ in 2100 relative to Scenario 1. RCPs 6 and 8.5 experience a

reduction in biochar production potential for all temperature projections applied. The maximum projected temperature increase for RCP 6 results in a reduction from the Scenario 1 biochar production potential of 256.9 Mt yr⁻¹ by 2100. The maximum projected temperature increase for RCP 8.5 results in a reduction of 713.2 Mt yr⁻¹ of biochar in 2100 relative to Scenario 1. Scenarios 9 a-c saw the impact on biochar production potential reduced, for all RCPs, with the application of simple adaptation methods (Figure 6-40).



Figure 6-40: Total biochar production (Gt yr⁻¹) in 2100 for Scenario 1 (dark grey) and the median climate change projection with adaptation (light grey). Where no dark grey is shown (i.e. RCP 2.6) there is no reduction in crop yields projected under the median climate change scenario (relative to Scenario 1). Error bars show the range of projected crop yield impacts relating to the minimum and maximum potential temperature change (5 % and 95 % values) related to each RCP scenario (see Table 2-3).

Table 6-3: The mitigation potential (in 2100) of simple adaptation measures on the impacts on biochar production of climate change related temperature change. The mitigation potential is displayed in Mt yr⁻¹ of biochar produced by Scenarios 9 a-c relative to the biochar production of Scenarios 8 a-c.

Temperature projection	Mitigation of impact on biochar production (Mt yr ⁻¹)			
	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5
Minimum temperature projection (9a)	0	31.4	102.8	198.8
Mean temperature projection (9b)	0	59.7	123.9	345.8
Maximum temperature projection (9c)	66.1	72.2	166.3	523.0

The effect of applying the maximum temperature change projection alongside adaptation measures, to RCP 4.5 reduces the impact of temperature change on biochar production from the no-adaptation scenario by 72.2 Mt yr⁻¹ in 2100. This results in a reduction in biochar production potential relative to Scenario 1 of 34.6 Mt yr⁻¹ in 2100. RCP 6 sees some impact on biochar production potential in 2100 even when the minimum projected temperature is applied. This reduces the impact, from the non-adaptation climate change scenario, by 102.8 Mt yr⁻¹ biochar in 2100 resulting in a reduction in 2100 of 6.1 Mt yr⁻¹ biochar relative to Scenario 1. Applying the maximum temperature change to RCP 6 in a scenario with adaptation reduces the effect of climate change on biochar production potential in 2100 relative to Scenario 1. The loss in biochar yield seen in Scenario 8 for RCP 8.5 is reduced by 198.8 Mt yr⁻¹ in 2100 through the adaptation measures applied under the minimum temperature projections, resulting in a reduction of 95 Mt yr⁻¹ relative to Scenario 1. The application of adaptation measures to the impacts of the maximum temperature projection of RCP 8.5 sees a reduction in crop yield impact of 523 Mt yr⁻¹ in 2100, resulting in a reduction in biochar production potential of 190.2 Mt yr⁻¹ in 2100.



Figure 6-41: Total cumulative biochar production (Gt) for Scenario 1 (black), Scenario 9b (dark grey) and Scenario 8b (light grey) over the 95 year period. Scenarios 9b and 8b are the mean temperature change projection for each RCP with adaptation and with no adaptation respectively. Where only light grey is shown (i.e. RCP 2.6) there is no reduction in crop yields projected under the median climate change scenario (relative to Scenario 1). Red bars show the range of projected crop yield impacts relating to the minimum and maximum potential temperature change (5 % and 95 % values) related to each RCP scenario with adaptation. Blue bars show the range of projected crop yield crop yield impacts relating to the minimum and maximum potential temperature change (5 % and 95 % values) related to each RCP scenario with adaptation. Blue bars show the range of projected crop yield impacts relating to the minimum and maximum and maximum potential temperature change (5 % and 95 % values) related to each RCP scenario with adaptation.

The reduction in biochar production potential caused by climate change, relative to Scenario 1, increases across the RCPs, from RCP 2.6 to RCP 8.5, when this potential is assessed for the 95 year period. RCP 2.6 sees no reduction in biochar production potential, except where the maximum temperature change is combined with assumptions of no adaptation, resulting in a reduction of 3.2 Gt biochar over the 95 year period. Assuming the minimum temperature projection and no adaptation (scenario 8a) resulted in a reduction in biochar production potential of 0.3, 1.4 and 6.7 Gt for RCPs 4.5, 6 and 8.5 respectively. This could be mitigated by 0, 0.3, 1.7 and 5.5 Gt biochar by employing simple adaptation measures (Scenario 9a). Mean temperature change and no adaptation (Scenario 8b) cause a reduction in biochar production

potential of 2.6, 4.3 and 11.2 Gt biochar over the 95 year period for RCPs 4.5, 6 and 8.5 respectively. This could be mitigated by 0, 2.6, 3.9 and 8.2 Gt biochar by employing adaptation measures (Scenario 9b). Maximum temperature projections alongside no adaptation measures (Scenario 8c) results in a reduction in biochar production potential of 3.2, 4.5, 7 and 16.7 Gt biochar respectively for the four RCPs, relative to Scenario 1, for the 95 year period. This could be mitigated by 3.2, 3.6, 5.3 and 12.4 Gt biochar production potential, using the minimum and maximum temperature projections of biochar production potential, using the minimum and maximum temperature projections, is smaller where adaptation measures are applied than where no adaptation is applied. These results indicate that any efforts to reduce the temperature change seen with climate change, and any applications of crop based adaptation methods will provide increased yields of biochar over the 95 year period.



6.10.1 Effect of crop yield change uncertainty

Figure 6-42: Annual biochar production (Gt yr⁻¹) for the four RCPs under the assumptions of Scenario 1, Scenario 8b (the mean yield and temperature change scenario using the IPCC projections of yield impacts) and Scenario 8f (using the crop yield impacts of climate change detailed in Kyle et al. (2014).

Using the crop yield impacts discussed by Kyle et al. (2014) illustrates again the increasing impact of the increasing temperature projections on crop yields and biochar potential, both over

time and across the RCPs (from RCP 2.6 to RCP 8.5). Scenario 8f sees a marked reduction in biochar production potential from that of Scenario 8b for RCPs 4.5, 6 and 8.5. No impact is seen on RCP 2.6 due to the low levels of temperature change projected for that scenario. The use of crop yield impact projections by Kyle et al. (2014) offers an alternative pathway for biochar production which more closely resembles the pathway of biochar production projected using the maximum temperature projection from the IPCC (Scenario 8c).



Figure 6-43: Total biochar production (Gt) for the four RCP scenarios under the assumptions of Scenario 1 (black), Scenario 8e2 (which assumes the highest yield projections discussed in Section 5.4.7.1, alongside the mean projected temperature change (dark grey)), and Scenario 8d2 (which assumes the lowest yield projections discussed in Section 5.4.7.1, alongside the mean projected temperature range (light grey)). Blue bars indicate the range where low yield projections are assumed with the low and high temperature projection assumptions. Red bars indicate the range when the high yield projections are assumed with the low and high temperature change assumptions. All of the scenarios shown in this figure assume no 'simple agricultural' adaptation measures are taken.

Biochar production in RCP 2.6 was not impacted, relative to Scenario 1, when using the mean temperature range with either the high or low yield projections. Low yield projections with the low temperature projections see a reduction in biochar production of 1.1, 4.7 and 13.6 Gt for RCPs 4.5, 6 and 8.5. Low yield projections combined with the mean temperature change makes this reduction in biochar potential greater, seeing 6.6, 9.7 and 19.0 Gt less biochar produced than in Scenario 1 for RCPs 4.5, 6 and 8.5. Low yield projections alongside the high temperature projections result in large reductions in biochar potential from those of Scenario 1, seeing 8.1, 8.9, 13.2 and 26.1 Gt less biochar for RCPs 2.6, 4.5, 6 and 8.5. Applying the high yield projections to the low temperature projections resulted in an increase in biochar production of 0.5, 1.4 and 0.3 Gt for RCPs 4.5, 6 and 8.5 respectively. High yields with mean temperature projections see an increase in biochar production of 1.2 and 1.1 Gt respectively for RCPs 4.5 and 6, and a decrease in biochar of 2.6 Gt for RCP 8.5. Applying the high yields and high temperature projection sees an increase of 1.5 Gt biochar for RCP 2.6, RCP 4.5 sees no change and RCPs 6 and 8.5 see a decrease of 0.5 and 6.3 Gt respectively.



Figure 6-44: Total biochar production (Gt) for the four RCP scenarios under the assumptions of Scenario 1 (black), Scenario 9e2 (which assumes the highest yield projections discussed in Section 5.4.7.1, alongside the mean projected temperature change (dark grey)), and Scenario 9d2 (which

assumes the lowest yield projections discussed in Section 5.4.7.1, alongside the mean projected temperature range (light grey)). Blue bars indicate the range where low yield projections are assumed with the low and high temperature projection assumptions. Red bars indicate the range when the high yield projections are assumed with the low and high temperature change assumptions. All of the scenarios shown in this figure assume that 'simple agricultural' adaptation measures are taken.

Applying the alternate yield scenarios to the scenarios of adaptation with climate change again sees no change in biochar production potential for RCP 2.6 for the minimum and mean temperature change. Biochar yield is impacted in RCP 2.6 when the low yields are combined with the maximum temperature, seeing a reduction of 3.7 Gt biochar. Low yields and low temperature projections see 0.6, 2.2 and 6.3 Gt less biochar than Scenario 1 for RCPs 4.5, 6 and 8.5. Low yields and medium projections reduce biochar yields by 3.0, 4.4 and 9.1 Gt and high temperatures reduce biochar production by 4.1, 6.3 and 12.8 Gt biochar, relative to Scenario 1, for RCPs 4.5, 6 and 8.5. High biochar yields produce either no change or increased yields of biochar, irrespective of temperature projection. High yields combined with low temperature projection sees 0, 1.1, 3.9 and 6.5 Gt increase in biochar for RCPs 2.6, 4.5, 6 and 8.5 respectively. Medium temperature projections result in increases of 4.5, 5.8 and 5.6 Gt for RCPs 4.5, 6 and 8.5. High temperature projections result in increased biochar production of 5.5, 4.0, 5.1 and 6.2 Gt for RCP 2.6, 4.5, 6 and 8.5.

6.11 Summary



Figure 6-45: Summary of biochar produced annually for the four RCPs under the different scenario assumptions for Scenarios 1 to 7. (N.B. Scales vary between plots).

See Annexe III for tables detailing biochar production summaries for these scenarios. Figure 6-45 shows the annual global biochar production, for each RCP, for Scenarios 1 to 7. Scenario 1, which assumes the main or mean parameter values from literature and experimental data, begins with annual production of 0.97 Gt yr⁻¹ for all RCPs, increasing to 1.84 Gt yr⁻¹, 1.57 Gt yr⁻¹, 3.02 Gt yr⁻¹ and 4.32 Gt yr⁻¹ for the four RCPs respectively in 2100. These scenarios project biochar production of 138.4 Gt, 132.3 Gt, 173.2 Gt and 217.9 Gt for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively across the 95 year period.

Figure 6-45 highlights the variation in biochar production which is projected when exploring the various parameters of the biochar analysis, and also the variation in biochar production between the RCPs. Variation between the projections of biochar production is larger in RCP 8.5, followed

by RCP 6 and this range between projections is more pronounced as the RCP timeline progresses. In 2100 the range in biochar production potential between the highest and lowest scenario projections for each RCP is 3356, 2830, 5513 and 7975 Mt yr⁻¹ for RCPs 2.6, 4.5, 6 and 8.5 respectively. This larger variation in the scenario projections for RCPs 6 and 8.5 is mostly attributable to the larger rate of crop yield increase within these scenarios. This results in a wider range between scenarios using these crop yield assumptions and the scenario which assumes alternative yields. Low assumptions of residue to product ratio resulted in the lowest biochar production quantities of all of the parameters explored, indicating that factors such as the future trends in RPR could have a large impact on the biochar production potential. Scenarios exploring high crop yield and biochar yield produced the highest projections of biochar production are possible in the future. Further research to narrow the range in projections for each RCP would benefit from better understanding of how factors such as crop yield, RPRs and available residues will change in the future.

Climate change induced changes in global mean surface temperature may have different levels of impact on biochar production, depending on the magnitude of climate change and on adaptation measures employed. The negative impacts on biochar production are seen to increase with increasing radiative forcing pathway, i.e. from lower impacts in RCP 2.6 to larger impacts in RCP 8.5. RCP 2.6 may not see an impact of biochar production, unless the changes in global mean temperatures seen are at the top end of the range of projections made within the literature. Even then the impacts on biochar production will be minimal when compared to impacts of the other RCP manifestations. All other RCPs would see impacts on the biochar production potential at some point in the scenario period, although the onset of these impacts occurs earlier with higher radiative forcing pathway. The mean climate change assumption (8b: mean crop yield impacts and mean temperature increase projection) saw a reduction in biochar production potential of 0 Gt, 2.6 Gt, 4.3 Gt and 11.2 Gt biochar over the 95 year period for RCPs 2.6, 4.5, 6 and 8.5 respectively. This could be mitigated by 0 Gt, 2.6 Gt, 3.9 Gt and 8.2 Gt of biochar by employing simple adaptation measures (Scenario 9b). This results in no reduction in biochar production potential, from that of Scenario 1, for RCP 2.6 and 4.5, and reductions of 0.4 Gt and 3 Gt over the scenario period for RCP 6 and RCP 8.5 respectively. The smallest (negative) climate impacts for the four RCPs over the 95 year period saw an increase in biochar production potential of 0 Gt, 0.5 Gt, 1.4 Gt and 0.3 Gt with assumptions of high yields and low temperature increase (Scenario 8e1), which could be increased to 0 Gt, 1.1 Gt, 3.9 Gt and 6.5 Gt of extra biochar by applying simple adaptation measures (Scenario 9e1). The largest negative impacts of climate induced increases in global mean surface temperature were a reduction of 8.1 Gt, 8.9 Gt, 13.2 Gt and 26.1 Gt of biochar across the scenario period when assuming the lowest crop yields and largest projected increase in global mean surface temperature (Scenario 8d3). This impact was mitigated to a reduction of 3.7 Gt, 4.1, 6.3 and 12.8 Gt, for the four RCPs over the scenario period, when simple adaptation measures were assumed. Simple crop based adaptation measures have the potential to mitigate much of the impacts of increased global mean surface temperature on biochar production, although these methods of adaptation are not explicitly stated by Porter et al, (2014) the may include actions such as irrigation and the planting of hardier cultivars. With these measures impacts are still seen for RCPs 6 and 8.5 under the full range of temperatures projected for these RCPs, although these impacts are greatly reduced from the non-adaptation scenarios. These results suggest that actions to mitigate changes in global mean temperature will see less impact on the production potential of biochar, though crop based adaptation measure may also have a large effect in reducing impact on biochar production even where temperature change does occur. The assessment of climate change impact here only addresses the impact of changes in global mean surface temperature on crop yields, and does not examine other potential climate change impacts such as changes in precipitation, localised surface temperature, pests and diseases or extreme weather events. See Section 5.5 for a discussion of the uncertainties and limitations associated with these results.

The projection of future scenarios will always be subject to elements of assumption and estimation as future scenarios depend on a number of as yet unknown drivers with a number of potential combinations and outcomes. As pathways become clearer, for example through long-term policy implementation, then prediction becomes easier, if still somewhat difficult. The structure of the biochar model would, in future research, allow for further alteration of parameters such as the contribution of each crop type to the total cropland utilized. Updates to the model would also be able to add further spatial and/or temporal detail to these parameters increasing the accuracy of regional projections. This would enable changes to be simulated and explored, in future research, as the uncertainty of future drivers is reduced.

7 Biochar carbon sequestration potential: Methodology

7.1 Introduction

One method of biochar utilization discussed within the literature is sequestering carbon over long time periods, mainly by means of the addition of biochars to soils. The stability of the carbon in biochar, upon this addition to soil, is still uncertain and dependent on a number of variables including the characteristics of the feedstock material, type and conditions of the biochar production process, and the environmental conditions. These environmental conditions are, for example climate and soil conditions such as moisture and microbial activity.

Chapter 4 examined the characteristics of biochars produced from different crop residue feedstocks and under different process conditions. Chapter 6 then used this experimental data in combination with the current biochar literature to develop and examine a number of scenarios which projected how much biochar could be produced from available crop residues, from the four prescribed cropland of the four RCPs, from 2005 to 2100. Following on from this work Chapter 7 details the methodology used to examine the potential recalcitrance of the biochar carbon in these biochar production scenarios, giving a potential quantity of carbon which could be stored long term within each scenario. The quantities of CO₂ which would be removed from the atmosphere and stored long-term in soils are examined. These scenarios of long-term carbon storage are compared to the carbon emission pathway of their respective RCPs, examining the mitigation potential of the biochar produced.

7.2 Methodology

7.2.1 Rates of biochar application

Within the scenarios detailed in Chapter 6 many different quantities of biochar are produced and are available, per hectare, for addition to soil. Some regions may have no localised biochar production capacity due to factors including limited agricultural production, production of low residue producing species, or high localised competition for residues. Other areas may produce more biochar than may be safely or beneficially added to soils within that locality. It was assumed here that surplus biochar produced within one region can be transported to regions with less than optimum production rates of biochar. As this assessment looks at the maximum technical potential for biochar production and soil addition, the economic or logistical issues associated with such assumptions were not addressed here but would be an insightful avenue of further research. For indicative purposes, the maximum and average quantities of biochar produced per hectare, under the assumptions of Scenario 1, were calculated to give an indication of the average and maximum rates of addition to soil if all biochar is added to soil in the same location as produced. These mean and maximum values were calculated for each of the four RCPs under the assumptions of Scenario 1 to indicate possible rates of biochar addition to soil within each RCP.

7.2.2 Carbon sequestration potential

A number of methods of assessing the recalcitrance of biochar carbon are discussed within the literature and summarised in Chapter 2, Section 2.2.5.2. These methods include using historical analogues, laboratory incubation tests, field tests and modelling techniques. There is often variation between parameters used in the different methodologies, for example some methods look at the recalcitrance of biochars in different soils, where others may look at degradation rates in other mediums or of biochar alone. This often makes direct comparison between studies difficult. A number of modelling based techniques, which draw on both the literature and primary experimental data, are used here to determine the potential carbon sequestration potential of the biochar quantities projected in the scenarios of Chapter 6. The three methods applied here are (1) the carbon sequestration potential (CS) methodology of Zhao et al. (2013), (2) an adaptation of this CS potential equation which was developed using correlations found through the experimental work of Chapter 4, and (3) the two-pool methodology of Woolf et al. (2010). The alternative CS equation, using correlation between the feedstock volatile content and the biochar R₅₀ value was developed to assess its potential as an alternative method of recalcitrance estimation.

7.2.2.1 Carbon sequestration methodology of Zhao et al (2013)

The carbon sequestration potential (CS) methodology, which builds on the biochar recalcitrance work (R_{50} recalcitrance index) of Harvey et al. (2012), was proposed by Zhao et al. (2013). This R_{50} recalcitrance index methodology (See Equation 3-1) was used in Chapter 4 to calculate the relative recalcitrance of the biochars produced experimentally within this study. The biochars produced were classified as either Class B or Class C, meaning under the classification system of Harvey et al. (2012) they would undergo 'minimal degradation' and 'more severe degradation' respectively, over time when added to soils. None of the biochars produced experimentally were found to be Class A biochars which are the 'most recalcitrant', though some biochars have been characterised as Class A within other studies (Harvey et al., 2012, Zhao et al., 2013). The R_{50} indexing of biochars by Harvey et al. (2012) does not indicate a specific timescale for the degradation of the biochar, or give a range of values for the amount of degradation which would occur, only a ranking against the recalcitrance of graphite which is highly stable in soils. Zhao et al. (2013) used the R_{50} index of Harvey et al. (2012) to develop a method of determining the

amount of carbon, from the original feedstock carbon content, which would be stored long term in soils upon addition of a biochar to soil. They termed this the carbon storage potential (CS) of the biochar.

$$CS(\%) = (M x Ch x C_{Ch} x R_{50}) / M x C_{F}$$
(7-1)

Where: M = mass of feedstock (g), Ch = yield of biochar (%), C_{Ch} = carbon content of the biochar (%), R_{50} = recalcitrance index, C_F = carbon content of the feedstock (%).

The CS potential equation of Zhao et al. (2013), was used here to determine the carbon storage potential of the biochar quantities produced within the scenarios projected within Chapter 6. The experimental data of Chapter 4 was used where relevant, with the ultimate analysis results for elemental carbon content for both the feedstocks and biochars (detailed in Table 4.2) used for carbon contents within the equation. Biochar yield (%) from the experimental data was also used. Values for the mass of feedstock available for biochar production were taken from the relevant biochar scenario as detailed in Chapter 3. Where more than one crop residue type for a particular crop group (see Table 5-2) was tested experimentally, the average biochar yield value was taken to determine average production for that crop group. The CS potential (%) of the biochar produced in each scenario was calculated using Equation 7-1 determining how much of the original feedstock carbon, per grid cell, would be retained long term in soil upon charring and addition to soils. As with the R₅₀ index of Harvey et al. (2012), no definition of the time period assumed to be 'long term' is given within the literature for the CS potential. As the Harvey et al. (2012) study, which developed the R_{50} index used by Zhao et al. (2013) to calculate CS, discussed, the lifetime of biochar in soils can range from 'under a century to several millennia'. It is therefore assumed here that the period of stability defined by Zhao et al. (2013) as 'long term' is longer than the 95 year assessment period used as the scenario period here. This was deemed to be an acceptable assumption as 95 years is at the lower end of the biochar lifetime range discussed in Harvey et al. (2012).

7.2.2.2 Recalcitrance from volatile content equation (RVC)

The discussion by Harvey et al. (2012) regarding the assessment of biochar lifetime also proposed a number of alternative methods, which use various biochar properties, to predict the lifetime of particular biochars, including using the thermal degradation and volatile content of biochars as indicators of biochar stability in soils. They proposed that these methods may provide estimated lifetimes for each biochar, and could potentially provide simplified methods of estimation where only some of the biochar characteristics are known, or if only simple analytical techniques can be applied due factors such as to cost or technical capacity. Further work is required to validate the effectiveness of these simple estimation methodologies. This discussion within literature on alternative indicators of biochar lifetime was expanded here by further development of the CS equation of Zhao et al. (2013) to use the feedstock volatile content as a predictor of biochar stability, in place of the R₅₀ index. The new equation uses feedstock volatile content (see Table 4.2) due to the correlation determined between R₅₀ and feedstock volatile content ($r^2 = 0.67$, p = 0.01) (see Figure 7-1).



Figure 7-1: Relationship between feedstock volatile content (% daf (dry, ash free)) and biochar R_{50} index values. Equation of line (y=mx+c) is shown, where m is the line gradient and c is the y intercept.

The equation of straight line produced from the correlation analysis of the two variables is:

$$y = -0.0073x + 1.1449 \tag{7-2}$$

Equation 7-2 was used in place of the biochar R_{50} values in Equation 7-1 resulting in the equation:

$$CS(\%) = (M \times Ch \times C_{Ch} \times (-0.0073x + 1.1449)) / M \times C_{F}$$
(7-3)

Where: x = the feedstock volatile content on a dry, ash free (daf) basis.

This new equation for the calculation of biochar carbon sequestration potential (Equation 7-3) is, from this point, termed the recalcitrance from volatile content (RVC) equation. The uncertainty of the regression slope (from Figure 7-1) was also calculated and used to determine whether the original projections of carbon sequestration potential (using Equation 7-1) lie within the projection range of carbon sequestration potential made using Equation 7-3 when applying this uncertainty of slope.

7.2.2.3 Two-pool calculation methodology

In their study of biochar carbon sequestration potential under current land-use regimes, Woolf et al. (2010) used a two-pool method of assessing the decay kinetics of biochars in soil. The equation (Equation 7-4) assumes that biochar consists of two fractions, one labile and one recalcitrant, which are modelled as two separate pools with different decay properties.

$$M(t) = M_0 \left[L \exp(-\ln(2)/t_{1/2L} t) + R \exp(-\ln(2)/t_{1/2R} t) \right]$$
(7-4)

Where:

M(t) = mass of carbon at time t, M_0 = initial mass of biochar carbon, L = labile fraction of biochar, R = recalcitrant fraction of biochar, $t_{1/2t}$ = labile half-life, $t_{1/2R}$ = recalcitrant half-life.

The method assumes that each pool follows an exponential decay curve, with the assumptions of the main scenario in Woolf et al. (2010) being that the labile fraction, constituting 15 % of the biochar carbon, has a half-life of 20 years, and the recalcitrant fraction (the remaining 85 % of the biochar carbon), has a half-life of 300 years. Variance in $t_{1/2L}$ of 1 to 25 years, and $t_{1/2R}$ of 50 – 1000 years is used by Woolf et al. (2010) to assess the effects of the range of degradation timescales reported within the literature. They discuss that using an upper limit of 1000 years for $t_{1/2R}$ may be a conservative upper estimate as some reporting in the literature would extend $t_{1/2R}$ beyond this value. They also explore uncertainty in the assumptions of the size of the labile and recalcitrant fractions of biochar carbon by applying values of L = 5 – 30 % (thus R = 95 – 70 % respectively).

The two-pool lifetime method was applied here to the biochar scenarios of Chapter 6 to assess whether the use of this alternative method for calculating biochar stability over time resulted in different projections from the carbon storage potential projections determined using the CS equation of Zhao et al. (2013) and the RVC equation. The main assumptions of L = 15 %, $t_{1/2L}$ = 20 and $t_{1/2R}$ = 300 from Woolf et al. (2010) were used, and the upper and lower uncertainty

estimates for L, $t_{1/2L}$ and $t_{1/2R}$ were also explored to see how variance in the assumptions of the size of the labile and recalcitrant fractions, and the half-lives of these fractions, affected the lifetime projections. The timeframe of assessment of the stored carbon was also altered, examining the quantity of biochar carbon remaining stable after, for example, 500 and 1000 years under different assumptions of recalcitrance. This gave an indication of the overall longevity of the biochars, not just the carbon stored over the 95 year period.

7.3 Summary

The methodologies for determining the efficacy of each biochar scenario at long-term carbon sequestration are detailed. The mean and maximum rates of biochar production per hectare of cropland were determined for each RCP under the assumptions of Scenario 1. This gives an indication of the amount of biochar which would be added to soils, per hectare. This is an indication of the addition to soil rates as the crop residues are likely to be collected to a more central point and then re-distributed, potentially to other regions, after biochar production. The three methods used to assess the carbon sequestration of the biochars in each scenario are the CS potential (Equation 7-1), the recalcitrance from volatile content (RVC) equation (Equation 7-3) and the two-pool method (Equation 7-4). The RVC equation, developed from the correlation between feedstock volatile content and the R_{50} index value of the biochar, was examined for effectiveness as an alternative biochar lifetime estimation tool. The CS and RVC equations were used to estimate the long-term carbon storage potential of the biochars in each scenario from Chapter 6. The two-pool method was used to assess the remaining carbon after the 95 year scenario period, and also to examine the long-term sequestration potential of biochars with different assumptions of labile and recalcitrant fraction size and decay rates. The CS equation (Equation 7-1) was then used to assess the long-term carbon storage of the biochars produced in all scenarios, for all RCPs. These values were used to determine the potential of each scenario for the mitigation of carbon emissions for each RCP. The results of these analyses are detailed and discussed in Chapter 8.

8 Biochar carbon sequestration potential: Results and discussion

8.1 Introduction

The maximum and mean rates of biochar production, per hectare, were determined to give an indication of the rates of biochar addition to soil which may occur within the scenarios discussed in Chapter 6. Three modelling methods were used to estimate the potential for long-term carbon storage of the scenarios of biochar production. These methods are the carbon sequestration (CS) equation of Zhao et al. (2013), the Recalcitrance from Volatile Content (RVC) equation developed in Section 7.2.2.2, and the two-pool equation methodology from Woolf et al. (2010). The projections of the CS and RVC equations were compared, assessing whether the uncertainty range of the projections made using the RVC equation encompassed the projections made using the CS equation. This comparison was used to determine the effectiveness of the RVC as an alternative recalcitrance estimation tool. The two-pool methodology was used to offer another comparison tool, and also to provide estimates of the overall recalcitrance timeframes of the biochars.

8.2 Rates of biochar addition to soil

The mean and maximum rates of biochar production were calculated for each RCP under the assumptions of Scenario 1 to give an indication of the mean and maximum rates of biochar application to soil which may occur within each RCP.

Table 8-1: Mean and maximum annual biochar per hectare for the four RCPs in 2005, 2100 and across the whole scenario period for Scenario 1.

	Rates of biochar production (tonnes per hectare (t ha yr ⁻¹)						
	20	2005		.00	Whole scenario		
RCP	Mean	Max	Mean	Max	Mean	Max	
2.6	0.10	1.51	0.17	2.20	0.14	2.20	
4.5	0.10	1.51	0.19	2.93	0.16	2.93	
6	0.10	1.51	0.27	3.81	0.16	3.81	
8.5	0.10	1.51	0.37	5.96	0.20	5.96	

The British Biochar Foundation (2014) recommend that biochar addition to soil is kept to a minimum of 3 kg m² until further research is conducted to investigate the safety of larger additions. The maximum rate of biochar production for any of the RCPs during the scenario period is almost 6 t ha yr^{-1} which is equivalent to 0.6 kg m² yr^{-1} . This is, therefore, well within the

recommended addition rates. This 0.6 kg m² yr⁻¹ rate is assuming that biochar addition is made at the point of residue collection and redistribution is even. This is, of course, a simplification of any potential distribution scenario but as the 0.6 kg m² yr⁻¹ is well below the recommended upper limit of 3 kg m² there is capacity for much of the biochar to be added where it is deemed most useful and economical, rather than by the even redistribution scenario detailed here. As this biochar accumulates, over time, to closer to the 3 kg m² threshold then alternative distribution sites could be used as required. As discussed previously, in reality all biochar is unlikely to be added to soil where the crop residues are produced. Depending on the scale of the biochar production system residues may be collected and transported to biochar production plants. Biochar may then be distributed to different locations from these plants, as required. This transport and distribution may add an emissions penalty to the biochar system which would be a useful focus for further research.

8.3 Carbon sequestration potential (CS) equation



8.3.1 Total carbon storage

Figure 8-1: Annual carbon storage potential (GtC yr⁻¹) of the biochar produced within the RCPs under the assumptions of Scenario 1 (left) and the cumulative carbon storage potential (GtC) of this biochar over the scenario period (right). 'Stored carbon' refers to the biochar carbon which would remain in stable form in soils from the point of addition to the scenario end point of 2100.

Under the assumptions of Scenario 1, 0.34 Gt yr⁻¹ of the biochar carbon added in 2005 would remain stable in soils for long time periods. For all RCPs the annual amount of carbon stored long term generally increases over time, resulting in 0.65, 0.54, 1.07 and 1.54 GtC yr⁻¹ in 2100 for RCPs 2.6, 4.5, 6 and 8.5 respectively. Cumulatively the addition of the biochar produced in each scenario to soils throughout the scenario time-period results in the long-term storage of 49 GtC,

46 GtC, 61 GtC and 77 GtC respectively for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5. The contribution of each biochar type is discussed in Section 8.4.2. The impact that this storage of carbon may have on the emission pathways of CO_2 for each RCP is discussed in Section 8.7.

Figure 8-2 shows the cumulative carbon storage potential of Scenarios 1 to 7 using the CS potential equation. The effects on carbon sequestration potential of uncertainty in crop yields, land-use change, crop RPRs, residue availability, biochar yields, and biochar carbon content are shown in the different panels. The effects of scenario uncertainty often led to larger ranges in the potential carbon storage projections as the RCPs increase from RCP 2.6 to RCP 8.5. This often means that the carbon storage potential of scenarios projected under the RCP2 pathway have smaller uncertainty ranges, with uncertainty increasing with increasing RCP up to RCP 8.5 where the largest uncertainty range is often seen.



Figure 8-2: Total carbon storage potential (GtC) of Scenarios 1 to 7 using the CS equation of Zhao et al. (2013). Plots show the cumulative carbon storage potential for the 95 year scenario period. The main legend is shown at the top of the figure. Individual legends shown inside a plot corresponds to that plot only.

If no crop yield increases are applied throughout the scenarios then total carbon storage potential of the RCPs are reduced to similar amounts for all RCPs, ranging from 36 GtC to 38 GtC. Keeping land use static at 2005 levels reduces the total carbon sequestration potential of each RCP, from that of Scenario 1, except for that of RCP 4.5 which is slightly increased by almost 5 GtC over the scenario period. The occurrence of very low RPR factors (Scenario 4a2) would greatly reduce the carbon sequestration potential of the RCPs, resulting in 20.3 GtC, 18.2 GtC, 23 GtC and 28.5 GtC stored long-term for the four RCPs respectively. These values are increased to nearer Scenario 1 levels with the application of medium RPR values (Scenario 4a1). The application of high average RPR values results in carbon storage similar to Scenario 1 for RCP 2.6 and RCP 6, slightly lower than Scenario 1 sequestration for RCP 4 and slightly above Scenario 1 sequestration for RCP 8.5. This variation in the effects of RPR may be due to the effect altering the RPRs of different crop types, which may have different weightings in the different RCPs due to the regional distribution of land over time in each RCP. When examining scenarios of residue availability both the minimum and medium availability scenario (5a and 5b) reduce the carbon sequestration potential of the RCPs relative to Scenario 1. The minimum carbon sequestration potential of the residue availability scenarios would see 29.3 GtC, 28.9 GtC, 37.1 GtC and 46.2 GtC stored long term for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively. Using both the high and low residue availability scenarios of Woolf et al. (2010) reduces the carbon sequestration potential relative to Scenario 1. The minimum carbon storage under these alternative residue availability scenarios is projected to be 30.5 GtC, 28.5 GtC, 37.8 GtC and 49.3 GtC for the four RCPs respectively. This is increased to 39.2 GtC, 36.2 GtC, 48.5 GtC and 62.3 GtC respectively when the high residue availability scenario of Woolf et al. (2010) is applied. Applying the low biochar yield assumptions reduces the carbon sequestration potential of Scenario 1 by between 10 GtC and 20 GtC for the four RCPs over the whole scenario period. Applying the highest biochar yield assumptions increases the potential of carbon sequestration greatly for all RCPs, resulting in between 90.1 GtC and 153.6 GtC stored (RCP 4.5 and RCP 8.5 respectively). The achievement of these high biochar yields from all crop residues for the full scenario period is unlikely compared to the other biochar yield scenarios. This is due to a number of factors, including the technological requirements necessary to achieve such yields, the variation in yields achievable from different feedstocks and the likely economic requirement to optimise the biochar producing process to also produce oil and/or gas as a fuel product. Altering the assumed biochar carbon content had little effect on the carbon sequestration of the RCPs in comparison to some of the other parameters. Scenarios 7b and 7c (medium and high average biochar carbon content values respectively) both have the potential to sequester more carbon

than Scenario 1. Using the low biochar carbon content assumptions (Scenario 7a) reduces the total carbon storage potential of the scenario to 41.9 GtC, 48.5 GtC, 63.2 GtC and 79.8 GtC for the four RCPs respectively.



8.3.2 The Impacts of climate change on carbon sequestration potential

Figure 8-3: Total carbon storage potential of the RCPs under the climate change impact assumptions of Scenario 8d and 8e. Scenarios 8d2 and 8e2 (light grey and dark grey respectively) are shown in comparison to the carbon storage potential of Scenario 1 (black). Scenario sub-set 8 details the projections for biochar potential under climate change where no adaptation measures are applied.

As discussed previously the impact on climate change is likely to reduce the potential of the biochar scenarios to sequester carbon, with increasing impact on the CS potential likely with increasing changes in climate. The climate impacts examined within these scenarios are changes in global mean surface temperature caused by the projected change in radiative forcing which would be expected from the emissions pathway of each RCP. The effects of these projected changes in global mean surface temperature were applied to the residue production potential of each scenario (in Chapter 6). The effect on carbon sequestration potential was then determined using the CS equation of Zhao et al. (2013) (Equation 7-1). Figure 8-3 and Figure 8-4 show the impacts of the range of crop yield impacts projected by Porter et al. (2014) on the CS potential of Scenario 1. The figures show only the projections of Scenarios 8 and 9 e and d as these scenarios

assess the maximum and minimum yield impacts, alongside the range of temperature projections. The projections of scenarios 8 and 9 a - c, which assess the mean yield projections alongside the range of temperatures, are encompassed within the range explored in Figure 8-3 and Figure 8-4.

The CS potential of RCP 2.6 was the same when the range of potential yield impacts under the mean temperature change scenario were applied (Figure 8-3). A reduction in the CS potential of RCP 2.6, when compared to that of Scenario 1, was seen under the scenario of low yield and high temperature assumptions (Scenario 8d3). The low yield projections combined with the mean projected temperature change resulted in a reduction in carbon sequestration, relative to that of Scenario 1, of 2.27 GtC, 3.45 GtC and 21.25 GtC respectively for RCP 4.5, RCP 6 and RCP 8.5. This highlights the increasing impact of global mean temperature change as the radiative forcing of the pathway increases (i.e. with increasing RCP). Applying high yield and mean temperature change assumptions resulted in small increases in total CS potential of 0.42 GtC and 0.38 GtC respectively for RCP 4.5 and RCP 6 relative to Scenario 1. The application of these assumptions to RCP 8.5 resulted in a reduction of 0.92 GtC for the total CS potential relative to Scenario 1. The impact of the range of temperature projections on CS potential for each RCP widens in range with increasing RCP (from RCP 2.6 up to RCP 8.5) when applied with assumptions of either low or high yield. Figure 8-4 shows the impact of simple adaptation measures to the CS potential of the climate change scenarios.



Figure 8-4: Total carbon storage potential of the RCPs under the climate change impact assumptions of Scenario 9d and 9e. Scenarios 9d2 and 9e2 (light grey and dark grey respectively) are shown in comparison to the carbon storage potential of Scenario 1 (black). Scenario sub-set 9 details the projections for biochar potential under climate change but where some simple adaptation measures are applied.

The application of simple crop based adaptation methods improved the carbon sequestration potential of biochar in all of the RCPs (Figure 8-4). No impact was seen in carbon sequestration potential where high or low yield projections were applied to the mean temperature projections for RCP 2.6. The high yield projection, alongside the high temperature projection, may provide some benefit in increased carbon sequestration due to the potential increase of crop yields with small increases in global mean temperature coupled with adaptation measures for RCP 2.6. The reduction in total carbon storage potential seen in RCP 4.5, RCP 6 and RCP 8.5 in the scenarios of low yield projections with no adaptation (Scenario 8d) was lessened when adaptation measures were employed (Scenario 9d). This brought the CS potentials much more in line with those of Scenario 1, seeing a reduction of only 1.55 GtC, 2.03 GtC and 1.99 GtC for the highest three RCPs respectively. This is a marked improvement on the reduction in CS potential seen in Scenario 8d2. The increases in CS potential which were seen with the high yield projections of Scenario 8e were further increased where adaptation measures are applied. These increases, relative to

Scenario 1, were: 1.03 GtC, 1.54 GtC and 3.23 GtC for the total scenario period for RCP 4.5, RCP 6 and RCP 8.5 respectively.



Figure 8-5: Total carbon storage potential of the RCPs under the climate change impact assumptions of Scenario 8f. Scenario 8f (dark grey) is shown in comparison to the carbon storage potential of Scenario 1 (black).

Using the climate change yield impact projections of Kyle et al (2014) as an alternative to the projections used in Scenarios 8 a to e and 9 a to e sees total CS potentials similar to those of Scenario 1. A small increase in CS potential of 0.43 GtC was seen in RCP 2.6, which can perhaps be attributed to increases in crop yields related to a small increase in global mean temperature. Reductions in CS potential are seen for the other three RCPs, with a reduction of 1.78 GtC, 2.08 GtC and 4.02 GtC for RCP 4.5, RCP 6 and RCP 8.5 respectively. This demonstrates an increasing negative impact on carbon sequestration potential as radiative forcing increases.

8.4 Recalcitrance from volatile content (RVC) equation

8.4.1 Projection of total carbon storage potential



Figure 8-6: Carbon storage potential of the four RCPs under the assumptions of biochar scenario 1 using the two variations of CS equation. The solid line denotes the carbon storage potential of Scenario 1 using the CS equation of Zhao et al. (2013) (Equation 7-1). The dashed line denotes the carbon storage potential of Scenario 1 using the RVC equation (7-3).

Using the RVC equation in place of the CS equation led to increases in the projected total carbon sequestration potential of the biochar produced within each RCP, under the assumptions of Scenario 1, of 2.85, 2.80, 3.65 and 4.36 GtC in for RCPs 2.6, 4.5, 6 and 8.5 respectively. The original projection of carbon storage potential (using Equation 7-1) was found to lie within the uncertainty range of the adapted equation (Equation 7-3). This indicates that Equation 7-3 can be used as an alternative equation to Equation 7-1 for the estimation of the long term carbon storage potential of biochars, as defined by Zhao et al. (2013). This may be of benefit where only limited data regarding the feedstock and biochar characteristics is available, with proximate analysis of the feedstock giving the feedstock volatile content, rather than requiring thermogravimetric analysis of the biochar to provide the R₅₀ index value. This may also be of

benefit if a number of feedstocks are available, and feedstock analysis can help to inform the choice of feedstock made.



8.4.2 Projection of carbon storage potential of different biochars

Figure 8-7: Carbon storage potential of the different biochar types under the assumptions of biochar scenario 1 using the two variations of CS equation. The solid line denotes the carbon storage potential of Scenario 1 using the CS equation of Zhao et al. (2013) (Equation 7-1). The dashed line denotes the carbon storage potential of Scenario 1 using the RVC equation (7-3). N.B. Scales vary between plots.

The carbon sequestration potential of the biochars produced from different crop groups varies greatly, as detailed in Figure 8-7. Cereal biochars have, throughout the scenario period, the greatest potential for long term carbon storage. This is followed by sugarcane bagasse. Cereal residues have high CS potential mainly due to the large residue quantities produced and available for biochar production, relative to the other residue types. Rice biochars are projected to have just over half of the potential of cereal biochars to sequester carbon long-term. The CS potential of oil biochars is around 10^{-2} smaller than that of cereal biochars. Fibre biochars have a relatively small CS potential, being of the order of 10^{-4} smaller than that of cereal biochars. Both oil and fibre crops have lower residue production and availability factors. Compared to the other crop groups, sugarcane bagasse has a relatively low R₅₀ index, and olive pomace has a relatively low C content, which contribute to the lower CS potentials of these groups.

Figure 8-7 shows the difference in projections of CS potential between the CS equation (Equation 7-1) and the RVC equation (Equation 7-3). The projections of CS potential made for the different biochar types using the RVC equation are well fitting with the projections made using the CS equation, except perhaps for that of cereal residues, where the difference in 2005 in CS potential is 0.02 GtC yr⁻¹, or 12.6 %. The difference projected in the sequestration potential of cereal biochars lies within the bounds of uncertainty determined for Equation 7-2, making it a suitable projection for use here. All other projections were an excellent fit between the two datasets. The differences between projections of total carbon stored over the full scenario period made using the RVC equation and CS equations, for the different biochar types, are detailed in Table 8-2.

Feedstock	Change in CS projection (GtC)					
	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Cereals	2.63	2.50	3.33	3.99		
Rice	0.37	0.41	0.49	0.62		
Oil	0.03	0.02	0.03	0.04		
Fibre	-0.0004	-0.00037	-0.00047	-0.00057		
Sugarcane	-0.18	-0.13	-0.21	-0.29		

Table 8-2: Difference in total CS projection (GtC) resulting from the use of the RVC equation(Equation 7-3) in place of R50 index in the CS equation of Zhao et al. (2013) (Equation 7-1).

8.5 Carbon sequestration potential – 2 pool method

8.5.1 95 Year sequestration potential



Figure 8-8: Annual carbon storage potential (GtC yr-1) (left) and the cumulative carbon storage potential (GtC) (right) of this biochar over the scenario period. Values were calculated, using the two pool method (Equation 7-4), for the biochar production projected under the assumptions of Scenario 1 (see Section 5.3). The carbon sequestration potential is shown using a) the range of biochar half-lives (recalcitrant fraction: 50 - 1000 years and labile fraction 1 - 25 years) and b) the range of labile fraction size (5 to 30 %) used by Woolf et al. (2010)). 'Stored carbon' refers to the biochar carbon which would remain in stable form in soils from the point of addition to the scenario end point of 2100.

The alternative two-pool calculation method resulted in different projections of the carbon storage potential of the scenarios, generally giving higher carbon storage potential values than where Equations 7-1 and 7-3 were used. These differences may be due, in large part, to the

specification of carbon storage timeframe (i.e. to 2100) using the two-pool method. In comparison, the CS equation only specifies that the carbon stored will be stable 'long-term'. As this 'long-term' carbon storage projection is likely to extend beyond this 2100 assessment point this makes direct comparison between the two assessments difficult. To examine the longer term carbon storage potential longer time periods were also assessed using the two-pool method (see Section 8.5.2).

More impact was seen on the biochar scenarios where the variation in half-life was explored than where labile fraction was varied (see Figure 8-8). Applying low half-life assumptions meant that much of the biochars applied to soils in the would quickly degrade, meaning that biochars added to soils towards the end of the scenario period would have much greater potential of storing carbon to the end of the scenario period (2100), but would quickly degrade past the end of the 95 year scenario period. Applying longer half-life values increased the carbon storage potential, at the end of the scenario period, of the biochars produced in the early part of the scenarios. This led to an increase in the cumulative stored carbon potential of the scenario when compared to the main half-life assumptions and the low half-life assumptions. Cumulative carbon sequestration potential across the 95 year period was increased by between 6.3 % and 7.2 % (across the four RCPs) when a low labile fraction was assumed, and decreased by 10.5 % and 12.2 % where a high labile fraction was assumed.

8.5.2 Longer-term carbon storage potential



Figure 8-9: The effect of variance in the assumed half-life of a) the recalcitrant fraction and b) the labile fraction of biochar on the long term carbon sequestration potential of the biochars produced in RCP 2.6 under the assumptions of Scenario 1.

The variance explored in the half-life of the recalcitrant fraction of biochar was 50 to 1000 years, with the main assumption being 300 years. Half-life assumptions for the labile fraction of biochar were 1 to 25 years, with the main assumption being 20 years. Where a half-life of 50 years was assumed for the recalcitrant fraction the biochar carbon was degraded within 100 years after addition. Where 1000 year recalcitrant fraction half-life was assumed the biochar maintained some stability past 2500 years from addition. The lifetime of the recalcitrant fraction had the largest effect on the overall stability of the biochar carbon under the main assumptions of fraction size. Where the shorter lifetime assumptions of the labile fraction were applied, an initial small reduction in the long term stability of the biochar carbon was seen.



Figure 8-10: The effect of variance in the size of the labile fraction of biochar on the long term carbon sequestration potential of the biochars produced in RCP 2.6 under the assumptions of Scenario 1.

The range in labile fraction size discussed by Woolf et al. (2010) was also tested, with the effects summarised in Figure 8-10. A larger labile fraction resulted in a faster degradation of biochar which then became more stable as the labile fraction was exhausted and the recalcitrant fraction remained.



Figure 8-11: The effect, on the long term carbon sequestration potential, of the minimum and maximum assumptions (labile and recalcitrant fraction half-life and fraction size) of biochar recalcitrance for biochars produced in RCP 2.6 under the assumptions of Scenario 1. The minimum assumptions (large labile fraction and short half-lives) are shown in black in plots A and B. The maximum assumptions (large recalcitrant fraction and long half-lives) are shown in plots A and C. N.B. Scales vary between plots.

Figure 8-11 shows the stability of the biochar produced in RCP 2.6, under the assumptions of Scenario 1, under the worst case and best case recalcitrance assumptions examined here. The difference between the two projections is large, and is in the most part influenced by the half-life of the recalcitrant fraction of biochar, where the recalcitrant fraction is large enough to have a real long-term influence. As discussed in Section 7.2.2.1, the CS equation (Equation 7-1) has no specific time frame for the lifetime of the stable carbon, only specifying the calculation of 'long-term carbon storage'. Comparison of the quantity of carbon remaining at the end of the scenario period, as projected using Equation 7-1, with the rates of decay projected using the worst case assumptions with the two-pool method (Equation 7-4) shows large differences in the
projections of remaining carbon. This indicates that the length of time projected for carbon sequestration using the CS equation is likely to be longer than the 95 year assessment period. There are a number of combinations of labile vs. recalcitrant fraction sizes and half-life assumptions which may result in the same quantity of carbon remaining in 2100 as projected using Equation 7-1, although these assumptions are not within the range assumed here or in Woolf et al. (2010). Using the assumptions, in Equation 7-4, of a labile fraction size of 5 %, alongside half-lives of 25 years and 1000 years for the labile and recalcitrant fraction respectively projects similar carbon storage quantities after 1000 years as the 'long-term' storage projections of Equation 7-1. This is also true after 750 years where a labile fraction of 15 % is assumed alongside labile and recalcitrant fraction half-lives of 20 and 1000 years and after 250 years where a labile fraction of 15 % is assumed alongside labile and recalcitrant half-lives of 1 and 300 years respectively. This highlights that uncertainty may arise when making very long term projections of biochar carbon storage potential using Equation 7-1 or 7-3, making a definite lifetime prediction difficult. If a detailed estimate of sequestration timeframe is required then the use of equation 7-4 may be more suitable.

8.5.3 Comparison of the three methodologies

Using the RVC equation (7-3) increased projections of carbon storage in 2100 by around 5 % relative to the projections made using the CS equation (7-1). The two-pool method (Equation 7-4) increased projections of stored carbon potential by between 38 % and 40 % for the different RCPs, relative to the CS equation (7-1). As discussed previously this large difference in the projections of stored carbon made between using either Equation 7-1 or 7-3 and Equation 7-4 is in large part due to the specification of decay period for Equation 7-4, where the other equations do not specify a decay period. They determine a 'long-term' carbon sequestration potential which may be far longer than the 95 year stability period specified for the main analysis using Equation 7-4. The assumption of longer assessment period in Equation 7-4 may better align the assessment periods of the two equations, reducing the variance seen in the projections of CS potential.

8.6 CO₂ Reduction Potential

Table 8-3, shows the stored carbon and CO_2 removal potential of the first and last years of the scenario period, highlighting change over time, and the total scenario period projected using the three methods of calculation.

Table 8-3: Carbon remaining at the end of the scenario period (in 2100 and for Scenario 1 assumptions) which was added to soil in 2005, 2100 and the across the total scenario period for the three methods of calculation (CS equation (Equation 7-1), RVC equation (Equation 7-3), and two-pool method (Equation 7-4)) and the related carbon in units of CO₂.

Stored carbon										
			GtC	yr⁻¹				GtC		
RCP	2005				2100			Total		
	CS	RVC	2 Pool	CS	RVC	2 Pool	CS	RVC	2 Pool	
RCP 2.6	0.34	0.36	0.46	0.65	0.69	1.29	49.01	51.90	80.13	
RCP 4.5	0.34	0.36	0.46	0.54	0.58	1.08	45.83	48.63	74.60	
RCP 6	0.34	0.36	0.46	1.07	1.13	2.12	60.93	64.58	101.34	
RCP 8.5	0.34	0.36	0.46	1.54	1.62	3.03	77.15	81.51	129.02	
			Carb	on stored	in unit of	CO ₂				
			Gt CC) ₂ yr ⁻¹				Gt CO ₂		
RCP 2.6	1.25	1.32	1.69	2.39	2.53	4.73	179.9	190.5	294.1	
RCP 4.5	1.25	1.32	1.69	1.98	2.13	3.96	168.2	178.5	273.8	
RCP 6	1.25	1.32	1.69	3.93	4.15	7.78	223.6	237.0	371.9	
RCP 8.5	1.25	1.32	1.69	5.65	5.95	11.12	283.1	299.1	473.5	

Using the two-pool methodology to determine the C storage potential indicates that between 274 Gt CO₂ and 474 Gt CO₂ could be removed from the atmosphere and stored for the 95 year scenario period. Depending on the size and decay periods of the labile and recalcitrant fractions of the biochars much of this CO₂ could be sequestered for long time scales. The reaction of the other carbon sinks to this removal may be to release some CO₂ back into the atmosphere (Lenton and Vaughan, 2009). The long term CO₂ sequestration potential of the four RCPs, calculated using the CS equation, is 180 Gt CO₂, 168 Gt CO₂, 224 Gt CO₂ and 283 Gt CO₂ for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively.

8.7 Impact of Biochar on RCP CO₂ Emissions Pathways

8.7.1 Scenario 1

Under the assumptions of Scenario 1, the mitigation impact of biochar production on the CO₂ emissions pathway of each RCP, throughout the 95 year period, shows a generally increasing impact from RCP 4.5 to RCP 8.5. As detailed in Table 8-3, projections for RCP 2.6 show more potential for biochar to reduce the RCP carbon emissions than RCP 4.5, which can be seen in



Figure 8-12. Figure 8-12 also illustrates the increasing potential of biochar to mitigate carbon emissions across the scenario period, with every RCP seeing greater potential in 2100 than 2005.

Figure 8-12: Carbon equivalent of the CO_2 emissions projections of the RCPs under the assumptions of Scenario 1 (left), and the potential reduction in emissions which can be achieved for each RCP using the three different equations for calculating long term carbon storage (right).

This increasing potential for carbon sequestration using biochar, across the scenario period, is due to assumptions in Scenario 1 such as increasing crop yields.

The total projected carbon emissions without any biochar systems in place, across the 95 year period, are 434.7 GtC, 837.7 GtC, 1208.7 GtC and 1856.4 GtC for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively. Under the assumptions of scenario 1, using the CS equation, these values may be reduced by 49.0 GtC, 45.8 GtC, 60.9 GtC and 77.2 GtC for the four RCPs respectively with the application of biochar carbon sequestration. These are reductions of 11 %, 5 %, 5 % and 4 % on the carbon emissions pathway of the four RCPs respectively. The CO₂ reduction seen where the CS and RVC equations are used are extremely similar projections, highlighting again that the use of Equation 7-3 is acceptable in the place of Equation 7-1. Where the carbon sequestration potential over a specific time period is required, the two-pool equation (Equation 7-4) can offer a more detailed projection. The use of Equations 7-1 and 7-3 is a more accurate indicator of the longer term CO₂ reduction potential of the RCP emission pathways than the use of Equation 7-4 would, in the longer term be reduced as more of the biochar degraded.

8.7.2 Mitigation potential of the alternative biochar scenarios



Figure 8-13: Potential emission reductions for RCP2.6, relative to the carbon emission pathway of projection RCP2.6 (black line). Blue line shows emissions reduction potential of Scenario 1 biochar assumptions. Other scenario assumptions shown are: top left, Scenario 2; top right, Scenario 3; mid-left, Scenario 4; mid-right, Scenario 5; bottom left, Scenario 6; bottom right, Scenario 7.

The mitigation potential of the alternative biochar scenarios was examined, looking at the potential reduction in emissions possible for each RCP. Figure 8-13 shows the potential emissions reduction, from the RCP 2.6 carbon emissions pathway, which is projected for the biochar scenarios 1 to 7. Figure IV-1 to Figure IV-3 in Annexe IV show the mitigation potentials of Scenarios 1 to 7 for RCPs 4.5, 6 and 8.5.

The different assumptions of biochar scenarios 1 to 7 have varied level of impact on the carbon emissions pathway of RCP 2.6. Where different potential crop yield changes are applied (Scenario 2), divergence between the impacts of different sub-scenarios becomes more evident from around 2045. This divergence then continues up to 2100. The application of a scenario of no land-use change (Scenario 3a) resulted in comparatively little change in CO₂ reduction potential, relative to Scenario 1, for RCP 2.6, indicating that land-use change in RCP 2.6 has little effect on the projections made using Scenario 1 assumptions. Exploring potential uncertainty in the RPR values of crops shows that variability in this parameter may have a large impact in the potential for carbon sequestration using biochar. The assumption of low RPR values throughout the scenario period in Scenario 4a2 had a large impact, reducing the carbon mitigation potential, relative to Scenario 1, for RCP 2.6. The assumption of high RPR values increased the mitigation potential, relative to Scenario 1, but not to the same magnitude as with the assumption of low RPR values. Variation in residue availability (Scenario 5) produced a smaller range of projections than the variation seen with Scenario 4 assumptions. Scenarios 5a to 5c produced relatively little variation around the projection of Scenario 1 for RCP 2.6. The application of both conservative and optimistic residue availability assumptions of Woolf et al. (2010) (Scenarios 5d and 5e respectively) both reduced the mitigation potential of biochar production within RCP 2.6 relative to both Scenario 1 and Scenarios 5a-c. Biochar yield may also have a large impact on the mitigation potential of the scenario. Where the high biochar yield was assumed (Scenario 6b) the mitigation potential of the scenario was approximately doubled relative to the potential of Scenario 1. Using the low yield assumptions (Scenario 6a) reduced the mitigation potential of Scenario 1 but with a much smaller impact than the high biochar yield assumptions. This highlights the benefit which could be derived from achieving the highest biochar yields possible, but it should also be noted that the high biochar yield values assumed in Scenario 6b are difficult to achieve and would only be achievable at the expense of oil and gas production from the process. It is highly unlikely that these high biochar yields would be achievable for all feedstocks in all regions throughout the full 95 year scenario period. The variation seen in biochar carbon content (Scenario 7) made relatively little difference to the carbon mitigation potential of Scenario 1 for RCP 2.6.

The same patterns of effect seen in RCP 2.6, from the assumptions of the different scenarios, are seen in RCPs 4.5, 6 and 8.5 (see Annexe IV for summary plots of the mitigation potential of Scenarios 1 to 7 in these RCPs). The maximum mitigation potential achievable under the assumptions of Scenarios 1 to 7, for each RCP, is a reduction of 97.9 GtC, 90.1 GtC, 121.1 GtC and 153.6 GtC for the four RCPs respectively. This is a reduction of 22.5 %, 10.8 %, 10.0 % and 8.3 % from the original carbon emissions pathways of the RCPs. These maximum potentials all result from the assumptions of Scenario 6b which sees maximum biochar yields of 63 % for all crop residues. The minimum mitigation potential of the biochar Scenarios 1 to 7 is a reduction of 20.3 GtC, 18.2 GtC, 23.0 GtC and 28.5 GtC respectively for the four RCPs. These are reductions of 4.7 %, 2.2 %, 1.9 % and 1.5 % from the original carbon emissions pathways of the RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively. The minimum mitigation potential is a result of the assumptions of Scenario 4a2 which assumes a very large decrease in the residue to product factor (RPR) of crops over time, reaching 0.14 in 2100. In reality, neither a uniform biochar yield of 63 % nor a uniform RPR of 0.14 is likely to manifest both spatially and temporally to 2100. These scenarios should, therefore be used as indicators of best and worst case scenarios, with a range of potential outcomes in between.

The potential variation of each parameter was applied to all of the RCPs, with no probability of actual manifestation applied to values within the range. In reality, each an assumed parameter value may be more likely to manifest in one RCP than another. For example crop residue availability in a world following the RCP 2.6 pathway may be low, whereas it may be high in an RCP 8.5 world due to the different underlying drivers within the scenarios. The detail of all of these drivers, such as diet, biofuel crop types, industrial processes and economic development was not sufficiently available within the background literature used in this study therefore all potential variance in parameters has been applied to all RCPs. This enables the effect of these uncertainties on all scenarios to be seen, showing the range of potential outcomes. A particular manifestation of each RCP may also be pulled out from the data, for example from using Figure 8-13 and its counterparts in Annexe IV, and assessed as required.

8.7.3 Climate change impacts on emissions reductions

The CS equation (Equation 7-1) was used to project the impact of climate change on the biochar mitigation potential of the scenarios. The CS equation was used as it could be employed using the experimental data of Chapter 4 which was produced using reliable, tested methodologies and was validated against data from the wider literature.



Figure 8-14: The impact of climate change on the carbon emissions reduction potential of the biochar sequestered in the four RCPs using the assumptions of Scenario 1 and Equation 7-1 for CS potential. The figure indicates the impacts on projected emissions mitigation potential for the scenarios of highest yield with lowest temperature change (8e1) and lowest yield with highest temperature change (8d3), which are deemed to be the minimum and maximum climate impact scenarios respectively.

The potential of biochar to mitigate the RCPs carbon emissions pathways is diminished when the impacts of climate change on crop yield are applied. Mitigation potential is decreased with increasing RCP. The impacts due to climate change also increase in severity over the scenario period, with a larger impact in 2100 as climate change induced temperature change becomes increasingly prominent across the scenario period. The largest impact of climate change, in 2100, is in RCP 8.5 where the mitigation potential may be reduced by 0.72 GtC yr⁻¹ by the assumptions of Scenario 8d3. The mitigation potential of RCP 6, RCP 4.5 and RCP 2.6

respectively may be reduced by 0.14, 0.061 and 0.059 GtC yr⁻¹ in 2100 under the same scenario assumptions.

Across the 95 year scenario period, the maximum impacts of climate change (Scenario 8d3) sees the mitigation potential of biochar, under the assumptions of Scenario 1, reduced from 49.0 GtC, 45.8 GtC, 60.9 GtC and 77.2 GtC respectively, to 46.2 GtC, 42.7 GtC, 56.3 GtC and 54.0 GtC respectively for the four RCPs. This is where the largest impacts on crop yields are combined with the highest temperature change projections for each RCP. Assuming the mean projections of temperature change impact on crop yields (Scenario 8b), sees the mitigation potential of the RCPs reduced to 49.0 GtC, 44.9 GtC, 59.4 GtC and 73.2 GtC respectively. Applying simple adaptation measures (Scenario 9b) sees these projections of mitigation potential increased to 49.0 GtC, 45.8 GtC, 60.8 GtC and 76.1 GtC for the four RCPs respectively over the scenario period.

The 95 year scenario timeframe here may be relatively short compared to timeframes of biochar degradation, and to the timeframes potentially needed for biochar production and addition to soil in order to continually provide some mitigation of carbon emissions. The trends seen in the impacts of climate change on biochar mitigation potential would be expected to continue after 2100 if the pathways remain on much the same course as pre-2100. This means that where increasing climate change reduces crop yields, the potential for biochar production from crop residues will also continue to decrease. These effects will potentially be heightened by other increasing impacts of climate change such as water scarcity, changes in crop pest/disease vectors and increased frequency and/or intensity of extreme weather events (IPCC, 2014c), none of which are currently incorporated into this assessment of biochar CO₂ mitigation potential.

8.8 Uncertainties and limitations

There are a number of areas of uncertainty within the scenarios of carbon sequestration here. Some of this uncertainty arises from the development of the biochar production scenarios, which is detailed in Chapter 5.

A number of generalisations must be made to project global scenarios across a 95 year timescale. The CS and RVC equations make generalised assumptions across each crop type group, where in reality more variation will exist in the carbon sequestration potential of these biochars. The range of values used for analysis using the two pool method are those detailed in Woolf et al. (2010) and are assumed to be uniform across all of the biochars assessed. There are, in reality, many possible combinations of the assumptions of size and half-life of the labile

and recalcitrant fractions. As uncertainty in the size and stability of the labile and recalcitrant fractions of biochars decreases with further research, the application of these values to Equation 7-4 will allow for the assessment of carbon sequestration to incorporate values for individual or groups of biochars. Further development of the knowledge of labile and recalcitrant fractions will also allow the time frame of the 'long-term' carbon storage potential of Equations 7-1 and 7-3 to be estimated by comparing results with the projections for carbon storage over time made using Equation 7-4.

There are a number of other mechanisms relating to a biochar system, which are not considered here, which may see increased or decreased emissions. These include the potential for increased net primary productivity through increased crop yields, reductions in emissions from fertilizer production due to increased fertilizer efficiency use, and the emissions related to the transportation of feedstocks and biochars within the biochar system. Research estimating these effects on the overall emissions balance would be useful further research which could not be undertaken here due to limitations including limited understanding within the wider literature and time constraints. Another consideration when assessing the removal of carbon from the atmosphere and long-term sequestration is the adjustment of the other land and ocean sinks in the carbon cycle, upon this removal, as these sinks reach new equilibria. The carbon cycle consists of a number of complex, inter-related mechanisms. An example of this is the exchange of CO_2 between the atmosphere and ocean which takes place through gas exchange and through respiration and photosynthesis of biota (Siegenthaler and Sarmiento, 1993). Lenton and Vaughan (2009) discussed that the effect of a removal of carbon from the atmosphere would degrade over time due to the reactions of other these other sinks. They used the Bern carbon cycle model to estimate that, for relatively small reductions in carbon, 92 % would still be removed after 1 year, 64 % after 10 years, 34 % after 100 years and 19 % after 1000 years. This indicates that the effect of carbon sequestration through the biochar scenarios detailed here will diminish over time as the other carbon sinks adjust to the atmospheric removal. The values detailed in Chapter 8 do not account for this reaction, but it should be assumed that the effect of atmospheric removal will be diminished over time. This effect could have co-benefits such as decreased ocean acidification as CO_2 is released from the oceans into the atmosphere as a new equilibrium is reached. The assessment of biochar production and carbon sequestration potential detailed here is, also, an assessment of the technical potential of biochars, with no consideration given to the economic, regulatory or social barriers which may prevent these maximum technical potential scenarios from being implemented. They should, therefore, be seen as maximum potential scenarios which may be impacted by these other considerations.

The scenario timeframe begins in 2005, which is now a historical date. This beginning point was used as each RCP begins from this point and so a uniform starting point could be implemented, with diversion from this point observed. This means that the maximum technical potential of each scenario could not, now in reality be achieved. This does not have a major impact on the scenarios over time, as divergence between the RCPs does not begin in earnest until around 2025 (see Figure 8-14).

8.9 Summary

The mean biochar production rates for the four RCPs, over the 95 year period, are 0.01, 0.02, 0.02 and 0.02 kg m² yr⁻¹. The maximum rate of biochar production is 0.6 kg m² yr⁻¹. These values indicate potential rates of biochar addition to soils if all biochar was added to soils at the point of the production. In reality, this is unlikely to happen due to the scale of biochar production technology, where feedstock is likely to be collected and converted in larger batches, then distributed to the point of addition. The mean and maximum values are, therefore, meant to be indicators and not actual distribution values. The maximum rate of 0.6 kg m² yr⁻¹ is also well below the advised upper limit for biochar addition to soils, allowing for addition over time. Once this upper limit for biochar addition is reached in an area the excess biochar can be used in other locations or stored until degradation allows for further addition.

The RVC equation, which was developed here using correlation found between feedstock volatile content and biochar R_{50} index, is a useful addition to the currently available tools for biochar stability estimation as it may offer a simple method of estimation which can be applied where only a small number of characteristic details are available for the biochar feedstocks. It could also offer a simple analysis method to help determine which feedstocks may be the most suitable for the long-term storage of carbon. Comparison of the RVC equation with the CS equation of Zhao et al. (2013) showed that projections made using the RVC equation are very similar to projections using the CS equation. When assessing the projections of long-term biochar carbon storage potential for the different crop groups assessed, the RVC equation made excellent projections for all crop groups except the cereals group, which was still seen to a good projection. Analysis using the uncertainty in slope from the correlation between volatiles and R_{50} showed that the CS projections lie within the uncertainty range of RVC projections.

Biochar has the potential to sequester carbon in all RCPs, and can reduce the carbon emissions of each RCP under all of the scenarios examined. The range of carbon mitigation projections made using Scenarios 1 to 7 is smallest for RCP 2.6 and increases across RCP 4.5 and RCP 6, up to the widest range for RCP 8.5 (See Figure 8-15).



Figure 8-15: Summary of the mitigation potential of the biochar scenarios, in relation to the carbon emission pathways of the RCPs as calculated using the CS equation. The Scenario 1 is shown in bold for each RCP. Scenarios 2 to 7 are shown in lighter tones to highlight the range of impacts projected for each RCP. The maximum impact of climate change on the carbon mitigation potential of Scenario 1 is also shown for each RCP.

Under the assumptions of Scenario 1, and as assessed using the CS equation, biochar systems have the potential to mitigate 49.0 GtC, 45.8 GtC, 60.9 GtC and 77.2 GtC across the 95 year scenario period, for RCP 2.6, RCP 4.5, RCP and RCP 8.5 respectively, and store this carbon for long time periods. The maximum reductions in carbon emissions, from the initial RCP emissions pathway, under biochar Scenarios 1 to 7 are: 97.9 GtC, 90.1 GtC, 121.1 GtC and 153.6 GtC carbon respectively for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 and result from the high biochar yield scenario (Scenario 6b: biochar yields of 63 %). The minimum mitigation potential of scenarios 1 to 7 seen in Scenario 4a2, the scenario of lowest RPR in 2100, was a reduction of 20.3 GtC, 18.2 GtC, 23.0 GtC and 28.5 GtC relative to the original RCP carbon emissions pathway for the four RCPs respectively. Projected impacts of climate change induced changes in global

mean temperature on crop yields has been seen to impact the mitigation potential of biochar scenarios. These impacts increase as the projected change in global mean temperature increases, seeing increasing impact, therefore, as radiative forcing scenario increases and also as time passes within each scenario. In 2100 the impacts of climate change (using the maximum impact climate change scenario of 8d3) may reduce the mitigation potential of biochar, under the assumptions of Scenario 1, by 0.059 GtC yr⁻¹, 0.061 GtC yr⁻¹, 0.14 GtC yr⁻¹ and 0.72 GtC yr⁻¹ for RCPs 2.6, 4.5, 6 and 8.5 respectively. This highlights the increasing impact of temperature change across the RCPs. Across the 95 year period the mitigation potential of Scenario 1 was reduced by the maximum climate change impact projections to 46.2 GtC, 42.7 GtC, 56.3 GtC and 54.0 GtC. Simple adaptation methods were found to reduce the impact of increases in global mean temperature on the mitigation of carbon emissions, resulting in a carbon mitigation potential for Scenario 1 of 49.0 GtC, 45.8 GtC, 60.8 GtC and 76.1 GtC for the four RCPs.

The RVC equation projected the maximum and minimum lifetimes of biochars as being from < 400 years to > 5000 years, highlighting the current uncertainty of biochar lifetime. This uncertainty could be reduced through further research into the size of labile and recalcitrant fraction size and decay rates.

9 Conclusions and recommendations for further work

Seven research objectives were outlined in the introduction to this work which aimed to address the overarching research question of quantifying the CS potential of biochars from crop residues using the land-use scenarios of the four RCPs. Below, each sub-objective is numbered and followed by discussion of how this study has addressed the research question. The discussion also highlights where further research beyond this study would add important detail and reduce uncertainty in areas that were either beyond the scope of this study, or where current uncertainty did not allow a more accurate analysis.

 To produce and characterise biochars from eight crop residues under uniform pyrolysis conditions, examining the effect of feedstock characteristics on subsequent biochar characteristics.

Biochars, bio-oils and syngas from the slow pyrolysis of the eight crop residues were characterised under uniform pyrolysis conditions, enabling analysis of the variance in the yields and the characteristics of these products attributable to the feedstock characteristics. Relationships were determined between feedstock and biochar nutrient content and ash content, with high feedstock quantities tending to remain in the biochars. The majority of biochars were slightly alkaline (pH 6.1 to pH 11.6), with relationships seen between K content and pH. Chan and Xu (2009) reported a pH range of 6.2 to 9.6 from a review of biochar nutrient content literature. Feedstocks with high lignin content produced biochars with high carbon content and with a relatively high recalcitrance. Increasing aromaticity of the biochar structure upon pyrolysis was identified through the decreasing H/C and O/C ratios. This decrease in ratios was also detailed in a review of the literature by Krull et al. (2009), who discuss that although decreasing ratios are seen upon pyrolysis the resulting ratios in biochars vary. CS potential of the biochars, between 21.3 % and 32.5 %, was influenced by the yields of biochar and carbon content, as well as the stability of this carbon content. This is within the range of 21.1 % to 47.1 % found by Zhao et al. (2013). Our results were at the lower end of the Zhao et al. (2013) range due to the higher biochar yields and higher recalcitrance of some of the biochars examined by Zhao et al. (2013). A number of the biochars produced by Zhao et al. (2013) were produced at different temperatures and from a variety of feedstock types, including manures and water weeds, making direct comparison with many of their biochars difficult. The CS potentials determined here add to the documentation of various biochars begun by Zhao et al. (2013). The biochar characterisation has added to the current biochar documenting literature, furthering knowledge of the variation and similarities which may be seen in biochars. Studies such as

Verheijen et al. (2010) and Lehmann and Joseph (2009) conducted literature reviews on various aspects of biochar properties, including characteristics and subsequent effects on crop yields. Both studies concluded that the variation seen in biochars requires the study and documentation of many more biochars, from various feedstocks and production regimes, is necessary to gain a rounded understanding of biochars and their effects in soils. The results presented here aid further understanding of biochars, including how feedstocks and production quantities affect biochar properties, and particularly understanding of the recalcitrance and CS potential of different crop residue biochars. Characterisation including both feedstock and biochar is limited within the current literature (Downie et al., 2009). The detailed characterisation of other feedstock types, for example animal wastes and municipal wastes, and their resulting biochars under a uniform pyrolysis regime would add further insight of feedstock effects on biochar properties. A small number of other characterisations, such as the cation exchange capacity, were beyond the scope of this study due to time constraints and irrelevance to CS potential. Characterisation of these biochar properties would further understanding of properties such as the nutrient properties of biochars in soils.

 To produce and characterise biochars from one crop residue, sugarcane bagasse, under varied pyrolysis conditions (peak temperature and heating rate), examining the effects of process conditions on subsequent biochar characteristics.

The effects of process conditions on biochar characteristics have been further documented by this study, using sugarcane bagasse to examine the effects of different peak temperatures and heating rates on biochar characteristics, building on the current literature (Amonette and Joseph, 2009, Bruun et al., 2011, Chan and Xu, 2009, Demirbas, 2004b, Demirbas, 2006, Downie et al., 2009, Hossain et al., 2011, Kim et al., 2012, Peng et al., 2011, Williams and Besler, 1996, Zhao et al., 2013). A strong correlation indicated a relationship between increased peak pyrolysis temperature and increased biochar pH. Feedstock carbon stored in biochar decreased with increasing peak pyrolysis temperature, and to a lesser extent with increased heating rate, although carbon recalcitrance increased with increasing peak temperature. Biochar CS potential decreased with increasing peak pyrolysis temperature and heating rate, likely caused by the decreasing yields seen. The current literature on biochar characteristics can be difficult to compare, study by study, due to the different process types, conditions and feedstocks used. This study aids this comparison by documenting the use of one process type for biochar production from a number of feedstocks and also for one feedstock under a range of conditions. This will enable biochars to be produced from the most suitable material and process conditions

to produce the characteristics which make it most suitable for its purpose. For example, pyrolysing the feedstock at high peak temperature will increase carbon stability thus making a biochar suitable for long term CS, whereas a low peak temperature biochar may have more suitable characteristics for other uses such as agronomy. Although the work presented here offers insight into how the pyrolysis process can be tailored to produce biochars with specific characteristics, the sample size of the analysis was small. Further benefit would be gained from determining whether the same effects are seen for different feedstocks. As detailed in Section 2.2.2 processing conditions may vary beyond those tested here, for example low temperature pyrolysis/torrefaction and high temperature gasification conditions often lie beyond the conditions tested here. These conditions may produce different biochar yields and characteristics to those observed here. Assessing the effects of a wider range of process conditions would be a beneficial next step.

3. To assess the recalcitrance of the biochars produced using the R_{50} Index described by Harvey et al. (2012).

The eight biochars produced under standard conditions were found to be either moderately or more highly degradable, with palm shell and wheat straw biochars having the highest and lowest recalcitrance values respectively. These trends are similar to those in Harvey et al (2012, although different feedstocks were assessed here, with woody and physically harder feedstocks having higher recalcitrance index values than grasses and straws. None of the biochars here were of the most recalcitrant classification. Supporting the results of Harvey et al (2012) biochar recalcitrance was found to increase with increasing pyrolysis temperature, with the R₅₀ of sugarcane bagasse increasing from 0.47 to 0.56 for biochars produced at 400 °C and 800 °C respectively. This reclassified the biochar from Class C, defined by Harvey et al (2012) as 'most susceptible to degradation', to Class B, 'some susceptibility to degradation', and indicates that higher pyrolysis temperature may be optimum for very long-term carbon storage. Similar to the findings of Harvey et al (2012) and Zhao et al (2013) the R_{50} index was found to be a useful tool for comparing the stability of biochars against other biochars, being a good tool for estimating which biochars may exhibit long-term stability. The CS equation of Zhao et al. (2013) projects carbon storage using the R₅₀ index, but the timeframe of this storage is not defined further than 'long-term'. Further investigation into the timeframe of CS and its relation to the R₅₀ index, for example through enhanced degradation experiments, would enable accurate projections of CS lifetime using these two factors.

4. To examine the potential influence of alkali metal content on biochar degradation, assessing possible conservative R₅₀ estimates in high alkali biochars.

The work on biochar characterisation has further developed the work on biochar recalcitrance by (Harvey et al., 2012). Thermogravimetric analysis of washed and unwashed wheat straw biochars demonstrated a catalytic effect of alkali metals such as potassium on the thermal degradation of biochars. These alkali metals have been seen to undergo rapid leaching in soils (Major et al., 2009). The recalcitrance of wheat straw biochar was increased after the washing out of alkali metals indicating that the R₅₀ index of Harvey et al. (2012) may underestimate the recalcitrance of biochars high in alkali metals. Further research examining this effect in biochars from different feedstocks would add further support to these conclusions.

5. The development of scenarios of biochar production using the land-use projections of the RCPs, examining the effects of various uncertainties and variation within the biochar literature and experimental work on these production potentials.

Following the biochar characterisation documentation, this study made an original assessment of the potential for biochar production and carbon sequestration in the four Representative Concentration Pathways (RCPs). A number of parameters which would influence the biochar production and carbon sequestration potential of the biochar system were tested, including land-use change, crop yields, residue factors, residue availability, biochar yield, carbon content and recalcitrance, and the impacts of climate change.

Biochar production varied over time in each RCP, with the greatest temporal variation seen in the biochar scenarios of RCP 6 and RCP 8.5, mainly due to the assumptions of rates of crop yield increase. Variation was also seen in each RCP across the seven scenario groups developed to explore uncertainty in model parameter inputs. Scenario 1, the scenario with main assumptions taken from experimental and literature assessment, projected biochar production of 138.4 Gt, 132.3 Gt, 173.2 Gt and 217.9 Gt biochar for RCP 2.6, RCP 4.5, RCP 6 and RCP 8.5 respectively across the 95 year period. Exploring the potential range in parameter values through Scenarios 1 to 7 resulted in a range of 218 Gt, 207 Gt, 277 Gt and 251 Gt biochar between the highest and lowest scenarios of biochar production, for RCPs 2.6, 4.5, 6 and 8.5 respectively across the 95 year period. In 2100 the range in annual biochar production projections, for the four RCPs respectively, varied by 0.34, 0.28, 0.55 and 0.80 Mt yr⁻¹ between the highest and lowest biochar production scenarios. Climate change was also projected to impact production of biochar, with the mean climate change scenario (Scenario 8b) seeing a reduction in biochar production

potential of 0 Gt, 2.6 Gt, 4.3 Gt and 11.2 Gt biochar over the 95 year period for RCPs 2.6, 4.5, 6 and 8.5 respectively. This could be mitigated by over 70% by employing simple adaptation measures (Scenario 9b), resulting in no reduction in biochar production potential for RCP 2.6 and 4.5, and reductions of 0.4 Gt and 3 Gt over the 95 year period for RCP 6 and RCP 8.5 respectively. These simple adaptation measures are not defined by Porter et al (2014), although they are likely to include measures such as planting hardier cultivars and irrigation systems. A range of impacts were explored around these mean impact projections, exploring the potential minimum and maximum impacts of climate change induced increases in global mean surface temperature and the mitigation potential of simple adaptation measures. Although a number of parameters were explored within the scenarios developed here, the work is a preliminary investigation which aims to draw together a number of facets of biochar and climate change mitigation research. A number of areas of further research have been highlighted throughout the study. The biochar scenarios could be further developed in a number of ways, using future developments in biochar literature to reduce uncertainty and to expand parameter exploration both spatially and temporally. The use of spatially specific temperature change projections would give an increased understanding of how crop yields may be impacted spatially throughout the scenarios. Incorporation of other climate change impacts, such as changes in the occurrence of extreme events or precipitation patterns, would also offer valuable insight as these impacts are all potential manifestations of climate change and may reduce the CS projections made here (IPCC, 2013).

 Development and evaluation of a new equation for estimating long-term carbon storage, based on the CS equation of Zhao et al. (2013) and incorporating the experimental data from the biochar characterisation.

Biochar characterisation has also added to the work on long-term carbon sequestration by Zhao et al (2013) whose CS equation used the recalcitrance index of Harvey et al. (2012) and other biochar characteristics. The assessment of relationships between feedstock and biochar characteristics in this thesis detailed a relationship between feedstock volatile content and the R₅₀ recalcitrance of biochars. From this the recalcitrance from volatile content (RVC) equation was developed and tested against the CS equation of Zhao et al. (2013). Projection of the CS potential of cereal crops using the RVC equation had the largest variation from the projections made using the CS equation. All projections of sequestration potential for the other crop types were extremely close fitting with the CS projections. The CS projections were found to be always within the uncertainty range of the RVC projections. The accuracy of long-term CS

projections, in relation to the projections made using the CS equation, were good, making the RVC equation a useful alternative methodology if the R_{50} value of the biochar is not determined. The RVC equation enables the CS potential to be estimated through characterisation of the feedstock, and knowledge of biochar yield and carbon content. This gives an alternative method of analysis which adds flexibility to the testing requirements of determining biochar CS. The use of a muffle furnace to determine the feedstock volatile content, rather than using TGA to calculate R_{50} values gives an assessment method which may be more accessible to many stakeholders. This would greatly improve the potential for assessing the recalcitrance of a biochar in many regions.

7. Assessment of the long-term carbon storage potential of the biochars produced within these scenarios, using the CS methodology of Zhao et al. (2013) and the two-pool methodology of Woolf et al. (2010).

Biochar has the potential to sequester carbon for long time periods in all of the RCPs, and across all of the biochar production scenarios to different extents. Three methodologies, the CS method of Zhao et al. (2013), the two-pool method of Woolf et al. (2010), and the RVC method developed in this thesis were used to calculate the long-term CS potential of the biochars produced for each RCP under the assumptions of Scenario 1. Projections from the three methods were compared, concluding, as discussed previously, that the RVC equation can be used in place of the CS equation of Zhao et al. (2013) to estimate the amount of feedstock carbon stored in soils long-term. The two-pool method of Woolf et al. (2010) is a useful tool for estimating carbon storage after a particular time period. The method, as used by (Woolf et al., 2010), currently uses one average value across all biochar types. Further research into the size and decay rates of the labile and recalcitrant fractions of different biochars would increase the accuracy of this methodology. Using Zhao et al. (2013)'s CS equation, due to its incorporation of individual biochar stabilities, compatibility with our experimental data, long-term projections, and validation within the literature, the main scenario assumptions (Scenario 1) projected that 49.0 GtC, 45.8 GtC, 60.9 GtC and 77.2 GtC of carbon would be sequestered over the 95 year scenario period in RCPs 2.6, 4.5, 6 and 8.5 respectively. The range of CS potential across all of the alternative parameter scenarios explored was 77.6 GtC, 71.9 GtC, 98.1 GtC and 125.1 GtC for the four RCPs respectively. The lowest projections were from the scenario of decreasing residue to product factor (Scenario 4a2) at 20.3 GtC, 18.2 GtC, 23.0 GtC and 28.5 GtC for the 95 year period. The maximum CS projections were from the scenario of high biochar yield assumptions, at 97.9 GtC, 90.1 GtC, 121.1 GtC and 153.6 GtC over 95 years for the four RCPs. In reality,

neither a uniform biochar yield of 63 % nor a uniform RPR of 0.14 is likely to manifest both spatially and temporally. These scenarios should, therefore be used as indicators of best and worst case scenarios, encompassing a range of potential outcomes.

Reductions in biochar production caused by increasing global mean temperature will potentially be heightened by other climate change impacts such as water scarcity, changes in crop pest/disease vectors and the increased frequency and intensity of extreme weather events (IPCC, 2013, IPCC, 2014c). The impacts of these other manifestations on the mitigation potential of the biochar scenarios are an important focus for further work. The values for the CS potential of the different scenarios detailed in Chapter 8 also do not account for the reactions of other carbon sinks to the removal of atmospheric CO_2 , but it should be assumed that the effect of atmospheric removal will be diminished over time. Lenton and Vaughan (2009) discuss that the response of other carbon sinks could reduce the impact of CO₂ removal by 66 % after 100 years and by 89 % after 1000 years. The projections made here would benefit from consideration of the reactions of other carbon sinks. The consideration of parameters such as potential effects of biochar addition to soil on crop yields (Verheijen et al., 2010) and any associated feedbacks from this increased net primary production would add to the study, as would the incorporation of wider emissions analysis such as other emissions from soil and potential emissions reductions due to increased efficiency of fertilizers from biochar addition (Woolf et al., 2010). The incorporation of transport infrastructure emissions analysis into the assessment would also be a useful future development as most biochar systems will require transport for the feedstock and/or biochars. The biochar systems assessed here have considered the maximum biochar production and CS potential for each scenario, we have not considered any potential economic, logistical, social or regulatory barriers to implementation of the biochar systems.

This thesis aimed to assess the global potential for carbon sequestration using biochar from crop residues, under four land-use scenarios, to 2100. Useful insights have been determined for a number of areas within the biochar research field including biochar characterisation, assessing biochar production potential and long-term carbon storage, and the development of a new method of carbon storage. The research is also a useful tool to direct further research as, due to its broad nature, it has identified a number areas which would benefit from further investigation.

I. Annexe 1

Macro-nutrient species (mg kg ⁻¹)							
Feedstocks							
	Sugarcane bagasse	Rice husk	Coconut shell	Wheat straw	Cotton stalk	Olive pomace	Coconut fibre
Р	447.1	835.1	0.0	165.5	1248.0	1287.6	530.0
К	4289.1	10140.0	415.0	17446.4	10900.8	20868.2	4110.2
Са	390.2	1995.1	370.9	1035.7	6694.6	8069.7	7852.8
Mg	258.2	879.9	133.4	438.0	3037.4	1860.8	621.9
Biochars							
Р	1728.4	263.8	0.0	957.4	4775.7	4155.7	2961.3
К	16055.9	4173.3	2223.9	60372.3	27570.9	51678.2	18210.1
Са	1730.6	751.0	779.3	5189.5	9546.8	26998.3	8368.8
Mg	1248.5	302.3	156.1	2514.9	4728.9	5333.1	4288.5

Table I-1: Macronutrient concentrations of the feedstocks and biochars (mg kg⁻¹)

Table I-2: Gas composition and calorific value (HHV) of syngas produced from the pyrolysis of the eight agricultural residue feedstocks under standard conditions. Species determined are: carbon dioxide (CO₂); permanent gases: hydrogen (H₂) and carbon monoxide (CO); and hydrocarbon gases: methane (CH₄), ethene (C₂H₄), ethane (C₂H₆), propene (C₃H₆), propane (C₃H₈), butene (C₄H₈), butane (C₄H₁₀)).

			Gas Characteristics									
	CV		Gas Composition (%)									
Feedstock	MJm⁻³	со	H ₂	CO ₂	CH_4	C_2H_4	$C_2 H_6$	C_3H_6	C_3H_8	C_4H_8	C_4H_{10}	
coconut												
shell	10.2	29.1	16.1	43.3	10.3	0.2	0.7	0.1	0.2	0.1	0.1	
cotton stalk	11.6	27.2	6.6	54.8	8.0	0.7	1.4	0.5	0.5	0.2	0.1	
palm shell	12.7	22.8	16.2	42.4	16.1	0.3	1.4	0.2	0.3	0.1	0.1	
coconut												
husk	12.3	18.1	11.3	51.5	15.5	0.5	1.8	0.4	0.4	0.3	0.3	
olive waste	14.2	13.4	15.0	50.9	13.8	0.7	3.4	0.6	1.1	0.6	0.5	
rice husk	15.0	23.8	8.6	48.7	14.2	0.6	1.5	0.5	0.3	1.5	0.3	
wheat straw	10.5	30.8	13.1	45.0	8.4	0.5	1.4	0.3	0.3	0.3	0.0	
s bagasse	12.3	22.4	11.0	48.4	14.9	0.5	1.6	0.4	0.3	0.2	0.3	

Table I-3: Gas composition and calorific value of syngas produced from the pyrolysis of sugarcane bagasse at different pyrolysis temperatures (top) and heating rates (bottom). Species determined are: carbon dioxide (CO_2); permanent gases: hydrogen (H_2) and carbon monoxide

(CO); and hydrocarbon gases: methane (CH₄), ethene (C_2H_4), ethane (C_2H_6), propene (C_3H_6), propane (C_3H_8), butene (C_4H_8), butane (C_4H_{10})).

		Gas Characteristics									
	CV	Gas Composition (%)									
	MJm⁻³	CO	H ₂	CO ₂	CH_4	C_2H_4	$C_2 H_6$	C_3H_6	C_3H_8	C_4H_8	C_4H_{10}
Altered Final Temperature											
400 °C	6.6	18.0	1.2	73.5	4.4	0.5	0.9	0.3	0.2	0.3	0.7
600 °C	12.3	22.4	11.0	48.4	14.9	0.5	1.6	0.4	0.3	0.2	0.3
800 °C	14.5	6.9	24.5	44.6	20.1	0.5	1.7	0.4	0.4	0.5	0.4
Altered Heat	ing Rate										
5 °C min⁻¹	12.3	22.4	11.0	48.4	14.9	0.5	1.6	0.4	0.3	0.2	0.3
20 °C min ⁻¹	11.2	17.7	8.0	56.6	14.3	0.6	1.6	0.4	0.3	0.3	0.1
50 °C min⁻¹	11.5	14.7	7.2	58.0	16.8	0.7	1.6	0.4	0.3	0.1	0.2

II. Annexe 2

a. Regional classifications

Table II-1: The 5 regional and 22 sub-regional groups used for the analysis, and the constituent countries of each group.

Region	Sub-Region	Country
Africa	Eastern Africa	Burundi, Comoros, Djibouti, Eritrea, Ethiopia, Ethiopia PDR, Kenya, Madagascar, Malawi, Mauritius, Mayotte, Mozambique,
		Reunion, Rwanda, Seychelles, Somalia, South Sudan, Uganda,
		United Republic of Tanzania, Zambia, Zimbabwe
	Middle Africa	Angola, Cameroon, Central African Republic, Chad, Congo,
		Democratic Republic of the Congo, Equatorial Guinea, Gabon,
		Sao Tome and Principe
	Northern Africa	Algeria, Egypt, Libya, Morocco, Sudan, Sudan (former), Tunisia,
		Western Sahara
	Southern Africa	Botswana, Lesotho, Namibia, South Africa, Swaziland
	Western Africa	Benin, Burkina Faso, Cabo Verde, Cote d' Ivoire, Gambia, Ghana,
		Guinea, Guinea-Bissau, Liberia, Maii, Mauritania, Niger, Nigeria,
		Leone Togo
Americas	Northern	Bermuda, Canada, Greenland, Saint Pierre and Miguelon.
	America	United States of America
	Central America	Belize, Costa Rica, El Salvador, Guatemala, Honduras, Mexico,
		Nicaragua, Panama
	Caribbean	Anguilla, Antigua and Barbuda, Aruba, Bahamas, Barbados,
		Bonaire, Sint Eustatius and Saba, British Virgin Islands, Cayman
		Islands, Cuba, Curacao, Dominica, Dominican Republic, Grenada,
		Guadeloupe, Haiti, Jamaica, Martinique, Montserrat,
		Netherlands Antilles, Puerto Rico, Saint Kitts and Nevis, Saint
		Lucia, Saint Vincent and the Grenadines, Sint Maarten (Dutch
		States Virgin Islands
	South America	Argentina Bolivia Brazil Chile Colombia Ecuador Falkland
	South America	Islands, French Guiana, Guyana, Paraguay, Peru, Suriname,
		Uruguay, Venezuela
Asia	Central Asia	Kazakhstan, Kyrgyzstan, Tajikistan, Turkmenistan, Uzbekistan
	Eastern Asia	China, Democratic People's Republic of Korea, Japan, Mongolia,
		Republic of Korea
	Southern Asia	Afghanistan, Bangladesh, Bhutan, India, Iran (Islamic Republic
		of), Maldives, Nepal, Pakistan, Sri Lanka
	South-Eastern	Brunei Darussalam, Cambodia, Indonesia, Lao People's
	Asia	Democratic Republic, Malaysia, Myanmar, Philippines,
		Singapore, Thailand, Timor-Leste, Viet Nam
	Western Asia	Armenia, Azerbaijan, Bahrain, Cyprus, Gaza Strip (Palestine),
		Georgia, Iraq, Israei, Jordan, Kuwait, Lebanon, Occupied
		Republic Turkey United Arab Emirates West Bank Venan
Furone	Fastern Furone	Relarus Bulgaria Czech Republic Czechoslovakia Hungary
Luiope	Lustern Lurope	beiding, buigana, ezech nepublic, ezechosiovania, huligary,

		Poland, Republic of Moldova, Romania, Russian Federation,
		Slovakia, Ukraine, USSR
	Northern Europe	Channel Islands, Denmark, Estonia, Faroe Islands, Finland,
		Guernsey, Iceland, Ireland, Isle of Man, Latvia, Lithuania,
		Norway, Sweden, United Kingdom
	Southern Europe	Albania, Andorra, Bosnia and Herzegovina, Croatia, Gibraltar,
		Greece, Holy See, Italy, Malta, Montenegro, Portugal, San
		Marino, Slovenia, Spain, The former Yugoslav Republic of
		Macedonia, Yugoslav SFR
	Western Europe	Austria, Belguim, Belgium-Luxembourg, France, Germany,
		Liechtenstein, Luxembourg, Monaco, Netherlands, Switzerland
Oceania	Australia & N Zld	Australia, New Zealand, Norfolk Island
	Melanesia	Fiji, New Caledonia, Papua New Guinea, Solomon Islands,
		Vanuatu
	Micronesia	Guam, Kiribati, Marshall Islands, Micronesia (Federated States
		of), Nauru, Northern Mariana Islands, Pacific Islands Trust
		Territory, Palau
	Polynesia	American Samoa, Cook Islands, French Polynesia, Niue, Pitcain
		Islands, Samoa, Tokelau, Tonga, Tuvalu, Wallis and Futuna
		Islands

Table II-2: Fraction of total cropland which is cultivated by each of the five crop types for each of the 22 sub-regions (where total cropland = 1).

		Fraction of total cropland							
		Cereal	Fibre	Oil					
Region	Sub-region	crops	crops	crops	Sugarcane	Total			
Africa	Eastern Africa	0.441	0.030	0.108	0.007	0.586			
	Middle Africa	0.277	0.021	0.116	0.009	0.423			
	Northern Africa	0.471	0.010	0.138	0.005	0.624			
	Southern Africa	0.301	0.002	0.055	0.025	0.383			
	Western Africa	0.450	0.028	0.154	0.001	0.633			
Americas	Northern America	0.327	0.025	0.203	0.002	0.557			
	Central America	0.342	0.005	0.026	0.034	0.407			
	Caribbean	0.126	0.004	0.032	0.097	0.258			
	South America	0.270	0.017	0.344	0.053	0.684			
Asia	Central Asia	0.555	0.081	0.107	0.000	0.744			
	Eastern Asia	0.608	0.041	0.211	0.010	0.871			
	Southern Asia	0.602	0.059	0.188	0.021	0.869			
	South-Eastern Asia	0.533	0.007	0.197	0.022	0.758			
	Western Asia	0.488	0.021	0.070	0.000	0.579			
Europe	Eastern Europe	0.396	0.001	0.077	0.000	0.473			
	Northern Europe	0.474	0.001	0.062	0.000	0.537			
	Southern Europe	0.377	0.011	0.164	0.000	0.553			
	Western Europe	0.492	0.003	0.102	0.000	0.597			
Oceania	Australia & New Zealand	0.394	0.006	0.032	0.009	0.441			

Polynesia	0.000	0.000	0.605	0.001	0.606
Micronesia	0.002	0.000	0.764	0.000	0.766
Melanesia	0.010	0.000	0.370	0.049	0.429

Table II-3: Developing countries and their regions, as classified by the World Bank (2013).

Region	Country
East Asia and Pacific	American Samoa, Cambodia, China, Fiji,
	Indonesia, Kiribati, Korea, Dem. Rep., LAO
	(People's Democratic Republic), Malaysia,
	Marshall Islands, Micronesia, Fed. Sts.,
	Mongolia
	Myanmar, Palau, Papua New Guinea,
	Philippines
	Samoa, Solomon Islands, Thailand, Timor-
	Leste, Tuvalu, Tonga, Vanuatu, Vietnam
Europe and Central Asia	Albania, Armenia, Azerbaijan, Belarus, Bosnia
	and Herzegovina, Bulgaria, Georgia,
	Kazakhstan, Kosovo, Kyrgyz Republic, Latvia,
	Lithuania, Macedonia, FYR, Moldova,
	Montenegro, Romania, Russian Federation,
	Serbia, Tajikistan, Turkey, Turkmenistan,
	Ukraine, Uzbekistan
Latin America and the Caribbean	Antigua and Barbuda, Argentina, Belize,
	Bolivia, Brazil, Chile, Colombia, Costa Rica,
	Cuba, Dominica, Dominican Republic, Ecuador,
	El Salvador, Grenada, Guatemala, Guyana,
	Haiti, Honduras, Jamaica, Mexico, Nicaragua,
	Panama, Paraguay, Peru, St. Lucia, St. Vincent
	and the Grenadines, Suriname, Uruguay,
	Venezuela, RB
Middle East and North Africa	Algeria, Djibouti, Egypt, Arab Rep., Iran, Islamic
	Rep., Iraq, Jordan, Lebanon, Libya, Morocco,
	Syrian Arab Republic, Tunisia, West Bank and
	Gaza, Yemen, Rep.
South Asia	Afghanistan, Bangladesh, Bhutan, India,
	Maldives, Nepal, Pakistan, Sri Lanka
Sub-Saharan Africa	Angola, Benin, Botswana, Burkina Faso,
	Burundi, Cameroon, Cape Verde, Central
	African Republic, Chad, Comoros, Congo, Dem.
	Rep., Congo, Rep., Cote d'Ivoire, Eritrea,
	Ethiopia, Gabon, Gambit, The, Ghana, Guinea,
	Guinea-Bissau, Kenya, Lesotho, Liberia,
	Madagascar, Malawi, Mali, Mauritania,
	Mauritius, Mozambique, Namibia, Niger,
	Nigeria, Rwanda, Sao Tome and Principe,
	Senegal, Seychelles, Sierra Leone, Somalia,
	South Africa, South Sudan, Swaziland,
	Tanzania, Togo, Uganda, Zambia, Zimbabwe

b. Crop yields within the RCPs

RCP 2.6

RCP 2.6 represents the lowest radiative forcing target of the RCPs, peaking at 3 W m² midcentury and then declining to 2.6 W/m² in 2100 (van Vuuren et al., 2011b). The aim of the scenario is to model a plausible pathway of a future which limits global average climate change to below 2 °C. The main assumptions and scenario drivers are detailed in van Vuuren et al. (2010) and van Vuuren et al. (2011b). Further discussion of the RCP scenario can be found in Section 2.4.1.1.

The baseline scenario used by the IMAGE modelling team to develop the RCP assumes an average rate of yield improvement for cereals of + 0.75 % yr^{-1} over the period to 2100. Van Vuuren et al. (2011b) discuss that agricultural land area in RCP 2.6 is increased in relation to the baseline scenario due to drivers such as an increased use of bioenergy and the reduction seen in the CO₂ fertilization effect. This reduction in CO₂ fertilization will potentially reduce crop yields within the RCP scenario. New values for the rates of crop yield change, relative to those in the baseline scenario, are not discussed in the RCP literature therefore estimations for the CO₂ fertilization effect from the literature have been used here to estimate the reduction in yields which may be seen in the RCP 2.6 pathway. Lobell and Field (2008) discuss that the average effect of CO₂ fertilization used regularly in crop models, for C₃ crops, is 0.1% yield increase for each 1 ppm CO₂ increase. They discuss that with the historical rate of CO₂ increase seen since 1960 (1.35 ppm yr⁻¹) a yield increase of 0.14 % yr⁻¹ would be expected. Analysis of the CO_2 fertilization effect on rice, wheat and maize crops by (Lobell and Field, 2008) was consistent with the current value of 0.1 % yield increase per 1 ppm CO_2 . The average value of 0.1 % yield was also determined by Kimball (1983), in a meta-analysis of 430 observations, who discussed that an average yield increase of 33% could be expected with a 330 ppm increase in CO₂ concentration. This equates to 0.1 % increase per 1 ppm CO_2 . This value of 0.1 % yield increase per 1 ppm CO_2 increase has therefore been assumed here as the CO_2 fertilization effect. To calculate the projected effect of this reduction in CO_2 fertilization effect in RCP 2.6, from the 0.75 % yr⁻¹ yield increase of the baseline scenario, CO_2 concentrations from the baseline scenario were used. Van Vuuren et al. (2007) projected atmospheric CO₂ concentrations of 708 ppm, in 2100, for the baseline scenario. Data for RCP 2.6 from the RCP database showed the CO_2 concentration for 2005 to be 378.81 ppm (IIASA, 2009). This value was used as the 2005 concentration for both the baseline and RCP 2.6 scenario. The average increase in CO₂ concentration for the baseline scenario was determined as 3.47 ppm yr⁻¹ using the 2005 and 2100 values. Using the 0.1 % plant yield increase discussed above, this equates to 0.35 % of the yield increase seen every year being attributable to the CO₂ fertilization effect. Subtracting this from the 0.75 % yr⁻¹ yield increase used in the baseline scenario, 0.40 % yr⁻¹ yield increase would have been expected without any CO₂ fertilization effect. Following this, the change in CO₂ concentration in RCP 2.6 was calculated for each year relative to the previous year, giving a change in atmospheric CO₂ concentration. This was multiplied by the 0.1 % yield increase factor per ppm CO₂ increase, resulting in a CO₂ fertilization factor for each year of the RCP. Added to the 0.4 % yield increase that would be expected in RCP 2.6 without CO₂ fertilization an average value of 0.45 % yr⁻¹ increase in crop yields was determined for RCP 2.6. The average crop yield increase for the period 2005-2050 was 0.55 % yr⁻¹ and the period 2051-2100 was 0.36 % yr⁻¹. These two periods were differentiated due to the peak in CO₂ concentration mid-century and then subsequent decline. It would be expected from this that there would be more of an enhanced CO₂ fertilization effect in the first half of the century, which would be reduced as atmospheric CO₂ declines. The two rates of crop yield increase (2005-2050 and 2051-2100) were used for the main RCP 2.6 biochar scenario.

RCP 4.5

RCP 4.5 uses global carbon emissions pricing to achieve a radiative forcing stabilization target of 4.5 W/m^2 in 2100. More discussion about the development of the scenario and its underlying assumptions can be found in Section 2.4.1.2. The supplementary literature around RCP 4.5 details the assumptions made by the modelling team about crop yields (Thomson et al., 2010). They discuss that for RCP 4.5, literature from the FAO, 'World Agriculture Towards 2015/2030: An FAO Perspective', was used by the RCP modelling team to determine crop yields to 2030 (FAO, 2003). This FAO report discusses an expected 67 % of the growth in annual crop production $(1.6 \% \text{ yr}^{-1})$ to come from increases in crop yields for developing countries to 2030. This results in ~ 1.1 % yr⁻¹ crop yield increase for the period. For the developed countries, a projected increase in crop production of 0.9 % yr⁻¹ to 2030 is determined, with the discussion that all of this increase will be achieved through increased yield and more intensive land use, rather than increased agricultural land area. These yield increases of 1.1 % yr⁻¹ and 0.9 % yr⁻¹, for developing and developed countries respectively, were used for the period 2005 to 2030 in the biochar scenarios. Developed and developing nation status is detailed in Annexe II a. The RCP 4.5 modelling team assumed, following 2030, a convergence to 0.25% yr⁻¹ within the second half of the century, which has also been assumed within the biochar scenario (Wise et al., 2009). The date of this convergence was not detailed within the literature. Convergence to 0.25 % yr⁻¹ yield increase in 2100 was implemented by an incremental change in the rate of yield change every 10 years, resulting in a yield increase of 0.25 % yr⁻¹ in 2100 for the main model scenarios. Assuming

this 10 year incremental change in rate of crop yield change enabled the complexity of yield assumptions to remain at similar levels to the other biochar scenarios, and remain within the bounds and time constraints of the study. These yield projections have been used to determine overall productivity data for RCP 4.5. Analysis of a scenario where the rate of crop yield increase converges to 0.25 % yr⁻¹ in 2050 was also conducted to determine the effects of applying the earliest and latest convergence dates (see section 5.4.1.3). Thomson et al., (2010) also examine scenarios of productivity extremes, applying yield changes of 0 % (from baseline yields) and 50 % greater increase than the standard scenario assumptions detailed above. These scenarios of no yield increase and 50 % extra yield increase were investigated for the RCP scenarios (see Section 5.4.1.).

RCP 6

RCP 6 uses climate policy to reduce emissions, limiting radiative forcing to 6 W/m^2 in 2100. The crop yield assumptions of RCP 6 are not directly indicated within the literature, and therefore have been derived here from the background scenario literature. RCP 6 was developed from the IPCC's Special Report on Emission Scenarios (SRES) B2 scenario. Arnell et al. (2004) discuss that to develop crop yields for the SRES B2 scenario, baseline rates of yield increase of 1 % yr⁻¹ for developed nations and 1.7 % yr⁻¹ for developing nations (global average of 1.2 % yr⁻¹) were taken from Parry et al. (1999) and then adapted to the assumptions of the B2 scenario. They determined a figure of 1 % yr⁻¹ yield increase for both developed and developing nations. This reduction in the projected rate of yield increase for developing nations is largely due to the assumption that much of the achievable yield increase through intensification and mechanisation has already been achieved and that diminishing rates of return are likely to be seen for further increases in input. They assume that increase in input is likely in some developing nations, but an eventual levelling off is highly likely. The assumptions made about yield increases within the previously detailed literature regarding RCP 6 are projected to 2080. Due to a lack of literature detailing other assumptions, these crop yield increases are assumed to remain constant for the period 2080 to 2100 for the RCP 6 biochar in Scenario 1.

RCP 8.5

RCP8.5 represents a scenario of radiative forcing based on the upper end of the range of radiative forcing pathways within the literature, reaching a radiative forcing of 8.5 W/m² in 2100. This RCP is also often used by the modelling communities as a baseline climate scenario as it does not include any specific climate mitigation actions (Riahi et al., 2011).

There is no direct discussion of crop yield projections within the RCP 8.5 literature therefore it was necessary to use the background literature to derive crop yields. RCP 8.5 was developed

from a revised version of the IPCC's SRES A2 scenario, named the A2r scenario (Nakicenovic et al., 2000, Riahi et al., 2007). Further details of the underlying assumptions of RCP 8.5 can be found in Chapter 2.4.1, Section 2.4.1.4. Arnell et al. (2004), who discussed crop yield changes for the SRES B2 scenario used for the development of crop yields for RCP 6 also discussed crop yield projections for the SRES A2 scenario. They detailed that increases of 1.0 % yr⁻¹ and 1.5 % yr⁻¹ for developed and developing countries respectively are projected within the A2 scenario. These values have therefore been assumed as annual crop yield changes for the main RCP 8.5 biochar scenario to 2100.

c. Residue availability

Cereal residues

Li et al. (1999) discuss that, in 1995, 2.3 % of residues in China were used for industrial processes, 24 % of this used as forage and between 29 % and 59 % used as fuel. Liu et al. (2008) states similar values, with 4 % of residues used within industrial processes, 23 % as forage and 0.5 % for biogas production. They also discuss that of the remaining residues, 37 % were burnt for fuel by farmers, 15 % were lost during the collection process and returned to the field, and 20.5 % were burnt in the field. This would give 20.5 % residue availability for biochar production if those cereal residues burnt in the field were utilized. A further 37 % could be potentially included (totalling 57.5 % of cereal residues) if farmers used pyrolysis systems, or other dual biochar-fuel producing systems, for energy production. For the main biochar scenarios developed here an assumption of 30 % cereal residue removal (from the field) rate has been made, in accordance with the work of Lal (2005) and Lindstrom (1986) (see Section 5.3.6.1). The 25 % and 10 % of crop residues currently burnt in the field, in developing and developed nations respectively, are also assumed to be available and collected here for biochar production. This makes the amount of cereal residues collected in the field 55 % and 40 % of total produced residues for developing and developed nations respectively. The values used here for developing nations (55 % removal rate) correspond well with the work of Liu et al. (2008) where 57.5 % of residues would be available using those residues currently burnt in the field and currently used for localised energy purposes. The values used here for cereal residue removal rates are optimistic compared to those used by Woolf et al. (2010), who use 25 %, 35 % and 45 % removal respectively in their scenarios, though the Woolf et al. (2010) values do not consider the use of those residues currently burnt in the field, or those currently used for localised energy purposes which could be converted to duel biochar-energy systems. The impacts on Scenario 1 of the residue availability assumptions of Woolf et al. (2010) are investigated in Scenario 5 (see Section 5.4.4).

Sugarcane residues

Smil (1999) discusses that sugarcane bagasse is often used as animal feed or industrial fuel sources within their processing streams. Woolf et al. (2010) assumed that all bagasse residues can be utilised for biochar production, as those currently used for power generation in the sugar production industry could be utilised in a dual energy-biochar production system. This would require conversion of energy systems, which would add further economic cost to a biochar system. They also discuss that sugarcane leaves are currently often left in the field rather than burnt, with the assumption that 50 % of this trash is currently recoverable, projected to rise to 75 % recoverability with improved technology and conservation practices. In the main biochar scenarios here (Scenario 1) 100 % bagasse utilization and 50 % field trash utilization are assumed.

Rice straw

100 % residue removal from the field is safely achievable for rice residues (Kim and Dale, 2004). Devendra (1997) discusses the utilization of rice straw in Asia, with an average value of 30 % determined for 9 Southeast Asian countries inclusive of China and the Philippines. Woolf et al., (2010b) determined that 26 % of rice straw produced is used for animal fodder. The average value of 30 % for Southeast Asia has been adopted here for the main scenarios due to the large share of total global rice production which is achieved by these nine Asian nations (FAO, 1998)). Following the work of Devendra (1997) an average value for the utilization of rice straw for animal feed of 30 % was assumed here for developing nations. This was applied to rice straw only, and as the majority of rice production occurs in Asia (over 90 % of the world's rice straw currently used for energy production could be utilised in dual biochar-energy producing pyrolysis systems, this leaves 70 % of rice straw residues available for biochar production.

Oil crop residues

The oil crop category includes crops such as cottonseed and rape, and also crops such as groundnuts, coconuts and olives. The diversity of the oil crop category makes applying an availability factor to the whole group difficult. As with cereal crops, a 30 % residue removal rate from fields is assumed here for Scenario 1, leaving 70 % of oil crop residues in-situ for soil quality purposes. Many of the residues considered here would become available after processing of the commodity, such as coconut shells and groundnut shells. Such residues may often be burnt for energy within the processing system. Residues from oil crops can be used for biofuel production (European Biofuels Technology Platform, 2014), though it is assumed within Scenario 1 that all

residues available, which would normally be used for energy production, are utilized in pyrolysis systems for dual biochar-energy production.

Fibre crop residues

The main fibre crops assessed here are cotton and jute. Cotton residues are mostly cotton stalk, of which on average 40 % are available as surplus. A conservative estimate of 30 % is assumed here for collection for pyrolysis leaving 10 % of the surplus for other uses or losses during collection. Saha and Sagorika (2013) detail that alongside jute production 5.43 Mt of dry leaf matter is produced each year. The only current use for this leaf matter is to leave it in-situ for soil conditioning purposes. It was assumed here for Scenario 1, as with cereals residues, that 30 % of the leaf litter could be collected for pyrolysis without detrimental effects on soil quality.

d. Crop yields

Table II-4: Summary of the annual crop yield change (% yr⁻¹) for the four RCPs under the crop yield changes of Scenario 2b. Developed and developing refer to the developed and developing nations respectively.

			Annual crop yield change (% yr ⁻¹)								
RCP		2010	2020	2030	2040	2050	2060	2070	2080	2090	2100
2.6	developed	0.825	0.825	0.825	0.825	0.825	0.54	0.54	0.54	0.54	0.54
2.6	developing	0.825	0.825	0.825	0.825	0.825	0.54	0.54	0.54	0.54	0.54
4.5	developed	1.35	1.35	1.35	1.215	1.065	0.93	0.795	0.66	0.51	0.375
4.5	developing	1.65	1.65	1.65	1.47	1.29	1.11	0.915	0.735	0.55	0.375
6	developed	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5
6	developing	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5
8.5	developed	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5	1.5
8.5	developing	2.25	2.25	2.25	2.25	2.25	2.25	2.25	2.25	2.25	2.25

e. Biochar carbon content

Table II-5: Elemental carbon content, from the literature, for biochars produced from various biomass materials. Process type and temperature for each biochar are shown, alongside the reporting basis of the values and reference.

		Carbon		
Process		Content		
Temp ([°] C)	Feedstock	(%)	Reference	
550	sugarcane bagasse	63.3 ^{° c}	(Cross and Sohi, 2011)	
550	sugarcane bagasse	62.43 ^{° c}		
550	sugarcane bagasse	58.31 ^{° c}		
550	sugarcane bagasse	59.34 ^{° c}		
550	sugarcane bagasse	63.37 ^{° c}		

550	sugarcane bagasse	45.31 ^{° c}	
550	papermill wastes	50 ^{° c}	(Van Zweiten et al., 2010)
550	papermill wastes	52 ^{° c}	
500	switchgrass	39.4 ^{° c}	(Brewer et al., 2009)
500	corn stover	62.8 ^{° c}	
500	hardwood	65.3 ^{° c}	
500	switchgrass	38.7 ^{ª d}	
500	corn stover	37.8 ^{ª d}	
500	hardwood	63 ^{ª d}	
760	switchgrass	42.8 ^{ae}	
550	tea waste	63 ^{bc}	(Demirbas, 2004b)
550	corncob	64 ^{bc}	
550	olive husk	65 ^{bc}	
500	Wood (pinus ponderosa)	81.9 ^{bc}	(Keiluweit et al., 2010)
600	wood (pinus ponderosa)	89 ^{b c}	
500	grass (festuca arundinacea)	82.2 ^{bc}	
600	grass (festuca arundinacea)	89 ^{b c}	
550	wood (pinus ponderosa)	79.2 ^{bc}	(Singh et al., 2012)
550	papermill sludge	83.6 ^{bc}	
550	leaf	71.9 ^{bc}	
500	wood	84 ^{b c}	(Vaccari et al., 2011)
500	wheat straw	60.2 ^{bd}	(Bruun et al., 2011)
525	wheat straw	64.3 ^{bd}	
550	wheat straw	67 ^{b d}	
575	wheat straw	69.2 ^b	
500	hazelnut	62 ^{bc}	(Enders et al., 2012)
500	pine	62 ^{bc}	
500	oak	68 ^{bc}	
^a = dry basis, ^l	$^{\circ}$ = dry, ash free basis, $^{\circ}$ = slow p	yrolysis, ^d = f	ast pyrolysis, ^e = gasification

f. Climate change impacts

Table II-6: Projected changes in crop yield (%) from the baseline crop yields of Scenario 1, with temperature increase (°C). Values shown are the mean, lowest and highest projected changes (%) in crop yield, both with and without adaptation, for each temperature increase (low and high values determined using the 95 % confidence interval data from Porter et al. (2014)).

Temp increase	Change in crop yield from baseline (%)						
	Mean			rojection	High yield p	High yield projection	
(°C)	No Adapt	Adapt	No Adapt	Adapt	No Adapt	Adapt	
1	-2	1	-8	-4	3.2	7.6	
1.1	-2.3	0.8	-8.1	-4	2.8	7.2	
1.2	-2.6	0.6	-8.3	-4	2.5	6.8	
1.3	-2.9	0.4	-8.4	-3.9	2.1	6.5	
1.4	-3.2	0.2	-8.5	-3.9	1.8	6.1	
1.5	-3.5	0	-8.7	-3.9	1.4	5.7	
1.6	-3.8	-0.2	-8.8	-3.9	1.1	5.3	
1.7	-4.1	-0.4	-8.9	-3.9	0.7	4.9	
1.8	-4.4	-0.6	-9.1	-3.8	0.4	4.6	
1.9	-4.7	-0.8	-9.2	-3.8	0	4.2	

2	-5	-1	-9.3	-4.1	-0.3	3.9
2.1	-5.3	-1.2	-9.7	-4.3	-0.6	3.7
2.2	-5.6	-1.4	-10	-4.6	-0.8	3.5
2.3	-5.9	-1.6	-10.3	-4.9	-1	3.2
2.4	-6.2	-1.8	-10.6	-5.1	-1.3	3
2.5	-6.5	-2	-10.9	-5.4	-1.5	2.7
2.6	-6.8	-2.2	-11.2	-5.6	-1.7	2.5
2.7	-7.1	-2.4	-11.6	-5.9	-2	2.3
2.8	-7.4	-2.6	-11.9	-6.2	-2.2	2
2.9	-7.7	-2.8	-12.2	-6.4	-2.4	1.8
3	-8	-3	-12.5	-6.6	-2.7	1.9
3.1	-8.5	-3	-13.1	-6.7	-3	2.1
,3.2	-9	-3	-13.8	-6.9	-3.4	2.2
3.3	-9.5	-3	-14.4	-7	-3.8	2.4
3.4	-10	-3	-15.0	-7.2	-4.1	2.5
3.5	-10.5	-3	-15.7	-7.3	-4.5	2.6
3.6	-11	-3	-16.3	-7.5	-4.9	2.8
3.7	-11.5	-3	-16.9	-7.6	-5.2	2.9
3.8	-12	-3	-17.6	-7.8	-5.6	3.1
3.9	-12.5	-3	-18.2	-8	-6	3.2
4	-13	-3	-18.8	-8.6	-6.3	3.3
4.1	-13.5	-3.2	-19.5	-9.2	-6.6	3.4
4.2	-14	-3.4	-20.2	-9.8	-6.9	3.6
4.3	-14.5	-3.6	-20.9	-10.4	-7.2	3.7
4.4	-15	-3.8	-21.6	-11	-7.5	3.8
4.5	-15.5	-4	-22.3	-11.7	-7.8	3.9
4.6	-16	-4.2	-23	-12.3	-8.1	4
4.7	-16.5	-4.4	-23.7	-12.9	-8.4	4.1
4.8	-17	-4.6	-24.4	-13.5	-8.7	4.3
4.9	-17.5	-4.8	-25.1	-14.1	-9	4.4
5	-18	-5	-25.8	-14.7	-9.3	4.5

III. Annexe 3: Biochar production: Results tables

Table III-1: Total cropland (in million hectares (Mha)) for the years 2005, 2025, 2050, 2075 and 2100 for the four RCPs.

	Total cropland	Total cropland (Mha yr ⁻¹)					
Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
2005	1554.98	1554.98	1554.98	1554.98			
2025	1765.54	1362.93	1580.05	1634.52			
2050	1909.01	1284.09	1651.59	1718.95			
2075	2036.09	1208.75	1776.50	1773.16			
2100	2097.55	1122.99	1930.81	1839.43			

	Total commodity production (Gt yr ⁻¹)				
Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5	
2005	3.06	3.06	3.06	3.06	
2025	3.66	3.60	3.81	4.14	
2050	4.45	4.46	5.11	6.00	
2075	5.10	4.86	7.01	8.61	
2100	5.66	4.85	9.49	12.46	

Table III-2: Total commodity production (in gigatonnes (Gt)) for the years 2005, 2025, 2050, 2075 and 2100 for the four RCPs.

Table III-3: Annual biochar production (Mt yr⁻¹), total carbon content of biochar (Pg yr⁻¹), and CO_2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 1.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79		
2025	1189.80	1147.03	1215.28	1349.65		
2050	1430.82	1439.53	1645.92	2010.89		
2075	1641.29	1575.01	2249.55	2932.67		
2100	1835.49	1570.53	3022.82	4322.58		
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)				
2005	0.68	0.68	0.68	0.68		
2025	0.84	0.80	0.85	0.95		
2050	1.01	0.99	1.16	1.42		
2075	1.16	1.09	1.58	2.07		
2100	1.30	1.09	2.13	3.05		
CO ₂ equivalent of car	bon in biochar (Petagr	rams (Pg CO₂ yr⁻¹))				
2005	2.49	2.49	2.49	2.49		
2025	3.08	2.92	3.12	3.48		
2050	3.71	3.65	4.24	5.20		
2075	4.27	3.99	5.80	7.59		
2100	4.78	3.98	7.83	11.20		

Table III-4: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 2a.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79		
2025	1066.18	926.17	995.98	1014.17		
2050	1119.98	922.57	1051.83	1054.38		
2075	1174.35	865.17	1120.98	1071.51		
2100	1201.11	793.75	1174.58	1098.01		
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)				
2005	0.68	0.68	0.68	0.68		
2025	0.75	0.64	0.70	0.71		
2050	0.79	0.64	0.74	0.74		
2075	0.83	0.60	0.79	0.76		
2100	0.85	0.55	0.83	0.78		
CO_2 equivalent of carbon in biochar (Petagrams (Pg CO_2 yr ⁻¹))						
2005	2.49	2.49	2.49	2.49		
2025	2.76	2.36	2.56	2.62		
2050	2.91	2.34	2.71	2.73		
2075	3.05	2.19	2.89	2.78		
2100	3.13	2.02	3.04	2.85		

Table III-5: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 2b.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5	
Total biochar produc	ction (Million tonnes (M	1t yr⁻¹))			
2005	969.79	969.79	969.79	969.79	
2025	1256.60	1275.48	1341.44	1555.15	
2050	1616.42	1795.63	2055.50	2772.01	
2075	1939.14	2121.58	3178.50	4843.12	
2100	2267.94	2205.38	4832.33	8564.66	
Total carbon in biochar (Petagrams (Pg C yr ⁻¹)) (1Pg = 1Gt)					
2005	0.68	0.68	0.68	0.68	
2025	0.89	0.88	0.94	1.09	
2050	1.14	1.24	1.44	1.95	

2075	1.37	1.46	2.23	3.41
2100	1.61	1.52	3.41	6.04
CO ₂ equivalent of carbo	n in biochar (Petagra	ms (Pg CO ₂ yr ⁻¹))		
2005	2.49	2.49	2.49	2.49
2025	3.25	3.24	3.45	4.01
2050	4.19	4.55	5.30	7.16
2075	5.04	5.37	8.20	12.52
2100	5.91	5.59	12.52	22.17

Table III-6: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 2c.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79		
2025	1189.80	1147.03	1215.28	1349.65		
2050	1430.82	1429.78	1645.92	2010.89		
2075	1641.29	1427.28	2249.55	2932.67		
2100	1835.49	1393.71	3022.82	4322.58		
Total carbon in bioch	nar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)				
2005	0.68	0.68	0.68	0.68		
2025	0.84	0.80	0.85	0.95		
2050	1.01	0.99	1.16	1.42		
2075	1.16	0.99	1.58	2.07		
2100	1.30	0.96	2.13	3.05		
CO ₂ equivalent of ca	rbon in biochar (Petagr	ams (Pg CO₂ yr⁻¹))				
2005	2.49	2.49	2.49	2.49		
2025	3.08	2.92	3.12	3.48		
2050	3.71	3.62	4.24	5.20		
2075	4.27	3.62	5.80	7.59		
2100	4.78	3.54	7.83	11.20		

Table III-7: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 3a.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5
Total biochar productio				

2005	969.79	969.79	969.79	969.79
2025	1082.23	1200.68	1183.33	1289.64
2050	1238.94	1511.47	1517.54	1844.38
2075	1355.39	1762.78	1946.14	2641.70
2100	1482.01	1915.98	2495.79	3788.86
Total carbon in bioch	har (Petagrams (Pg	C yr ⁻¹)) (1Pg = 1Gt)		
2005	0.68	0.68	0.68	0.68
2025	0.76	0.84	0.83	0.90
2050	0.87	1.06	1.06	1.29
2075	0.95	1.23	1.36	1.85
2100	1.04	1.34	1.75	2.65
CO ₂ equivalent of ca	rbon in biochar (Pe	tagrams (Pg CO ₂ yr	1))	
2005	2.49	2.49	2.49	2.49
2025	2.78	3.09	3.04	3.31
2050	3.19	3.89	3.90	4.74
2075	3.49	4.53	5.01	6.78
2100	3.81	4.92	6.42	9.71

Table III-8: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 3b. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	968.71	969.79	969.79	969.79		
2025	1156.77	1111.83	1164.37	1349.65		
2050	1379.21	1295.85	1526.03	2010.89		
2075	1550.33	1286.54	2062.50	2932.67		
2100	1704.09	1178.22	2338.71	4322.58		
Total carbon in biocha	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)				
2005	0.68	0.68	0.68	0.68		
2025	0.82	0.77	0.81	0.95		
2050	0.97	0.89	1.06	1.42		
2075	1.10	0.89	1.43	2.07		
2100	1.21	0.81	1.65	3.05		
CO_2 equivalent of carbon in biochar (Petagrams (Pg CO_2 yr ⁻¹))						
2005	2.49	2.49	2.49	2.49		
2025	2.99	2.82	2.97	3.48		
2050	3.58	3.27	3.89	5.20		
2075	4.03	3.25	5.25	7.59		
------	------	------	------	-------		
2100	4.43	2.98	6.08	11.20		

Table III-9: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 4a1.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	784.20	784.20	784.20	784.20		
2025	988.96	881.59	971.56	1106.35		
2050	1174.80	1080.66	1318.86	1663.34		
2075	1333.88	1164.38	1772.53	2404.46		
2100	1479.28	1152.39	2386.55	3546.26		
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)				
2005	0.60	0.60	0.60	0.60		
2025	0.77	0.67	0.75	0.86		
2050	0.92	0.82	1.02	1.30		
2075	1.04	0.89	1.37	1.88		
2100	1.16	0.88	1.86	2.78		
CO ₂ equivalent of car	bon in biochar (Petagr	rams (Pg CO₂ yr⁻¹))				
2005	2.22	2.22	2.22	2.22		
2025	2.83	2.47	2.75	3.15		
2050	3.36	3.01	3.75	4.76		
2075	3.82	3.25	5.04	6.89		
2100	4.24	3.22	6.81	10.19		

Table III-10: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 4a2.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	677.04	677.04	677.04	677.04		
2025	699.25	637.89	695.09	785.19		
2050	655.70	618.55	740.98	924.74		
2075	435.38	390.21	583.49	780.95		
2100	247.89	198.39	402.69	590.00		
Total carbon in biochar (Petagrams (Pg C yr ⁻¹)) (1Pg = 1Gt)						

2005	0.51	0.51	0.51	0.51
2025	0.53	0.48	0.52	0.59
2050	0.50	0.46	0.56	0.70
2075	0.33	0.29	0.44	0.59
2100	0.19	0.15	0.31	0.45
CO ₂ equivalent of cark	oon in biochar (Petagrar	ns (Pg CO ₂ yr ⁻¹))		
2005	1.87	1.87	1.87	1.87
2025	1.95	1.74	1.92	2.18
2050	1.83	1.68	2.05	2.58
2075	1.22	1.06	1.62	2.18
2100	0.69	0.54	1.12	1.65

Table III-11: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 4b.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar product	Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	677.04	677.04	677.04	677.04			
2025	987.55	900.90	981.67	1108.92			
2050	1349.43	1272.98	1524.93	1903.12			
2075	1836.30	1645.79	2461.02	3293.84			
2100	2266.44	1813.88	3681.74	5394.32			
Total carbon in bioch	ar (Petagrams (Pg C yr	¹)) (1Pg = 1Gt)					
2005	0.51	0.51	0.51	0.51			
2025	0.75	0.67	0.74	0.84			
2050	1.03	0.94	1.15	1.45			
2075	1.40	1.22	1.86	2.51			
2100	1.73	1.35	2.80	4.12			
CO ₂ equivalent of car	bon in biochar (Petagi	rams (Pg CO₂ yr⁻¹))					
2005	1.87	1.87	1.87	1.87			
2025	2.75	2.46	2.71	3.08			
2050	3.77	3.46	4.23	5.31			
2075	5.14	4.48	6.83	9.21			
2100	6.35	4.95	10.26	15.12			

Table III-12: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 5a.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar produ	Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	632.26	632.26	632.26	632.26			
2025	763.76	764.65	793.45	872.66			
2050	915.05	967.36	1068.73	1290.96			
2075	1044.41	1056.94	1457.91	1876.00			
2100	1163.58	1050.76	1939.51	2751.70			
Total carbon in bioc	har (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)					
2005	0.41	0.41	0.41	0.41			
2025	0.50	0.49	0.51	0.57			
2050	0.60	0.62	0.69	0.84			
2075	0.68	0.67	0.95	1.22			
2100	0.76	0.67	1.27	1.78			
CO ₂ equivalent of ca	irbon in biochar (Petagi	rams (Pg CO₂ yr⁻¹))					
2005	1.50	1.50	1.50	1.50			
2025	1.82	1.80	1.88	2.08			
2050	2.19	2.27	2.54	3.07			
2075	2.51	2.48	3.47	4.46			
2100	2.80	2.46	4.64	6.55			

Table III-13: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 5b.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar produc	Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	909.53	909.53	909.53	909.53			
2025	1103.72	1080.67	1137.94	1252.59			
2050	1328.51	1354.53	1535.10	1848.56			
2075	1521.83	1479.43	2098.45	2680.63			
2100	1700.39	1473.57	2818.35	3925.59			
Total carbon in bioch	har (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)					
2005	0.63	0.63	0.63	0.63			
2025	0.76	0.74	0.78	0.86			
2050	0.92	0.92	1.06	1.27			
2075	1.06	1.00	1.45	1.85			

2100	1.18	1.00	1.95	2.71	
CO_2 equivalent of carbon in biochar (Petagrams (Pg CO_2 yr ⁻¹))					
2005	2.30	2.30	2.30	2.30	
2025	2.80	2.70	2.87	3.17	
2050	3.38	3.37	3.88	4.68	
2075	3.88	3.68	5.31	6.78	
2100	4.34	3.67	7.17	9.94	

Table III-14: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 5c.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar product	Total biochar production (Million tonnes (Mt yr ⁻¹))					
2005	1186.80	1186.80	1186.80	1186.80		
2025	1443.68	1396.68	1482.43	1632.52		
2050	1741.96	1741.70	2001.48	2406.16		
2075	1999.25	1901.92	2739.00	3485.25		
2100	2237.19	1896.38	3697.18	5099.49		
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)				
2005	0.84	0.84	0.84	0.84		
2025	1.03	0.98	1.05	1.16		
2050	1.24	1.22	1.42	1.71		
2075	1.43	1.33	1.95	2.48		
2100	1.60	1.33	2.64	3.63		
CO ₂ equivalent of car	bon in biochar (Petagr	ams (Pg CO₂ yr⁻¹))				
2005	3.09	3.09	3.09	3.09		
2025	3.78	3.61	3.86	4.26		
2050	4.57	4.48	5.22	6.28		
2075	5.25	4.89	7.15	9.10		
2100	5.88	4.88	9.69	13.33		

Table III-15: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 5d.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	596.91	596.91	596.91	596.91		

2025	741.24	713.06	751.63	843.11
2050	883.87	902.62	1022.43	1272.46
2075	1012.13	988.69	1393.74	1868.18
2100	1129.62	984.36	1858.53	2775.34
Total carbon in bio	char (Petagrams (Pg C	yr ⁻¹)) (1Pg = 1Gt)		
2005	0.40	0.40	0.40	0.40
2025	0.50	0.47	0.50	0.57
2050	0.60	0.59	0.68	0.86
2075	0.69	0.64	0.93	1.26
2100	0.77	0.64	1.25	1.88
CO ₂ equivalent of o	carbon in biochar (Peta	agrams (Pg CO ₂ yr ⁻¹	¹))	
2005	1.46	1.46	1.46	1.46
2025	1.84	1.71	1.84	2.08
2050	2.19	2.16	2.51	3.15
2075	2.52	2.36	3.43	4.63
2100	2.82	2.36	4.60	6.90

Table III-16: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 5e.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar produc	Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	749.92	749.92	749.92	749.92			
2025	928.95	887.35	941.68	1053.01			
2050	1111.79	1115.69	1279.63	1580.83			
2075	1275.23	1220.98	1746.97	2313.44			
2100	1424.40	1216.43	2343.00	3425.39			
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)					
2005	0.52	0.52	0.52	0.52			
2025	0.65	0.60	0.65	0.73			
2050	0.78	0.76	0.89	1.10			
2075	0.89	0.83	1.21	1.62			
2100	1.00	0.83	1.64	2.40			
CO ₂ equivalent of car	bon in biochar (Petagr	rams (Pg CO ₂ yr ⁻¹))					
2005	1.90	1.90	1.90	1.90			
2025	2.38	2.22	2.39	2.68			
2050	2.85	2.77	3.25	4.04			
2075	3.28	3.03	4.45	5.93			
2100	3.67	3.03	6.00	8.80			

Table III-17: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO_2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 6a.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5
Total biochar product	tion (Million tonnes (N	1t yr ⁻¹))		
2005	749.49	749.49	749.49	749.49
2025	923.19	879.76	938.47	1044.39
2050	1111.68	1100.42	1272.81	1557.58
2075	1277.20	1204.18	1740.82	2272.13
2100	1429.99	1201.86	2347.61	3351.11
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)		
2005	0.54	0.54	0.54	0.54
2025	0.67	0.62	0.67	0.75
2050	0.80	0.78	0.91	1.12
2075	0.93	0.85	1.25	1.64
2100	1.04	0.85	1.70	2.42
CO ₂ equivalent of car	bon in biochar (Petagr	ams (Pg CO₂ yr⁻¹))		
2005	1.97	1.97	1.97	1.97
2025	2.45	2.29	2.47	2.76
2050	2.95	2.86	3.36	4.13
2075	3.40	3.13	4.60	6.02
2100	3.81	3.12	6.23	8.90

Table III-18: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO_2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 6b.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar product	Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	1888.71	1888.71	1888.71	1888.71			
2025	2326.44	2216.99	2364.94	2631.86			
2050	2801.44	2773.06	3207.49	3925.10			
2075	3218.56	3034.54	4386.88	5725.78			
2100	3603.56	3028.69	5915.99	8444.80			
Total carbon in biochar (Petagrams (Pg C yr ⁻¹)) (1Pg = 1Gt)							
2005	1.36	1.36	1.36	1.36			
2025	1.68	1.57	1.70	1.89			

2050	2.03	1.96	2.31	2.83	
2075	2.33	2.15	3.16	4.14	
2100	2.62	2.15	4.28	6.11	
CO_2 equivalent of carbon in biochar (Petagrams (Pg $CO_2 yr^{-1}$))					
2005	4.97	4.97	4.97	4.97	
2025	6.17	5.78	6.22	6.95	
2050	7.44	7.20	8.46	10.40	
2075	8.56	7.88	11.58	15.18	
2100	9.60	7.87	15.69	22.43	

Table III-19: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 7a.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79		
2025	1189.80	1147.03	1215.28	1349.65		
2050	1430.82	1439.53	1645.92	2010.89		
2075	1641.29	1575.01	2249.55	2932.67		
2100	1835.49	1570.53	3022.82	4322.58		
Total carbon in bioch	nar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)				
2005	0.58	0.58	0.58	0.58		
2025	0.72	0.69	0.73	0.81		
2050	0.86	0.87	0.99	1.21		
2075	0.99	0.95	1.35	1.77		
2100	1.10	0.95	1.82	2.60		
CO ₂ equivalent of car	rbon in biochar (Petagr	ams (Pg CO₂ yr⁻¹))				
2005	2.14	2.14	2.14	2.14		
2025	2.63	2.53	2.68	2.98		
2050	3.16	3.18	3.64	4.44		
2075	3.63	3.48	4.97	6.48		
2100	4.06	3.47	6.68	9.55		

Table III-20: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 7b.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5

Total biochar production (Million tonnes (Mt yr ⁻¹))				
2005	969.79	969.79	969.79	969.79
2025	1189.80	1147.03	1215.28	1349.65
2050	1430.82	1439.53	1645.92	2010.89
2075	1641.29	1575.01	2249.55	2932.67
2100	1835.49	1570.53	3022.82	4322.58
Total carbon in bioch	nar (Petagrams (Pg C y	r ⁻¹)) (1Pg = 1Gt)		
2005	0.70	0.70	0.70	0.70
2025	0.86	0.83	0.88	0.98
2050	1.04	1.04	1.19	1.46
2075	1.19	1.14	1.63	2.13
2100	1.33	1.14	2.19	3.13
CO ₂ equivalent of car	rbon in biochar (Petag	rams (Pg CO ₂ yr ⁻¹))		
2005	2.58	2.58	2.58	2.58
2025	3.17	3.05	3.23	3.59
2050	3.81	3.83	4.38	5.35
2075	4.37	4.19	5.99	7.80
2100	4.88	4.18	8.04	11.50

Table III-21: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 7c.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5	
Total biochar product	ion (Million tonnes (N	1t yr ⁻¹))			
2005	969.79	969.79	969.79	969.79	
2025	1189.80	1147.03	1215.28	1349.65	
2050	1430.82	1439.53	1645.92	2010.89	
2075	1641.29	1575.01	2249.55	2932.67	
2100	1835.49	1570.53	3022.82	4322.58	
Total carbon in biocha	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)			
2005	0.86	0.86	0.86	0.86	
2025	1.06	1.02	1.08	1.20	
2050	1.27	1.28	1.46	1.79	
2075	1.46	1.40	2.00	2.61	
2100	1.63	1.40	2.69	3.85	
CO_2 equivalent of carbon in biochar (Petagrams (Pg CO_2 yr ⁻¹))					
2005	3.17	3.17	3.17	3.17	
2025	3.89	3.75	3.97	4.41	
2050	4.67	4.70	5.38	6.57	

2075	5.36	5.14	7.35	9.58
2100	6.00	5.13	9.87	14.12

Table III-22: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO_2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 8a. Values shown for each RCP use the land-use and lowest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5
Total biochar product	ion (Million tonnes (M	/lt yr⁻¹))		
2005	969.79	969.79	969.79	969.79
2025	1189.80	1147.03	1215.28	1349.65
2050	1430.81	1439.53	1645.91	2010.89
2075	1641.29	1575.01	2249.55	2821.23
2100	1835.48	1539.12	2926.09	4028.64
Total carbon in biocha	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)		
2005	0.68	0.68	0.68	0.68
2025	0.84	0.80	0.85	0.95
2050	1.01	0.99	1.16	1.42
2075	1.16	1.09	1.58	1.99
2100	1.30	1.06	2.07	2.84
CO ₂ equivalent of carl	bon in biochar (Petagr	$\operatorname{rams}(\operatorname{Pg}\operatorname{CO}_2\operatorname{yr}^{-1}))$		
2005	2.49	2.49	2.49	2.49
2025	3.08	2.92	3.12	3.48
2050	3.71	3.65	4.24	5.20
2075	4.27	3.99	5.80	7.30
2100	4.78	3.90	7.58	10.44

Table III-23: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 8b. Values shown for each RCP use the land-use and average temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79		
2025	1189.80	1147.03	1215.28	1349.65		
2050	1430.81	1439.53	1645.92	1952.58		
2075	1641.29	1519.88	2170.82	2742.04		

2100	1835.48	1501.43	2862.61	3847.10		
Total carbon in biochar (Petagrams (Pg C yr ⁻¹)) (1Pg = 1Gt)						
2005	0.68	0.68	0.68	0.68		
2025	0.84	0.80	0.85	0.95		
2050	1.01	0.99	1.16	1.38		
2075	1.16	1.05	1.53	1.93		
2100	1.30	1.04	2.02	2.72		
CO ₂ equivalent of carbo	on in biochar (Petagra	ams (Pg CO ₂ yr ⁻¹))				
2005	2.49	2.49	2.49	2.49		
2025	3.08	2.92	3.12	3.48		
2050	3.71	3.65	4.24	5.05		
2075	4.27	3.85	5.60	7.10		
2100	4.78	3.81	7.42	9.97		

Table III-24: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 8c. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar produc	Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79			
2025	1189.80	1147.03	1215.28	1349.65			
2050	1397.91	1393.47	1598.18	1934.48			
2075	1583.84	1491.53	2130.32	2683.39			
2100	1765.74	1463.74	2765.88	3609.35			
Total carbon in bioch	nar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)					
2005	0.68	0.68	0.68	0.68			
2025	0.84	0.80	0.85	0.95			
2050	0.99	0.96	1.12	1.36			
2075	1.12	1.03	1.50	1.89			
2100	1.25	1.01	1.95	2.55			
CO ₂ equivalent of car	rbon in biochar (Petagr	ams (Pg CO₂ yr⁻¹))					
2005	2.49	2.49	2.49	2.49			
2025	3.08	2.92	3.12	3.48			
2050	3.63	3.53	4.12	5.00			
2075	4.12	3.78	5.49	6.94			
2100	4.60	3.71	7.17	9.35			

Table III-25: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 8d1. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar produc	Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79			
2025	1189.80	1147.03	1215.28	1349.65			
2050	1430.81	1439.53	1645.91	2010.89			
2075	1641.29	1575.01	2249.55	2674.59			
2100	1835.48	1444.89	2765.88	3838.45			
Total carbon in bioch	ar (Petagrams (Pg C yr	¹)) (1Pg = 1Gt)					
2005	0.68	0.68	0.68	0.68			
2025	0.84	0.80	0.85	0.95			
2050	1.01	0.99	1.16	1.42			
2075	1.16	1.09	1.58	1.89			
2100	1.30	1.00	1.95	2.71			
CO ₂ equivalent of car	rbon in biochar (Petagr	rams (Pg CO ₂ yr ⁻¹))					
2005	2.49	2.49	2.49	2.49			
2025	3.08	2.92	3.12	3.48			
2050	3.71	3.65	4.24	5.20			
2075	4.27	3.99	5.80	6.92			
2100	4.78	3.67	7.17	9.95			

Table III-26: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 8d2. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5	
Total biochar product	ion (Million tonnes (N	1t yr ⁻¹))			
2005	969.79	969.79	969.79	969.79	
2025	1189.80	1147.03	1215.28	1349.65	
2050	1430.81	1439.53	1645.92	1841.98	
2075	1641.29	1437.98	2053.84	2613.01	
2100	1835.49	1427.61	2729.61	3618.00	
Total carbon in biochar (Petagrams (Pg C yr ⁻¹)) (1Pg = 1Gt)					
2005	0.68	0.68	0.68	0.68	

2025	0.84	0.80	0.85	0.95
2050	1.01	0.99	1.16	1.30
2075	1.16	0.99	1.44	1.84
2100	1.30	0.99	1.93	2.55
CO ₂ equivalent of carbon in bio	char (Petagrams (Pg	CO ₂ yr ⁻¹))		
2005	2.49	2.49	2.49	2.49
2025	3.08	2.92	3.12	3.48
2050	3.71	3.65	4.24	4.76
2075	4.27	3.64	5.30	6.76
2100	4.78	3.62	7.07	9.38

Table III-27: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 8d3. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79		
2025	1189.80	1147.03	1215.28	1349.65		
2050	1314.92	1317.17	1507.66	1833.93		
2075	1498.50	1422.23	2031.34	2548.49		
2100	1673.96	1394.63	2626.83	3298.13		
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)				
2005	0.68	0.68	0.68	0.68		
2025	0.84	0.80	0.85	0.95		
2050	0.93	0.91	1.06	1.29		
2075	1.06	0.98	1.43	1.80		
2100	1.19	0.96	1.85	2.33		
CO ₂ equivalent of car	rbon in biochar (Petagr	rams (Pg CO₂ yr⁻¹))				
2005	2.49	2.49	2.49	2.49		
2025	3.08	2.92	3.12	3.48		
2050	3.41	3.34	3.88	4.74		
2075	3.90	3.60	5.24	6.60		
2100	4.36	3.54	6.81	8.55		

Table III-28: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100,

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar product	Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79			
2025	1189.80	1147.03	1215.28	1349.65			
2050	1430.81	1439.53	1645.91	2010.89			
2075	1641.29	1575.01	2249.55	2964.93			
2100	1835.48	1620.79	3077.23	4249.10			
Total carbon in biocha	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)					
2005	0.68	0.68	0.68	0.68			
2025	0.84	0.80	0.85	0.95			
2050	1.01	0.99	1.16	1.42			
2075	1.16	1.09	1.58	2.09			
2100	1.30	1.12	2.17	3.00			
CO ₂ equivalent of carl	bon in biochar (Petagr	rams (Pg CO₂ yr⁻¹))					
2005	2.49	2.49	2.49	2.49			
2025	3.08	2.92	3.12	3.48			
2050	3.71	3.65	4.24	5.20			
2075	4.27	3.99	5.80	7.67			
2100	4.78	4.11	7.97	11.01			

for each of the RCPs assessed under biochar Scenario 8e1. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Table III-29: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 8e2. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar produc	Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79			
2025	1189.80	1147.03	1215.28	1349.65			
2050	1430.81	1439.53	1645.92	2053.12			
2075	1641.29	1597.06	2281.05	2888.68			
2100	1835.49	1576.81	3004.69	4110.77			
Total carbon in bioch	nar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)					
2005	0.68	0.68	0.68	0.68			
2025	0.84	0.80	0.85	0.95			
2050	1.01	0.99	1.16	1.45			
2075	1.16	1.10	1.60	2.04			

2100	1.30	1.09	2.12	2.90
CO ₂ equivalent of carbo	on in biochar (Petagrai	ms (Pg CO ₂ yr ⁻¹))		
2005	2.49	2.49	2.49	2.49
2025	3.08	2.92	3.12	3.48
2050	3.71	3.65	4.24	5.31
2075	4.27	4.05	5.88	7.48
2100	4.78	4.00	7.79	10.65

Table III-30: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 8e3. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar produc	Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79			
2025	1189.80	1147.03	1215.28	1349.65			
2050	1470.88	1465.45	1680.48	2033.01			
2075	1664.27	1565.56	2236.05	2844.69			
2100	1855.68	1543.83	2932.14	3959.48			
Total carbon in bioch	nar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)					
2005	0.68	0.68	0.68	0.68			
2025	0.84	0.80	0.85	0.95			
2050	1.04	1.01	1.18	1.43			
2075	1.18	1.08	1.57	2.01			
2100	1.32	1.07	2.07	2.80			
CO_2 equivalent of ca	rbon in biochar (Petagr	rams (Pg CO ₂ yr ⁻¹))					
2005	2.49	2.49	2.49	2.49			
2025	3.08	2.92	3.12	3.48			
2050	3.82	3.71	4.33	5.26			
2075	4.33	3.97	5.77	7.36			
2100	4.83	3.92	7.60	10.26			

Table III-31: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 8f. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5

Total biochar production (Million tonnes (Mt yr ⁻¹))				
2005	969.79	969.79	969.79	969.79
2025	1183.99	1166.09	1202.72	1366.37
2050	1421.01	1367.01	1607.21	1952.76
2075	1638.16	1443.11	2050.68	2612.74
2100	1818.54	1389.20	2740.77	3722.19
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)		
2005	0.68	0.68	0.68	0.68
2025	0.84	0.82	0.85	0.97
2050	1.02	0.96	1.15	1.41
2075	1.18	1.02	1.47	1.91
2100	1.30	0.98	2.00	2.75
CO ₂ equivalent of car	bon in biochar (Petagr	ams (Pg CO ₂ yr ⁻¹))		
2005	2.49	2.49	2.49	2.49
2025	3.09	3.00	3.11	3.56
2050	3.73	3.52	4.22	5.17
2075	4.31	3.74	5.41	7.00
2100	4.79	3.61	7.33	10.08

Table III-32: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO_2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 9a. Values shown for each RCP use the land-use and lowest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79		
2025	1189.80	1147.03	1215.28	1349.65		
2050	1430.81	1439.53	1645.91	2010.89		
2075	1641.29	1575.01	2249.55	2926.80		
2100	1835.48	1570.53	3028.87	4227.48		
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)				
2005	0.68	0.68	0.68	0.68		
2025	0.84	0.80	0.85	0.95		
2050	1.01	0.99	1.16	1.42		
2075	1.16	1.09	1.58	2.06		
2100	1.30	1.09	2.14	2.99		
CO_2 equivalent of carbon in biochar (Petagrams (Pg $CO_2 \text{ yr}^{-1}$))						
2005	2.49	2.49	2.49	2.49		

2025	3.08	2 92	3 12	3 48
2050	0.00	2.02	4.04	5.10
2075	3.71	3.05	4.24	5.20
2075	4.27	3.99	5.80	7.57
2100	4.78	3.98	7.85	10.96

Table III-33: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO_2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 9b. Values shown for each RCP use the land-use and average temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 8.5					
Total biochar product	Total biochar production (Million tonnes (Mt yr ⁻¹))							
2005	969.79	969.79	969.79	969.79				
2025	1189.80	1147.03	1215.28	1349.65				
2050	1430.81	1439.53	1645.92	2018.94				
2075	1641.29	1575.01	2249.55	2874.01				
2100	1835.49	1561.11	2986.55	4192.90				
Total carbon in biochar (Petagrams (Pg C yr ⁻¹)) (1Pg = 1Gt)								
2005	0.68	0.68	0.68	0.68				
2025	0.84	0.80	0.85	0.95				
2050	1.01	0.99	1.16	1.42				
2075	1.16	1.09	1.58	2.03				
2100	1.30	1.08	2.11	2.96				
CO ₂ equivalent of carb	oon in biochar (Petagr	rams (Pg CO ₂ yr ⁻¹))						
2005	2.49	2.49	2.49	2.49				
2025	3.08	2.92	3.12	3.48				
2050	3.71	3.65	4.24	5.22				
2075	4.27	3.99	5.80	7.44				
2100	4.78	3.96	7.74	10.87				

Table III-34: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 9c. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	CP 2.6 RCP 4.5 RCP 6		RCP 8.5		
Total biochar production (Million tonnes (Mt yr ⁻¹))						
2005	969.79	969.79	969.79	969.79		
2025	1189.80	1147.03	1215.28	1349.65		

2050							
2030	1442.26	1442.41	1652.50	2006.87			
2075	1641.29	1556.11	2222.56	2844.69			
2100	1831.82	1535.98	2932.14	4132.39			
Total carbon in biochar (Petagrams (Pg C yr ⁻¹)) (1Pg = 1Gt)							
2005	0.68	0.68	0.68	0.68			
2025	0.84	0.80	0.85	0.95			
2050	1.02	1.00	1.16	1.41			
2075	1.16	1.07	1.56	2.01			
2100	1.30	1.06	2.07	2.92			
CO ₂ equivalent of car	rbon in biochar (Petagr	ams (Pg CO ₂ yr ⁻¹))					
2005	2.49	2.49	2.49	2.49			
2025	3.08	2.92	3.12	3.48			
2050	3.74	3.65	4.26	5.19			
2075	4.27	3.94	5.73	7.36			
2100	4.77	3.90	7.60	10.71			

Table III-35: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 9d1. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5
Total biochar product	tion (Million tonnes (N	1t yr ⁻¹))		
2005	969.79	969.79	969.79	969.79
2025	1189.80	1147.03	1215.28	1349.65
2050	1430.81	1439.53	1645.91	2010.89
2075	1641.29	1575.01	2249.55	2818.29
2100	1835.48	1507.71	2904.93	4080.52
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)		
2005	0.68	0.68	0.68	0.68
2025	0.84	0.80	0.85	0.95
2050	1.01	0.99	1.16	1.42
2075	1.16	1.09	1.58	1.99
2100	1.30	1.04	2.05	2.88
CO ₂ equivalent of car	bon in biochar (Petagr	ams (Pg CO₂ yr⁻¹))		
2005	2.49	2.49	2.49	2.49
2025	3.08	2.92	3.12	3.48
2050	3.71	3.65	4.24	5.20
2075	4.27	3.99	5.80	7.29
2100	4.78	3.83	7.53	10.58

Table III-36: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 9d2. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 8.5				
Total biochar production (Million tonnes (Mt yr ⁻¹))							
2005	969.79	969.79	969.79	969.79			
2025	1189.80	1147.03	1215.28	1349.65			
2050	1430.81	1439.53	1645.92	1932.47			
2075	1641.29	1513.58	2161.82	2774.30			
2100	1835.49	1510.85	2892.84	3998.39			
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)					
2005	0.68	0.68	0.68	0.68			
2025	0.84	0.80	0.85	0.95			
2050	1.01	0.99	1.16	1.36			
2075	1.16	1.05	1.52	1.96			
2100	1.30	1.04	2.04	2.82			
CO ₂ equivalent of car	bon in biochar (Petagr	$rams (Pg CO_2 yr^{-1}))$					
2005	2.49	2.49	2.49	2.49			
2025	3.08	2.92	3.12	3.48			
2050	3.71	3.65	4.24	5.00			
2075	4.27	3.84	5.57	7.18			
2100	4.78	3.83	7.50	10.36			

Table III-37: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 9d3. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar production (Million tonnes (Mt yr ⁻¹))							
2005	969.79	969.79	969.79	969.79			
2025	1189.80	1147.03	1215.28	1349.65			
2050	1373.58	1383.39	1581.73	1932.47			
2075	1577.28	1507.28	2152.82	2736.18			
2100	1763.90	1482.58	2820.29	3764.97			
Total carbon in biochar (Petagrams (Pg C yr ⁻¹)) (1Pg = 1Gt)							

2005	0.68	0.68	0.68	0.68		
2025	0.84	0.80	0.85	0.95		
2050	0.97	0.95	1.11	1.36		
2075	1 12	1 04	1 51	1 93		
2100	1.12	1.04	1.91	2.66		
CO_2 equivalent of carbon in biochar (Petagrams (Pg CO_2 yr ⁻¹))						
2005	2.49	2.49	2.49	2.49		
2025	3.08	2.92	3.12	3.48		
2050	3 56	3 50	4 08	5.00		
2075	4 10	3.82	5 55	7.08		
2100	4.59	3.76	7.31	9.76		

Table III-38: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 9e1. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5			
Total biochar production (Million tonnes (Mt yr ⁻¹))							
2005	969.79	969.79	969.79	969.79			
2025	1189.80	1147.03	1215.28	1349.65			
2050	1430.81	1439.53	1645.91	2010.89			
2075	1641.29	1575.01	2249.55	3088.10			
2100	1835.48	1689.89	3207.22	4430.64			
Total carbon in biocha	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)					
2005	0.68	0.68	0.68	0.68			
2025	0.84	0.80	0.85	0.95			
2050	1.01	0.99	1.16	1.42			
2075	1.16	1.09	1.58	2.18			
2100	1.30	1.17	2.26	3.13			
CO_2 equivalent of carb	oon in biochar (Petagr	rams (Pg CO₂ yr⁻¹))					
2005	2.49	2.49	2.49	2.49			
2025	3.08	2.92	3.12	3.48			
2050	3.71	3.65	4.24	5.20			
2075	4.27	3.99	5.80	7.99			
2100	4.78	4.29	8.31	11.48			

Table III-39: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100,

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5				
Total biochar product	Total biochar production (Million tonnes (Mt yr ⁻¹))							
2005	969.79	969.79	969.79	969.79				
2025	1189.80	1147.03	1215.28	1349.65				
2050	1430.81	1439.53	1645.92	2141.60				
2075	1641.29	1664.78	2377.78	3011.85				
2100	1835.49	1642.78	3134.67	4443.61				
Total carbon in bioch	ar (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)						
2005	0.68	0.68	0.68	0.68				
2025	0.84	0.80	0.85	0.95				
2050	1.01	0.99	1.16	1.51				
2075	1.16	1.15	1.67	2.12				
2100	1.30	1.14	2.21	3.14				
CO_2 equivalent of car	bon in biochar (Petagr	rams (Pg CO₂ yr⁻¹))						
2005	2.49	2.49	2.49	2.49				
2025	3.08	2.92	3.12	3.48				
2050	3.71	3.65	4.24	5.54				
2075	4.27	4.22	6.13	7.79				
2100	4.78	4.17	8.12	11.52				

for each of the RCPs assessed under biochar Scenario 9e2. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Table III-40: Annual biochar production (Mt yr-1), Total carbon content of biochar (Pg yr-1), and CO2 equivalent of the carbon stored in biochar for the years 2005, 2025, 2050, 2075 and 2100, for each of the RCPs assessed under biochar Scenario 9e3. Values shown for each RCP use the land-use and highest temperature projections for that particular RCP.

Year	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5
Total biochar production	on (Million tonnes (M	1t yr ⁻¹))		
2005	969.79	969.79	969.79	969.79
2025	1189.80	1147.03	1215.28	1349.65
2050	1533.84	1527.35	1752.90	2117.47
2075	1734.84	1633.28	2332.78	2994.25
2100	1932.77	1609.80	3086.30	4499.81
Total carbon in biochar	r (Petagrams (Pg C yr	⁻¹)) (1Pg = 1Gt)		
2005	0.68	0.68	0.68	0.68
2025	0.84	0.80	0.85	0.95
2050	1.08	1.05	1.23	1.49
2075	1.23	1.13	1.64	2.11

2100	1.37	1.11	2.18	3.18		
CO_2 equivalent of carbon in biochar (Petagrams (Pg CO_2 yr ⁻¹))						
2005	2.49	2.49	2.49	2.49		
2025	3.08	2.92	3.12	3.48		
2050	3.98	3.87	4.52	5.48		
2075	4.51	4.14	6.02	7.75		
2100	5.03	4.08	8.00	11.66		

	Units	95 year total values			
Scenario 1		RCP 2.6	RCP 4.5	RCP 6	RCP 8.5
Total biochar	Gt	138.38	132.25	173.17	217.89
Total carbon	PgC	97.87	91.48	121.69	153.56
Total CO_2 equivalent of carbon in	$PgCO_2$	359.20	335.75	446.61	563.55
Scenario 2					
2a					
Total biochar	Gt	107.90	86.00	102.26	100.53
Total carbon	PgC	76.29	59.57	71.82	70.91
Total CO_2 equivalent of carbon in	PgCO ₂	280.00	218.61	263.58	260.25
2b					
Total biochar	Gt	157.48	166.19	231.02	338.34
Total carbon	PgC	111.39	114.89	162.39	238.32
Total CO_2 equivalent of carbon in	PgCO ₂	408.82	421.66	595.97	874.65
2c					
Total biochar	Gt	As S1	125.78	As S1	As S1
Total carbon	PgC	As S1	87.04	As S1	As S1
Total CO ₂ equivalent of carbon in	PgCO ₂	As S1	319.42	As S1	As S1
Scenario 3					
3a	.				
Total biochar	Gt	118.58	144.35	155.10	198.57
Total carbon	PgC	83.12	101.11	108.72	138.86
Total CO_2 equivalent of carbon in	PgCO₂	305.06	371.08	398.99	509.62
3b					
Total biochar	Gt	131.91	115.61	155.94	217.89
Total carbon	PgC	93.20	79.78	108.79	153.56
Total CO_2 equivalent of carbon in	PgCO ₂	342.04	292.81	399.25	563.55
Scenario 4					
4a1	C 1				
Total blochar	Gt	113.11	99.39	137.50	178.80
Total carbon	PgC	40.20	36.04	45.47	56.20
Total CO_2 equivalent of carbon in	PgCO ₂	147.54	132.27	166.88	206.27
	<u> </u>				
i otal biochar	GT	52.88	48.43	60.22	73.99
	PgC	28.03	25.29	31.77	39.16
rotal CO_2 equivalent of carbon in	PgCO ₂	102.86	92.80	116.59	143.72
40					

Table III-41: Total biochar production (Gt), biochar carbon content (GtC) and CO_2 equivalent of carbon content (Gt CO_2) for each scenario, over the 95 year period.

Total biochar	Gt	140.03	125.17	176.04	229.65
Total carbon	GtC	106.62	93.01	133.11	174.78
Total CO_2 equivalent of carbon in	GtCO ₂	391.30	341.33	488.52	641.43
Scenario 5					
5a					
Total biochar	Gt	88.39	88.46	112.22	139.68
Total carbon	GtC	57.70	56.60	72.85	90.56
Total CO_2 equivalent of carbon in	GtCO ₂	211.76	207.72	267.36	332.37
5b					
Total biochar	Gt	128.46	124.33	161.65	199.82
Total carbon	GtC	89.07	84.54	111.48	137.78
Total CO_2 equivalent of carbon in	GtCO ₂	326.89	310.25	409.13	505.64
5c					
Total biochar	Gt	168.53	160.19	211.08	259.96
Total carbon	GtC	120.44	112.47	150.11	184.99
Total CO_2 equivalent of carbon in	GtCO ₂	442.01	412.78	550.90	678.91
5d		-	-		
Total biochar	Gt	85.48	82.70	107.12	138.23
Total carbon	GtC	57.85	54.03	71.74	93.26
Total CO_2 equivalent of carbon in	GtCO ₂	212.31	198.29	263.27	342.27
5e					0.2.2.
Total biochar	Gt	107.58	102.44	134.38	171.49
Total carbon	GtC	75.31	69.56	93.24	119.61
Total CO_2 equivalent of carbon in	GtCO ₂	276.37	255.28	342.19	438.98
Scenario 6					
6a					
Total biochar	Gt	107.57	101.26	134.03	168.78
Total carbon	GtC	77.86	71.75	96.42	121.84
Total CO_2 equivalent of carbon in	GtCO ₂	285.75	263.34	353.86	447.17
6b					
Total biochar	Gt	271.07	255.16	337.75	425.32
Total carbon	GtC	196.21	180.82	242.98	307.05
Total CO_2 equivalent of carbon in	GtCO ₂	720.09	663.62	891.73	1126.86
Scenario 7					
7a					
Total biochar	Gt	138.38	132.25	173.17	217.89
Total carbon	GtC	83.31	79.61	104.25	131.17
Total CO_2 equivalent of carbon in	GtCO ₂	305.74	292.18	382.59	481.39
7b			-		
Total biochar	Gt	138.38	132.25	173.17	217.89

Total carbon	GtC	100.33	95.88	125.55	157.97
Total CO_2 equivalent of carbon in	GtCO ₂	368.21	351.88	460.76	579.74
7c					
Total biochar	Gt	138.38	132.25	173.17	217.89
Total carbon	GtC	123.16	117.70	154.12	193.92
Total CO_2 equivalent of carbon in	GtCO ₂	452.00	431.96	565.62	711.68

Table III-42: Total biochar production (Gt), biochar carbon content (Gt) and CO_2 equivalent of carbon content (Gt) for each climate change with no adaptation scenario (Scenario 8), over the 95 year period.

	Units	95 year total values			
Scenario 8		RCP 2.6	RCP 4.5	RCP 6	RCP 8.5
8a (minimum temp)					
Total biochar	Pg	138.38	131.96	171.73	211.21
Total carbon	PgC	97.87	91.29	120.68	148.84
Total CO_2 equivalent of carbon in	$PgCO_2$	359.20	335.03	442.88	546.25
8b (median temperature)					
Total biochar	Pg	138.38	129.69	168.90	206.73
Total carbon	PgC	97.87	89.72	118.69	145.68
Total CO_2 equivalent of carbon in	$PgCO_2$	359.20	329.27	435.59	534.65
8c (maximum temperature)					
Total biochar	Pg	135.22	127.72	166.17	201.18
Total carbon	PgC	95.63	88.36	116.77	141.77
Total CO_2 equivalent of carbon in	$PgCO_2$	350.97	324.26	428.53	520.30
8d1 (Lowest yield projections, minimu	m tempe	erature)			
Total biochar	Pg	138.38	131.11	168.50	204.33
Total carbon	PgC	97.87	90.70	118.41	143.99
Total CO_2 equivalent of carbon in	PgCO ₂	359.20	332.87	434.55	528.45
8d2 (lowest yield projections, mean te	mperatu	re)			
Total biochar	Pg	138.38	125.69	163.45	198.87
Total carbon	PgC	97.87	86.95	114.85	140.14
Total CO_2 equivalent of carbon in	PgCO ₂	359.20	319.11	421.50	514.32
8d3 (lowest yield projections, maximu	m tempe	erature)			
Total biochar	Pg	130.31	123.34	159.99	191.79
Total carbon	PgC	92.16	85.34	112.42	135.15
Total CO_2 equivalent of carbon in	PgCO ₂	338.22	313.18	412.59	495.99
8e1 (highest yield projections, minimu	m tempe	erature)			
Total biochar	Pg	138.38	132.70	174.61	218.23
Total carbon	PgC	97.87	91.80	122.71	153.80

Total CO_2 equivalent of carbon in	PgCO ₂	359.20	336.90	450.33	564.44
8e2 (highest yield projections, mean te	mperature)			
Total biochar	Pg	138.38	133.46	174.24	215.31
Total carbon	PgC	97.87	92.32	122.44	151.74
Total CO_2 equivalent of carbon in	PgCO ₂	359.20	338.83	449.36	556.87
8e3 (highest yield projections, maximu	m tempera	iture)			
Total biochar	Pg	139.84	132.25	172.68	211.63
Total carbon	PgC	98.91	91.49	121.34	149.14
Total CO_2 equivalent of carbon in	PgCO ₂	362.99	335.76	445.33	547.33
8f					
Total biochar	Pg	137.83	124.79	163.74	200.61
Total carbon	PgC	98.45	87.80	117.33	145.53
Total CO_2 equivalent of carbon in	PgCO ₂	361.31	322.23	430.60	534.09

Table III-43: Total biochar production (Gt), biochar carbon content (Gt) and CO_2 equivalent of carbon content (Gt) for each climate change with adaptation scenario (Scenario 9), over the 95 year period.

	Units	95 year to	95 year total values		
Scenario 9		RCP 2.6	RCP 4.5	RCP 6	RCP 8.5
9a (minimum temp)					
Total biochar	Pg	138.38	132.25	173.38	216.71
Total carbon	Pg	97.87	91.48	121.84	152.73
Total CO_2 equivalent of carbon in	Pg	359.20	335.75	447.16	560.51
9b (median temperature)					
Total biochar	Pg	138.38	132.26	172.79	214.90
Total carbon	Pg	97.87	91.49	121.42	151.45
Total CO_2 equivalent of carbon in	Pg	359.20	335.79	445.63	555.81
9c (maximum temperature)					
Total biochar	Pg	138.40	131.35	171.50	213.59
Total carbon	PgC	97.89	90.87	120.52	150.52
Total CO_2 equivalent of carbon in	$PgCO_2$	359.24	333.48	442.29	552.42
9d1 (Lowest yield projections, minimu	m tempe	erature)			
Total biochar	Pg	138.38	131.68	170.93	211.64
Total carbon	PgC	97.87	91.09	120.11	149.15
Total CO_2 equivalent of carbon in	$PgCO_2$	359.20	334.31	440.81	547.37
9d2 (lowest yield projections, mean te	mperatu	re)			
Total biochar	Pg	138.38	129.27	168.81	208.79
Total carbon	PgC	97.87	89.43	118.63	147.14
Total CO_2 equivalent of carbon in	PgCO ₂	359.20	328.20	435.36	540.00

9d3 (lowest yield projections, maximur	n tempera	ature)			
Total biochar	Pg	134.72	128.13	166.85	205.11
Total carbon	PgC	95.28	88.64	117.25	144.55
Total CO_2 equivalent of carbon in	PgCO ₂	349.68	325.30	430.29	530.48
9e1 (highest yield projections, minimur	m temper	ature)			
Total biochar	Pg	138.38	133.33	177.09	224.34
Total carbon	PgC	97.87	92.23	124.46	158.11
Total CO_2 equivalent of carbon in	PgCO ₂	359.20	338.48	456.76	580.26
9e2 (highest yield projections, mean te	mperatur	e)			
Total biochar	Pg	138.38	136.74	178.93	223.51
Total carbon	PgC	97.87	94.58	125.74	157.52
Total CO_2 equivalent of carbon in	PgCO ₂	359.20	347.12	461.47	578.12
9e3 (highest yield projections, maximum temperature)					
Total biochar	Pg	143.85	136.21	178.29	224.06
Total carbon	PgC	101.75	94.22	125.29	157.91
Total CO_2 equivalent of carbon in	PgCO ₂	373.41	345.78	459.83	579.52

IV. Annexe 4: Carbon storage: Results tables

Table IV-1: Total carbon stored long-term after the addition of biochar to soils for the different scenarios developed in Chapter 3. The stored carbon values were calculated from the biochar production scenarios using the CS equation of Zhao et al. (2013).

	Total carbon stored long-term (GtC)				
Scenario	RCP 2.6	RCP 4.5	RCP 6	RCP 8.5	
1	49.0	45.8	60.9	77.2	
2a	38.2	29.8	36.0	35.6	
2b	55.8	57.6	81.3	119.9	
2c	49.0	43.6	60.9	77.2	
3a	41.6	50.6	54.4	69.6	
3b	46.7	40.0	54.4	77.2	
4a1	45.0	38.5	54.2	71.3	
4a2	20.3	18.2	23.0	28.5	
4b	53.9	46.9	67.3	88.7	
5a	29.3	28.9	37.1	46.2	
5b	44.4	42.3	55.6	69.0	
5c	59.6	55.7	74.2	91.8	
5d	30.5	28.5	37.8	49.3	
5e	39.2	36.2	48.5	62.3	
6a	38.8	35.8	48.1	61.0	
6b	97.9	90.1	121.1	153.6	
7a	41.9	40.2	52.5	66.3	
7b	50.5	48.4	63.2	79.8	
7c	62.0	59.4	77.6	97.9	

Table IV-2: Long term carbon storage potential of biochars using the two pool assessment methodology of Woolf et al. (2010) under different assumptions. The assumptions made are variances in the size of the labile and recalcitrant fractions, and in the half-lives of these pools.

RCP	L	R	T _{1/2L}	T _{1/2R}	Т	CS
2.6			(years)	(years)	(years)	(GtC)
Variation in	labile half-life					
Low labile ha	alf-life					
	0.15	0.85	1	300	100	66.0
	0.15	0.85	1	300	250	46.7
	0.15	0.85	1	300	500	26.2
	0.15	0.85	1	300	750	14.7
	0.15	0.85	1	300	1000	8.3
	0.15	0.85	1	300	1250	4.6
	0.15	0.85	1	300	1500	2.6
	0.15	0.85	1	300	1750	1.5
Main labile ł	nalf-life assum	ption				

0.15	0.85	20	300	100	66.5
0.15	0.85	20	300	250	46.7
0.15	0.85	20	300	500	26.2
0.15	0.85	20	300	750	14.7
0.15	0.85	20	300	1000	8.3
0.15	0.85	20	300	1250	4.6
0.15	0.85	20	300	1500	2.6
0.15	0.85	20	300	1750	1.5
Long labile half-life					
0.15	0.85	25	300	100	66.9
0.15	0.85	25	300	250	46.7
0.15	0.85	25	300	500	26.2
0.15	0.85	25	300	750	14.7
0.15	0.85	25	300	1000	8.3
0.15	0.85	25	300	1250	4.6
0.15	0.85	25	300	1500	2.6
0.15	0.85	25	300	1750	1.5
Variation in recalcitrant h	alf-life				
Low recalcitrant fraction	half-life				
0.15	0.85	20	50	100	21.3
0.15	0.85	20	50	150	10.5
0.15	0.85	20	50	200	5.2
0.15	0.85	20	50	250	2.6
Main recalcitrant fraction	half-life as	sumptions			
0.15	0.85	20	300	100	66.5
0.15	0.85	20	300	250	46.7
0.15	0.85	20	300	500	26.2
0.15	0.85	20	300	750	14.7
0.15	0.85	20	300	1000	8.25
High recalcitrant fraction	half-life				
0.15	0.85	20	1000	100	78.1
0.15	0.85	20	1000	250	70.0
0.15	0.85	20	1000	500	58.8
0.15	0.85	20	1000	750	49.5
0.15	0.85	20	1000	1000	41.6
0.15	0.85	20	1000	1250	35.0
0.15	0.85	20	1000	1500	29.4
0.15	0.85	20	1000	1750	24.7
0.15	0.85	20	1000	2000	20.8
0.15	0.85	20	1000	2500	14.7
0.15	0.85	20	1000	3000	10.4
0.15	0.85	20	1000	4000	5.2
Variation in labile fraction	า				
High labile fraction					
0.3	0.7	20	300	100	55.3
0.3	0.7	20	300	250	38.5

	0.3	0.7	20	300	500	21.6
	0.3	0.7	20	300	750	12.1
	0.3	0.7	20	300	1000	6.8
	0.3	0.7	20	300	1250	3.8
Low labile fr	raction					
	0.05	0.95	20	300	100	74.0
	0.05	0.95	20	300	250	52.2
	0.05	0.95	20	300	500	29.3
	0.05	0.95	20	300	750	16.4
	0.05	0.95	20	300	1000	9.2
	0.05	0.95	20	300	1250	5.2
	0.05	0.95	20	300	1500	2.9
Best and Wo	orst Case Assur	nptions				
'Worst case	' assumptions					
	0.3	0.7	1	50	100	17.1
	0.3	0.7	1	50	150	8.6
	0.3	0.7	1	50	200	4.3
	0.3	0.7	1	50	250	2.1
	0.3	0.7	1	50	300	1.1
'Best case' a	ssumptions					
	0.05	0.95	25	1000	100	87.1
	0.05	0.95	25	1000	500	65.7
	0.05	0.95	25	1000	1000	46.5
	0.05	0.95	25	1000	2000	23.2
	0.05	0.95	25	1000	3000	11.6
	0.05	0.95	25	1000	4000	5.8
	0.05	0.95	25	1000	5000	2.9



Figure IV-1: Potential emission reductions for RCP4.5, relative to the carbon emission pathway of projection RCP4 (black line). Blue line shows emissions reduction potential of Scenario 1 biochar assumptions. Other scenario assumptions shown are: top left, Scenario 2; top right, Scenario 3; mid-left, Scenario 4; mid-right, Scenario 5; bottom left, Scenario 6; bottom right, Scenario 7.



Figure IV-2: Potential emission reductions for RCP6, relative to the carbon emission pathway of projection RCP6 (black line). Blue line shows emissions reduction potential of Scenario 1 biochar assumptions. Other scenario assumptions shown are: top left, Scenario 2; top right, Scenario 3; mid-left, Scenario 4; mid-right, Scenario 5; bottom left, Scenario 6; bottom right, Scenario 7.



Figure IV-3: Potential emission reductions for RCP8.5, relative to the carbon emission pathway of projection RCP8 (black line). Blue line shows emissions reduction potential of Scenario 1 biochar assumptions. Other scenario assumptions shown are: top left, Scenario 2; top right, Scenario 3; mid-left, Scenario 4; mid-right, Scenario 5; bottom left, Scenario 6; bottom right, Scenario 7.

V. Annexe 5: Figure licences

Figure 6-2: Percentage grid cell area under cropland in 2000. Adapted from SEDAC (2014) was licenced for use here under the Creative Commons 3.0 Attribution Licence: http://creativecommons.org/licenses/by/3.0/legalcode

Figures by Shackley et al. (2010) (in Chapter 1) are available to use under © Crown Copyright 1012

http://www.defra.gov.uk/site-info/copyright

License Number	3503691327427
License date	Nov 07, 2014
Licensed content publisher	Springer
Licensed content publication	Mitigation & Adaptation Strategies for Global Change
Licensed content title	Bio-char Sequestration in Terrestrial Ecosystems – A Review
Licensed content author	Johannes Lehmann
Licensed content date	Jan 1, 2006
Volume number	11
Issue number	2
Type of Use	Thesis/Dissertation
Portion	Figures
Author of this Springer article	No
Order reference number	None
Original figure numbers	Figure 1
Title of your thesis / dissertation	Assessing the global potential of biochars from agricultural residues to sequester carbon
Expected completion date	Sep 2015
Estimated size(pages)	300
Total	0.00 GBP

License Number	3443040762258
License date	Aug 06, 2014
Licensed content publisher	Nature Publishing Group
Licensed content publication	Nature
Licensed content title	Closing yield gaps through nutrient and water management
Licensed content author	Nathaniel D. Mueller, James S. Gerber, Matt Johnston, Deepak K. Ray, Navin Ramankutty, Jonathan A. Foley
Licensed content date	Aug 29, 2012
Volume number	490
Issue number	7419
Type of Use	reuse in a dissertation / thesis
Requestor type	academic/educational
Format	print and electronic
Portion	figures/tables/illustrations
Number of figures/tables/illustrations	1
High-res required	no
Figures	Figure 1 Average yield gaps for maize, wheat and rice. These were measured as a percentage of the attainable yield achieved circa the year 2000. Yield gap in each grid cell is calculated as an area-weighted average across the crops and is displayed on the top 98% of growing area.
Author of this NPG article	no
Your reference number	None
Title of your thesis / dissertation	Assessing the global potential of biochars from agricultural residues to sequester carbon
Expected completion date	Sep 2015
Estimated size (number of pages)	300
Total	0.00 GBP

License Number	3443050549334
License date	Aug 06, 2014
Licensed content publisher	Nature Publishing Group
Licensed content publication	Nature Communications
Licensed content title	Recent patterns of crop yield growth and stagnation
Licensed content author	Deepak K. Ray, Navin Ramankutty, Nathaniel D. Mueller, Paul C. West, Jonathan A. Foley
Licensed content date	Dec 18, 2012
Volume number	3
Type of Use	reuse in a dissertation / thesis
Requestor type	academic/educational
Format	print and electronic
Portion	figures/tables/illustrations
Number of figures/tables/illustrations	1
High-res required	no
Figures	Figure 2 Global maps of current crop yield trends. At each political unit where (a) maize, (b) rice, (c) wheat and (d) soybean crop yields were tracked globally, we determined the status of their current yield trend. The trends were divided into the six categories and colour coded.We show in the maps only those areas in the political unit where the crop was harvested.
Author of this NPG article	no
Your reference number	None
Title of your thesis / dissertation	Assessing the global potential of biochars from agricultural residues to sequester carbon
Expected completion date	Sep 2015
Estimated size (number of pages)	300
Tota	0.00 USD

VI. Bibliography

- AFIONIS, S. 2011. The European Union as a negotiator in the international climate change regime. *Int Environ Agreements*, 11, 341-360.
- AGUILAR, R., RAMIREZ, J. A., GAROTTE, G. & VAZQUEZ, M. 2002. Kinetic study of the acid hydrolysis of sugar cane bagasse. *J Food Eng*, 55, 309-318.
- AKPINAR, O., AK, O., KAVAS, A., BAKIR, U. & YILMAZ, L. 2007. Enzymatic Production of Xylooligosaccharides from Cotton Stalks. *J Agr Food Chem*, 55, 5544-5551.
- ALCAMO, J., KOK, K., BUSCH, G. & PRIESS, J. A. 2008. Searching for the Future of Land: Scnearios from the Local to Global Scale. *In:* LAMBIN, E. F. & GEIST, H. J. (eds.) *Land-Use and Land-Cover Change: Local Processes and Global Impacts.* Berlin Heidelberg: Springer.
- AMONETTE, J. E. & JOSEPH, S. 2009. Characteristics of Biochar: Microchemical Properties. *In:* LEHMANN, J. & JOSEPH, S. (eds.) *Biochar for Environmental Management: Science and Technology.* London: Earthscan.
- ANDREWS, S. S. 2006. Crop Residue Removal for Biomass Energy Production: Effects on Soils and Recommendations. USDA National Resource Convention Centre.
- ANTAL, M. J. 1985. Biomass Pyrolysis: A Review of the Literature Part 2 Lignocellulose Pyrolysis. *In:* BOER, K. W. & DUFFIE, J. A. (eds.) *Advances in Solar Energy.* Springer US.
- ANTAL, M. J. 2003. The Art, Science, and Technology of Charcoal Production. *Ind. Eng. Chem. Res.*, 42, 1619-1640.
- APPENZELLER, T. 1998. Evolution or Revolution? *Science*, 282, 1451.
- ARNAB. Overcoming constraints to the efficient utilization of agricultural by products as animal feed. Fourth Annual Workshop, 20-27 October 1987 1989 Institute of Animal Research, Mankon Station, Bamenda, Cameroun. ARNAB, Addis Ababa, Ethiopia.
- ARNELL, N. W., LIVERMORE, M., KOVATS, S., LEVY, P. E., NICHOLLS, R., PARRY, M. L. & GAFFIN, S. R. 2004. Climate and socio-economic scenarios for global-scale climate change impacts assessments: characterising the SRES storylines. *Global Environmental Change*, 14, 3-20.
- ASADULLAH, M., ZHANG, S. & LI, C. 2010. Evaluation of structural features of chars from pyrolysis of biomass of different particle sizes. *Fuel Process Technol*, 91, 877-881.
- ATKINSON, C. J., FITZGERAKI, J. D. & HIPPS, N. A. 2010 Potential mechanisms for achieving agricultural benefits from biochar application to temperate soils: A review. *Plant and Soil*, 337, 1-18.
- AUSTRALIAN GOVERNMENT. 2014. *Carbon Farming Initiative* [Online]. Available: http://www.climatechange.gov.au/reducing-carbon/carbon-farming-initiative [Accessed 10/11/2014].
- AYRILMIS, N. & BUYUKSARI, U. 2010. Utilization of olive mill sludge in manufacture of lignocellulosic/polyproylene composite. *J Mater Sci*, 45, 1336-1342.
- AZARGOHAR, R. & DALAI, A. K. 2006. Biochar as a Precursor of Activated Carbon. *Appl Biochem Biotech*, 129-132, 762-773.
- BALCI, S., DOGU, T. & YUCEL, H. 1993. Pyrolysis Kinetics of Lignocellulosic Materials. *Industrial and Engineering Chemistry Research*, 32, 2573-2579.
- BASU, P. 2010. Other Gasification-Related Properties of Biomass. *Biomass Gasification and Pyrolysis: Practical Design.* Massachusetts: Academic Press.
- BERNDES, G., HOOGWIJK, M. & VAN DEN BROEK, R. 2003. The contribution of biomass in the future global energy supply: A review of 17 studies. *Biomass Bioenerg*, 25, 1-28.
- BEURSKENS, J., FAAJI, A., FRAENKEL, P., FRIDLEIFSSON, I., LYSEN, E., MILLS, D., MOREIRA, J.
 R., NILSSON, L. J., SCHAAP, A. & SINKE, W. C. 2000. Renewable Energy Technologies.
 World Energy Assessment: Energy and the Challenge of Sustainability. New York:
 United Nations Development Programme.
BIEDERMAN, L. A. & HARPOLE, S. 2013. Biochar and its effects on plant productivity and nutrient cycling: a meta-analysis. *Glob Change Biol: Bioenergy*, **5**, 202-214.

BIOMASS ENERGY CENTRE. 2014a. *Gasification* [Online]. Available: http://www.biomassenergycentre.org.uk/ [Accessed 19/01/2014].

BIOMASS ENERGY CENTRE. 2014b. *Typical calorific values of fuels* [Online]. Available: http://www.biomassenergycentre.org.uk/ [Accessed 09/01/2014].

BRENTRUP, F., KUSTERS, J., LAMMEL, J., BARRACLOUGH, P. & KUHLMANN, H. 2004. Environmental impact assessment of agricultural production systems using the life cycle assessment (LCA) methodology II. The application to N fertilizer use in winter wheat production systems. *Eur J Agron*, 20, 265-279.

BREWER, C. E., SCHMIDT-ROHR, K., SATRIO, J. A. & BROWN, R. C. 2009. Characterization of biochar from fast pyrolysis and gasification systems. *Environmental Progress & Sustainable Energy*, 28, 386-396.

BRIDGEWATER, A. V., CZERNIK, S. & PISKORZ, J. 2001. An Overview of Fast Pyrolysis. *In:* BRIDGEWATER, A., V. (ed.) *Progress in Thermochemical Biomass Conversion: IEA Bioenergy.* Blackwell Science.

BRIDGEWATER, A. V., MEIER, D. & RADLEIN, D. 1999. An overview of fast pyrolysis of biomass. *Organic Geochemistry*, 30, 1479-1493.

BRIDGWATER, A. V. 2003. Renewable fuels and chemicals by thermal processing of biomass. *Chem Eng J*, 91, 87-102.

BRITISH BIOCHAR FOUNDATION 2013. Biochar Quality Mandate (BQM) v. 1.0. British Biochar Foundation.

BRITISH BIOCHAR FOUNDATION. 2014. *Biochar FAQs* [Online]. Available: http://www.britishbiocharfoundation.org/?page_id=52#eight [Accessed 02/12/2014].

- BROWN, R. 2009. Biochar Production Technology. *In:* LEHMANN, J. & JOSEPH, S. (eds.) *Biochar for Environmental Management: Science and Technology.* Oxford: Earthscan.
- BROWNSORT, P. 2009. Biomass Pyrolysis Processes: Performance Parameters and their Influence on Biochar System Benefits. University of Edinburgh.
- BRUNAUER, S., EMMETT, P. H. & TELLER, E. 1938. Adsorption of Gases in Multimolecular Layers. *J Am Chem Soc*, 60, 309-319.
- BRUUN, E. W., HAUGGAARD-NIELSEN, H., IBRAHIM, N., EGSGAARD, H., AMBUS, P., JENSEN, P. A. & DAM-JOHANSEN, K. 2011. Influence of fast pyrolysis temperature on biochar labile fraction and short-term carbon loss in a loamy soil. *Biomass & Bioenerg*, 35, 1182-1189.
- BTG. 2011. *BTG's Pyrolysis Technology* [Online]. Available: http://www.btg.cz/en/aboutbiomass/fast-pyrolysis/btg-s-pyrolysis-technology [Accessed 21/10/2011].

CAPAREDA, S. C. 2011. Biomass Energy Conversion. *In:* NAYERIPOUR, M. (ed.) *Sustainable Growth and Applications in Renew Energ Sources.* Rijeka, Croatia: InTech.

- CERAM. 2013. *BET Surface Area Analysis & BJH Pore Size and Volume Analysis* [Online]. Available: http://www.ceram.com/testing-analysis/techniques/brunauer-emmettteller-surface-area-analysis-barrett-joyner-halenda-pore-size-and-volume-analysis [Accessed 25/06/2013].
- CHAN, K. Y., VAN ZWEITEN, L., MESZAROS, I., DOWNIE, A. & JOSEPH, S. Assessing the agronomic values of contrasting char materials on Australian hardsetting soil. Proceedings of the Conference of the International Agrichar Initiative, 2007 Terrigal, NSW, Australia.
- CHAN, K. Y., VAN ZWEITEN, L., MESZAROS, I., DOWNIE, A. & JOSEPH, S. 2008. Using poultry litter biochars as soil amendments. *Soil Research*, 46, 437-444.
- CHAN, K. Y. & XU, Z. 2009. Biochar: Nutrient Properties and Their Enhancement. *In:* LEHMANN, J. & JOSEPH, S. (eds.) *Biochar for Environmental Mangement: Science and Technology.* London: Easthscan.
- CHEAH, S., MALONE, S. C. & FEIK, C. J. 2014. Speciation of Sulphur in Biochar Produced fomr Pyrlysis and Gasification of Oak and Corn Stover. *Environ Sci Technol*, In Press.

- CHENG, C. H., LEHMANN, J., THIES, J. E. & BURTON, S. D. 2008. Stability of black carbon in soils across a climatic gradient. *Journal of Geophysical Research: Biogeosciences*, 113.
- CLARKE, L., EDMONDS, J., JACOBY, H., PITCHER, H., REILLY, J. & RICHELS, R. 2007. CCSP synthesis and assessment product 2.1, Part A: scenarios of greenhouse gas emissions and atmospheric concentrations. Washington, DC.
- COLLINS, M., KNUTTI, R., ARBLASTER, J., DUFRESNE, J.-L., FICHEFET, T., FRIEDLINGSTEIN, P., GAO, X., GUTOWSKI, W. J., JOHNS, T., KRINNER, G., SHONGWE, M., TEBALDI, C., WEAVER, A. J. & WEHNER, M. 2013. Long-term Climate Change: Projections, Commitments and Irreversibility. *In:* STOCKER, T. F., QIN, D., PLATTNER, G. K., TIGNOR, M., ALLEN, S. K., BOSCHUNG, J., NAUELS, A., XIA, Y., BEX, V. & MIDGLEY, P. M. (eds.) *Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change.* Cambridge, United Kingdom and New York, NY, USA.
- CORNELL UNIVERSITY. 2006. Amazonian Terra Preta Can Transform Poor Soils into Fertile [Online]. ScienceDaily. Available: http://www.sciencedaily.com/releases/2006/03/060301090431.htm [Accessed 01/01/2012].
- CROSS, A. & SOHI, S. P. 2011. The priming potential of biochar products in relation to labile carbon contents and soil organic matter status. *Soil Biology & Biochemistry*, 43, 2127-2143.
- CROWN 1993. Clean Air Act. In: GOVERNMENT, H. (ed.) 1993 c.11.
- DAS, K. C., GARCIA-PEREZ, M., BIBENS, B. & MELEAR, N. 2008. Slow pyrolysis of poultry litter and pine woody biomass: Impact of chars and bio-oils on microbial growth. *Journal of Environmental Science and Health, A*, 714-724.
- DAUD, W. M. A. W. & ALI, W. S. W. 2004. Comparison on pore development of activated carbon produced from palm shell and coconut shell. *Bioresource Technol*, 93, 63-69.
- DE LEEUW, P. N. 1997. Crop residues in tropical Africa: trends in supply, demand and use. *In:* RENARD, C. (ed.) *Crop Residues in Sustainable Mixed Crop/Livestock Farming Systems.* Oxon, UK and NY, USA: CAB International.
- DE MEESTER, S., CALLEWAERT, C., DE MOL, E., VAN LANGENHOVE, H. & DEWULF, J. 2011. The resource footprint of biobased products: a key issue in the sustainable development of biorefineries. *Biofuels Bioproducts & Biorefining-Biofpr*, 5, 570-580.
- DECC. 2011. Energy and Emissions Projections [Online]. Available: http://www.decc.gov.uk/en/content/cms/about/ec_social_res/analytic_projs/en_e mis_projs/en_emis_projs.aspx [Accessed 01/11/2011].
- DEMIRBAS, A. 2001. Biomass resource facilities and biomass conversion processing for fuels and chemicals. *Energ Convers Manage*, 42, 1357-1378.
- DEMIRBAS, A. 2004a. Determination of calorific values of bio-chars and pyro-oils from pyrolysis of beech trunkbarks. *J Anal Appl Pyrol*, 72, 215-219.
- DEMIRBAS, A. 2004b. Effects of temperature and particle size on bio-char yield from pyrolysis of agricultural residues. *J Anal Appl Pyrol*, 72, 243-248.
- DEMIRBAS, A. 2006. Effect of temperature on pyrolysis products from four nut shells. *J Anal Appl Pyrol*, 76, 285-289.
- DEMIRBAS, A. & ARIN, G. 2002. An Overview of Biomass Pyrolysis. *Energy Sources*, 24, 471-482.
- DEMIRBAS, M. F. 2009. Biorefineries for biofuel upgrading: A critical review. *Applied Energy*, 86, 151-161.
- DEVENDRA, C. Crop residues for feeding animals in Asia: technology development and adoptnio in crop/livestock systems. *In:* RENARD, C., ed. Crop Residues in Sustainable Mixed Crop/Livestock Farming Systems, 1997. International Crops Research Institute, 241-268.
- DIBLASI, C. 1996. Kinetics and heat transfer control in the slow and flash pyrolysis of solids. *Industrial and Engineering Chemical Research*, 35, 37-46.

- DOMINGUEZ, A., MENDEZ, J. A., FERNANDEZ, Y., PIS, J. J., VALENTE NABAIS, J. M., CARROTT, P. J. M. & RIBEIRO CARROTT, M. M. L. 2007. Conventional and microwave induced pyrolysis of coffee hulls for the production of a hydrogen rich fuel gas. *Jounal of Analytical and Applied Pyrolysis*, 79, 128-135.
- DOWNIE, A., CROSKY, A. & MUNROE, P. 2009. Physical Properties of Biochar. *In:* LEHMANN, J. & JOSEPH, S. (eds.) *Biochar for Environmental Management: Science and Technology.* London: Easrthscan.
- DOWNIE, A. E., VAN ZWEITEN, L., SMERNIK, R. J., MORRIS, S. & MUNROE, P. R. 2011. Terra Preta Australis: Reassessing the carbon storage capacity of temperate soils. *Agr Ecosyst Environ*, 140, 137-147.
- ENDERS, A., HANLEY, K., WHITMAN, T., JOSEPH, S. & LEHMANN, J. 2012. Characterisation of biochars to evaluate recalitrance and agronomic performance. *Bioresource Technol*, 114, 644-635.
- ERICSSON, K. & NILSSON, L. J. 2006. Assessment of the potential biomass supply in Europe using a resource-focussed approach. *Biomass Bioenerg*, **30**, 1-15.
- EUROPEAN BIOCHAR FOUNDATION 2013. The European Biochar Certificate. Arbaz, Switzerland: European Biochar Foundation [EBC].
- EUROPEAN BIOFUELS TECHNOLOGY PLATFORM. 2014. *Agricultural residues as feedstocks for biofuels production* [Online]. Available: http://www.biofuelstp.eu/residues.html [Accessed 05/03/2014].
- EUROPEAN COMMISSION. 2011a. *The Common Agricultural Policy after 2013* [Online]. Available: http://ec.europa.eu/agriculture/cap-post-2013/index_en.htm [Accessed 29/12/2011].
- EUROPEAN COMMISSION. 2011b. *Waste Framework Directive: The European List of Waste* [Online]. Available: http://ec.europa.eu/environment/waste/framework/list.htm [Accessed 26/01/2012].
- FAGBEMI, L., KHEZAMI, L. & CAPART, R. 2001. Pyrolysis products from different biomasses: application to the thermal cracking of tar. *Applied Energy*, 69, 293-306.
- FAO 1998. Report of the Fifth External Programme and Management Review of the International Rice Research Institute. FAO.
- FAO 2000. Bridging the Rice Yield Gap in the Asia-Pacific Region. *In:* PAPADEMETRIOU, M. K., DENT, F. J. & HERATH, E. M. (eds.). Bangkok, Thailand: Regional Office for Asia and the Pacific, FAO.
- FAO 2003. World Agriculture: Towards 2015/2030, An FAO Perspective. *In:* BRUINSMA, J. (ed.). London.
- FAO. 2013a. *FAOSTAT* [Online]. Available: http://faostat3.fao.org/home/index.html#HOME [Accessed 12/03/2013].
- FAO. 2013b. FAOSTAT: Glossary [Online]. Available: http://faostat.fao.org/site/379/DesktopDefault.aspx?PageID=379 [Accessed 01/03/2013].
- FISCHER, G., TEIXEIRA, E., HIZSNYIK, E. T. & VAN VELTHUIZEN, H. 2008. Land use dynamics and sugarcane production. *In:* ZUURBIER, P. J. P. & VAN DE VOOREN, J. (eds.) *Sugarcane Ethanol: Contributions to Climate Change Mitigation and the Environment.* Wageningen, Germany: Wageningen Academic Publishers.
- FLATO, G., MAROTZKE, J., ABIODUN, B., BRACCONOT, P., CHOU, S. C., COLLINS, W., COX, P., DRIOUECH, F., EMORI, S., EYRING, V., FOREST, C., GLECKLER, P., GUILYARDI, E., JAKOB, C., KATTSOV, V., REASON, C. & RUMMUKAINEN, M. 2013. Evaluation of Climate Models. *In:* STOCKER, T. F., QIN, D., PLATTNER, G. K., TIGNOR, M., ALLEN, S. K., BOSCHUNG, J. & NAUELS, A. (eds.) *Climate Change 2013: The Physical Science Basis. Contribution of Working Gorup I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change.* Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- FOEREID, B., LEHMANN, J. & MAJOR, J. 2011. Modeling black carbon degradation and movement in soil. *Plant Soil*, 345, 223-236.

- FOREST STEWARDSHIP COUNCIL. 2014. FSC Standards [Online]. Available: https://ic.fsc.org/standards.340.htm [Accessed 19/12/2014.
- FREDDO, A., CAI, C. & REID, B. J. 2012. Environmental contextualisation of potential toxic elements and poycyclic aromatic hydrocarbons in biochar. *Environmental Pollution*, 171, 18-24.
- GAUNT, J. 2012. *Production* [Online]. Available:
 - http://biocharfarms.org/biochar_production_energy/ [Accessed 22/01/2012].
- GAUNT, J. & LEHMANN, J. 2008. Energy balance and emissions associated with biochar sequestration and pyrolysis bioenergy production. *Environmental Science Technology*, 42, 4152-4158.
- GENESIO, L., MIGLIETTA, F., LUGATO, E., BARONTI, S., PIERI, M. & VACCARI, F. P. 2012. Surface albedo following biochar application in durum wheat. *Environ. Res. Lett.*, 7, 1-8.
- GLASER, B., HAUMAIER, L., GUGGENBERGER, G. & ZECH, W. 2001. The 'Terra Preta' Phenomenon: a model for sustainable agriculture in the humid tropics. *Naturwissenschaften*, 88, 37-41.
- GLASER, B., PARR, M., BRAUN, C. & KOPOLO, G. 2009. Biochar is carbon negative. *Nature Geoscience*, 2.
- GOKSU, E. L., KARAMANLIOGLU, M., BAKIR, U., YILMAZ, L. & YILMAZER, U. 2007. Production and Characterisation of Films from Cotton Stalk Xylan. *J Agr Food Chem*, 55, 10685-10691.
- GOLD STANDARD. 2014. *The Gold Standard* [Online]. Available: http://www.goldstandard.org/ [Accessed 10/11/2014].
- GRABER, E. R., TSECHANSKY, L., GERSTL, Z. & LEW, B. 2012. High surface area biochar negatively impacts herbicide efficacy *Plant and Soil*, 353, 95-106.
- HAMER, U., MARSCHNER, B., BRODOWSKI, S. & AMELUNG, W. 2004. Interactive priming of black carbon and glucose mineralisation. *Organic Geochemistry*, 35, 823-830.
- HAMMOND, J., SHACKLEY, S., SOHI, S. & BROWNSORT, P. 2011. Prospective life cycle carbon abatement for pyrolysis biochar systems in the UK. *Energy Policy*, 39, 2646-2655.
- HARTMANN, D. L., KLEIN TANK, A. M. G., RUSTICUCCI, M., ALEXANDER, L. V., BRONNIMANN, S., CHARABI, Y., DENTENER, F. J., DLUGOKENCKY, E. J., EASTERLING, D. R., KAPLAN, A., SODEN, B. J., THORNE, P. W., WILD, M. & ZHAI, P. M. 2013. Observations:
 Atmosphere and Surface. . *In:* STOCKER, T. F., QIN, D., PLATTNER, G. K., TIGNOR, M., ALLEN, S. K., BOSCHUNG, J., NAUELS, A., XIA, Y., BEX, V. & MIDGLEY, P. M. (eds.) *Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change.*Cambridge, United Kingdom and New York, NY, USA.: Cambridge University Press.
- HARVEY, O. R., KUO, L. J., ZIMMERMAN, A. R., LOUCHOUARN, P., AMONETTE, J. E. & HERBERT, B. E. 2012. An Index-Based Approach to Assessing Recalcitrance and Soil Carbon Sequestration Potential of Engineered Black Carbons (Biochars). *Environ Sci Technol*, 46, 1415-1421.
- HARYANTO, A., FERNANDO, S. D., PORDESIMO, L. O. & ADHIKARI, S. 2009. Upgrading of syngas derived from gasification: A thermodynamic analysis. *Biomass & Bioenerg*, 33, 882-889.
- HILSCHER, A., HEISTER, K., SIEWERT, C. & KNICKER, H. 2009. Mineralisation and strucutral changes during the intial phase of microbial degradation of pyrogenic plant residues in soil. *Organic Geochemistry*, 40, 332-342.
- HOOGWIJK, M., A FAAIJ, B DEVRIES & W TURKENBURG 2009. Exploration of regional and global cost-supply curves of biomass energy from short-rotation crops at abandoned cropland and rest land under four IPCC SRES land use scenarios. *Biomass and Bioenergy*, 33, 26-43.
- HOOGWIJK, M., FAAIJ, A., EICKHOUT, B., DE VRIES, B. & TURKENBURG, W. 2005. Potential of biomass energy out to 2100, for four IPCC SRES land-use scenarios. *Biomass and Bioenergy*, 29, 225-257.

- HOOGWIJK, M., FAAIJ, A., VAN DEN BROEK, R., BERNDES, G., GIELEN, D. & TURKENBURG, W. 2003. Exploration of the ranges of the global potential of biomass for energy. *Biomass Bioenerg*, 25, 119-133.
- HOSHI, T. Growth promotion of tea trees by putting bamboo charcoal in soil. Proceedings of 2001 International Conference on O-cha (tea) Culture and Science, 2001 Tokyo, Japan. 147-150.
- HOSOYA, T., KAWAMOTO, H. & SAKA, S. 2008. Pyrolysis gasification reactivities of primary tar and char fractions from cellulose and lignin as studied with a closed ampoule reactor. *J Anal Appl Pyrol*, 83, 71-77.
- HOSSAIN, M. K., STREZOV, V., CHAN, K. Y., ZIOLKOWSKI, A. & NELSON, P. F. 2011. Influence of pyrolysis temperature on production and nutrient properties of wastewater sludge biochar. *J. Environ. Manage.*, 92, 223-228.
- HUEHN, M. 1993. Harvest index versus grain/straw-ratio. Theoretical comments and experimental results on the comparison of variation. *Euphytica*, 68, 27-32.
- HURTT, G. C., CHINI, L. P., FROLKING, S., BETTS, R., FEDEMMA, J., FISCHER, G., GOLDEWIJK, K. K., HIBBARD, K., KINOSHITA, T., RIAHI, K., SHEVLIAKOVA, E., SMITH, S., STEHFEST, E., THOMSON, A., THORNTON, P., VAN VUUREN, D. & PING WANG, Y. 2009. Harmonisation of global land-use scenarios for the period 1500-2100 for IPCC-AR5. *iLEAPS*, 7, 6-8.
- IEA BIOENERGY. 2011. *Pyrolysis- Reactors* [Online]. Available: http://www.pyne.co.uk/?_id=69 [Accessed 11/10/11].
- IIASA. 2009. *RCP Database (version 2.0)* [Online]. Available: http://www.iiasa.ac.at/web-apps/tnt/RcpDb/dsd?Action=htmlpage&page=welcome [Accessed 29/03/2012].
- IMHOFF, M. L., BOUNOUA, L., RICKETTS, T., LOUCKS, C., HARRISS, R. & LAWRENCE, W. T. 2004. Global patterns in human consumption of net primary production. *Nature*, 429, 870-873.
- IN-YONG, E., KWANG-HO, K., JAE-YOUNG, K., SOO-MIN, L., HWAN-MYUNG, Y., IN-GYU, C. & JOON-WEON, C. 2011. Characterization of primary thermal degradation features of lignocellulosic biomass after removal of inorganic metals by diverse solvents. *Bioresource Technol*, 102, 3437-3444.
- INTERNATIONAL BIOCHAR INITIATIVE. 2010. *Biochar Recalcitrance in Soil* [Online]. Available: http://www.biochar-international.org/sites/default/files/IBI-RSrecalcitrance-2-Apr-2010..pdf [Accessed 10/11/2011].
- INTERNATIONAL BIOCHAR INITIATIVE. 2011a. *Biochar Production Units* [Online]. Available: http://www.biochar-international.org/technology/production [Accessed 03/11/2011].
- INTERNATIONAL BIOCHAR INITIATIVE. 2011b. *Feedstocks* [Online]. Available: http://www.biochar-international.org/technology/feedstocks [Accessed 11/10/11].
- INTERNATIONAL BIOCHAR INITIATIVE. 2012a. *Biochar and Sustainability* [Online]. Available: http://www.biochar-international.org/sustainability [Accessed 10/04/2012].
- INTERNATIONAL BIOCHAR INITIATIVE. 2012b. *Climate Change and Carbon Sequestration: How Can Biochar Be Carbon Negative?* [Online]. Available: http://www.biocharinternational.org/biochar/carbon [Accessed 10/04/2012].
- INTERNATIONAL BIOCHAR INITIATIVE. 2012c. Developing Guidelines for Specifications of Biochars [Online]. Available: http://www.biochar-
- international.org/characterizationstandard [Accessed 10/03/2012]. INTERNATIONAL BIOCHAR INITIATIVE. 2012d. *Environmental Benefits of Biochar* [Online]. Available: http://www.biochar-international.org/biochar/benefits [Accessed 14/03/2012].
- INTERNATIONAL BIOCHAR INITIATIVE. 2012e. *IBI Biochar Standards and Testing Guidelines* [Online]. Available: http://www.biochar-international.org/characterizationstandard [Accessed 15/07/2014].

INTERNATIONAL BIOCHAR INITIATIVE. 2014. *Biochar Use in Soils* [Online]. Available: http://www.biochar-international.org/biochar/soils [Accessed 07/11/2014].

INTERNATIONAL ENERGY AGENCY 2010. *World Energy Outlook: Executive Summary*, IEA. IPCC 2000. IPCC Special Report on Emission Scenarios: Summary for Policymakers.

IPCC 2007. Climate Change 2007: Synthesis Report. Contribution of Working Groups I, II and III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. *In:* PACHAURI, R. K. & REISINGER, A. (eds.). Geneva, Switzerland.

IPCC 2013. Climate Change 2013: The Physical Science Basis, Contribution of Working Group I to the Fifth Assessment Report of the Intergovenmental Panel on Climate Change, Cambridge, United Kingdom and New York, NY, USA, Cambridge University Press.

- IPCC. 2014a. *Fifth Assessment Report (AR5)* [Online]. Available: http://www.ipcc.ch/report/ar5/ [Accessed 02/12/2014].
- IPCC 2014b. Summary for Policymakers. In: EDENHOFER, O., PICHS-MADRUGA, R., SOKONA, Y., FARAHANI, E., KADNER, S., SEYBOTH, K., ADLER, A., BAUM, I., BRUNNER, S., EICKEMEIER, P., KRIEMANN, B., SAVOLAINEN, J., SCHLOMER, S., VON STECHOW, C., ZWICKEL, T. & MINX, J. C. (eds.) Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Campridge, United Kingdom and New York, NY, USA.: Cambridge University Press.
- IPCC 2014c. Summary for Policymakers. In: FIELD, C., BARROS, V. R., DOKKEN, D. J., MACH, K. J., MASTRANDREA, M. D., BILIR, T. E., CHATTERJEE, M., EBI, K. L., ESTRADA, Y. O., GENOVA, R. C., GIRMA, B., KISSEL, E. S., LEVY, A. N., MACCRACKEN, S., MASTRANDREA, P. R. & WHITE, L. L. (eds.) Climate Change 2014: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom, New York, NY, USA: Cambridge University Press.
- JAUHIAINEN, J., MARTIN-GULLON, I., CONESA, J. A. & FONT, R. 2005. Emissions from pyrolysis and combustion of olive oil solid waste. *J Anal Appl Pyrol*, 74, 512-517.
- JIANG, L., HU, S., SUN, L., SU, S., ZXU, K., HE, L. & XIANG, J. 2013. Influence of different demineralization treatments on physicochemical structure and thermal degradation of biomass. *Bioresource Technol*, 146, 254-260.
- JOINT GLOBAL CHANGE RESEARCH INSTITUTE. 2013. *GCAM RCP Scenario Results* [Online]. Available: http://www.globalchange.umd.edu/gcamrcp/ [Accessed 12/03/2013].
- JOLLI, D. & GILJUM, S. 2005. Unused biomass extraction in agriculture, forestry and fishery. Vienna: Sustainable Europe Research Institute.
- JOSEPH, S., PEACOCKE, C., LEHMANN, J. & MUNROE, P. 2009 Developing a Biochar Classification and Test Methods. *In:* LEHMANN, J. & JOSEPH, S. (eds.) *Biochar for Environmental Management: Science and Technology.* London: Earthscan.
- JUSTIZ-SMITH, N. G., JUNIOR VIRGO, G. & BUCHANAN, V. E. 2008. Potential of Jamaican banana, coconut coir and bagasse fibres as composite materials. *Materials Characterisation*, 59, 1273-1278.
- KADAM, K. L., FORREST, L. H. & JACOBSON, W. A. 2000. Rice straw as a lignocellulosic resource: collection, processing, transportation and environmental aspects. *Biomass Bioenerg*, 18, 369-389.
- KANG, S., XIAO, L., MENG, L., ZHANG, X. & SUN, R. 2012. Isolation and Structural Characterisation of Lignin from Cotton Stalk Treated in an Ammonia Hydrothermal System. *Int J Mol Sci*, 13, 15209-15226.
- KAPARJU, P., SERRANO, M., THOMSEN, A. B., KONGJAN, P. & ANGELIDAKI, I. 2009. Bioethanol, biohydrogen and biogas production from wheat straw in a biorefinery concept. *Bioresource Technol*, 100, 2562-2568.
- KEILUWEIT, M., NICO, P. S., JOHNSON, M. G. & KLEBER, M. 2010. Dynamic Molecular Structure of Plant Biomass-Derived Black Carbon (Biochar). *Environ Sci Technol*, 44, 1247-1253.

- KHEDARI, J., NANKONGNAB, N., HIRUNLABH, J. & TEEKASAP, S. 2004. New low-cost insulation particleboards from mixture of durian peel and coconut coir. *Building and Environment*, 39, 59-65.
- KIM, K. H., KIM, J. Y., CHO, T. S. & CHOI, J. W. 2012. Influence of pyrolysis temperature on pyhsicochemical properties of biochar obtained from the fast pyrolysis of pitch-pine (Pinus rigida). *Bioresource Technol*, 118, 158-162.
- KIM, S. & DALE, B. E. 2004. Global potential production from wasted crops and crop residues. *Biomass Bioenerg*, 26, 361-375.
- KIMBALL, B. A. 1983. Carbon Dioxide and Agricultural Yield: An Assemblage and Analysis of 430 Prior Observations. *Agronomy Journal*, 75, 779-788.
- KIRKBY, E. 2012. Introduction, Definition and Classification of Nutrients. *In:* MARSCHENER, P. (ed.) *Marschner's Mineral Nutrition of Higher Plants.* San Diego, USA: Academic Press.
- KIRTMAN, B., POWER, S. B., ADEDOYIN, J. A., BOER, G. J., BOJARIU, R., CAMILONI, I., DOBLAS-REYES, F. J., FIORE, A. M., KIMOTO, M., MEEHL, G. A., PRATHER, M., SARR, A., SCHAR, C., SUTTON, R., VAN OLDENBORGH, G. J., VECCHI, G. & WANG, H. J. 2013. Near-term Climate Change: Projections and Predictability. *In:* STOCKER, T. F., QIN, D., PLATTNER, G. K., TIGNOR, M., ALLEN, S. K., BOSCHUNG, J., NAUELS, A., XIA, Y., BEX, V. & MIDGLEY, P. M. (eds.) *Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovermental Panel on Climate Change.* Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- KLEIN GOLDEWIJK, K., BEUSEN, A., DE VOS, M. & VAN DRECHT, G. 2011. The HYDE 3.1 spatially explicit database of human induced land use change over the past 12,000 years. *Global Ecology and Biogeography*, 20, 73-86.
- KLOPPFER, W. 1997. Life cycle assessment: From the beginning to the current state. *Environmental Science and Pollution Research International*, 4, 223-228.
- KLOSE, W. & WIEST, W. 1999. Experiments and mathematical modeling of maize pyrolysis in a rotary kiln. *Fuel*, 78, 65-72.
- KOOKANA, R. S., SARMAH, A. K., VAN ZWEITEN, L., KRULL, E. & SINGH, B. 2011. Biochar Application to Soil: Agronomic and Environmental Benefits and Unintended Consequesnces. *In:* SPARKS, D. L. (ed.) *Advances in Agronomy.* San Diego: Elsevier.
- KRISTENSEN, J. B., BORJESSON, J., BRUUN, M. H., TJERNELD, F. & JORGENSEN, H. 2007. Use of surface active additives in enzymatic hydrolysis of wheat straw lignocellulose. *Enzyme Microb Tech*, 40, 888-895.
- KROEZE, C., MOSIER, A., NEVISON, C., OENEMA, O., SEITZINGER, S. & VAN CLEEMPUT, O. 1996. Agriculture. *Revised 1996 IPCC Guidelines for National Greenhouse Gas Inventories: Reference Manual.* Intergovernmental Panel on Climate Change.
- KRULL, E., LEHMANN, J., SKJEMSTAD, J. O., BALDOCK, J. A. & SPOUCER, L. 2008. The global extent of black C in soils: is it everywhere? *In:* SCHRODER, H. G. (ed.) *Grasslands: Ecology, Management and Restoration.*: Nova Science Publishers, Inc.
- KRULL, E. S., BALDOCK, J. A., SKJEMSTAD, J. O. & SMERNIK, R. J. 2009. Characteristics of Biochar: Organo-chemical Properties. *In:* LEHMANN, J. & JOSEPH, S. (eds.) *Biochar for Environmental Management: Science and Technology.* Sterling, USA and London, UK: Earthscan.
- KUWAGAKI, H. & TAMURA, K. 1990. Aptitude of wood charcoal to a soil improvement and other non-fuel use. *In:* DIXON, R. (ed.) *Mitigation and Adaptation Strategies for Global Change.* New York: Springer.
- KUZYAKOV, Y., SUBBOTINA, I., CHEN, H., BOGOMOLOVA, I. & XU, X. 2009. Black carbon decomposition and incorporation into soil microbial biomass estimated by 14C labeling. *Soil Biology and Biochemistry*, 41, 210-219.
- KYLE, P., MULLER, C., CALVIN, K. & THOMSON, A. 2014. Meeting the radiative forcing targets of the representative concentration pathways in a world with agricultural climate impacts. *Earth's Future*, *2*, 83-98.

- LAL, R. 2005. World crop residues production and implications of its use as a biofuel. *Environ Int*, 31, 575-584.
- LANGE, J. P. 2007. Lignocellulose conversion: an introduction to chemistry, process and economics. *Biofuels, Bioproducts and Biorefining,* **1**, 39-48.
- LE QUERE, C., RAUPACH, M. R., CANADELL, J. G. & MARLAND, G. 2009. Trends in the sources and sinks of carbon dioxide. *nature geoscience*, **2**, 831-839.
- LEAKEY, A. D. B. 2009. Rising atmospheric carbon dioxide concentration and the future of C4 crops for food and fuel. *Proceedings of the Royal Society B: Biological Sciences*, 276, 2333-2343.
- LEE, J. F., HAWKINS, B., DAY, D. M. & REICOSKY, D. C. 2010. Sustainability: The capacity of smokeless biomass pyrolysis for energy production, global carbon capture and sequestration. *Energy & Environmental Science*, **3**, 1695-1705.
- LEE, Y., EUM, P. R. B., RYU, C., PARK, Y. K., JUNG, J. H. & HYUN, S. 2013. Characteristics of biochar produced from slow pyrolysis of Geodae-Uksae 1. *Bioresource Technol*, 130, 345-350.
- LEHMANN, J. 2007. A handful of carbon. Nature, 447, 143-144.
- LEHMANN, J., CZIMCZIK, C., LAIRD, D. & SOHI, S. 2009. Stability of Biochar in Soil. *In:* LEHMANN, J. & JOSEPH, S. (eds.) *Biochar for Environmental Management: Science and Technology.* London: Earthscan.
- LEHMANN, J., GAUNT, J. & RONDON, M. 2006. Bio-char Sequestration in Terrestrial Ecosystems – A Review. *Mitigation and Adaptation Strategies for Global Change*, 11, 395-419.
- LEHMANN, J. & JOSEPH, S. 2009. *Biochar for Environmental Management: Science and Technology*, London, Earthscan.
- LEHMANN, J., KERN, D. C., GLASER, B. & WOODS, W. I. 2003. *Amazonian Dark Earths: Origin, Properties, Management,* The Netherlands, Kluwer Academic Publishers.
- LEHMANN, J. & RONDON, M. 2006. Bio-char soil management on highly weathered soils in the humid tropics *In:* UPHOFF, N. (ed.) *Biological Approaches to Sustainable Soil Systems.* Boca Raton, FL.: CRC Press.
- LENTON, T. M. & VAUGHAN, N. E. 2009. The radiative forcing potential of different climate geoengineering options. *Atmos. Chem. Phys.*, 9, 5539-5561.
- LI, J., BAI, J. & OVEREND, R. 1999. Assessment of Biomass Resource Availability in China, Beijing, China Environmental Science Press.
- LIESCH, A. M., WEYERS, S. L., GASKIN, J. W. & DAS, K. C. 2010. Impact of Two Different Biochars on Earthwork Growth and Survival. *Annals of Environmental Science*, 4, 1-9.
- LINDSTROM, M. J. 1986. Effects of residue harvesting on water runoff, soil erosion and nutrient loss. *Agr Ecosyst Environ*, 16, 103-112.
- LIU, H., JIANG, G. M., ZHUANG, H. Y. & WANG, K. J. 2008. Distribution, utilization structure and potential of biomass resources in rural China: With special references of crop residues. *Renewable and Sustainable Energy Reviews*, 12, 1402-1418.
- LOBELL, D. B. & FIELD, C. B. 2008. Estimation of the carbon dioxide (CO2) fertilization effect using growth rate anomalies of CO2 and crop yields since 1961. *Global Change Biology*, 14, 39-45.
- LORENZ, A. J., GUSTAFSON, T. J., COORS, J. G. & DE LEON, N. 2010. Breeding Maize for a Bioeconomy: A Literature Survey Examining Harvest Index and Stover Yield and Their Relationship to Grain Yield. *Crop Science*, 50, 1-12.
- MAATHUIS, F. J. M. & DIATLOFF, E. 2013. Roles and Functions of Plant Mineral Nutrients. *Method Mol Biol*, 953, 1-21.
- MAJOR, J., RONDON, M., MOLINA, D., RIHA, S. J. & LEHMANN, J. 2010. Maize yield and nutrition during 4 years after biochar application to a Colombian savanna oxisol. *Plant and Soil*, 333, 117-128
- MAJOR, J., STEINER, C., DOWNIE, A. & LEHMANN, J. 2009. Biochar Effects on Nutrient Leaching. *In:* LEHMANN, J. & JOSEPH, S. (eds.) *Biochar for Environmental Management: Science and Technology.* London: Earthscan.

- MANTE, O. D., AGBLEVOR, F. A., OYAMA, S. T. & MCCLUNG, R. 2012. The influence of recycling non-condensable gases in the fractional catalytic pyrolysis of biomass. *Bioresource Technol*, 111, 482-490.
- MANYA, J. J. 2012. Pyrolysis for Biochar Purposes: A Review to Establish Current Knowledge Gaps and Research Needs. *Environ Sci Technol*, 46, 7939-7954.
- MARRIS, E. 2006. Black is the new green. Nature, 442, 624-626.
- MASUI, T., MATSUMOTO, K., HIJIOKA, Y., KINOSHITA, T., NOZAWA, T., ISHIWATARI, S., KATO, E., SHUKLA, P. R., YAMAGATA, Y. & KAINUMA, M. 2011. An emissions pathway for stabilization at 6 Wm-2 radiative forcing. *Climatic Change*, 109, 59-76.
- MATOVIC, D. 2011. Biochar as a viable carbon sequestration option: Global and Canadian perspective. *Energy*, 36, 2011-2016.
- MCCARL, B. A., PEACOCKE, C., CHRISMAN, R., KUNG, C. C. & SANDS, R. D. 2009. Economics of Biochar Production, Utilization and Greenhouse Gas Offsets. *In:* LEHMANN, J. & JOSPEH, S. (eds.) *Biochar for Environmental Management: Science and Technology.* Oxford: Earthscan.
- MCHENRY, M. P. 2009. Agricultural bio-char production, renewable energy generation and farm carbon sequestration in Western Australia: Certainty, uncertainty and risk. *Agr Ecosyst Environ*, 129, 1-7.
- MCKENDRY, P. 2002. Energy production from biomass (part 1): overview of biomass. *Bioresource Technol*, 83, 37-46.
- MEYER, S., BRIGHT, R. M., FISCHER, D., SCHULZ, H. & GLASER, B. 2012. Albedo impact on the Suitability of Biochar Systems to Mitigate Global Warming. *Environmental Science and Technology*, 46, 12726-12734.
- MILLENNIUM ECOSYSTEM ASSESSMENT. 2005. *Millennium Ecosystem Assessment* [Online]. Available: http://www.unep.org/maweb/en/index.aspx [Accessed 22/03/2013].
- MOHAN, D., RAJPUT, S., SINGH, V. K., STEELE, P. H. & PITTMAN JR, C. U. 2011. Modeling and evaluation of chromium remediation from water using low cost bio-char, a green adsorbent. *J Hazard Mater*, 188, 319-333.
- MUELLER, N. D., GERBER, J. S., JOHNSTON, M., RAY, D. K., RAMANKUTTY, N. & FOLEY, J. A. 2012. Closing yield gaps through nutrient and water management. *Nature*, 490, 254-257.
- NAKICENOVIC, N., ALCAMO, J., DAVIS, G., DE VRIES, B., FENHANN, J., GAFFIN, S., GREGORY, K., GRUBLER, A., YONG JUNG, T., KRAM, T., LEBRE LA ROVERE, E., MICHAELIS, L., MON, S., MORITA, T., PEPPER, W., PITCHER, H., PRICE, L., RIAHI, K., ROEHRL, A., ROGNER, H.-H., SANKOVSKI, A., SCHLESINGER, M., SHUKLA, P., SMITH, S., SWART, R., VAN ROOIJEN, S., VICTOR, N. & DADI, Z. 2000. Special Report on Emission Scenarios. *In:* NAKICENIVOC, N. & SWART, R. (eds.). Intergovernmental Panel on Climate Change.
- NATIONAL GRID. 2013. *Calorific Value* [Online]. Available: https://www.nationalgrid.com/uk/Gas/Data/misc/reports/description/ [Accessed 29/04/2013].
- NATIONAL SCIENCE FOUNDATION. 2014. *Lignocellulosic Biomass* [Online]. Michigan Technological University. Available: http://www.sfi.mtu.edu/FutureFuelfromForest/LignocellulosicBiomass.htm
- [Accessed 12/01/2014]. NEVES, D., THUNMAN, H., MATOS, A., TARELHO, L. & GOMEZ-BAREA, A. 2011. Characterization and prediction of biomass pyrolysis products. *Progrss in Energy Combustion Science*, 37, 611-630.
- NGUYEN, T. A., KIM, K. R., HAN, S. J., CHO, H. Y., KIM, J. W., PARK, S. M. & SIM, S. J. 2010. Pretreatment of rice straw with ammonia and iconic liquid for lignocellulose conversion to fermentable sugars. *Bioresource Technol*, 101, 7432-7438.
- NONIER, M. F., VIVAS, N., VIVAS DE GAULEJAC, N., ABSALON, C., SOULIE, P. & FOUQUET, E. 2006. Pyrolysis-gas chromatography/mass spectrometry of Quercus sp. wood:

Application to structural elucidation of macromolecules and aromatic profiles of different species. *J Anal Appl Pyrol*, 75, 181-193.

- NOON, C. E. & DALY, M. J. 1996. GIS Based Biomass Resource Assessment with BRAVO. *Biomass and Bioenergy*, 10, 101-109.
- NOWAKOWSKI, D. J., JONES, J. M., BRYDSON, R. M. D. & ROSS, A. B. 2007. Potassium catalysis in the pyrolysis behaviour of short rotation willow coppice. *Fuel*, 86, 2389-2402.
- OKIMORI, Y., MAKOTO, O. & FUMIO, T. 2003. Potential of CO2 emission reductions by carbonizing biomass waste from industrial tree platation in South Sumatra, Indonesia. *Mitigation and Adaptation Strategies for Global Change*, 8, 261-280.
- OKOROIGWE, E. C. & SAFFRON, C. M. 2012. Determination of bio-energy potential of Palm Kernel Shell by physicochemical characterisation. *Nigerian Journal of Technology*, 31.
- OREGON STATE UNIVERSITY. 2004. *World Food Crops: Oil Crops* [Online]. Available: http://oregonstate.edu/instruct/css/330/seven/index2.htm#OilDistribution [Accessed 19/09/2014].
- OSBORNE, M. 2004. *Adsorption Isotherms* [Online]. Available: http://www.sussex.ac.uk/Users/kaf18/SurfSci2.pdf [Accessed 15/04/2012].
- PAGNANELLI, F., MAINELLI, S. & TORO, L. 2008. New biosorbent materials for heavy metal removal: Product development guided by active site characterisation. *Water Res*, 42, 2953-2962.
- PANDEY, A., SOCCOL, C. R., NIGAM, P. & SOCCOL, V. T. 2000. Biotechnological potential of agro-industrial residues. 1: Sugarcane bagasse. *Bioresource Technol*, 74, 69-80.
- PARRY, M. L., ROSENZWEIG, C., IGLESIAS, A., FISCHER, G. & LIVERMORE, M. 1999. Climate change and world food security: a new assessment. *Global Environmental Change*, 9, 51-67.
- PAUL, H., ERNSTING, A., SEMINO, S., GURA, S. & LORCH, A. 2009. Agriculture and Climate Change - Real Problems, False Solutions. Oxford, UK.
- PENG, X., YE, L. L., WANG, C. H., ZHOU, H. & SUN, B. 2011. Temperature- and durationdependent rice straw-derived biochar: Characteristics and its effects on soil properties of an Ultisol in southern China. *Soil Till Res*, 112, 159-166.
- PENN STATE COLLEGE OF AGRICULTURAL SCIENCES. 2012. Crop residue production of different crops in rotation [Online]. Available: http://extension.psu.edu/agronomy-guide/cm/tables/table-1-1-3 [Accessed 28/09/12].
- PERLACK, R. D., WRIGHT, L. L., TURHOLLOW, A., GRAHAM, R. L., STOKES, B. & ERBACH, D. C. 2005. Biomass as a Feedstock for a Bioenergy and Bioproducts Insudtry: The Technical Feasibility of a Billion-Tonne Annual Supply. US Department of Energy and US Department of Agriculture,.
- PETERS, B. 2011. Prediction of pyrolysis of pistachio shells based on its components hemicellulose, cellulose and lignin. *Fuel Process Technol*, 92, 1993-1998.
- PHONPHUAK, N. & THIANSEM, S. 2012. Using charcoal to increase properties and durability of fired briquettes. *Construction and Building Materials*, 29, 612-618.
- PORTER, J. R., XIE, L., CHALLINOR, A., COCHRANE, K., HOWDEN, M., IQBAL, M. M., LOBELL, D. B. & TRAVASSO, M. I. 2014. Food Security and Food Production Systems. In: AGGARWAL, P. & HAKALA, K. (eds.) Working Group II contribution to the IPCC Fifth Assessment Report: Climate Change 2014: Impacts, Adaptation and Vulnerability. Stanford: Intergovernmental Panel on Climate Change.
- PRETO, F. Pyrolysis, Char and Energy. The Canadian Biochar Initiative, 12/12/2008 2008 Ste Anne de Bellevue, Canada.
- QI, Z., JIE, C., TIEJUN, W. & YING, X. 2007. Review of biomass pyrolysis oil properties and upgrading research. *Energy Conversion and Management*, 48, 87-92.
- RAMACHANDRA, T. V., JOSHI, N. V. & SUBRAMANIAN, D. K. 2000. Present and prospective role of bioenergy in regional energy system. *Renewable and Sustainable Energy Reviews*, 4375-430.

- RAMANKUTTY, N., EVAN, A. T., MONFREDA, C. & FOLEY, J. A. 2008. Farming the planet: 1. Geographical distribution of agricultural lands in the year 2000. *Global Biogeochem. Cy.*, 22, 1-19.
- RAMASWAMY, V., BOUCHER, O., HAIGH, J., HAUGLUSTAINE, D., HAWYWOOD, J., MYHRE, G., NAKAJIMA, T., SHI, G. Y. & SOLOMON, S. 2001. Radiative Forcing of Climate Change. *Climate Change 2001: The Scientific Basis.* Cambridge UK, and New York, NY: Cambridge University Press.
- RAY, D. K., RAMANKUTTY, N., MUELLER, M. P., WEST, D. C. & FOLEY, J. 2012. Recent patterns of crop yield growth and stagnation. *Nat. Commun.*, 3, 1-7.
- REDDY, B. V. S., REDDY, P. S., BIDINGER, F. & BLUMMEL, M. 2003. Crop management factors influencing yield and quality of crop residues. *Field Crop Res*, 84, 57-77.
- REID, G. 2009. Delivering sustainable practice? A case study of the Scottish Active Schools programme. *Sport Education and Society*, 14, 353-370.
- REID, P. 2008. Biosphere carbon stock management: addressing the threat of abrupt climate change in the next few decades: an editorial essay. *Climatic Change*, 87, 305-320.
- REID, P. & PARSHOTAM, A. 2008. Holistic greenhouse gas management strategy (With reviewers' comments and authors' rejoinders). *Institute of Policy Studies Working Paper 07/01.* Victoria University of Wellington, Wellington, New Zealand.
- RETAN, G. A. 1915. Charcoal as a means of solving some nursery problems. *Forestry Quarterly*, 13, 25-30.
- RIAHI, K., GRUBLER, A. & NAKICENOVIC, N. 2007. Scenarios of long-term socio-economic and environmental development under climate stabilization. *Technological Forecasting & Social Change*, 74, 887-935.
- RIAHI, K., RAO, S., KREY, V., CHO, C., CHIRKOV, V., FISCHER, G., KINDERMANN, G., NAKICENOVIC, N. & RAFAJ, P. 2011. RCP 8.5 - A scenario of comapritively high greenhouse gas emissions. *Climatic Change*, 109, 33-57.
- ROBERTS, K. G., GLOY, B. A., JOSEPH, S., SCOTT, N. R. & LEHMANN, J. 2010. Life Cycle Assessment of Biochar Systems: Estimating the Energetic, Economic, and Climate Change Potential. *Environ Sci Technol*, 44, 827-833.
- RODRIGUEZ, L., SALAZAR, P. & PRESTON, T. R. 2009. Effect of biochar and biodigester effluent on growth of maize in acid soils. *Livestock research for rural development*, 21.
- ROGULSKA, M. & KUNIKOWSKI, G. 2006. *Biomass potential assessment and land use change* [Online]. Available:

http://ec.europa.eu/research/energy/pdf/gp/gp_events/biorefinery/bs1_05_roguls ka_en.pdf [Accessed 29/12/2011].

RONDON, M., MOLINA, D., HURTADO, M., RAMIREZ, J., LEHMANN, J., MAJOR, J. & AMEZQUITA, E. Enhancing the productivity of crops and grasses while reducing greenhouse gas emissions through bio-char amendments to unfertile tropical soils. 18th World Congress in Soil Science, 9-15 July 2006 Philadelphia, PA.

ROY, C., DARMSTADT, H., BENALLAL, B., CHAALA, A. & SCHWERDTFEGER, A. E. 2011. *Vacuum pyrolysis of used tires* [Online]. Available: http://www.p2pays.org/ref/11/10504/html/biblio/html4/pyh3.htm [Accessed 21/10/2011].

- ROY, C., LABRECQUE, B. & DE CAUMIA, B. 1990. Recycling of scrap tyres to oil and carbon black by vacuum pyrolysis. *Resources, Conservation and Recycling*, 4, 203-213.
- ROYAL SOCIETY 2009. Geoengineering the climate: science, governance and uncertainty. *In:* SHEPHERD, J. (ed.).
- SAHA, C. K. & SAGORIKA, S. 2013. Carbon Credit of Jute and Sustainable Environment. *Jute Matters*, 1, 1-2.
- SALDARRIAGA, J. G. & WEST, D. C. 1986. Holocene fires in the northern Amazon basin. *Quaternary Research*, 26, 358-366.
- SCHELLER, H. V. & ULVSKOV, P. 2010. Hemicelluloses. *Annual Review of Plant Biology*, 61, 263-289.

- SCHIMMELPFENNIG, S. & GLASER, B. 2012. One Step Forward toward Characterization: Some Important Material Properties to Distinguish Biochars. *Journal of Environmental Quality*, 41, 1001-1013.
- SCHMIDT, M. W. I., SKJEMSTAD, J. O. & JAGER, C. 2002. Carbon isotope geochemistry and nanomorphology of soil black carbon: Black chernozemic soilsin central Europe originate from ancient biomass burning. *Global Biogeochem Cy*, 16, 701-708.
- SCHNEIDER, D., ESCALA, M., SUPAWITTAYAYOTHIN, K. & TIPPAYAWONG, N. 2011. Characterization of biochar from hydrothermal carbonization of bamboo. *International Journal of Energy and Environment*, 2, 647-652.
- SCHULZ, H., DUNST, G. & GLASER, B. 2013. Positive effects of composted biochar on plant growth and soil fertility. *Agron. Sustain. Dev.*, 33, 817-827.
- SCHULZ, H. & GLASER, B. 2012. Effects of biochar compared to organic and inorganic fertilizers on soil quality and plant growth in a greenhouse experiment. *Journal of Plant Nutrition and Soil Science*, 175, 410-422.
- SCHUMACHER, B. A. 2002. Method for the determination of total organic carbon (TOC) in soils and sediments. Las Vegas: United States Environmental Protection Agency.
 SEDAC. 2014. *Global Agricultural Lands* [Online]. Available:
- http://sedac.ciesin.columbia.edu/data/collection/aglands [Accessed 11/09/2014].
- SHACKLEY, S., SOHI, S., BROWNSORT, P., CARTER, S., COOK, J., CUNNINGHAM, C., GAUNT, J., HAMMOND, J., IBARROLA, R., MASEK, O., SIMS, K. & THORNLEY, P. 2010. An Assessment of the Benefits and Issues Associated with the Application of Biochar to Soils. *In:* SHACKLEY, S. & SOHI, S. (eds.). UK Biochar research Centre,.
- SHINOGI, Y., YOSHIDA, H., KOIZUMI, T., YAMAOKA, M. & SAITO, T. 2002. Basic characteristics of low-temperature carbon products from waste sludge. *Advances in Environmental Research.*, 7, 661-665.
- SIEGENTHALER, U. & SARMIENTO, J. L. 1993. Atmospheric carbon dioxide and the ocean. *Nature*, 365, 119-126.
- SINCLAIR, T. R. 1998. Historical Changes in Harvest Indexd and Crop Nitrogen Accumulation. *Soil Science*, 38, 638-643.
- SINGH, B. P., COWIE, A. L. & SMERNIK, R. J. 2012. Biochar carbon stability in a clayey soil as a function of feedstock and pyrolysis temperature. *Environmental Science and Technology*, 46, 11770-11778.
- SLADE, R., SAUNDERS, R., GROSS, R. & BAUEN, A. 2011. *Energy from Biomass: The size of the global resource.*, London, Imperial College Centre for Energy Policy and Technology and UK Energy Research Centre.
- SMIL, V. 1999. Crop Residues: Agriculture's Largest Harvest. *Bioscience*, 49, 299-308.

SOCIETY FOR BIOLOGICAL ENGINEERING. 2011. Producing fuels and chemicals from lignocellulosic biomass [Online]. Available: http://www.aiche.org/uploadedFiles/SBE/MemberCenter/0808S10(2).PDF [Accessed 13/10/11].

- SOHI, S., E LOEZ-CAPAL, E KRULL & R BOL 2009. Biochar's roles in soil and climate change: A review of research needs. *CSIRO Land and Water Science Report.*
- SOHI, S. P., KRULL, E., LOEZ-CAPEL, E. & BOL, R. 2010. Chapter 2- A Review of Biochar and Its Use and Function in Soil. *Advances in Agronomy*, 105, 47-82.
- SOMBROEK, W., RUIVO, M. L., FEARNSIDE, P. M., GLASER, B. & LEHMANN, J. 2003. Amazonian Dark Earths as carbon stores and sinks. *In:* LEHMANN, J., KERN, D. C., GLASER, B. & WOODS, W. I. (eds.) *Amazonian Dark Earths: Origin, Properties, Management.* The Netherlands: Klewer Academic Publishers.
- STEINER, C. 2010. Biochar: Mitigation of Climate Change and Soil Restoration. Deutsche Gesellschaft fur Internationale Zusammenarbeit (GIZ), FOrests and Climate Change Programme (FORCLIME).
- STERN, D. I. 2004. The Rise and Fall of the Environmental Kuznets Curve. *World Development*, 32, 1419-1439.

- SUD, D., MAHAJAN, G. & KAUR, M. P. 2008. Agricultural waste material as potential adsorbent for sequestering heavy metal ions from aqueous solutions- A review. *Bioresource Technol*, 99, 6017-6027.
- TANAKA, S. 1963. Fundamental Study on wood carbonization. *Bulletin of Experimental Forestry of Hokkaido University*.
- THOMSON, A. M., CALVIN, K. V., CHINI, L. P., HURTT, G. C., EDMONDS, J. A., BOND-LAMBERTY, B., FROLKING, S., WISE, M. A. & JANETOS, A. C. 2010. Climate mitigation and the future of tropical landscapes. *Proceedings of the Natioanl Academy of Sciences*, 107, 19633-19638.
- THOMSON, A. M., CALVIN, K. V., SMITH, S. J., KYLE, P., VOLKE, A., PATEL, P., DELGADO-ARIAS, S., BOND-LAMBERTY, B., WISE, M. A., CLARKE, L. E. & EDMONDS, J. A. 2011. RCP 4.5: a pathway for stabilization of radiative forcing by 2100. *Climatic Change*, 109 77-94.
- TOMCZAK, F., SYDENSTRICKER, T. H. D. & SATYANARAYANA, K. G. 2007. Studies on lignocellulosic fibers of Brazil. Part II: Morphology and properties of Brazilian cononut fibers. *Composites Part A: Applied Science and Manufacturing*, 38, 1710-1721.
- TRIMBLE, W. H. 1851. On charring wood. Plough, the Loom and the Anvil.
- UCHIMIYA, M., WARTELLE, L. H., KLASSON, K. T., FORTIER, C. A. & LIMA, I. M. 2011. Influence of Pyrolysis Temperature on Biochar Property and Function as a Heavy Metal Sorbent in Soil. *J Agr Food Chem*, 59, 2501-2510.
- UK BIOCHAR RESEARCH CENTRE 2010. An Assessment of the Benefits and Issues Associated with the Application of Biochar to Soils. *In:* SHACKLEY, S. & SOHI, S. (eds.).
- UK BIOCHAR RESEARCH CENTRE. 2011. UK Biochar Research Centre: reducing and removing CO2 whilst improving soils: A significant and sustainable response to climate change [Online]. Available: http://www.biochar.org.uk/ [Accessed 10/10/11].
- ULBIG, P. & HOBURG, D. 2002. Determination of the calorific value of natural gas by different methods. *Thermochimica Acta*, 382, 27-35.
- UNCCD 2009. Required policy actions to include carbon contained in soils including the use of biochar to replenish soil carbon pools, and restore soil fertility and sequester CO2.
- UNEP 1995. IPCC Guidelines for National Greenhouse Gas Inventories.
- UNEP. 2012. *The Montreal Protocol on Substances that Deplete the Ozone Layer* [Online]. Available: http://ozone.unep.org/new_site/en/montreal_protocol.php [Accessed 07/05/2012].
- UNFCCC. 2012. *Kyoto Protocol* [Online]. Available:

http://unfccc.int/kyoto_protocol/items/2830.php [Accessed 27/03/2012].

UNFCCC. 2014. *Background on the UNFCCC: The international response to climate change* [Online]. Available: http://unfccc.int/essential_background/items/6031.php [Accessed 02/12/2014].

UNITED NATIONS ECONOMIC COMMISSION FOR EUROPE. 1979. *The 1979 Geneva Convention on Long-range Transboundary Air Pollution* [Online]. Available: http://www.unece.org/env/lrtap/lrtap_h1.html [Accessed 10/11/2014].

UNITED NATIONS ENVIRONMENT PROGRAMME. 1992. *Rio Declaration on Environment and Development* [Online]. Available: http://www.unep.org/Documents.Multilingual/Default.asp?documentid=78&article id=1163 [Accessed 10/04/2012].

UNITED NATIONS STATISTICS DIVISION. 2013. Composition of macro geographical (continental) regions, geographical sub-regions, and selected economic and other groupings [Online]. Available:

http://unstats.un.org/unsd/methods/m49/m49regin.htm [Accessed 18/04/2013].

UNIVERSITY OF EAST ANGLIA. 2011. Aromatic compounds and aromaticity [Online]. Available: ≤http://www.uea.ac.uk/~c286/aromaticnotes.htm≥ [Accessed 29/10/11].

- UNIVERSITY OF WATERLOO. 2012. *Calorimetry: Measuring Heats of Reactions* [Online]. Available: http://www.science.uwaterloo.ca/~cchieh/cact/c120/calorimetry.html [Accessed 15/04/2012].
- US BIOCHAR INITIATIVE. 2011. *Biochar Sustainability Protocols* [Online]. Available: http://www.biochar-

international.org/sites/default/files/Biochar_Sustainability_Protocols_March_2011_ Draft.pdf [Accessed 10/04/2011].

- US NATIONAL RESEARCH COUNCIL BOARD ON AGRICULTURE 1983. Crop Residues. Underutilized Resources as Animal Feedstuffs. Washington: National Academy Press.
- VACCARI, F. P., BARONTI, S., LUGATO, E., GENESIO, L., CASTALDI, S., FORNASIER, F. & MIGLIETTA, F. 2011. Biochar as a strategy to sequester carbon and increase yield in durum wheat. *European Journal of Agronomy*, 34, 231-238.
- VAN VUUREN, D., DEN ELZEN, G. J., LUCAS, P. L., EICKHOUT, B., STRENGERS, B. J., VAN RUIJVEN, B., WONINK, S. & VAN HOUDT, R. 2007. Stabilizing greenhouse gas concentrations at low levels: an assessment of reduction strategies and costs. *Climatic Change*, 81, 119-159.
- VAN VUUREN, D. P., EDMONDS, J., KAINUMA, M., RIAHI, K., THOMSON, A., HIBBARD, K., HURTT, G. C., KRAM, T., KREY, V., LAMARQUE, J.-F., MASUI, T., MEINSHAUSEN, M., NAKICENOVIC, N., SMITH, S. J. & ROSE, S. K. 2011a. The representative concentration pathways: an overview *Climatic Change*, 109, 5-31.
- VAN VUUREN, D. P., STEHFEST, E., DEN ELZEN, G. J., VAN VLIET, J. & ISAAC, M. 2010. Exploring IMAGE model scenarios that keep greenhouse gas radiative forcing below 3 W/m2 in 2100. *Energy Economics*, 1105-1120.
- VAN VUUREN, D. P., STEHFEST, E., DEN ELZEN, M. G. J., KRAM, T., VAN VLIET, J., DEETMAN, S., ISAAC, M., GOLDEWIJK, K. K., HOF, A., BELTRAN, A. M., OOSTENRIJK, R. & VAN RUIJVEN, B. 2011b. RCP2.6: exploring the possiblity to keep global mean temperature increase below 2oC. *Climatic Change*, 109, 95-116
- VAN ZWEITEN, L. 2012. *Primary Industries Science and Research: Biochar* [Online]. Available: http://www.dpi.nsw.gov.au/research/topics/biochar [Accessed 01/01/2012].
- VAN ZWEITEN, L., KIMBER, S., DOWNIE, A., CHAN, K. Y., COWIE, A., WAINBERG, R. & MORRIS, S. Papermill char: Benefits to soil health and plant production. Conference of the International Agrichar Initiative, 30 April-20th May 2007 Terrigal, NSW, Australia.
- VAN ZWEITEN, L., KIMBER, S., MORRIS, S., CHAN, K. Y., DOWNIE, A., RUST, J., JOSEPH, S. & COWIE, A. 2010. Effects of biochar from slow pyrolysis of papermill waste on agronomic performance and soil fertility. *Plant Soil*, 327, 235-246.
- VAN ZWEITEN, L., SINGH, B., JOSEPH, S., KIMBER, S., COWIE, A. & CHAN, K. Y. 2009. Biochar and Emissions of Non-CO2 Greenhouse Gases from Soil. *In:* LEHMANN, J. & JOSEPH, S. (eds.) *Biochar for Environmental Management: Science and Technology.* London, UK and Stirling, USA: Earthscan.
- VCS. 2014. *Verified Carbon Standard* [Online]. Available: http://www.v-c-s.org/ [Accessed 10/11/2014].
- VERHEIJEN, F., JEFFERY, S., BASTOS, A. C., VAN DER VELDE, M. & DIAFAS, I. 2010. Biochar Application to Soils: A Critical Scientific Review of Effects on Soil Properties, Process and Functions. *JRC Scientific and Technical Reports*. European Communities.
- VERVERIS, C., GEORGHIOU, K., CHRISTODOULAKIS, N., SANTAS, P. & SANTAS, R. 2004. Fiber dimensions, lignin and cellulose content of various plant materials and their suitability for paper production. *Industrial Crops and Products*, 10, 245-254.
- WAGNER, A. & KAUPENJOHANN, M. 2014. Suitability of biochars (pyro- and hydrochars) for metal immobilization on former sewage-field soils. *European Journal of Soil Science*, 65, 139-148.
- WALCOTT, J. J., CHAUVIROJ, M., CHICHEST, A., CHOTICHEUY, P., FERRARIS, R. & NORMAN, B.
 W. 1977. Long-term Productivity of Intense Rice Cropping Systems of the Central Plain of Thailand. *Expl. Agriculture*, 13, 305-315.

- WARNOCK, D. D., LEHMANN, J., KUYPER, T. W. & RILLIG, M. C. 2007. Myorrhizal responses to biochar in soil concepts and mechanisms. *Plant Soil*, 300, 9-20.
- WILLIAMS, P. T. 2013. Other Waste Treatment Technologies: Pyrolysis, Gasification, Combined Pyrolysis-Gasification, Composting, Anaerobic Digestion: Pyrolysis. *Waste Treatment and Disposal.* 2nd ed. Chichester: Wiley.
- WILLIAMS, P. T. & BESLER, S. 1996. The Influence of Temperature and Heating Rate on the Slow Pyrolysis of Biomass. *Renew Energ*, 7, 233-250.
- WINDEATT, J. H., ROSS, A. B., WILLIAMS, P. T., FORSTER, P. M., NAHIL, M. A. & SINGH, S. 2014. Characteristics of biochars from crop residues: Potential for carbon sequestration and soil amendment. J. Environ. Manage., 15, 189-197.
- WISE, M., CALVIN, K., THOMSON, A., CLARKE, L., BOND-LAMBERTY, B., SANDS, R., SMITH, S. J., JANETOS, A. & EDMONDS, J. 2009. Implications of Limiting CO2 Concentrations for Land Use and Energy. *Science*, 324, 1183-1186.
- WOODS, W. I. 2003. Development of Anthrosol Research. *In:* LEHMANN, J., KERN, D. C., GLASER, B. & WOODS, W. I. (eds.) *Amazonian Dark Earths: Origin, Properties, Management.* Netherlands: Kluwer Academic Publishers.
- WOOLF, D., AMONETTE, J. E., STREET-PERROTT, A. F., LEHMANN, J. & JOSEPH, S. 2010.
 Sustainable biochar to mitigate global climate change. *Nature Communications*, 1, 1-9.
- WORLD BANK. 2011. *Country Income Groups* [Online]. Available: http://chartsbin.com/view/2438 [Accessed 15/04/2013].
- WORLD BANK. 2013. *Country and Lending Groups* [Online]. Available: http://data.worldbank.org/about/country-classifications/country-and-lendinggroups [Accessed 15/04/2013].
- WU, C. F., NAHIL, M. A., SUN, X., SINGH, S., CHEN, J. H., SHEN, B. X. & WILLIAMS, P. T. 2012. Novel application of low temperature biochar from biomass pyrolysis for low temperature selective catalytic reduction. *Journal of the Energy Institute*, 85, 236-239.
- XU, G., YINGCHUN, L., JUNNE, S., HONGBO, S. & LINLIN, W. 2012. Recent Advances in Biochar Applications in Agricultural Soils: Benefits and Environmental Implications. *Clean-Soil Air Water*, 40, 1093-1098.
- XU, R., FERRANTE, L., HALL, K., BRIENS, C. & BERRUTI, F. 2011. Thermal self-sustainability of biochar production by pyrolysis. *J Anal Appl Pyrol*, 91, 55-66.
- YAMAN, S. 2004. Pyrolysis of biomass to produce fuels and chemical feedstocks. *Energ Convers Manage*, 45, 651-671.
- YANAI, Y., TOYOTA, K. & OKAZAKI, M. 2007. Effects of charcoal addition on N20 emissions from soil resulting from rewetting air dried soil in short-term laboratory experiments. *Science and Plant Nutrition*, 53, 181-188.
- YAO, F. X., CAMPS, A. M., VIRQEL, S., BLANCO, F., AROSTEQUI, J., MACIA-AQULLO, J. A. & MACIAS, F. 2010. Simulated geochemical weathering of a mineral ash-rich biochar in a modified Soxhlet reactor. *Chemosphere*, 80, 724-732.
- YIN CHAN, K. & ZHIHONG, X. 2009. Biochar: Nutrient Properties and Their Enhancement. *In:* LEHMANN, J. & JOSEPH, S. (eds.) *Biochar for Environmental Management: Science and Technology.* London, UK and Stirling, USA: Earthscan.
- ZAFAR, S. 2011. *Biomass Pyrolysis* [Online]. Available: http://www.altenergymag.com/emagazine.php?issue_number=09.02.01&article=py rolysis [Accessed 22/01/2012].
- ZHANG, A., CUI, L., PAN, G., LI, L., HUSSAIN, Q., ZHANG, X., ZHENG, J. & CROWLEY, D. 2010. Effect of biochar amendment on yield and methane and nitrous oxide emissions from a rice paddy from Tai Lake plain, China. *Agr Ecosyst Environ*, 139, 469-475.
- ZHANG, Z. 2011. In what format and under what timeframe would China take on climate commitments? A roadmap to 2050. *International Environmental Agreements : Politics, Law and Economics,* 11, 245-259.

- ZHAO, L., CAO, X., MASEK, O. & ZIMMERMAN, A. 2013. Heterogeneity of biochar properties as a function of feedstock sources and production temperatures. *J Hazard Mater.*, 256, 1-9.
- ZHAO, X., ZHANG, J., SONG, Z., LIU, H., LI, L. & MA, C. 2011. Microwave pyrolysis of straw bale and energy balance analysis. *J Anal Appl Pyrol*, 92, 43-49.
- ZHENG, W., SHARMA, B. K. & RAJAGOPALAN, N. 2010. Using Biochar as a Soil Amendment for Sustainable Agriculture. Illinois Sustainable Technology Centre, University of Illinois.
- ZIMMERMAN, A. R. 2010. Abiotic and Microbial Oxidation of Laboratory_Produced Black Carbon (Biochar). *Environ Sci Technol*, 44, 1295-1301.